Finding a Place in the Journalistic Field:

An examination of two digital native news organisations, BuzzFeed and Vice

Paul David Stringer

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

The University of Leeds School of Media and Communication

September 2018

The candidate confirms that the work submitted is his own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

The right of Paul Stringer to be identified as Author of this work has been asserted by him in accordance with the Copyright, Designs and Patents Act 1988.

© 2018 The University of Leeds and Paul Stringer

Acknowledgements

There are so many people to thank – tangentially and directly – for their support, encouragement, friendship, patience, and loyalty during this process.

Mum: thank you for allowing me to move back home again and I'm sorry I have not cooked us anything the last two months. But I am forever grateful for your love and support and most of all your patience. You have, by very definition, lived this process with me. I simply couldn't have completed this work without you. Thank you.

Shanice: meeting you has been my single greatest joy of the PhD process. Thank you, from the bottom of my heart, for your words of encouragement and your never-ending belief in me. I love you and I can't wait to start our new chapter.

Stephen and Cat: I am not combining you because you're twins. But because your joint Stringer-brand of humour has kept me alive this whole process and made me see the funny side of everything. Thank you for the whisky, too.

Chris and Julie: thank you both for your patience, wisdom, and guidance during this process. I am indebted to each of you for the endless hard work you put in to getting me over the line. This thesis has been a collaborative effort from start to finish and so this work feels as much yours as mine.

Smithy, *AJ:* your friendship has been the salve to this whole process. I look forward to not cancelling so many of our plans in the future.

*Finally...*thank you to Leeds University and everyone from the School of Media and Communication. I was an undergraduate student at the school almost exactly ten years ago today, and what a wonderful decade of education it has been.

Abstract

Over approximately the last decade, an interesting counterpoint to the dominant narrative of "crisis" has been the rise and entrenchment of digital native news organisations. In a fraught and challenging news environment, companies such as *BuzzFeed, The Huffington Post, Vice*, and *Vox* have grown in size and stature to the point where they now compete with legacy news media for attention and advertising.

Despite this, digital natives remain conspicuous by their absence in the journalism studies literature. As a consequence, much of what we know about these organisations relates their so-called status as "innovators" in news. By virtue of being new to the field it is often taken-for-granted that these organisations are *different* from traditional journalism. However, current discourse rarely expands upon this observation to explain precisely *how*.

This thesis focuses on two North American digital-native news organisations: *BuzzFeed* and *Vice*. As two of the largest and most popular digital natives in the world, these organisations merit closer critical attention. Adopting a mixed methods approach, this research combines qualitative interviews with 24 journalists and a comparative quantitative content analysis to examine the organisation and production of news at *BuzzFeed* and *Vice*.

Despite having reputations for being innovators in digital news, this research shows that both organisations remain surprisingly wedded to traditional norms, stemming from a combination of resource constraints, and a desire to be recognised as legitimate by peers and the public at large. Using Bourdieu's field theory as a theoretical framework, the results of this research also provide the stimulus for a broader consideration about the field of journalism; both how it is changing, and, how, in many ways, it remains the same.

Table of Contents

Acknowledgements	iii	
Abstract	iv	
Table of Contents	v	
List of Tables	viii	
1. Introduction		
1.1. Research Gap		
1.2. Research Questions	12	
1.3. Thesis Outline	13	
2. Theoretical Framework	16	
2.1. Field Theory	16	
2.2. The Professionalization of Journalism	19	
2.3. Professionalism, Ideology and Field Theory	24	
2.4. The Digitization of the Field	25	
3. Literature Review		
3.1. Gatekeeping Research		
3.2. Objectivity, Emotionality, and Transparency		
3.3. Summary		
4. Research Design and Methods		
4.1. Case Study Selection		
4.1.1. Vice		
4.1.2. BuzzFeed		
4.2. Research Design		
4.2.1. Mixed Methods		
4.3. In-depth Interviews		
4.4. Quantitative Content Analysis		
4.4.1. Design and Rationale		
4.4.2. Coding Variables		
4.4.3. Coding Scheme and Codebook	57	
4.4.4. Unit of Analysis		
4.4.5. Pilot Study	58	
4.4.6. Results and Analysis		
4.5. Limitations	59	
5. Building News: BuzzFeed and Vice's arrival in the journalistic field	62	
5.1. Green Shoots		
5.2. Investing in News	66	
5.3. News Content at BuzzFeed and Vice	69	
5.3.1. Authorship	69	
5.3.2. Format	71	
5.3.3. News Topic		
5.3.4. Word Count	75	
6. Organising News: The Structure and Organisation of News at BuzzFeed and Vice		
6.1. Motivating Factors	78	

6.2. Funding and Partnerships	80	
6.2.1. Inter-Firm Integration: Investments, Partnerships, and Deals		
6.2.2. Intra-Firm Integration: Native Advertising		
6.3. Staff		
6.3.1. Staff Demographics		
6.3.2. Hiring Practices		
6.3.3. Senior / Experienced Staff		
6.3.4. Young Staff		
6.4. Topics and Beats 6.4.1. Vice		
6.4.2. BuzzFeed		
6.4.3. Traditional Beats		
6.4.4. Niche Topics		
6.5. Conclusion		
7. Producing News: The Routines of News Production at <i>BuzzFeed</i> and <i>Vice</i>	117	
7.1. Routines		
7.1.1. The Structure of Daily Work		
7.1.2. Multiskilling		
7.1.3. Autonomy (Part I)		
7.1.4. Autonomy (Part II)		
7.1.5. Audience Analytics and Metrics		
7.2. Conclusion		
8. Making News: News Production Practices at BuzzFeed and Vice	140	
8.1. Exclusivity and Gatekeeping		
8.1.1. Scoops and Exclusives		
8.1.2. Redefining Exclusivity		
8.2. Sourcing Practices		
8.2.2. Sourcing Rhetoric		
8.2.3. Sourcing Practices		
8.2.4. Analysis and Discussion		
8.3. Objectivity		
8.3.1. Transparency	179	
8.3.1.1. Disclosure Transparency	181	
8.3.1.2. Participatory Transparency	190	
8.3.3. Emotion		
8.4. Conclusion	209	
9. Conclusion: <i>BuzzFeed</i> and <i>Vice</i> in the Field of Journalism	213	
9.1. Overview of research findings		
9.2. Journalistic Autonomy or "What is Happening to the Field?"		
Appendix A: Sample email requesting newsroom access	223	
Appendix B: Interview guide		
Appendix C: Sample email requesting interview		
Appendix D: Participant Information Sheet and Consent Form		
Appendix E: Interviewee List	231	
Appendix F: Coding Scheme23		
Appendix G: Codebook	239	

9. B	bliography.	
------	-------------	--

List of Tables

Table 1: Author gender of news articles 70
Table 2: Author ethnicity of news articles 71
Table 3: Format of news articles
Table 4: Topic of news articles 73
Table 5: Word count of news articles
Table 6: Word count of news reports
Table 7: Source exclusivity in news articles 148
Table 8: Source exclusivity in news articles (opinion and comment articles re-coded as
non-exclusive)149
Table 9: Example headlines in news articles published by BuzzFeed and Vice
Table 10: Presence of interpretive journalism in news articles
Table 11: Interpretive journalism versus topic
Table 12: Interpretive journalism in news reports
Table 13: Summary of sources in news articles
Table 14: Type of sources in news articles 168
Table 15: Gender of sources in news articles
Table 16: Age of sources in news articles 171
Table 17: Ethnicity of sources in news articles 173
Table 18: Authorial information in news articles
Table 19: Time stamps in news articles 182
Table 20: Hyperlinks in news stories
Table 21: Hyperlinks in news stories (adjusted)186
Table 22: Hyperlinks in news stories (UK and US split)187
Table 23: Internal and external hyperlinks in news stories
Table 24: Participatory features in news articles 192
Table 25: Examples of forward referencing in headlines199
Table 26: Headline language in news articles
Table 27: Emotive headline versus news format
Table 28: Lead paragraph in news articles
Table 29: Personalised storytelling in news articles 203
Table 30: Emotive lead paragraph versus news format
Table 31: Personalised storytelling versus news format
Table 32: Emotive story lead versus news format
Table 33: Personalised storytelling versus news subject

Part 1 Introduction

1. Introduction

In the decade following the first phase of web development in newsrooms – which stretched roughly from 1994 to 2004 – a second wave of technological change began to unfurl (Bell et al. 2017). The spread and diffusion of broadband internet access, Web 2.0 technologies, and mobile devices precipitated a shift toward an increasingly digital, mobile, and social media environment that brought new challenges and opportunities to the field of journalism (Nielsen et al. 2016). While legacy media were generally slow to respond to these tectonic pressures, a wave of new, digitally native media companies would put these new technologies at the heart of their businesses. Among the most notable of these were US-based organisations such as *The Huffington Post, BuzzFeed, Vox, Business Insider, Quartz, Mashable,* and *Vice* – which arrived with aspirations to "disrupt" the industry by introducing new ideas and practices to the field (Carlson and Usher 2016).

Despite pursuing distinct editorial strategies, many of these aforementioned digital natives have pursued an "expansionist strategy", optimising their content for search engines and social media platforms to build large, international audiences whose attention can subsequently be sold to advertisers (Nicholls et al. 2016). Since their launch in the mid-2000s, several digital natives have grown to occupy a prominent position in the news ecosystem, to the point that they now compete with legacy media for attention and advertising (Jurkowitz 2014; Nicholls et al. 2016; Nicholls et al. 2017; Newman et al. 2016; 2017; 2018). The growth and entrenchment of digital natives has served as an interesting counter-narrative to the notion of "crisis" that has permeated much of the discourse about journalism (Zelizer 2015). It is also for this reason that digital natives have become an object of industry fascination, with various journalists, scholars, and public intellectuals heaping praise on them for being "explicitly innovative and forward-looking" (Carlson and Usher 2016: 569).

While the buzz generated by digital natives is understandable, a fixation on "innovation" in the scholarship on digital journalism has often had the effect of homogenising the discourse about these organisations (Kreiss and Brennen 2016). Little remains known, for instance, about the types of content that digital natives produce, or the routines, cultures, and practices that undergird their different sites of production. Given their expanding role in the contemporary information ecosystem, it is vital that scholarship does more to take this large and disparate group of social actors into account.

1.1. Research Gap

To address this gap in research, this study focuses on two North American digital native media organisations: *BuzzFeed* and *Vice*. Over the last decade, these organisations have grown become two of the most popular sources for news and entertainment on the web; particularly among younger generations (Moses 2014; comScore 2017; Newman et al. 2018). Despite this, they have received almost no critical attention from scholars¹. Instead, research has primarily focused on understanding change through the prism of traditional news organisations (Craft et al. 2014; Steensen and Ahva 2015). While this work has generated important insights, it has largely failed to address issues of "difference, divergence, and diversity" in the contemporary news ecosystem (Witschge et al. 2017: 2), thus leaving many important actors unchecked.

This shared parochialism – which is a longstanding issue in journalism studies (Wahl-Jorgensen 2009; Zelizer 2009) – is becoming less tenable as the field grows ever more heterogeneous. In this regard, there is an urgent need for scholarship to embrace *complexity* (Witschge et al. 2018), and look for fresh or novel ways to explain the ongoing evolution of the field. Digital native news organisations would appear to present such an opportunity. Amidst larger media upheaval and a lack of an established paradigm for digital news, these sites have been gifted with an elevated platform to "compete to define what digital news should look like, reestablish the boundaries of journalism, and

¹ For a few minor exceptions, see: Riordan (2014); Küng (2015); Tandoc and Jenkins (2017); Tandoc (2017).

determine strategies for legitimating news content" (Carlson and Usher 2016: 564).

1.2. Research Questions

With this context in mind, the purpose of this thesis is threefold. The first is to provide a holistic description of news production at *BuzzFeed* and *Vice*, paying close attention to the various norms and practices that structure news work at both organisations. The second is to consider the different endogenous and exogenous forces that have shaped *BuzzFeed* and *Vice's* approach to journalism, such as technological, economic, and cultural factors. The third and final aim is to reflect on what these findings tell us about the field at large, and how it might, or might not, be changing.

These three interrelated research goals are reflected in the primary research questions of the thesis, which are as follows:

- 1) What are the major characteristics of news production at *BuzzFeed* and Vice? And how, if at all, do they differ from traditional journalism?
- 2) How do various endogenous and exogenous forces shape *BuzzFeed* and *Vice's* approach to journalism?
- 3) What do these findings tell us about the cultural, economic, social and symbolic capital of the journalistic field? And how, if at all, it might be shifting?

To answer these questions, this project adopts a mixed methods research design. The intent of mixed methods is to "obtain different but complementary data on the same topic" (Morse 1991: 122). In this study, qualitative interviews with journalists from *BuzzFeed* and *Vice* were combined with data from a quantitative content analysis of news content. This triangulation of method was deemed essential to building a holistic picture of news production at *BuzzFeed* and *Vice*, particularly because of the well-documented gap between rhetoric and practice in journalism (Mellado and Van Dalen 2013).

Despite calls to "blow up" the newsroom (Anderson 2011a), this project has also retained the organisational setting as the chief locus of research. The criticism that newsroom research tends to under-emphasise the external factors that shape news content is helpfully addressed by the Bourdieu's theory of "fields" (*champs*). As a theoretical framework, field theory is concerned with bridging the divide between macro-societal and micro-organisational approaches to examine "how meaning is produced *relationally*" (Benson 1999: 486, emphasis added). It is precisely by taking this reflexive approach that I hope to draw links between journalistic practices and the broader macro level structural forces that shape everyday news work (Willig 2012). Studying these relations provides the essential context for asking deeper questions about transformations in the field at large.

1.3. Thesis Outline

This study is divided into two distinct parts. The first part constructs the theoretical and methodological framework for the study. Chapter 2 discusses the development of journalism's professional ideology during the 20th century and introduces Bourdieu's field theory as the framework for the study. Chapter 3 reviews the contemporary literature on digital journalism, tracing the effects of digitization on the core norms and practices of the field. Chapter 4 describes the research design and methods used in this research project and introduces digital natives as a distinct object of study.

The second part of the study discusses the major findings from the research, combining data from qualitative interviews with a comparative content analysis to address the key research questions of the project. Chapter 5 describes *BuzzFeed* and *Vice's* arrival into the journalistic field and establishes some basic facts about the key characteristics of their news content. Chapter 6 describes the structure and organisation of news at *BuzzFeed* and *Vice*: their business model, sources of funding, hiring practices, and news coverage. These different facets are critically interpreted using a Bourdieusian framework. Chapter 7 then goes inside the news departments of *BuzzFeed* and *Vice* to examine the routines and structure of daily work, focusing predominantly on

issues of journalistic autonomy and the impact of the audience in the news process. Chapter 8, the final empirical chapter of the thesis, pays critical attention to the selection and presentation of news, focusing in particular on the norms of gatekeeping and objectivity and their interaction with the broader logic of digital media and culture. Finally, Chapter 9 summarises and reflects on the major findings from the thesis as well as offering some potential avenues for future research.

Part 2 Theoretical and Methodological Background

2. Theoretical Framework

To examine the norms, values and practices of news workers at digital native news organizations, this study combines Pierre Bourdieu's field theory (1991; 1993;1998) and theory from the literature on the sociology of professions (Sarfatti-Larson 1977; Gieryn 1983; Abbott 1988; Waisbord 2013). This chapter begins by introducing field theory before moving on to discuss journalism's project of professionalization. These two strands of theory are then brought together and linked to recent research in the field.

2.1. Field Theory

Beginning with field theory, Bourdieu follows from Max Weber and Emile Durkheim in describing modernity as a "process of differentiation into semiautonomous and increasingly specialized spheres of action - 'fields' (Willig et al. 2015: 3). The reason fields are *semi*-autonomous, rather than completely autonomous, is that they exist within a larger field of forces, the most significant of which in the case of journalism, are the economic and political fields (Vos 2016). For Bourdieu (1993: 162), the field is "an independent social universe with its own laws of functioning", or as he later put it, a social space with "specific logics or demands that differ from the conditions and sets of rules of other fields" (Bourdieu 2000: 99). These "rules", or doxa, to use Bourdieu's terminology, can be thought of as a set of shared understandings that help govern practices and define behavior within the field. As Bourdieu (1998: 37) puts it, those who inhabit a particular field share a "system of presuppositions inherent in membership in a field". In the case of journalism, the "rules" or doxa of the field include accepted institutional roles (e.g. adversarial, watchdog, disseminator), epistemological frameworks (e.g. objectivity, empiricism) and ethical standards (e.g. accuracy, fairness, respect of privacy) (see Hanitzsch 2007 for a detailed breakdown of these different constituents of journalistic culture). These shared presuppositions constitute the *cultural capital* of the

field². Cultural capital is the primary source journalism's legitimacy; and it is this legitimacy, as Vos (2016: 391) writes, that "ultimately earns journalism its autonomy and power". Broad acceptance of these explicit and implicit rules and ways of "doing" journalism helps produce a certain degree of homogeneity within the field (Deuze 2005; Benson 2006). Though crucially, it is individual agents within the field who must accept the journalistic doxa in order for certain ways of doing journalism to become enshrined. The link between the field and its agents, between the micro and macro levels, is defined by the concept of habitus. Habitus refers to one's historical trajectory in the field, "a collection of personal and professional experiences accumulated from social positions that produces knowledge and understanding of the game" (Tandoc 2014: 562). An individual's habitus can both be a source of resistance to the established structures of the field, but also conformity. This helps explain why newer entrants to the field (both individuals and organisations), who are arguably less socialized to the "rules of the game" and come equipped with different forms of cultural capital, can be an important source of transformation (Vos 2016). Compared with print journalists, for example, online journalists have emphasized different norms, such as speed over rigor and greater loyalty to audiences than sources (Cassidy 2005; Steensen 2009; Agarwal and Barthel 2015). However, it is important to note that more often than not, activities by agents within the field largely reproduce its structure (Benson and Neveu 2005). It is largely because of this that we can speak of a dominant occupational ideology³ in journalism, which crystallized as the field underwent a process of professionalization during the course of the 20th century (Deuze 2005).

The *doxa* of the field, however, should not imply the presence of a monolithic or static journalistic culture. As Bourdieu (1985: 734) argues, "every field is the site of more or less overt struggle over the definition of the legitimate principles of the division of the field". This "struggle" can be thought of as both an internal

² Cultural capital is defined elsewhere as a collection of competencies, skills, expertise, knowledge, and other similar characteristics (e.g. Benson and Neveu 2005; Siapera and Spyridou 2012)

³ In this context, ideology is defined as "a collection of values, strategies and formal codes characterizing professional journalism and shared most widely by its members" (Deuze 2005: 445).

battle to preserve or transform the "rules of the game" (Bourdieu 2005), and also, a battle for autonomy against various exogenous forces (chiefly the economic and political fields).

Throughout the course of history, journalism has participated in numerous struggles, some of which have changed the doxa and autonomy of the field. Western journalism in the 19th century, for example, was more explicitly partisan than it is today (Nerone 2009), and hence was dominated by the political field. Since the 1970s, following the period of "high-modernism"⁴ (Hallin 1992) in journalism, the field has become steadily more commercialised, and thus, subject to greater influence from the economic field (McManus 1994; Nadler 2016). There are concerns that as a result of these changes, news values, which are an important part of the journalistic *doxa* (Tandoc 2014), have become subordinate to commercial values, altering the form of news and compromising its democratic purpose (see Schudson 2011; McChesney 2013). Whether the internet has truncated or extended the commoditization of news is an issue that remains contested and will be discussed further later.

One of the biggest strengths of Bourdieu's field theory is that it is a meso-level concept (Willig et al. 2015). The field environment is between the organizational level (micro) and the societal level (macro), which invites us to consider both the journalistic field in relation to other fields, and the individual agents within a field and their relation to one another. This can help transcend well-known binaries such as that between structure and agency; a notable issue in studies of news production (Wahl-Jorgensen 2009; Willig 2012). Perhaps one weakness of field theory, however, is that it does not explain in sufficient detail how fields come into being or where the shared rules or *doxa* of a field emerge (Anderson 2008). Anderson (ibid.) suggests one potential answer to this problem can be found in the work on the sociology of professions, and more specifically, the work of Abbott (1988), who argues that:

⁴ 'High-modernism' refers to a period of journalism that roughly began during the start of the 20th century and culminated in the 1960s, when journalism arguably enjoyed its highest degree of autonomy from the political and economic spheres (see Hallin 1992)

"A professional field begins when an occupational group or groups attempt to seize jurisdiction over a form of expertise via cultural work. The journalistic field emerges, in this instance, when an occupational group ('journalists') attempt to claim for themselves a monopoly on the provision of everyday public knowledge" (cited in Anderson 2008: 257).

Jurisdiction, to borrow Abbot's (1988) terminology, is the link between an occupation's "abstract" knowledge (what it *knows*) and its daily work (what it *does*). Successful jurisdictional claims are important because they grant professions exclusive *rights* that can be converted into various material and symbolic rewards. In the case of journalism, it has been observed that during the course of the 20th century, the occupation managed to establish jurisdiction over the collection and distribution of "factual" information about current events (or "news"). Quite how journalism achieved this has been the subject of extensive and intensive speculation (Schudson 1978; Chalaby 1996; Tumber and Prentoulis 2005; Waisbord 2013). Some of this work, including the literature on the sociology of professions, is discussed further in the following section.

2.2. The Professionalization of Journalism

One could argue that the journalism's professionalization project began around the mid-19th century, when legislative changes, technological innovation, and a subsequent period of commercialisation and industrialisation heralded the emergence of a mass press in the United States and Europe (Barnhurst and Nerone 2009). Journalism began to separate itself from other fields of related activity such as printing, literature, and politics, and develop its own norms and practices (Waisbord 2013). It was also during this period that "journalistic content became less centered on the dissemination of ideas and opinion...and more oriented to the reporting of recent happenings" (Waisbord 2013: 133). This, in effect, resulted in "news" becoming the primary subject of journalism during the middle-to-late-19th century (Barnhurst and Nerone 2009). It was also in this period that the press began to take on a more central role in the political

process, by acting as a "representative" of the people, which helped transform journalism into a recognisable "public good" (Hampton 2001). Perhaps the most important idea to emerge in this period, was the notion of journalism as a "fourth estate" – a term coined by Edmund Burke to describe the role of journalists in democracy as *watchdogs*, collectively tasked with keeping check on the activities of the powerful on behalf of society. It was through this more explicit orientation towards serving the public that journalism arguably made its strongest claim to professional status (Singer 2003).

In the early part of the 20th century, Western journalism began to professionalize further. Press clubs and associations, and schools of journalism were founded, as well as general codes of ethics (Barnhurst and Nerone 2009). The adoption of ethics in newsrooms, based on principles of public service, social responsibility, neutrality and fairness, was particularly important to journalism's claim to professional status (Waisbord 2013). Of all these ethical principles, however, it was objectivity that became the "cornerstone of the professional ideology of journalists" (Lichtenberg 1996: 225). While objectivity was "[not] the only occupational norm to both emerge from and buttress the professional project" (Schudson and Anderson 2009: 93), many scholars consider it to be the most important element in the professional development of Anglo-American journalism (Summers 1994; Banning 1999). It is beyond the scope of this thesis to discuss objectivity in at great length (see Schudson 1978; Kaplan 2002; Maras 2013); however, the establishment of this principle had three major effects. As an ethical ideal, it gave journalism the illusion of scientific rigour, where a more explicit separation of fact from opinion helped distance the occupation from government propaganda, marketeers, publicity agents, and other actors who were seen to pedal a "distorted" version of the truth (Bourdieu 1996; Schudson 2001). Secondly, objectivity performed a social function; by enhancing journalism's status as an independent institution, working for the good of society; and, in a Durkheimian sense, fostering internal solidarity and a sense of shared identity between group members (Schudson 2001). Finally, from a commercial standpoint, it was also critical in enabling newspapers to reach the broadest possible readership, and thus maximize revenue earned from advertising (Hackett 1984).

In reality, objectivity was always a contentious norm, for it implied journalists were neutral transmitters of events, who could provide a "true" account of reality to the public (Hackett 1984). This "idealised image of the journalist" (Anderson 2008: 250) formed the basis of many of the substantive critiques of news production that emerged during the 1970s and 1980s (e.g. Epstein 1973; Tuchman 1978; Schlesinger 1978; Fishman 1980; Soloski 1989; Gans 2004). Nonetheless, objectivity has remained a key element in the professional self-perception of journalists (Schudson 1978; 2001; Reese 1990; Mindich 1998).

While at least in Anglo-American journalism, objectivity came to be recognised as a core tenet of journalistic professionalism, it should be noted that this is not the only, or necessarily most important norm in journalism around the world (Chalaby 1996; Donsbach and Patterson 2004; Hallin and Mancini 2004). As several of the major surveys conducted on journalists around the globe have indicated, professional norms and standards are partly contingent on the media systems, news organisations, political structures, and cultures that are present in different countries (e.g. Weaver 1998; 2012; Deuze 2002; Donsbach and Patterson 2004). This being said, scholars have identified some similarities between journalists around the world. For example, research has indicated that journalists are relatively homogenous in terms of their key demographics: socio-economic background, educational level, age and ethnicity (e.g. Weaver 1998; Deuze and Paulussen 2002). In the West, the majority of journalists hold a degree, come predominantly from the established and dominant cultural group in society, and are typically male and between 25-44 years-old (Weaver 2005).

Meanwhile, the literature on journalists' role perceptions indicates similarities and differences in the ways journalists describe and understand their work. In a well-known study on role perceptions, Janowitz (1975) identified the *advocate* and *gatekeeper*. The *advocate* believes their major task is to act on behalf of the audience, and consequently they select news according to its use-value for the social groups they choose to support. Conversely, the *gatekeeper* assumes the audience is more autonomous and able to pursue their own needs, and thus selects news "exclusively according to their own professional criteria, such as the perceived news value" (Donsbach 2008). In a survey of American journalists, Johnstone et al. (1976) identified similar roles in the *neutral* and *participant*. Hanitzsch (2007) suggests these roles exist on a pole of interventionism, which ranges from passive (low) to interventionist (high). The *neutral*, along with the *gatekeeper* tend to merely perceive themselves as "disinterested transmitters of the news", emphasizing neutrality and impartiality in reporting (Hanitzsch 2007: 372). The *advocate* and *participant*, by contrast, adopt a more "active and assertive role in reporting", often acting on the behalf of a particular social group to advocate for, or promote change (Hanitzsch 2007: 373). This latter role underpins some of the thinking behind the public journalism movement of the early 90s, and also, the genres of peace journalism and development journalism.

Within the role conceptions literature, other scholars have tried, from a normative standpoint; to describe the social tasks that journalism should perform. Patterson (1995) highlighted four different roles, the signaler, common carrier, watchdog, and public representative. The signaler acts as an early warning system for society, while as the *common carrier*, journalists act as vessels of information between the government and the public. As watchdogs, journalists keep tabs on institutions, drawing attention to wrongdoing. And finally, as a *public representative*, journalists are spokespersons for the public and representatives of public opinion. These role conceptions carry different normative expectations about the relationship between journalists and the government. The *public representative* and *common carrier* are arguably more deferent to authority figures, accepting information provided by government sources as authoritative, legitimate and credible (Hanitzsch 2007). The signaler and *watchdog*, on the other hand, adopt a more adversarial role: they are more critical of social institutions and generally more skeptical of authority figures. In democratic countries such as the UK, Canada, Finland and the US, investigating government claims (an adversarial role) has traditionally been understood as extremely important (Weaver 2005).

More recent empirical research has highlighted that there might be a difference in the ways traditional and online journalists think of their roles. In interviews with digital journalists, Ferrucci and Vos (2017) found less support for the adversarial role, with digital journalists "defining their profession in terms of the close connection with a citizen-audience" (Ferrucci and Vos 2017: 880). They link this more with the *populist-mobilizer* role (Weaver et al. 2007), which is concerned with empowering the public with information so they can actively participate in public life. These findings, however, conflict with Agarwal and Barthel's (2015) research, which found strong support for the *adversarial* role among online journalists. Hence, the roles of online journalists remain very much under construction.

Hanitzsch (2007) adds an important third dimension to the role of journalists, which concerns their relationship with the public. He labels this "market orientation". In journalism cultures where market orientation is high, there is an emphasis on providing news that champions the values of consumerism, focusing on people's everyday life issues and individual needs:

"Audiences are not addressed in their role as citizens concerned with the social and political issues of the day but in their role as clients and consumers whose personal fears, aspirations, attitudes, and emotional experiences become the center of attention" (Hanitzsch 2007: 375)

There is a strong argument to be made that since the 1970s, journalism has become more explicitly market oriented, treating people less like citizens and more like consumers (e.g. McManus 1994; Nadler 2016). Again though, much like the other roles described, the market orientation of different journalistic cultures is highly contextual, and based on several interlinked factors. Journalists therefore identify, to varying degrees, with different role conceptions and often believe they perform more than one role at the same time (Agarwal and Barthel 2015). This helps explains why there remains a lack of universal consensus on the different roles of journalism, both from a material perspective (what journalism 'is') and from a normative perspective (what journalism's main tasks should be). Nonetheless, they have remained an important part of how journalists understand and give meaning to their work (Ferrucci and Vos 2017).

2.3. Professionalism, Ideology and Field Theory

Whilst journalism does not necessarily possess many of the archetypal "traits" of a profession, over the course of the 20th century, journalism certainly managed to make a strong claim to professional *status* (Tumber and Prentoulis 2005). As discussed above, important factors in this process included the emergence of journalism as a separate field of practice, and then following this, the gradual creation of ethical codes, norms, routines, and organisational structures to manage the production of news (Dooley 1997). For Deuze (2005: 444), the professionalization of journalism during the 20th century is typified by the "consolidation of a consensual occupational ideology amongst journalists in different parts of the world". This ideology, according to Deuze (2005), can be categorized as a set of five discursively constructed ideal-typical values: *public service, objectivity, immediacy, autonomy* and *ethics* (Deuze 2005: 447). Deuze (2005) describes this ideology as the "cultural cement" of journalism, meaning they form the foundation of journalism's identity (Hanitzsch 2007).

Connecting this with Bourdieu's field theory, it is possible to make the case that professionalism, and its accompanying ideology, was integral to the formation of the contemporary *doxa* of the journalistic field. As a reminder, the *doxa* of the journalistic field refers to "a system of presuppositions inherent in membership in the field" (Bourdieu 1998: 37). As Hellmueller et al. (2013: 2) note, "these shared presuppositions – a shared understanding of institutional roles, epistemologies, and ethical ideologies (Hanitzsch, 2007) – constitute the cultural capital of the field". Journalism's professional ideology, which includes assumptions about institutional roles (providing a public service), epistemologies (objectivity, empiricism) and ethical ideologies (a commitment to truth and sense of fairness) thus form an essential part of journalism's primary source of legitimacy, autonomy and power (Vos 2016), and is what makes it "autonomous or distinct from other fields" (Vos and Craft 2016: 1507).

Over the course of the 20th century, journalism's cultural capital remained fairly consistent amidst constant pressure from various endogenous and exogenous forces (Champagne and Marchetti 2005). This, along with several important structural factors, helped journalism effectively establish a monopoly over the system of news provision. As Lewis (2012: 838) writes:

"For much of the twentieth century, both the business model and the professional routines of journalism in developed nations were highly stable and successful enterprises because they took advantage of scarcity, exclusivity, and control. In the local information market, news media dominated the means of media production, access to expert source material, and distribution to wide audiences – which translated to tremendous capital, both in gatekeeping authority (Shoemaker & Vos 2009) and economic power".

Journalists benefited from this exclusivity and control, both materially (e.g. income, access to sources, audiences, and legal rights), and symbolically (e.g. social prestige, credibility, and authority). Only very recently has this exclusivity once again come under serious threat.

2.4. The Digitization of the Field

As a result of media digitization, information is "no longer scarce, hard to produce, nor difficult to repurpose and share" (Lewis 2012: 4). Citizens also no longer consume information in a linear way, but rather, "assemble information associatively by interacting with it online" (Peters and Broersma 2013: 4). A great deal of this information is "free" because of the huge reductions in costs of transaction and distribution online. These changes have upended the traditional business model of journalism, which relied on maintaining exclusivity, scarcity and control in the information market (Lewis 2012). Moreover, the new environment has given rise to a large number of new information providers, many of which seek to obtain a position within the journalistic field by openly challenging established norms (Broersma 2010). As

a result of these changes, many scholars and critics argue that journalism's "occupational monopoly on telling the news" is diminishing as journalists' exclusivity over different aspects of news production and distribution weakens (Anderson 2008: 248).

Exclusivity lies at the heart of any successful claim to authority and status made by an occupation. As Carlson (2007: 277) notes, "authority must be exclusive in order for journalists to separate their work from other modes of public information". This exclusivity, ultimately depends on journalists' ability to maintain control, "rhetorically and materially, over a body of knowledge" (expertise), not claimed by any other occupation (Lewis 2012: 5). In a Bourdieusian sense, this "body of knowledge" can be thought of as the field's cultural capital, which has long been predicated on journalists' maintaining a significant degree of *control* over news content (Lewis 2012). Digital media, by contrast, carry affordances that tend to privilege values such as transparency, participation, and interactivity. These changes, at least in theory, provide ordinary users with more power when it comes to creating, filtering and distributing information, blurring the lines between users and producers, amateurs and professionals, and news and other forms of communication.

It is for this reason that many have argued journalism is undergoing a period of "deprofessionalization" and beginning to lose its distinctiveness as a field of practice (Ryfe 2012). Yet at this stage it might be too early to suggest with such any degree of certainty that journalism is fading to irrelevance. As Bourdieu (1982: 470) argues, "every field is the site of more or less overt struggle over the definition of the legitimate principles of the division of the field", and in this respect, much like in response to television and radio, the cultural capital of journalism might be *shifting*, rather than disappearing altogether in the digital environment (Hellmueller et al. 2013; Zelizer 2015). The question of *how* it is shifting though, is key. For example, how are individual journalists and news organisations attempting to adapt to the new environment? What new norms, values and practices are gaining currency? How do these changes in journalistic capital help or undermine the field as it tries to protect its authority and preserve its public legitimacy? And finally, what, if anything, remains the

same? Some of the research conducted on online journalism has attempted to address these questions and this becomes the focus of the next section.

3. Literature Review

Changes to the field over the last two decades have been "substantive, pervasive and multifaceted" (Singer 2014: 2), and the wealth of academic literature on digital journalism reflects this. Of the many transformations that journalism has experienced, one of the most consequential has been the altering nature of news production and changes to the actual work of journalists (Picard 2014). A recurrent theme in this work has been on the subject of "convergence"; a somewhat vague term used to describe the "blurring of the limits between different media, professional skills and roles" (Domingo 2006: 3). Trends towards convergence have been enormously varied and highly dependent on different the contextual factors that shape technological adoption (e.g. Boczkowski 2004; Boczkowski and Ferris 2005; Klinenberg 2005; Colson and Heinderyckx 2008; Singer 2008).

First and foremost, scholars have been interested in how journalists' work has been affected by what are seen to be the three core technological characteristics of the Internet: hypertext, interactivity, and multimedia (Dahlgren 1996; Deuze and Paulussen 2002; Deuze 2003; Domingo 2006). Findings commonly suggest that the Internet has created new pressures on journalists, as they are required to incorporate more tasks into their daily work and master new storytelling techniques, which undermines their ability to perform their normal functions (Bromley 1997). In addition, new pressures on speed and immediacy in reporting have been seen to lead to a general rise in the circulation of non-original material, either copied verbatim from news wire services and press releases, or "cannibalized" from other news websites, as journalists struggle to cope with the demands of publishing more stories perday (Klinenberg 2005; Davies 2008; Fenton 2010; Bakker 2012; Philips 2012). As a result of these pressures, scholars have observed that online journalism is increasingly desk-bound, with being provided less time to go out in the field and conduct original newsgathering (O'Sullivan and Heinonen 2008; Fenton 2010).

Research has shown that processes of convergence have generally created the most tensions among print journalists, and at news organizations with deepseated "traditional" cultures. Even as journalists have begun to adopt new technologies such as Twitter into their working routines, and engage in new practices like interacting with social media users, Lasorsa et al. (2012) has observed that the trend is toward "normalization", whereby new technologies and processes are coerced to fit existing organisational routines and professional norms (Singer 2005; Domingo et al. 2008; Singer et al. 2011; Hermida 2011). This is also true of sourcing practices, with research indicating a general reluctance among journalists to give up on traditional news sourcing techniques (Lecheler and Kruikemeier 2015). While the use of online sources, including those from social media, is common, they are seen to generally play a supportive role, with mainly "offline", elite sources remaining centre stage (Machill and Beiler 2009; Knight 2012; Van Leuven et al. 2014). This has challenged early assumptions that the Internet would necessarily expand the public sphere by increasing the presence of non-elite sources in news coverage (see Cottle 2000; Borger et al. 2013).

3.1. Gatekeeping Research

Beyond the news production process, the term convergence also refers to the technological structure of the internet, with journalists operating in a "converged" media space alongside other online users, who also have the means to produce, filter, reshape and distribute content. According to Quandt and Singer (2009: 132), this "broadening of the media space through user and community participation represents a form of convergence that is likely to be an even greater challenge to journalists than the one posed by the need to master new tools and techniques". In research examining the changing relationship between journalists and audiences, scholars have often invoked notions of the journalist as "gatekeeper": a term first introduced to journalism studies by White (1950) to describe journalists' subjective power in the process of deciding what news and information should be disseminated to the public. The theory of gatekeeping has been revitalized over the past decade or so, as the jurisdictional claim by journalists to decide news for the public has been

seen to diminish. As Deuze (2005: 451) writes, "new media technologies challenge one of the most fundamental 'truths' in journalism, namely: the professional journalist is the one who determines what publics see, hear and read about the world" (2014: 13) In research on this topic, an important underlying question has been what level of gatekeeping control journalists are willing to relinquish over news content, in order to foster greater levels of participation from audiences in the news process (Lewis 2012).

What studies in this area have indicated so far is a general unwillingness on behalf of journalists to give up on established practices of publishing, filtering and selecting information (Robinson 2007; Domingo et al. 2008; Thurman 2008; Hermida et al. 2011). In this regard, Lewis (2012: 836) argues that journalists remain "caught in the professional impulse toward one-way publishing control". This also helps to explain why user-generated content (UGC) has so far only occupied a marginal space in news coverage (Domingo et al. 2008). Such behavior is thought to be the result of journalists wanting to maintain control over familiar processes of newsgathering, in the fear that they might otherwise lose some of their public legitimacy and cultural authority (Lowrey 2011). There are also concerns regarding the accuracy and quality of UGC, and the ability of journalists to verify this content (Carlsson and Nilsson 2015). When user contributions are used, it is said to be mainly in the context of breaking news and "soft" news stories (Usher 2017). Similarly, when users are invited to directly participate in the news process, scholars have noted how this is usually in "token" form, for example, via "most-read" lists, "have-yoursay" invitations, audience polls, "comments" sections or "send-in-your-pictures" requests (Peters and Witschge 2015). The current absence of more meaningful forms of participation has raised questions about the democratising potential of the internet and also highlighted the continuing power of journalists as gatekeepers. Usher (2017) argues that when it comes to selecting news content and deciding on the contributions of ordinary users, professional journalists remain overwhelmingly in control.

Nevertheless, other research has argued that the audience-journalist relationship might be slowly changing, and with it, the gatekeeping function of

journalism (Gillmor 2004; Lowrey, 2006; Singer 2006; Bruns 2008). This is partly attributed to the growing importance of social media in journalism, which has become an important channel of news distribution and also a new tool for journalists to engage with the audience day-to-day in their work (Hermida 2013). Distribution in these social networks relies heavily on users sharing content within their own personal networks, and in this respect, Singer (2013) has argued that Internet users have acquired more power in determining what news is circulated in wider society. Singer (ibid.) argues that the audience now effectively function as secondary-gatekeepers, who choose news "not only for their own consumption but also for the consumption of others, including those within their personal circle of acquaintances and those who are part of an undifferentiated online public" (Singer 2013: 22). She argues this shift toward "user-generated visibility" suggests a "new way of looking at one of the oldest conceptualizations of the journalist's role in our society", namely, that of the gatekeeper (Singer 2013: 22).

For Bruns (2008) the gatekeeping function of journalists is shifting to a new practice of "gatewatching": "a form of reporting and commenting on the news which does not operate from a position of authority...but works by harnessing the collective intelligence and knowledge of dedicated communities to filter the newsflow and to highlight and debate salient topics of importance to the community" (Bruns 2008: 176-77). What these debates suggest is that audiences might slowly be reconfiguring the autonomy and authority of journalists, which might eventually lead to a more collaborative relationship between journalists and their publics. At this stage, however, it might be too early to conclude this is happening. Anderson (2011b: 564) is equally cautious, arguing that the present rhetoric around audience empowerment and participation, at least in newsrooms, often takes place in the context of audience analytics and metrics.

According to Tandoc (2014), it is in the context of economic instability and shrinking audiences, that news organisations are increasingly turning to web analytics to understand the preferences of their audiences, with the aim of increasing traffic to their websites and earning more revenue from advertising.

31

Those optimistic about such changes, like Usher (2010: 1), have argued that analytics enable journalists to "figure out who their audiences are, learn what they want, and in real time, track their behaviours in order to be more responsive to their needs". However, others such as Anderson (2011b), Tandoc and Thomas (2015), and Cohen (2015), have expressed concern that the growing presence of web analytics in newsrooms is influencing news selection in ways that undermine traditional editorial judgement and journalistic values of autonomy. In his study of the Philadelphia news website, Philly.com, Anderson (2011b: 561) observed that "website traffic often appeared to be primary ingredient in Philly.com news judgement", resulting in sports, gossip, and human-interest stories being selected, sometimes at the expense of more serious and complex news stories. Similarly, in a study of three online newsrooms, Tandoc (2014) noticed news stories and headlines were often being selected according to a "consumer-driven logic", with audience metrics playing a key role in influencing editorial decisions. The debate over the use of metrics in newsrooms captures a central tension concerning journalism and its role in society. In their traditional roles, journalists select news based on their own professional judgement (a trustee model). The growing influence of web metrics, however, challenges this position, and encourages journalists to select news based on what audiences want (a market model) (see Welbers et al. 2016).

3.2. Objectivity, Emotionality, and Transparency

Aside from the research on gatekeeping, a significant amount of attention has been directed towards studying the journalistic norm of objectivity. Many scholars have questioned the continuing relevance of objectivity in the digital environment, arguing that a paradigmatic shift towards transparency might be necessary (Karlsson 2011). The reasons for this are multifaceted, but essentially stem from criticisms of objectivity as a means of truth-telling. While objectivity is often associated with impartiality and detachment, transparency is linked with greater openness and accountability (Hellmeuller et al. 2013). A shift to transparency would mean journalists being more open about different news processes, by "embedding in the news reports a sense of how the story came to be and why it was presented as it was" (Kovach and Rosenstiel 2001: 83). Singer (2008: 74) links transparency with the transition to a networked digital environment, arguing that detachment, long associated with the objectivity norm, is "deeply isolating" when one inhabits "a world of fluid and interconnected information". Some also suggest that transparency would help journalists establish a more mutually beneficial relationship with their audiences, revitalising their capacity to act as an "authoritative source of information" (Karlsson 2011: 292). It has also been suggested that transparency would allow journalists to cope better with the increased pressures of immediacy in online news, whereby audiences would be more tolerant of errors or unverifiable information in rolling news if journalists became more open about their use of sources (Karlsson 2010; Phillips 2010).

In empirical research, scholars have found some support for transparency, though it remains poorly defined. Studies that have analysed news content, for example, have cited features such as external links, time stamps, invitations to participate (e.g. commenting), author information, and an acknowledgement of errors, as signs of increased transparency in news reporting (Karlsson 2010). Other researchers like Robinson (2007) have conducted interviews with journalists to show how an "unfiltered look at the news" can build trust and credibility among audiences. In recent research, Vos and Craft (2016) examined the discursive construction of transparency in US trade journalism publications, finding an increasing level of support for transparency among journalists and a waning level of support for the traditional norm of objectivity. That being said, they also drew attention to one criticism of transparency, which is that it is sometimes associated with journalists being overly deferential towards audiences. A similar criticism has come from Lowrey and Anderson (2005), and also Robinson (2007), who have suggested that allowing "readers" to peek past the institutional curtain to see the working parts of newsgathering", might result in journalism inadvertently diluting its own authority. Vos and Craft (2016: 13) are more upbeat in their conclusions, arguing that "transparency is a form of cultural capital whereby the field foregoes a measure of autonomy to gain legitimacy". While the idea of transparency is still relatively new in

traditional journalism, it is a fairly widespread practice in blogging (Thorson 2008) and also seems to be a common norm among emerging news providers, including digital natives (Riordan 2014).

Alongside transparency, a second major challenge to the professional norm of objectivity is seen to come from the rise of more emotional and personalised forms of storytelling in news (Steensen 2016; Wahl-Jorgensen 2016). Similar to transparency, the context for this trend has, in part, been precipitated by a shift to a "networked" media system. In this environment, traditional journalistic epistemologies that favour objectivity and distancing have been challenged by the rise of more personal technologies (e.g. mobile phones) and more personal forms of communication (e.g. social media) (see Castells 2007; Kormelink and Meijer 2014; Wahl-Jorgensen 2014; Beckett and Deuze 2016). While it is important to emphasise that the use of emotion has long been recognised as central to component of storytelling in news, it has often been associated with a decline in the standards of journalism and thus heavily criticised and neglected as an object of study (Bird 2009; Wahl Jorgensen 2013). However, in the online space, it has been noted that "many of the traditional dichotomies associated with journalism, such as hard/soft. fact/opinion and information/entertainment are becoming progressively blurred" (Peters 2011: 298). Peters (2011: 301) connects this trend with an "upsurge in news alternatives" and "an increased variety of "valid" news styles", many of which set out to challenge established conventions of objectivity by shunning journalistic neutrality and detachment in favour of generating a deeper emotional connection with readers.

Though empirical research is still in short supply, some research studying the content of news stories has highlighted the important role that emotion can play in: deepening citizenship and engagement (Wahl-Jorgensen 2016); reducing the distance between journalists and audiences (Steensen 2016); and fostering emotional "counterpublic spheres" that blend public/private and political/public issues and concerns (Papacharissi 2016; Zou 2018). These various roles are seen as particularly crucial in the digital environment, where news organisations have to battle harder to attract and maintain users' attention and 34

foster engagement (Singer 2013; Beckett and Deuze 2016). This also points to one issue concerning the rise of emotion in journalism, which is that the types of stories that are selected to drive emotional engagement are often seen to be provocative, attention-grabbing, and salacious, and bear little or no relation to issues of general public concern (Tandoc 2014; van Dijck and Poell 2013). However, in the absence of more research, it remains somewhat of an open question whether greater use of emotion in journalism is a normatively desirable development in the field.

3.3. Summary

The preceding literature review has attempted to sketch some of the major ways in which the cultural capital of journalism might be shifting in response to new media. What should be clear from the discussion that the news industry has undergone a period of profound and deeply unsettling change, leading to rapid changes in news work and "straining how journalists think about the field and its core values" (Ferrucci and Vos 2017: 869). New pressures on speed and immediacy in reporting have given rise to a number of practices that are said to undermine journalism's core functions. Meanwhile, new technologies and platforms have had only limited impact in newsrooms, with most being coopted to fit old norms and practices (Singer 2005). This is particularly evident in the research exploring online sourcing techniques, which points to the continuing importance of elite, 'offline' sources in coverage, despite the new possibilities afforded to journalists to include a "different cast of voices" in news (Harcup 2003: 360).

Along with sourcing techniques, other research has shown that journalists have continually struggled to break out of "the professional impulse toward one-way publishing control" (Lewis 2012: 836). This is most clear in the body of work examining the relationship between journalists and audiences, which has shown a clear reluctance on the behalf of journalists to "open up" the news process to ordinary users. This resistance is somewhat understandable, given the fact that much of journalism's professional legitimacy and sense of autonomy is premised on a significant degree of control over news content. Perhaps the area where the audience has made the most impact is via the use of analytics in the newsroom, which has raised concerns about journalists ability to independently select the news (e.g. Anderson 2011b; Tandoc 2014; Welbers et al. 2016).

In spite of this evidence, other research has indicated that a "slow philosophical shifting" (Robinson 2010: 140) might be occurring in journalism, which could eventually lead to a revised logic for the field:

"[A revised logic] that preserves certain ethical practices and boundaries that lend legitimacy, abandons jurisdictional claims that have lost their currency in the new environment, and embraces fresh values, such as open participation, that are more compatible with the logic of digital media and culture" (Lewis 2012: 852).

Proof of such changes is perhaps most evident in the debates over objectivity with scholars indicating increased support for transparency as a new norm in journalism (Vos and Craft 2016). The nascent work on emotion in journalism also forms part of this questioning of the "meta-narrative of objectivity" (Peters 2011). One important and unanswered question in this debate is whether such changes would lead to a *reconstitution* or *dilution* journalism's cultural authority (Anderson 2008), which, at this stage in online journalism's development, is difficult to answer.

While many questions loom large about the shifting state of journalism's cultural capital, one major issue with current research is its overwhelming focus on change at traditional news organisations. While this work has undoubtedly been valuable, such a narrow focus has largely failed to adequately capture the highly diverse and dispersed ways in which news is now produced, distributed and consumed online. At a time when journalism remains in a state of flux (Ryfe 2012), new actors are entering the fore in a bid to definite the legitimate principles of the field. Many of these new entrants, including digital natives, are less socialized to the traditional "rules of the game", and thus come equipped
with different forms of cultural capital that have the potential to disrupt established ways of "doing" journalism (Vos 2016).

As Carlson and Usher (2016: 564) put it, these new players are "compet[ing] to define what digital news looks like, [trying to] reestablish the boundaries of journalism, and determine strategies for legitimating news content". Studying these organisations – including *BuzzFeed* and *Vice* – is important because it has the potential to provide us with a fresh perspective on how news, and the various activities and discourses surrounding its production, is evolving online. This, in turn, might go some way to explaining what *new* boundaries journalists and news organisations are attempting to draw, in a bid to enhance and protect their seemingly vulnerable authority.

4. Research Design and Methods

To examine news production at *BuzzFeed* and *Vice*, this project adopts the case study as a research strategy. According to Eisenhardt's (1989: 534) oftquoted definition, "the case study is a research strategy which focuses on understanding the dynamics present within single settings". For this reason, it has been widely used in studies of organisations, where researchers have been interested in examining how phenomena occur within a bounded context (Hartley 2004). Case studies can involve single or multiple cases, though fewer cases are generally seen as preferable, given their propensity to generate enormous volumes of data (Gomm et al. 2000). This is partly because case studies typically combine multiple methods of data collection, which can be "either qualitative, quantitative, or both" (Hartley 2004: 324). As such, researchers adopting a case study strategy have used methods as diverse as questionnaires, interviews, participant observation, and archival research to examine complex social phenomena within specific social or organisational settings (Stake 2003). This includes more recent research into digital news production (e.g. Boczkowski 2004; Painter 2008; Tandoc 2013; Hermida et al. 2014; Brooks 2016).

Case studies, however, are not so much defined by their research methods as their *theoretical orientation* (Hartley 2004). As Yin (1994: 27) argues, "theory development prior to the collection of any case study data is an essential step in doing case studies". This, he suggests, is one point of difference from related methods like ethnography and 'grounded theory', both of which typically "avoid specifying any theoretical propositions at the outset of an inquiry" (Yin 1994: 27). This particular project has used Bourdieu's field theory as a theoretical framework⁵, as well as borrowing key concepts from the literature on journalistic professionalism. For Hartley (2004), the presence of a theoretical framework in

⁵ Though scholars have rarely made the connection, the reflexive sociology of Pierre Bourdieu would seem particularly suited to case study research, given their overlapping aims. Case study research "places emphasis on understanding processes alongside their (organizational and other) contexts" (Hartley 2004: 324), while field theory is concerned with "bridg[ing] the epistemological divide between agent and structure and between micro and macro" (Willig 2012: 381).

case study research helps enhance the internal validity and generalisability of the results. This also draws attention to a major strength of case study research, which is that it is concerned with generating concepts and theories that have fidelity to reality (Yin 1994; Flyvbjerg 2006). For this reason, Eisenhardt (1989: 548-49) argues that case study research is "particularly wellsuited to new research areas or research areas for which existing theory seems inadequate". Hence, it is particularly relevant to this research project, which is interested in providing a fresh perspective on media change by advancing our understanding of digital native news organisations.

The remainder of this chapter is divided into sections that roughly correspond with Eisendhardt's (1989: 533) procedure for doing case study research. The first section (4.1) deals with my selection of cases, and includes a brief overview of *BuzzFeed* and *Vice* as companies. The second section (4.2) discusses the methodological approach and research design used in this study, as well as issues surrounding access. The third section (4.3) describes the interview method used in this project, including the process of data collection and analysis. The fourth section (4.4) explains the quantitative content analysis used in the study, including the key decisions made about which variables to investigate. The fifth and final section (4.5) addresses some of the limitations of this research.

4.1. Case Study Selection

According to Eisenhardt (1989: 537), the concept of a population is crucial when selecting cases, because the selection of an appropriate population: 1) "defines the set of entities from which the research sample is to be drawn", and, 2) "controls extraneous variation and helps to define the limits for generalising the findings". In this study, the population is defined as digital native news organisations. Digital natives are generally understood as media organisations founded in the internet-era, that produce news and other content for a predominantly online audience. It is difficult to determine precisely how many digital native news organisations now exist, but at least in the US and European

context there appear to be several hundred (Jurkowitz 2014; Nicholls et al. 2017).

There is huge variation between digital native news organisations in terms of size, target audience, the topics and issues they cover, and the ways in which they choose to deliver and present news. In a Pew study that analysed 438 different digital native news organisations in the US, Jurkowitz (2014) found that over half had a focus on producing local news, and that nearly three-quarters employed three or fewer full-time editorial staff. At the other end of the spectrum, there are large US-based digital media companies like *BuzzFeed*, *The Huffington Post*, and *Vice*, which employ more than 100 full-time editorial staff and tend to focus on producing national and international news stories for a global audience (Jurkowitz 2014)⁶.

To limit my population, I chose to focus on US-based digital native news organisations, and specifically, those residing in the sub-field of large-scale or 'mass' production⁷. I was initially drawn to this subset of organisations because of the wider impact they appeared to be having upon the field of journalism. At a time when notions of "crisis" permeated much of the discourse about journalism (Zelizer 2015), digital natives like *The Huffington Post, BuzzFeed, Vice, Mashable, Quartz,* and *Vox,* appeared to be on the rise, growing in terms of size, audience, and revenue. However, while these organisations were generating plenty of excitement *within* the industry, they were virtually absent from academic discussions about journalism. Instead, scholars seemed generally more interested in theorising about digital journalism from the perspective of traditional organisations⁸.

⁶ To my knowledge, this is still the most extensive survey of North American digital natives to date.

⁷ In Bourdesian terms, the sub-field of large-scale production or mass production is "oriented towards the making of 'commercial' cultural goods" for large audiences (Hesmondhalgh 2006: 214). This stands in contrast to the subfield of small-scale production, which tends to be more autonomous and oriented towards the production of 'pure' artistic products" for smaller, or more niche audiences (Hesmondhalgh 2006: 214).

⁸ It should be noted that this parochialism is a longstanding issue in journalism studies (Zelizer 2009).

To address this gap in research, I decided to study news production at two digital natives. While a larger study might have afforded me more resources to study multiple organisations, for practical and logistical reasons, I chose to limit my study to two cases: *BuzzFeed* and *Vice*⁹. It is important to provide a very brief overview of both organisations, before moving on to discuss the research methods used in this study.

4.1.1. Vice

Vice, which initially began as a news and culture magazine in the 1990s in Canada, has grown to become an expansive global media company with a valuation of \$5.7 billion (Garrahan 2017). Vice News is a relatively recent addition to the Vice Media network, having only been founded in December 2013. The Vice News channel is one of 11 different digital channels that Vice own, and chiefly presents news in text and video format. *Vice News* is perhaps most famous for its video content, which is typically shot in a "lo-fi" documentary style, is presented by a revolving cast of young, multicultural reporters, and has explored topics as diverse as: life in North Korea (2011); the rise of the Islamic State (ISIS) in Syria and Iraq (2014); the sex industry in Bangladesh (2015); and the state of the mental health industry in America (2015); and the Ukrainian civil war (2016). Vice are seen to focus on reporting stories or issues that they deem to have been ignored by mainstream news organisations, which sometimes courts controversy (see Widdicombe 2013). Many of these videos have garnered millions of hits. For example, Vice's 2014 documentary on the Islamic State has received nearly 13.2 million views on YouTube at the time of writing (Vice 2018). What would appear to make *Vice News* distinct from other news providers is its presentation style, which openly challenges the journalistic norm of impartiality to present news in a more subjective and involved way (Riordan 2014).

⁹ Note: in case study research, "random selection [of cases] is neither necessary, nor even preferable" (Eisenhardt 1989: 537).

This aesthetic has proven particularly popular with younger audiences, who are *Vice's* target market (Ip 2015a). Precisely why *Vice* content seems to resonate with young audiences is difficult to determine, but Costera Meijer (2007; 2013), who has conducted several studies on the news tastes of young people, has found they enjoy the experience of feeling "closer" to an event, and also tend to dislike the presentation style of conventional news. This in part, helps to explain the popularity of *Vice News*, however, another crucial factor relates to the fact they appear less burdened by the overheads, bureaucracy, regulations, and procedures of a legacy media organisation. This has enabled *Vice News* to expand its operations quickly, as well as continuously experiment with different ways of delivering content to users on new and emerging social platforms (See Küng 2015; Painter et al. 2017).

4.1.2. BuzzFeed

Buzzfeed is different to *Vice* in the sense that its business has always thrived on the use of social media platforms to promote content¹⁰. Initially, the company rose to prominence through the mass publication of 'listicles' (e.g. "21 signs you're overly empathetic"), and quizzes (e.g. "Which pop star are you based on your Zodiac sign"), both of which still form a key part of the organisation's output. However, more recently, Buzzfeed has started to invest more seriously in news and journalism. The motivation for establishing a news service, according to Jonah Perretti, is that *Buzzfeed* has "worked out that news content, not just lists, can be shareable and viral" (cited in Riordan 2014: 48). While such a statement risks trivialising Buzzfeed's ambitions to become a 'serious' news provider, it belies the seriousness of BuzzFeed's approach in understanding *why* content is shared. As one of the few researchers to examine *BuzzFeed's* operation in detail, Küng (2015) is worth quoting here at length:

"At heart, BuzzFeed is about data science, about analysing user data to decode how and why content is shared and distributed. User data are

¹⁰ One article published by *BuzzFeed* estimated that 75% of its total website traffic was derived from social media platforms (Isaac 2014).

captured, analysed and manipulated in a perpetual loop of analysis, interpretation, experimentation, feedback, and refinement. BuzzFeed's goal is to identify which characteristics have a predictive relationship with virality, maximise these and thus accelerate the 'spread rate' of its content" (Küng 2015: 58).

Crucially, *BuzzFeed* takes what it learns from this process and applies it to social advertising campaigns that it runs for an ever-expanding roster of clients. In this respect, news, alongside other, more popular entertainment content, all work in tandem to generate insights about the audience: their consumption habits, basic demographic profiles, tastes, search history etc, which are subsequently sold on to advertisers. Whether this trade-off of free content for data is fair or ethical, is something that has attracted a great deal of debate and criticism (e.g. van Dijck and Nieborg 2009; Fuchs 2011; Freedman 2012).

The news division of *Buzzfeed* has garnered a reputation for focusing on news stories relevant to its young audience, particularly issues related to social justice movements such as LGBT (lesbian, gay, bisexual transsexual) rights, gender rights, and civil rights (Burrell 2014). Similar to *Vice, BuzzFeed* have placed a heavy emphasis on experimentation, with several teams constantly exploring new ways of presenting and sharing content using different media platforms (Lafrance 2012). Also like *Vice*, the company has expanded internationally at a fairly rapid rate, and now has editorial operations in countries such as Australia, Spain, Mexico, Brazil, India, and Germany, as well as New York-based BuzzFeed Español, which targets a Latin-American audience (Painter et al. 2017). A common strategy pursued in each of these countries is to start by creating cheap, highly shareable, "viral" content, and then use the revenue earned from this to fund more serious news production (Lichterman 2015).

4.2. Research Design

As mentioned, one feature of case study research is that it typically relies on using a combination of methods (Yin 1994; Gomm et al. 2000). Initially,

ethnography and content analysis were determined as the best approach for answering the research questions. It was via these methods that I hoped to observe *BuzzFeed* and *Vice* journalists in their natural settings, comparing their "social meanings and ordinary activities" (Brewer 2000: 10) to the actual news content they were producing. I was also encouraged by the early ethnographies of news production, which, as Wahl-Jorgensen (2009: 21) notes, have "contributed tremendously to knowledge about news production processes and newsroom cultures, providing a rich description of journalists' ways of life and work" (e.g. Epstein 1973; Schlesinger 1978; Tuchman 1978; Fishman 1980; Soloski 1989; Gans 2004).

In arranging to conduct ethnography at *BuzzFeed* and *Vice*, I entered myself into a fairly long and protracted negotiation to gain access at their (UK and US) sites of news production. This process, which began in around July 2015, lasted approximately nine months and ultimately ended without success. I was invited by the Editor-in-Chief of *BuzzFeed UK*, Janine Gibson, to spend one day in the London office of *BuzzFeed* to conduct some non-participant observation. However, I was unable to secure long-term access to either organisation. My repeated requests to gain access were either ignored or rejected by the people I had identified as the key 'gatekeepers'¹¹ (see Appendix A for an example email requesting access). Based on the experiences of previous researchers, I had expected I might encounter this issue (e.g. Schlesinger 1980; Tuchman 2002; Puijk 2008; Usher 2014; Ryfe 2016). According to Lindlof (1995), the problem tends to be more acute at commercial organisations, which are less publically accountable, and like to "keep up their guards very high, perceiving proposed research as an intrusion into the proprietary nature of their activities" (Lindlof 1995: 107). A further hindrance to the process of gaining access might have been my own professional inexperience in the field of journalism. As Paterson and Zoellner (2010: 97) note, prior professional experience in the field of study is "increasingly becoming an essential criterion in gaining access for long-term ethnographic investigations".

¹¹ By 'gatekeepers', I am referring to the key people within both organisations who I determined had the power to to grant me access on behalf of everyone to the key sites of news production at *BuzzFeed* and *Vice* (see Lindlof 1995: 106).

4.2.1. Mixed Methods

Taking these challenges into account, I was able to build significant contingency into the research plan. To study news production at *BuzzFeed* and *Vice*, I decided upon a mixed methods research design. In their extensive synthesis of the literature on mixed methods, Johnson et al. (2007: 123) provide the following definition:

"Mixed methods research is the type of research in which a researcher or team of researchers combines elements of qualitative and quantitative research approaches (e.g., use of qualitative and quantitative viewpoints, data collection, analysis, inference techniques) for the broad purposes of breadth and depth of understanding and corroboration".

Mixed methods is often referred to as a 'third' methodological or research paradigm, and is based on the premise "that the use of quantitative and qualitative approaches in combination provides a better understanding of research problems that either approach alone" (Creswell and Plano Clark 2007: 5). Hence, mixed methods research is defined by a strong emphasis on triangulation, whereby a "combination of methodologies [are used] in the study of the same phenomenon" (Denzin 1978: 291). Morse (1991) describes two types of triangulation: simultaneous or sequential. Simultaneous triangulation occurs when the quantitative and qualitative methods of the study are conducted independently, and only mixed during the phase of interpretation. Sequential triangulation is "when a direct interaction exists between the quantitative and qualitative strands of the study. Through this direct interaction, the two methods are mixed before the final interpretation" (Creswell and Plano Clark 2007: 65). Triangulation, according to Mathison (1988), can produce three different results in findings: convergence, consistency, or contradiction. Each of these outcomes can lead to a superior understanding of the phenomena being studied and improve the general validity of results (Mathison 1988).

Given the well-noted gap between rhetoric and practice in journalism, or between "what journalists say they do" and "what they actually do" (Paterson 2008: 2), there is a strong case to make that triangulation is even more essential when studying news production. In this study, I chose to combine in-depth interviews with a quantitative content analysis of news content. On the interaction between these two methods, this research followed what Creswell and Plano Clark (2007) describe as an "exploratory sequential design". The exploratory design, according to Creswell and Plano Clark (ibid: 86) is a "twophase sequential design that can be recognized because the researcher starts by qualitatively exploring a topic before building to a second, quantitative phase". While in my case, I knew of some of the basic variables I would like to examine in the quantitative phase of the project, I wanted to remain open to incorporating new variables into the content analysis as I progressed through the qualitative phase of my research. As Eisenhardt (1989: 539) argues, "overlapping data analysis with data collection not gives the researcher a head start in the analysis but... [gives them] the freedom to make adjustments during the data collection process". This approach was particularly valuable in light of the fact I was studying new phenomena and therefore wanted to remain flexible during the process of discovery.

In the following two sections, I provide more detail about the two methods used in this project, discussing the rationale behind their selection, as well as the general process of data collection.

4.3. In-depth Interviews

The interview is an established and widely used method of data gathering in social science research (King 2004). According to one classic definition, the interview is a "conversation with a purpose" (Bingham and Moore 1931: 3). Perhaps it would be more accurate to say that the interview is a conversation that can fulfil several purposes. On the subject of qualitative interviews, Lindlof and Taylor (2010: 173) make the case that they are "particularly well suited to *understanding the social actor's experience, knowledge, and worldviews*" (emphasis in original). Or as King (2004: 21) puts it, "the qualitative research

interview is ideally suited to examining topics in which different levels of meaning need to be explored". The interview thus seemed to fit well with my own research aims, which were partly concerned with understanding journalists' interpretation of their own practices and routines. Indeed, one of the strengths of qualitative interviews is that they aspire to understand participants "on their own terms and how they make meaning of their own lives, experiences, and cognitive processes" (Brenner 2006: 357). This endeavour can also generate an enormous wealth of data, particularly in the case of semi-structured or unstructured interviews where the conversation between the interviewer and interviewee is only loosely scripted.

In the case of this research, I chose to pursue semi-structured interviews because of my interest in pursuing "an active, open ended dialogue with interviewees" (Deacon et al. 2007: 67). Given the exploratory nature of my research, it seemed important to adopt an interview procedure that would allow me to be responsive to the situation at hand, and pursue interesting topics as they arose in conversation (Berger 1998). An obvious drawback to this approach was that interviewees could digress on one or more subjects for too long without addressing the core concerns of the research. To help mitigate against this problem, a comprehensive interview guide was developed (see Appendix B). The development of the guide was informed by the broader literature on digital journalism and news production, as well as my own exploratory work into the news cultures of *BuzzFeed* and *Vice*. To investigate the key research questions of the project (see section 1.2), interviewees were asked questions regarding their previous experience and current journalistic role; their organisation's approach to news; their daily routine and working practices; their use of social media in news work; their relationship with the audience; their use of data and metrics in news work; and finally, their thoughts on the role and purpose of journalism in society. Following the advice of King (1994) and others (Deacon et al. 2007), interviews generally started with a brief and polite introduction to the research, with general, non-threatening questions preceding more complex and detailed questions. At the end of each interview, participants were asked if they could recommend any other colleagues for interview. This technique of "snowball" sampling (Yin 2011: 89) proved to be an effective means of recruiting more participants to the study.

Based on an assessment of my email and social media communication, I estimate that during the course of my fieldwork (July 2015 - March 2017 approx.), I attempted to contact more than 90 different staff working for BuzzFeed and Vice in the US and UK (see Appendix C for an example email requesting interview). These staff varied significantly in seniority and experience, but were predominantly drawn from a purposive sample of editors and reporters, rather than technologists (e.g. programmers, project managers) and other technical or support staff working in news production (see Nielsen 2012). The recruitment phase of the project was challenging on a number of fronts. Frequently, journalists ignored my requests for interview. On other occasions, they would decline to participate in the study, either because they were not interested, or because they were concerned about the ramifications of speaking to an outsider about their organisation. This was particularly an issue when trying to contact US-based journalists, who generally seemed more guarded than their UK counterparts¹². In fact, almost no current journalists from Vice US responded to my request for interview. Rather, I had to settle on interviewing mainly former employees, who, on the condition of anonymity, agreed to speak about their work for the company. As a rule, I offered anonymity to all participants in the study. This was to help protect their identities and prevent any potential blowback in case they disclosed any sensitive information during the interview process. The information sheet and participant form I sent to my participants can be found in Appendix D.

Overall, 24 journalists were interviewed for the project: 14 from *BuzzFeed*, 8 from *Vice*, plus 2 freelance journalists who had extensive experience writing for either or both organisations¹³ (see Appendix E). Participants varied in terms of age and experience, ranging from the level of junior reporter to senior editor. For reasons already cited, most of these journalists were based in the UK. At

¹³ Note: this excludes the many conversations I had with various *BuzzFeed* staff during the limited time I was able to conduct non-participant observation.

BuzzFeed, however, a few of the staff I interviewed were on secondment from the US, and therefore possessed detailed knowledge about the US operations of the company. For practical and logistical reasons, 13 interviews took place over the phone. However, I was able to interview nine journalists face-to-face. Interviews lasted anywhere from 18 minutes to 101 minutes, averaging around 55 minutes. This added up to approximately 1303 minutes, or 21.7 hours of interview content in total.

Following each interview, I got in the habit of taking some brief notes on the most interesting points of conversation, including any early themes and issues that seemed to be emerging. This evolved into what grounded theorists commonly describe as a *constant comparative* approach (e.g. Glaser and Strauss 1967; Strauss and Corbin 1990; Charmaz 2006). Emerging data from interviews was continuously analysed for similarities and differences and placed into tentative categories. This process of 'open' coding (Corbin and Stauss 1990: 12) also helped shape ongoing qualitative work by giving me a stronger notion of the specific topics or areas of interest I wanted to pursue in interviews. In the week after conducting an interview, audio recordings were typed up into formal transcripts on *Microsoft Word* and uploaded to the qualitative data analysis software package, *NVivo*. After immersing myself in the data by reading and re-reading the interview transcripts, I conducted an *iterative* analysis of the data (Tracy 2013). An iterative analysis, according to Tracy (2013: 184), is distinct from the grounded theory approach in that it:

"[A]Iternates between emic, or emergent, readings of the data and an etic use of existing models, explanations, and theories. Rather than grounding the meaning solely in the emergent data, an iterative approach also encourages reflection upon the active interests, current literature, granted priorities, and various theories the researcher brings to the data. Iteration is... a reflexive process in which the researcher visits and revisits the data, connects them to emerging insights, and progressively refines his/her focus and understandings".

In the case of this research, the 'etic' component of analysis was derived chiefly from the theoretical framework of Bourdieu as well as the literature on the sociology of journalism. Following the iterative framework of Tracy (2013), I coded the interview transcripts on *NVivo* using a two-step process. During the *primary-cycle coding* phase, interesting words, sentences, and phrases that appeared in interviews were assigned broad, mainly descriptive codes, or 'nodes' to use *NVivo's* terminology. Using a constant comparative approach, codes were continuously created, reviewed, and modified to fit new data. Some of the codes generated at this stage also built on the tentative categories and themes that were developed during the data gathering phase of the project.

Following this first cycle of coding, I engaged in *secondary-cycle coding*. In this process, first-level codes were analysed in conjunction with the broader theory and literature on journalism, and organised, synthesised, and categorized "into interpretive concepts" (Tracy 2013: 194). Also referred to as 'axial' coding (Corbin and Strauss 1990), this procedure was primarily about identifying patterns among open codes, and then grouping these various codes under hierarchical 'umbrella' categories that made conceptual sense (Tracy 2013: 195). These 'umbrella' categories related to key concepts in the literature on journalism including objectivity, audiences, routines, gatekeeping, and transparency. The process of second-cycle coding ended when I decided I had reached a point of *theoretical saturation*, where no additional data could be found to further develop the properties of specific categories (See Glaser and Strauss 1967: 61).

4.4. Quantitative Content Analysis

As outlined in the research design, the second phase of this project utilised a quantitative content analysis for the purposes of testing and generalising findings from the qualitative part of the study. Sociologists have used content analysis to examine media content since the early 20th century (Hansen et al. 1998). To borrow Berelson's (952: 18) oft-quoted description, content analysis "a research technique for the objective, systematic and quantitative description of the manifest content of communication". While this description has been

subject to much criticism, it provides a useful starting point for understanding the broad purpose of content analysis. Building on this, Deacon et al. (2007: 119) suggest that the goal of content analysis is to "quantify salient and manifest features of a large number of texts". It is, therefore, especially useful when one wishes to produce a wide ranging, descriptive insight into the characteristics and form of media content. This was one of my primary objectives in researching *BuzzFeed* and *Vice*, partly for the reason that – as was clear from my literature review – no formal study of their news content had ever been undertaken.

This aside, it is worth recalling that individual news stories are the outcome of a variety of influences, including market pressures, editorial resources, journalist-source interactions, and decisions about presentational style, structure, emphasis, and language (See Riffe et al. 2014: 9). Thus, from the choice of headline in a news article to the decisions made about which actors to quote, studying the "materiality" of news (Carlson 2017) has the potential to provide an alternative glimpse into the routines, practices, and values that shape everyday news production. As Hansen et al. (1998: 91) and others (e.g. Neuendorf 2002; Cottle 2007) have argued, content analysis can be particularly effective when triangulated with evidence and findings from other methods. For Cottle (2007: 5) a general issue in the sociological literature on journalism occurs when "critics of the media make an illicit leap from a critical reading of media content to inferences about the motivations or explanations accounting for this output and, on this basis, quite often 'get it wrong'". This issue of "inference" was helpfully ameliorated by the data I had gathered from interviews, with the responses of journalists allowing me to transition from the 'manifest' meanings of news texts to an analysis of their 'latent' functions (Merton 1949).

At this stage, it would be remiss not to mention that the method of content analysis has come under scrutiny in recent years, with some questioning its appropriateness for studying online content (e.g. Karlsson 2012; Karpf 2016; Widholm 2016). Karlsson and Sjøvaag (2015: 180-181) surmise that content analysis methods on the web are problematic because:

- 1) Data no longer exists prior to the investigation so it has to be collected continuously or in real-time
- Contrary to analogue analyses, *ex post* digital data collection is problematic because digital content is not necessarily preserved in the form it was published
- 3) Digital data collection is neither non-reactive nor unobtrusive. The fact that data processing of digital media content is performed by programming and can be stored on computer hardware puts greater demands on the researcher to be in possession of the tools that enable data protection and processing
- 4) Data structures are not neutral. Unlike analogue formats, digital data can go missing, automatic collection can be interrupted, systems can be hacked and news organisations can erect digital walls around content.

While all four of these points must be taken seriously, I was confident that with a few adjustments, it would be possible to produce a content analysis research design that could overcome each of these issues. Other scholars have made a similar argument, noting that with a few minor adaptations, content analysis can still work effectively in analyses of digital media content (e.g. Weare and Lin 2000; McMillan 2001; Herring 2010). It should also be noted that at the stage of planning my methods, newer approaches to studying online content such as *Big Data Analysis* (Lewis et al. 2013) and *Liquid Content Analysis* (Karlsson 2012) were still in their relative infancy, and thus lacked the established protocols, procedures and rigor of more established methods (see Karlsson and Sjøvaag 2015). In the following sub-sections, I outline the design of my content analysis in more detail.

4.4.1. Design and Rationale

As a reminder, the overarching goal of this project was to produce a holistic description of news production at *BuzzFeed* and *Vice*. Based on the fact no systematic analysis of *BuzzFeed* and *Vice* news content had ever been taken,

it was decided that the focus of the content analysis should be individual news stories published online by both organisations. In addition, two traditional, 'legacy' news organisations were selected as supplementary cases for the content analysis. The two additional cases selected were the online versions of *The Guardian* and *BBC*. Commonly regarded as bastions of 'quality' and professional journalism, I determined that these two legacy organisations would make for an interesting point of comparison with *BuzzFeed* and *Vice*, and help address the issue of how, if at all, news at *BuzzFeed* and *Vice* might be *different* from traditional journalism.

From a broader population of news articles by *BuzzFeed, Vice, The Guardian*, and *BBC*, published between September 2016 and March 2017, two separate weeks of news coverage were selected for study. These individual weeks were picked months apart to decrease the likelihood of the same kinds of stories being repeated across coverage. Given my interest in studying the UK and US news operations of *BuzzFeed* and *Vice*, content was selected from the UK and US websites of all four organisations in the study. To help narrow down the sample, the unit of data collection was limited to the first 10 stories that appeared on the homepage of each website, in order of prominence. The websites of all four organisations in the sample used some form of grid system to assign importance to stories, with the most important story typically being featured at the top, or top-left corner of the homepage, and also being given more physical space (in pixels).

In an effort to preserve news content in the form it was captured, I used the screenshotting tool, *NCapture* (by *NVivo*), to download every individual news article from each website in its entirety (images, video, etc.). A backup screenshotting tool was also used to limit any loss of data. To help "freeze the flow" of online news (Karlsson and Strömbäck 2010), the homepages of each of the websites were loaded in my internet browser at the same time, before being consecutively mined for news content. Overall, the process of capturing ten new articles from eight websites for one week (Monday – Friday) took me approximately one hour per day. For the purposes of consistency, these articles were collected at the same time every day, between approximately 3-4pm

GMT. This time was also selected because it was toward the end of the working day – a period when I anticipated the (UK) websites of each organisation would be changed or updated with less regularity. This did not apply to the US websites of each organisation, which inadvertently highlights one difficulty of doing comparative research in real-time. To highlight one other issue, during the coding procedure, the *Vice News* website underwent a redesign and became subsumed under a single *Vice.com* homepage. For this reason, the second week of content collected from *Vice* contained some cultural content as well as news content.

This procedure, performed over two separate weeks, in October 2016 and March 2017, yielded a total of 800 articles (100 per organisation in each country).

4.4.2. Coding Variables

To design the appropriate general and medium-specific variables for the content analysis, I relied upon a combination of theory, past research, and a "grounded" or "emergent" process of variable identification (see Neuendorf 2002: 95-110). In the first instance, coding variables were developed that corresponded with the very basic features of news stories: their publisher, date of publication, authorial information, headline, word count, and URL. Given both digital natives' reputations for employing a young and diverse editorial staff, it was deemed important to record the gender and ethnicity of journalists for every story.

Beyond this, I sought to design coding variables that would allow me to make inferences about the routines and practices of news workers at *BuzzFeed* and *Vice*. To assess the editorial priorities, division of labour, and availability of resources at both organisations, mutually exclusive variables were developed to examine the topic, format, storytelling style, and sources of news stories. To learn more about the division of news coverage at *BuzzFeed* and *Vice*, an exhaustive list of news topics was developed for the 'news topic' variable, with a final category of "can't tell" being used in case of uncertainty. The same

process was followed for the 'news format' variable, with a list of commonly occurring formats (reports, features, investigations, interviews, etc.) being developed for the coding procedure. Here, I was particularly interested in exploring to what extent both digital natives experimented with content formats in daily news production. For this reason, unconventional categories such as 'quiz', 'game', and 'listicle' were added to the list of formats to be coded. A third variable was developed to examine the storytelling style of news stories. This was borne from an interest in interpretive journalism (Strömbäck and Salgado 2012; Fink and Schudson 2014), or journalistic "sense-making" (Singer 2008), with scholars arguing that in the digital era, journalists have "not been replaced but displaced, moved higher up the editorial chain from the production of initial observations to a role that emphasises verification and interpretation" (Anderson et al. 2012: 22). For this reason, a variable was developed to assess the dominant 'style' of news reports. Under this variable, stories were categorised according to whether they were *mainly* descriptive or *mainly* interpretive, with a final category of 'can't tell' being added in case of uncertainty. These categories were informed by the previous work of Strömbäck and colleagues (e.g. Strömbäck and Shehata, 2007; Strömbäck and Aalberg 2008; Strömbäck and Salgado 2012).

A fourth set of variables were developed to examine sourcing in individual news stories. In the broadest sense, a source is an individual, organisation, or document, that provides a journalist with information for a news story. There is an extensive literature demonstrating that journalists gravitate toward 'elite' sources when constructing news stories (Cottle 2000). In this respect, I was interested in exploring whether *BuzzFeed* and *Vice* mirrored or challenged these practices. For the purposes of clarity, this project defined a source as a provider of *attributed* textual information (e.g. the president *said*, it was *reported*, the officer *believed*), that appeared in *quotation* marks in a news story (any attributed text that *not* quoted was not recorded). A variable was developed to count the number of quoted sources that appeared in each article, and also the number of sources from social media that were quoted (including embedded content from specific social media accounts). Sources were also categorised according to *name, type, gender, age, ethnicity, exclusivity*

(exclusive or non-exclusive), and whether or not they were obtained from *social media*. This process would be time-intensive and so a smaller sample of stories (n=160) was selected for coding sources.

Based on an extensive reading of the literature on digital journalism (see Chapter 3), several coding variables were also developed to examine key normative issues relating to objectivity, gatekeeping, transparency, and participation. Concerning objectivity, I was particularly interested in exploring the extent to which BuzzFeed and Vice embraced more subjective, personal, and emotional forms of storytelling in news. Building on the work of other scholars (Wahl-Jorgensen 2013; Blom and Hansen 2015; Palau-Sampio 2016), several variables were developed to examine the presence of emotionality in news content. The first of these concerned *headline language*, with headlines being categorised as either 'straight' or 'emotional', depending on the presence of different linguistic techniques such as forward-referencing, personalization, and special punctuation. A second variable was designed to examine the lead paragraph of news stories. Here, stories were categorised according to whether they employed a conventional, objective 'inverted pyramid' lead paragraph, or used a more emotional 'anecdotal' or 'narrative' lead. A final third variable was developed to assess the presence of personalised storytelling in news stories, described by Wahl-Jorgensen (2013: 135) as a "narrative form which draws on the personal experience of a particular individual caught up in a story to dramatize a broader social issue".

Finally, several variables were developed to explore the related issues of gatekeeping, transparency, and participation in news. It is often assumed that by virtue of being 'native' to the digital environment, digital native news organisations have embraced new values like transparency and participation in reporting. However, these assertions have rarely been quantified or supported with other forms of empirical evidence. Following Karlsson (2010) and colleagues (Hellmueller et al. 2013; Hedman 2016), I developed variables that corresponded with two different forms of transparency: *disclosure* and *participatory* transparency. Disclosure transparency "is concerned with whether news producers are being open about how news is being produced thus 56

relating to making journalistic routines discernible (Tuchman, 1972) and communicating standards to but not necessarily with the audience" (Karlsson 2010: 537). Meanwhile, participatory transparency "aims at getting the audience *involved* in the news production process in various ways" (Karlsson 2010: 538, emphasis added).

To examine disclosure transparency, a number of exclusive variables were developed to study transparency features of news stories. The first variable related to the use of time stamps in news articles, with individual stories being assessed for how much information they provided about the time of their publication. The second variable concerned the amount of authorial information disclosed in news stories, ranging on an ordinal scale from no authorial information, to an author's name and contact details. A third, fourth and fifth variable were also developed to examine the use of hyperlinks in news stories. The first of these variables simply counted the number of hyperlinks that appeared per article. The second, split across two variables, assessed whether these were internal or external links, and counted the number of each. Moving on to participatory transparency, one variable was developed to examine various ways in which the audience was invited to participate in the news, via activities like *commenting* and *sharing* (see Peters and Witschge 2015). This variable was divided into seven categories, corresponding with different degrees of participation

In total, 37 mutually exclusive variables were developed to analyse content produced by *BuzzFeed*, *Vice*, *The Guardian*, and *BBC*.

4.4.3. Coding Scheme and Codebook

Following the development of coding variables and corresponding categories, a formal coding scheme was constructed to assist with the process of data collection (see Appendix F). This was accompanied by a codebook that explained the goals of the study, specified the content to be examined, and provided definitions of each of the variables, including instructions on how to code content to particular variables and categories (see Appendix G). This project relied on human coding, rather the computer coding. However, the actual process of coding was performed using the statistical software package, *SPSS*.

4.4.4. Unit of Analysis

The unit of analysis for this research project is defined as a single news story.

4.4.5. Pilot Study

Reliability, according to Neuendorf (2002: 141), "can be defined as the extent to which a measuring procedure yields the same results on repeated trials". Reliability is considered essential in quantitative content analysis; partly because of its emphasis on the objectivity of the researcher and partly because "without reliability, a measure cannot be considered valid" (Neuendorf 2002: 141). To establish reliability, a pilot study was conducted on a small sub-sample of news articles. This was performed with the help of two coders external to the project. Coders were trained with the help of the codebook, and any key concepts or variables that required further clarification were explained and updated accordingly. Coding was performed on a randomised sample of 40 news articles (10 from each outlet), or 5% of the total sample. The test resulted in a percentage agreement ranging from 72.5% to 100% across variables. The most contentious variable was source exclusivity (72.5%), where coders sometimes had difficulty identifying whether a source was obtained exclusively by the journalist, or indirectly, via another source. One category measuring ethnicity of journalists and sources ('mixed race') was dropped because of imprecision in the coding. Also, several variables measuring news values (see Harcup and O'Neill 2001), were dropped from the study because of poor reliability.

4.4.6. Results and Analysis

Once all news content was coded, SPSS was used to generate various descriptive statistics about the data. These results were initially studied in

isolation, before being recombined with the data from the qualitative interviews. This triangulation of methods was performed to look for evidence of convergence, consistency, or contradiction in the findings (Mathison 1988).

4.5. Limitations

Before proceeding to discuss the empirical findings, it is important to acknowledge some limitations of this study. First, although every effort has been made to provide a *holistic* description of news production at *BuzzFeed* and *Vice*, the problems of gaining access to observe journalists' work first-hand, means that some detail on the minutiae of everyday news work is missing from this study. These interactions, as Ryfe (2016: 46) notes, can provide "an extraordinary window into the culture of journalism". On this point, there is also a strong case to make that the culture of journalism has expanded in recent years to include a more diverse cast of workers, in particular technologists (project managers, computer programmers, user-experience designers, etc.), whose position in news production has been elevated in a converged media environment (Nielsen 2012). These actors – who were certainly present in the news production process at *BuzzFeed* and *Vice* – were not included in this study. Instead, reporters and editors were prioritised as interview subjects.

On the quantitative strand of this study, one limitation relates to the fact the content analysis only focused on news articles that appeared on the homepage of the websites of *BuzzFeed, Vice, The Guardian,* and *BBC*. This does not accurately represent the full breadth of news content published by all four organisations. In a single day, a large online news organisation will typically post hundreds of individual pieces of content (text, images, video, etc.) across a multitude of platforms. This is particularly true of *BuzzFeed* and *Vice*, which, by virtue of being 'native' to the digital environment, have always relied upon platforms like *Twitter, Facebook,* and *Google,* to reach audiences (Nicholls et al. 2017). Hence, this study can only claim to offer a snapshot – albeit an important one – of the news content published by both organisations. Finally, the conclusions drawn in this study about *BuzzFeed* and *Vice,* do not speak for

all digital native news organisations. Simply put, this group of actors are too heterogeneous to be ever be discussed or represented in a unitary fashion.

Part 3 Findings and Discussion

5. Building News: BuzzFeed and Vice's arrival in the journalistic field

There is no tried and tested formula for building a news organisation in the digital era. Rather, in this period of "mind-blowing uncertainty in the evolution of journalism" (Domingo et al. 2015), the field has been the subject of manifold experiments, ranging from small to large; from local to global (Jurkowitz 2014; Carlson and Usher 2016; Nicholls et al. 2016). Among the most noteworthy of these are *BuzzFeed* and *Vice*, which, along with other digital natives like *The Huffington Post, Quartz, First Look,* and *Vox,* have been widely praised for being "innovators" in digital news (Küng 2015; Carlson and Usher 2016). As a catch-all term, innovation is used expansively in the literature to refer to "business models, collaborations, technologies, practices, and content" (Kreiss and Brennen 2016: 306). It is often taken-for-granted that digital natives are doing something *different* in these areas, though usually in the absence of any serious or sustained critical analysis.

To help remedy these shortcomings, this chapter provides a detailed overview of *BuzzFeed* and *Vice's* trajectory into the field of journalism: their origins, business models, audiences, and content strategies. This is followed by a description of some of the formal characteristics of *BuzzFeed* and *Vice's* news content, based on results from the content analysis. The findings from this exploratory chapter help foreground the rest of the study, and are referenced at different points throughout the thesis.

5.1. Green Shoots

Perhaps the first thing to note about *BuzzFeed* and *Vice* is they are "reverseentrants" to news, meaning they "discovered news while focusing on other strategic agendas" (Küng 2015: 4). For several years, both organisations remained outside the field of journalism – though still inside the larger field of mass cultural production – producing mainly entertainment content for young audiences aged approximately 18-34¹⁴. Although Vice is often recognised as a digital native, it began as a free magazine in Montreal in 1994, reporting on the counterculture of the 1990s in a typically "adolescent, male, and proudly boorish" manner (Widdicombe 2013). In 2006 - the same year BuzzFeed launched – Vice signed a deal with MTV to produce video content, which was made accessible for free on the $VBS.tv^{15}$ online platform and also distributed by Vice across several other online platforms, most notably, YouTube. Vice's first foray into online video was widely regarded as a turning point for the organisation, and ultimately led to the production of several short documentary films, focusing on subjects as diverse as heavy metal music in Baghdad (2010), North Korean labour camps (2011), the weapons trade in Pakistan (2011), and cannibalism in Liberia (2012). These videos were typically shot in a "lo-fi" documentary style, and employed an "ostensibly raw aesthetic" (Ip 2015a) to draw the viewer closer to the action. Vice have described this documentary style as "immersionism", which Kevin Sutcliffe, the former head of programming for Vice News, explains as follows:

"You might call it more relaxed. It's involved, it's embedded, it's responsive, it has an emotional connection between the reporter and the people who are consuming it. Those things make it seem fresh and less part of a corporate enterprise, which a lot of news feels like" (cited in Riordan 2014: 53).

If Vice's merging with digital technologies was somewhat serendipitous, *BuzzFeed*'s was more deliberate, with Jonah Peretti, founder of *BuzzFeed* and former co-founder of the *Huffington Post*, engaging in a series of experiments

¹⁴ This age-group has colloquially been referred to as "millennials", though the term is becoming less accurate as time progresses. According to Dimock (2018), a millennial would qualify as "anyone born between 1981 and 1996 (ages 22-37 in 2018)". In this respect, it would be more accurate to say *BuzzFeed* and *Vice* target a cross-generational audience, including those born *after* 1996 who nonetheless still comfortably fit within the 18-34 age bracket.

¹⁵ *VBS.tv* was a joint-funded venture between *Vice* and *Viacom*, who also own *MTV* (Levine 2007).

to learn more about how media could spread and "go viral" on the Internet¹⁶. These experiments were part of a side project known as *BuzzFeed Labs*, which Peretti set-up while still working for the *Huffington Post*. One of the first products he created was *Buzz Bot*, an instant messaging programme designed to track trends in the online blogosphere and then send its users a link to the most popular stories of the day (Shontell 2012). The website, *BuzzFeed.com*, was launched in 2006, and after AOL purchased the *Huffington Post* in 2011, Peretti left the organisation to focus full-time on growing his own business. In this same period, *BuzzFeed* also began moving away from aggregation to producing its own original content. In keeping with Jonah Peretti's interest in virality, this was underpinned by a strong emphasis on data science. As Küng (2015) explains:

"At heart, BuzzFeed is about data science, about analysing user data to decode how and why content is shared and distributed. User data are captured, analysed and manipulated in a perpetual loop of analysis, interpretation, experimentation, feedback, and refinement. BuzzFeed's goal is to identify which characteristics have a predictive relationship with virality, maximise these and thus accelerate the 'spread rate' of its content" (Küng 2015: 58).

This translated into the creation of multiple viral content experiments, most notably "listicles" (e.g. '21 signs you're overly empathetic'), and quizzes (e.g. 'Which pop star are you based on your Zodiac sign?'). An essential ingredient in this process was social media networks, particularly *Facebook*, which became the bedrock of *BuzzFeed's* "distributed" approach to content (Nguyen 2016). By 2015, Jonah Peretti had arrived at the conclusion that the best way to maximise the spread and reach of online content was to aggressively

¹⁶ The period in which Peretti conducted these experiments has sometimes been referred to as a "second-wave" of technological development, when the wider diffusion of broadband and rise of web 2.0 technologies (including social media platforms) made it possible for media organisations to reach new audiences and interact with them in new ways (see Bell et al. 2017: 16).

promote it off-site, on social media platforms¹⁷. At one point, it was estimated that 77% of *BuzzFeed's* content views occurred outside its website (Morrissey 2016). More recently, the company has been trying to diversify its approach (Peretti 2017a).

In the first decade of *BuzzFeed* and *Vice's* online development, spanning 2006-2016, both companies grew to the point where they began attracting huge amounts of publicity and significant commercial interest. This materialised in several major investments from venture capital firms, as well as strategic partnerships with various media brands and digital intermediaries. Beyond these irregular injections of economic capital, a more sustained source of revenue for *BuzzFeed* and *Vice* has been income earned from advertising. As their primary source of revenue, advertising has largely dictated *BuzzFeed* and *Vice's* business model, which as Nicholls et al. (2016: 7) note:

"[U]ses a combination of on-site and off-site distribution, often involving aggressive search engine optimisation and social media promotion coupled with content that is free at the point of consumption, to build large audiences across multiple countries, generally on the basis of a much leaner organisation than most legacy media".

Perhaps what distinguishes *BuzzFeed* and *Vice* from other advertising-based media is that they attract a young audience¹⁸. *Vice*, for instance, have described themselves as a "network…built around millennial passions", and claim to reach twice as many 18-34-year-olds compared to other media channels (Vice 2016a). *BuzzFeed*, meanwhile, has a global audience of approximately 650 million and claims to reach three out of five U.S millennials monthly (BuzzFeed

¹⁷ While social media sites also make up a significant portion of *Vice*'s total traffic (Corcoran 2017), the company has a more diversified publishing strategy, producing its own monthly magazine as well as content for various television networks around the world, including its own channel, *VICELAND*, which launched in the UK and US in 2016.

¹⁸ The main target audience of *BuzzFeed* and *Vice* – young people – represent an attractive and highly coveted demographic for advertisers, partly because of their size in terms of overall population, and secondly, because of their estimated purchasing power (Bazaarvoice 2012; comScore 2012; James 2016).

2018a). For this reason, they have been particularly attractive to advertisers, who have offered significant amounts of revenue in exchange for audience data¹⁹ and the use of *BuzzFeed* and *Vice's* production expertise and platform. In recent years, for example, the advertising divisions of *BuzzFeed* and *Vice* have partnered with major brands like *Intel*, *Verizon*, *Samsung*, *Warner Brothers*, and *Nike* to create quizzes, games, videos, articles, and other forms of sponsored content²⁰ using their signature style.

5.2. Investing in News

Buoyed by the growth of their entertainment businesses, in December 2011 and December 2013, respectively, *BuzzFeed* and *Vice* began investing in news, first at home, in the US²¹, and then later, abroad. This decision seemed to be based on a simple premise: that young people had particular needs and interests that could be exploited for news, as well as entertainment purposes. According to the leaders of both organisations, there was an obvious gap in the market for a youth-focused news brand, with the current crop of media being criticised for failing to adequately cater to the needs and interests of young people²²:

"For news, I was saying to my staff, it's sort of a crazy time to be alive, but no better time to launch a news platform. We're one of the fastest growing news platforms in the world for a reason, and it's because there's a lot of news out there. Like the Cold War built *CBS*, *ABC* and *NBC*, and the invasion of Afghanistan and Iraq built *CNN*, this sort of

¹⁹ By user data I am referring to "information about their uploaded data, social networks, their interests, demographic data, their browsing and interaction behaviour", which "is sold to the advertisers as a commodity" (Fuchs 2012: 704).

²⁰ I discuss the issues surrounding sponsored content in a separate section entitled "Funding and Partnerships" (6.2) which can be found in the next chapter.

²¹ Although *Vice* was founded in Montreal, Canada, its headquarters and main news operation is located in Brooklyn, New York.

²² There is at least some evidence to suggest that these critiques have an empirical foundation. Several studies in recent decades have pointed to the issue of younger audiences being increasingly turned off by the style, presentation, content, and delivery of traditional news (e.g. Buckingham 2000; Mindich 2005; Costera Meijer 2007; Banaji and Buckingham 2013; Costera Meijer and Kormelink 2014).

economic turmoil, this social chaos and this generational de-stratification is what is fuelling the rise of *Vice News*... I think that where we find our place in that is, basically, just being the voice of a very disenfranchised [population] ... I think that going forward, as Generation Y gets more and more powerful, we're going to be [their] voice" (Smith 2016a).

"Despite the struggles of the traditional media, there remains an insatiable desire for great reporting, entertaining content, and powerful storytelling. Facebook, Twitter, and other Silicon Valley-based social sites are amazing distribution platforms, but user generated content alone isn't enough to fill the hole left by the ongoing decline of print newspapers and magazines. The world needs sustainable, profitable, vibrant, content companies staffed by dedicated professionals; especially content for people that grew up on the web, whose entertainment and news interests are largely neglected by television and newspapers" (Peretti 2013).

On this basis, BuzzFeed and Vice began investing in several different areas of news coverage, focusing on a mixture of traditional beats such as politics, crime, business, international affairs, and technology, as well as subjects more specific to their audience, for example, issues concerning lesbian, gay, bisexual and transgender (LGBT) rights, gender equality, drugs, and civil rights issues (Ip 2015b; Küng 2015; Painter et al. 2017). This investment was supported by the recruitment of a large number of "young, versatile, tech-savvy, high energy staff" (cited in Phillips et al. 2009: 78), as well as several high profile and experienced journalists, who were hired to help lead their respective news operations. For example, former *Politico* reporter, Ben Smith, was appointed editor-in-chief of BuzzFeed News (US) in 2011 and former deputy editor of The Guardian, Janine Gibson was hired in 2015 to lead BuzzFeed's UK news team (its first foreign venture). Other major hires included Mark Schoofs, a Pulitzerprize winning journalist who joined the BuzzFeed US from ProPublica in 2013, and the award-winning Sunday Times journalist, Heidi Blake, who was hired in 2015 to lead *BuzzFeed*'s investigative journalism team in the UK. Similarly, early on in the development of Vice News, several experienced reporters and

editors were hired from legacy media such as *Channel 4*, *NME*, *The Guardian*, *The New Yorker* and *The Wall Street Journal*.

As "reverse-entrants" to the field (Küng 2015), BuzzFeed and Vice looked to import many of the principles that had led to their success as entertainment producers, and apply them to news. For Vice, this meant a strong emphasis on video-based content, and on so-called "news from the edge"23, typically centring on dangerous and war-torn locations around the world to explore topics like: political defectors in North Korea (2013); The rise of the Islamic State in Syria and Iraq (2014); The heroin epidemic in Iran (2014); and the civil war in Ukraine (2014). Much of this content was presented in Vice's signature "immersionist" style, which has drawn comparisons with the gonzo journalism of the 1970s, popularised by figures like Hunter S. Thompson (see Widdicombe 2013). By contrast, BuzzFeed appeared more interested in applying their expertise in data science to journalism, because, according to Riordan (2014: 48), Jonah Peretti had "worked out that news content, not just lists, could be shareable and viral". This gave rise to a news process shaped strongly by the affordances and algorithms of social platforms; leading to an interest in creating news that people wanted to *share* (Küng 2015). Much like Vice and their style of immersionism, this motive seemed to push BuzzFeed toward pursuing a style and form of news that eschewed traditional journalistic conventions, emphasising, for example, subjective, emotional and personal narratives over more "objective" styles of news (Riordan 2014; Tandoc 2017).

The main point here is that – owing to a combination of technological factors and audience considerations – *BuzzFeed* and *Vice* have entered journalism with ideas and values that would appear to "clash with the prevailing norms of production and the expectations of the field" (Bourdieu 1993: 57). In the second half of this chapter, I begin to provide an empirical basis for this hypothesis by examining some of the basic features of news content at *BuzzFeed* and *Vice*.

²³ This phrase comes from the tagline *Vice* used in their *HBO* television documentary series.

This is compared with news produced by two traditional news organisations: *The Guardian* and *BBC*.

5.3. News Content at BuzzFeed and Vice

As outlined in Chapter 4 of this thesis, the quantitative portion of this study is based on an analysis of 800 news articles, gathered over two separate weeks in October 2016 and March 2017. These articles were collected from the UK and US websites of *BuzzFeed, Vice, The Guardian* and the *BBC*, and coded according to a pre-defined set of variables (see Appendix F). Stories that were repeated by the same outlet across more than one day were discounted from the sample, which left a total of 753 news articles. In the results below, data from the UK and US outlets of each news website are combined, except in instances where significant differences between the two outlets have been recorded. It should also be noted that the following analysis is mainly descriptive, with a more detailed discussion of these findings taking place in the other empirical chapters of this thesis. Articles are examined according to authorship, format, topic, and word count.

5.3.1. Authorship

The first variable examined concerned the authorship of news articles. The objective here was to see if there were any notable differences in the demographic makeup of journalists' working for *BuzzFeed* and *Vice*. Where possible, the personal bylines of news articles were coded according to gender and ethnicity. The age of authors proved too difficult to validate, and so this variable was not used in the final coding procedure. Articles that contained more than one author were also coded, and form part of the combined totals below:

Table 1: Author gender	of news articles
------------------------	------------------

	BuzzFeed (n=179)	VICE (n=174)	Guardian (n=200)	BBC (n=200)
Male (%)	46.2	60.8	60.2	65.7
Female (%)	53.3	39.2	39.8	34.3
Can't tell (%)	0.5	0	0	0
TOTAL	100	100	100	100

At *Vice*, *The Guardian*, and *BBC*²⁴, male journalists authored more stories than female journalists, by a difference of around 20%. These findings are largely consistent with the existing literature on gender in (Anglo-American) journalism, which has shown that the profession is generally dominated by males (National Council for the Training of Journalists 2013; Willnat and Weaver 2014; Thurman et al. 2016). In this regard, *BuzzFeed* represented somewhat of an outlier, with female-authored reports outweighing male-authored reports by justo over seven percentage points. This difference was more extreme at *BuzzFeed US*, where female-authored news articles made up 60.6% of total coverage; almost an exact inverse of the male-female ratio at the other three outlets in the sample. Following gender, authors of news stories were also categorised according to their ethnicity:

²⁴ One important caveat here is that the *BBC* generally published no authorial information in their news articles, meaning the sample size of authors for the *BBC* was very small (35 authors in total).

	BuzzFeed	VICE	Guardian	BBC
	(n=179)	(n=174)	(n=200)	(n=200)
White (%)	78.0	83.3	84.9	82.8
Black (%)	5.6	1.9	0.9	0
Asian and Indian (%)	10.8	3.1	6.2	0
Hispanic (%)	1.4	1.2	0.9	2.9
Can't tell (%)	4.2	10.5	7.1	14.3
TOTAL	100	100	100	100

Table 2: Author ethnicity of news articles

As the data shows, white authors dominated across all outlets, featuring the most in stories published by *The Guardian*, followed by *Vice, BBC*, and *BuzzFeed*. Again, this is largely consistent with the data on the demographic composition of journalists, which has shown that the profession is dominated by whites (National Council for the Training of Journalists 2013; Willnat and Weaver 2014; Thurman et al. 2016). *BuzzFeed*, however, did publish a notably higher number of Asian, Indian, and black journalists, compared to the other three outlets in the sample. The results of this analysis closely match diversity statistics published by *BuzzFeed* this year, which indicated that 7.9% of its US news team was black, and 13.8% Asian (Peretti 2018a). Demographic data on its UK operation has not been made public.

5.3.2. Format

Secondly, articles were categorised by format. Based on digital natives' reputation for experimenting with non-traditional content formats (Carlson and

Usher 2016; Usher 2017), I hypothesised that greater format variation might be found in news articles published by *BuzzFeed* and *Vice*.

	BuzzFeed	VICE	Guardian	BBC
	(n=179)	(n=174)	(n=200)	(n=200)
Report (%)	71	34.5	83.5	85
Feature / investigation (%)	21.2	29.9	7	2.5
Interview (%)	1.1	10.3	0.5	1
Comment / opinion (%)	0.6	12.5	2	3.5
Review (%)	0	1.2	0	0
Listicle (%)	3.9	0	0	1
Quiz (%)	0	0	0	0
Game (%)	0	0	0	0
Aggregated content (%)	1.1	5.8	0.5	4
Live blog (%)	1.1	0.6	5.5	1.5
Other / can't tell (%)	0	5.2	1	1.5
TOTAL	100	100	100	100

Table 3: Format of news articles

Generally speaking, both digital natives leaned toward using traditional news formats in daily coverage, while unconventional formats such as "listicles" and quizzes barely featured. This being said, both *BuzzFeed* and *Vice* showed
greater variation *between* formats compared to *The Guardian* and the *BBC*. For example, while the standard "report" format dominated news published by the two legacy organisations, *BuzzFeed* and *Vice* devoted significantly more resources to formats like features, investigations, interviews, and opinion or comment pieces. This was particularly true of *Vice*, which only employed the "report" format in about a third of news stories.

5.3.3. News Topic

News articles were also coded by topic. The purpose here was to see if coverage by *BuzzFeed* and *Vice* varied significantly by subject matter, compared to the two legacy organisations in the sample.

Table 4: Topic of news articles

	BuzzFeed (n=179)	VICE (n=174)	Guardian (n=200)	BBC (n=200)
Government / politics (%)	31.9	22.4	37.5	37
Social issues, rights, and protests (%)	18.4	22.4	11	7
Crime and terrorism (%)	8.9	14.4	17	21.5
Science, education, and technology (%)	8.9	1.7	4	4.5
Health (%)	2.8	1.7	2	2.5
Business and economy (%)	8.4	2.9	14.5	11
Entertainment and arts (%)	12.3	18.4	8.5	9

Lifestyle (trends, fashion, travel) (%)	0	6.9	0	0
Accidents and disasters (%)	2.8	0.6	1	3.5
Energy and environment (%)	1.7	0	3.5	1
Other / can't tell (%)	3.9	8.6	1	3
TOTAL	100	100	100	100

There were some notable similarities and differences in news topics covered by each organisation. First, all organisations dedicated a substantial portion of resources to covering what might very broadly be described as "hard" news topics: government, politics, business, the economy²⁵. Still, *The Guardian* and BBC covered these subjects more often than BuzzFeed and Vice. Conversely, both digital natives published significantly more news about social rights, issues, and protests. Vice also featured more news about arts and entertainment – a difference which can be explained by the fact that stories were collected from the *Vice.com* website, rather than the *Vice News* website, during the second week of coding²⁶. Taking only the first week of *Vice* content into account, stories on the subject of entertainment and arts did not feature at all on either their UK or US website. Outside these observations, there were no significant differences in news topic between each news organisation. BuzzFeed covered slightly more news about science and technology. Meanwhile, stories about crime and terrorism featured heavily in coverage by The Guardian, BBC, and Vice. News about lifestyle, health, energy and the environment, all hardly featured.

²⁵ Although there is wide disagreement on what constitutes "hard" and "soft" news, Reinemann et al. (2011: 11) suggest that, generally speaking: "Foreign and domestic politics, economy and finance are usually regarded as hard news. News about sports, celebrities, royal families, crime, scandals and service are regarded as soft news". ²⁶ See Chapter 4 (section 4.4).

5.3.4. Word Count

Finally, articles were assessed by word count²⁷. Based on *BuzzFeed* and *Vice's* interest in producing news for mobile devices and for social media, it was hypothesised that news stories by *BuzzFeed* and *Vice* would be shorter in length than *The Guardian* and *BBC*.

	BuzzFeed (n=177)	VICE (n=173)	Guardian (n=189)	BBC (n=197)
Mean word count	1139.9	990.3	812.9	522.5
Median word count	648.5	941	715.5	499

Table 5: Word count of news articles

The results generally seemed to disprove this hypothesis. In fact, *BuzzFeed's* articles had the highest *mean* word count (1139.9), while *Vice's* articles had the highest *median* word count (941). One obvious explanation for this was the relatively high number of features and investigations published by both organisations, which typically had higher word counts than standard news reports (see section 5.3.2.). Comparing the mean word counts of news reports in isolation, the following results emerge:

²⁷ Note: news stories in the "live blog" format were discounted from this sample because of their ever-changing word counts.

Table 6: Word count of news reports

	BuzzFeed	VICE	Guardian	BBC
	(n=127)	(n=60)	(n=167)	(n=170)
Mean word count	576.5	483.4	725.8	492.63

As shown, there was no substantial difference between *BuzzFeed, Vice*, and the *BBC*, although *Vice* did technically publish the shortest news reports on average. *The Guardian* was the major outlier, averaging around 150-200 more words per article compared to the other three outlets in the sample.

6. Organising News: The Structure and Organisation of News at BuzzFeed and Vice

Having traced *BuzzFeed* and *Vice*'s passage into journalism and described some of the basic features of their news content, this chapter moves on to examine the structure and organisation of news at *BuzzFeed* and *Vice*. Several points from the preceding discussion are revisited, including *BuzzFeed* and *Vice*'s motivations, goals, business model, audience, staff, and news coverage. Collectively, these "invisible structures" play an important role in shaping journalistic norms and practices, and have the potential to reveal more about "the objective power relations that structure the field" (Bourdieu 1998: 39-40). Their exposition also provides the groundwork for the later chapters of this thesis. Key aspects of the enquiry include:

- a) How BuzzFeed and Vice 'arrived' at news, their journalistic ethos and continuing trajectory in the field
- b) The underlying motives behind staff recruitment and the arrangement of news into particular 'beats' and topic areas
- c) The position of *BuzzFeed* and *Vice* in the journalistic field at large.

The findings of this chapter introduce qualitative data from interviews conducted with 24 news staff at *BuzzFeed* and *Vice*. This data has been combined with a variety of secondary sources, which have proven particularly valuable in light of the already-documented problems gaining access to *BuzzFeed* and *Vice*²⁸. Nonetheless, while this information has been helpful, the general lack of public information about *BuzzFeed* and *Vice* (the number of staff they employ, their yearly revenue, their various sources of funding, etc.), has come at a cost to providing a more detailed explanation of the operations of both companies. This would seem indicative of a general issue in researching commercial, privately-owned media, as noted in Chapter 4 of this thesis.

²⁸ For a more detailed discussion of the issues pertaining to access, please refer back to section 4.2 of this thesis.

Despite these limitations, this chapter makes an important intervention by revealing the "hybrid" (Chadwick 2013) structure of news at *BuzzFeed* and *Vice*. As two organisations regularly lauded for being "innovators" in digital news (Küng 2015), this analysis shows that *BuzzFeed* and *Vice* place an equally high emphasis on culturally sanctified journalistic forms and practices, in an effort to boost their professional and public legitimacy. The opening section of this chapter (6.1) begins by describing the key motivating factors behind *BuzzFeed* and *Vice's* entry into journalism. Following this, *BuzzFeed* and *Vice's* primary sources of funding are broken down and analysed (6.2). From here, the news operations of both organisations are examined in reference to their staff (6.3), and division of news coverage (6.4), principally from a Bourdieusian perspective. Finally, I provide a short conclusion (6.5) summarising the key findings from the chapter.

6.1. Motivating Factors

"When we first started, the way that people were connecting with each other online was internet memes and humor and cute animals...Then what we saw was that social became much bigger, and people started to do that with news... When people started to share news we said, 'Wow, we'd love to be in the news industry. We'd love to make news'. We didn't think we could because people weren't sharing news. All of a sudden, we started to see news on Facebook and Twitter, and it made us realize we could go into that business. We've evolved along with the way consumers have evolved and the way social interactions have evolved online" – Jonah Peretti, founder and CEO of *BuzzFeed* (Peretti 2017b).

"Our audience is actually saying make more news. We tell stories that a lot of other people don't tell, and we tell them in a different way. That's what's really been resonating with our audience. So, we're going to double down" – Shane Smith, co-founder and former CEO of *Vice* (Smith 2013).

When considering first about *BuzzFeed* and *Vice's* motivations for entering the field, it is helpful to view these factors in Bourdieu's terms, as related to the pursuit of different forms of *capital*²⁹, or resources. The first and most obvious of these is economic capital, with both organisations recognising news as a potentially effective way of enhancing their public recognition, "measured by numbers of readers, listeners, or viewers, and therefore, in the final analysis, by sales and profits" (Bourdieu 1998: 70). In seeking to capitalise and expand their existing audience, BuzzFeed and Vice have also been motivated to introduce new forms of cultural capital to the field; that is, specialist skills, knowledge, expertise, and tastes - honed during the process of making entertainment content – which were deemed to have significant migratory value in the journalistic field. A third and final and important motivating factor relates to the symbolic capital³⁰ of the journalistic field *in toto*. Discussing this aspect, Painter et al. (2017: 18) argue that the economic promise of journalism was secondary to BuzzFeed and Vice's assumption that being associated with its practice might quickly "add reputation and credibility to their brands". There is a certain logic to this if one considers that the subfield of journalism is generally higher in symbolic and cultural capital compared to the broader field of cultural production (Hesmondhalgh 2006). In an interview with Columbia Journalism Review, Jonah Peretti acknowledged this, recognising that news brought certain "reputational benefits" to BuzzFeed's brand that would be harder to earn if the company was solely focusing on entertainment (Peretti 2018b). Although Shane Smith of Vice has been less explicit in making this argument, many of my interviewees from Vice similarly cited the symbolic benefits of journalism as a major factor behind Vice's initial decision to invest in news³¹.

²⁹ As a reminder, "capital" concerns not only material resources like currency and assets, but also less tangible forms of capital like knowledge, reputation and prestige (Bourdieu 1986).

³⁰ As a reminder, symbolic capital refers to "symbolic forms of power, such as prestige and status" (Siapera and Spyridou 2012: 81).

³¹ I discuss these responses in more detail later in this chapter.

6.2. Funding and Partnerships

In principle, *BuzzFeed* and *Vice* have relied on two major sources of revenue to fund their businesses: 1) external investment, chiefly in the form of venture capital and content partnerships, and 2) native advertising, elsewhere referred to as sponsored or branded content. A potentially useful way of thinking about BuzzFeed and Vice's approach to funding and partnerships is in terms of integration. As Gade and Raviola (2009) write, in response to economic uncertainty, changing markets, and new technologies, media firms have tried to become more "structurally integrated...[reducing] organizational barriers both within and among firms in ways that spur innovation" (Gade and Raviola 2009: 88). This shift has had implications on how organisations interact with one another (inter-firm integration), and structure themselves internally (intra*firm integration*). In this section, I consider these two aspects of integration from the perspective of field theory, looking at how partnerships and sources of funding are premised on different exchanges of *capital*, or resources, which have brought various benefits to *BuzzFeed* and *Vice*, while also being a portend of wider change in the field.

6.2.1. Inter-Firm Integration: Investments, Partnerships, and Deals

Since their launch, *BuzzFeed* and *Vice* have relied on a range of investments, strategic partnerships and deals to help support their operations. A major portion of this funding has come from venture capitalists and legacy media organisations. For example, *Vice's* investors now include established media such as 21^{st} Century Fox, who own a 5% stake in the business, and A&E *Networks* (50% of which is owned by *Disney*), who paid \$250 million for a 10% share of *Vice* in 2014 (Küng 2015). Similarly, *BuzzFeed* has attracted investment from *NBCUniversal*, a subsidiary of the telecommunications conglomerate, *Comcast*, which has invested \$200m in the companyon two occasions, most recently in 2016 (Kafka 2016). In recent years, both companies have also agreed to various partnerships with other media brands such as *HBO*, *The Guardian*, *NBC Universal*; and also with digital intermediaries including *Twitter*, *Facebook*, and *YouTube*. These "strategic alliances" have

involved "sharing knowledge, resources, and access to markets" (Gade and Raviola 2009: 90), or various forms of *capital*, to put it in Bourdieu's terms.

For Vice and BuzzFeed, these alliances have typically been premised on an exchange of editorial expertise for money, or cultural capital for economic capital. For instance, in 2014, the video sharing platform, *YouTube*, launched a major campaign to promote the *Vice News* channel on its own platform, both online, and via press and television (Griffith 2014). The following year, *Vice* announced a major partnership deal with the television network, *HBO*, that allowed the company to "vastly expand its reporting capabilities" by providing the funding for a daily newscast as well as numerous video documentaries which were made available on *HBO's NOW* online streaming service (*Vice* 2015a). Many journalists I interviewed described *Vice's* deal with *HBO* as a major turning point for the organisation, a "graduation of sorts into the [journalistic] field" that led to *Vice* being recognised as a "potential player in the emerging digital sphere" (Respondent E, editor, *Vice US*). As one journalist put it:

"HBO came with big money for a deal, and so [*Vice*] chased that and the *YouTube* money went away...I also think that Shane [Smith] is driven a lot by prestige. I think he's got a chip on his shoulder, where, you know, he feels like no one takes him seriously as a journalist or as a business man and *HBO* is a real credible platform, it's a kind of blue-chip prestige channel here in the States...So first and foremost *HBO* gave [*Vice*] a tonne of money I'm sure, but secondly it feeds into his desire to be legitimate" (Respondent J, editor, *Vice US*).

Similar to Vice, BuzzFeed also signed a number of deals with other media to help propel the growth of their business. Given the centrality of social media to their business model, many of these deals have involved entering partnerships with large social platforms, with BuzzFeed delivering fresh entertainment and news content to these sites in exchange for new advertising opportunities and

expanded audience reach³² (economic capital). Very recently, *BuzzFeed* agreed a partnership with the social platform, *Twitter*, to produce a weekday morning show called *AM to DM*, which debuted in September 2017 and is said to receive approximately one million viewers per episode (McAteer 2018).

Outside these deals, *BuzzFeed* and *Vice* have occasionally entered into partnerships with legacy news organisations, sharing expertise, revenues, and audiences³³. For example, in 2016, *The Guardian* announced a partnership with *Vice*, which included the production of various "co-branded special reports" as well as other forms of news content targeted towards younger audiences (Sweney 2016a). Likewise, *BuzzFeed* have entered into agreements with legacy organisations like the *BBC* to produce a series of special, one-off, joint news investigations (Burrell 2016; Ridley 2016). One journalist from *BuzzFeed* outlined the various ways in which this partnership benefited each organisation:

"[The *BBC*] get the advantage of having a properly resourced investigations team... who can, you know go through the documents, track people down; all those key investigative skills...The other thing we bring to traditional press, like the *BBC*, is a [young] audience...So we are reaching an audience they are struggling to reach...We've never regretted a collaboration with the *BBC* because of the impact and weight of having them on side...I think having the *BBC* backing has gotten [stories] to such a wide audience...So it's been really good in terms of reaching influential people in order to have some kind of impact" (Respondent A, reporter, *BuzzFeed UK*).

In this manner, journalists reflected on *BuzzFeed's* partnership with the *BBC* much in the same way journalists from *Vice* discussed the business deal with

³² For a useful account of the relationship between news organisations and social platforms, see Bell et al. (2017).

³³ The trend of inter-organisational partnerships is a phenomenon that has been remarked upon in the media industries at large. Mendelson (2000) makes the distinction between the present "information-age architecture" of organisations and the "industrial-age architecture", prevalent for much of the 20th century. Information-age architecture, he argues, involves "network[s] of firms bound together by information sharing and reciprocity, rather than ownership and control" (Mendelson 2000: 520).

HBO, drawing particular attention to the symbolic benefits of partnering with news organisations that already occupied a powerful position in the field³⁴. One might consider this an example of what Brooks (2016: 131) calls "the proximity paradigm", which describes the way new entrants to the field gain "reputation, distinction, and/or prestige" through affiliation with individuals or organisations rich in symbolic capital.

6.2.2. Intra-Firm Integration: Native Advertising

Alongside external forms of support and funding, *BuzzFeed* and *Vice* have contributed a significant amount of their own economic capital to fund their respective news operations. A major source of this capital has been revenue earned from native advertising³⁵ (sometimes referred to as 'sponsored' or 'branded' content). Native advertising is distinct from other forms of advertising because of the way it is designed to mimic editorial content. For this reason, it has been controversial from its outset, particularly in the field of journalism where some scholars have argued that it might contribute to the erosion of institutional credibility by weakening journalistic autonomy (Carlson 2015; Schauster et al. 2016; Amazeen and Muddiman 2017; Amazeen and Wojdynski 2018). Despite these concerns, native advertising has steadily grown to become a significant source of revenue for many new and established media organisations, and an important part of the online advertising ecosystem in general (Fullerton 2017). The success of native advertising is a product of several interlinked trends which it is only possible to sketch here. First, native advertising tends to yield higher returns on investment compared to traditional online formats like display advertising. Second, there has been a general rise in the number of people using ad-blocking software, which native advertising

³⁴ In Bourdieu's terms, the most powerful actors within any social field are those that manage to combine high economic capital with high symbolic capital (Marliére 1998). It is these players, according to Bourdieu, who have the greatest power to "consecrate", that is, "deform the space" around themselves so that their particular ideas and values are considered "worthy of wider consideration" by other members of the field (Benson 1999: 469).

³⁵ According to one report in the *Wall Street Journal*, the lion's share of total revenue earned by *BuzzFeed* and *Vice* in 2016, was derived from native advertising (Marshall and Alpert 2016).

can evade by seamlessly blending with other editorial content. Third, more people are accessing news and other information via their mobile phones, which has encouraged a move away from advertising formats that feel overly intrusive or disruptive to the experience of the individual user (Newman and Austin 2015). Finally, and perhaps more tangentially, there is a general sense among advertisers that people have become more sceptical about advertising and brand messages, and in this regard, native advertising has been seen as a better way of getting past people's so-called "bullshit detectors" (Benady 2014).

As two pioneers of native advertising, BuzzFeed and Vice have often been credited for introducing this format to other actors in the field (e.g. Oakes 2015; Rodriguez 2016). Perhaps to avoid controversy, journalists at *BuzzFeed* and Vice rarely acknowledged or spoke about these issues. This might also been because they were also not generally asked any direct questions about native advertising in interviews (see Appendix B). One editor from BuzzFeed UK insisted that they knew "nothing about the mechanics of the advertising side other than they seem to be making money", suggesting that there remained a "strong Chinese wall" between the business and editorial sides of the organisation (Respondent L, editor, *BuzzFeed UK*). At *Vice*, a couple of editors from the US suggested that the company had practiced "questionable ethics" in its fledgling years, which improved as the organisation matured (Respondent D, reporter, Vice US). Referring to this early period, one editor mentioned that when they first joined the organisation, they had received an email from the advertising department about a story they were working on because it "implicated [redacted brand name] and other Western brands" (Respondent J, editor, Vice US). The story was never published although they could not confirm whether this was directly because of commercial concerns.

Publically, there have also been moments at both companies where the "Chinese Wall" between business and editorial has become porous. To highlight one case, in April 2015, *BuzzFeed's* editor-in-chief, Ben Smith, shared an internal memo he sent to employees with the news and entertainment website *Gawker*. In the memo, he acknowledged that on at least three

occasions, published posts had been deleted following complaints by advertisers (Trotter 2015). In the UK, *BuzzFeed* were also reprimanded by the Advertising Standards Authority in 2016 for failing to adequately label an advert for laundry detergent as a piece of sponsored content (Sweney 2016b). In the past, *Vice* has also been accused by ex-employees of terminating stories because of "brand-partner concerns", and encouraging writers to seek approval from senior management before writing about brands (Sterne 2014). *Vice* responded to some of these criticisms in a strongly-worded post entitled "*Vice* to *Gawker: Fuck You and Fuck Your Garbage Click-Bait Journalism*", though the company never directly refuted the claims made against them (Vice 2015b).

Despite the lack of clarity concerning *BuzzFeed's* and *Vice's* relationship with advertisers, the limited evidence available would suggest that both organisations have sometimes overstepped traditional boundaries between advertising and editorial. While this alone might be a cause for concern, a larger issue relates to the broader *cultural* impact of *BuzzFeed* and *Vice*, and the role they have played in normalising closer integration between editorial and advertising in journalism. This was evident in the discourse of some interviewees, who seemed to broach native advertising with a degree of acceptance, recognising it as one potential path toward financial salvation. For example:

"[If you] are given lots money by Ford to make some adverts and then you go and take that money and invest it in real journalism, it has its quandaries, it has its questions, concerning that model [of native advertising], but it's one of the only ways that *Vice* has been able to keep itself going" (Respondent F, editor, *Vice UK*).

"You know, for the past five years [here's been] cuts in newsrooms everywhere...I don't know much about the commercial side because we're kept separate but obviously we use native advertising, we use sponsored ads...that's quite a new funding model...So, I think we're going to have to look at ways of attracting audiences in different ways, beyond viewing figures and circulation, to keep the money coming in" (Respondent A, reporter, *BuzzFeed UK*).

These comments are consistent with observations made by other scholars, who have similarly noted a creeping degree of acceptance among journalists regarding native advertising and other forms of intra-firm integration (Gade and Raviola 2009; Carlson 2015). For Cornia et al. (2018) this amounts to a full-blown normative shift: from *separation* (between editorial and business interests) to *integration*, as journalists and organisations become more comfortable with "combining established editorial values with the values of collaboration, adaptation and business interests has always been a somewhat of a tenuous commitment (Coddington 2014), these findings would collectively seem to suggest that editorial autonomy might also be waning as a normative aspiration, as organisations trade in some of their cultural capital and autonomy in exchange for greater economic sustainability.

6.3. Staff

Having spent some time discussing funding and partnerships at *BuzzFeed* and *Vice*, this section moves on to examine their recruitment strategies. As mentioned in the previous chapter, *BuzzFeed* and *Vice* have experienced fairly rapid growth, and in the process, hired large teams of staff to help support their operations (Jurkowitz 2014). According to their own website, *BuzzFeed* currently employ around 1300 staff globally (BuzzFeed 2018b). One recent report in the *Atlantic* suggested that around 460 of these employees worked in editorial, though this figure has been difficult to corroborate (Meyer 2016). *BuzzFeed's* own website suggests there are 250 reporters and editors working in their US newsroom, as well as editorial teams of various sizes working across different international territories (BuzzFeed 2018b). In the UK, *BuzzFeed* employs 204 staff, with 138 of these working in editorial³⁶ (BuzzFeed 2016).

³⁶ Note: this data is based on financial statements filed to *Companies House* by *BuzzFeed UK* in 2016.

One recent report, citing spokespeople from *BuzzFeed's* management team, claimed that 76 of these staff worked specifically in news (Kanter 2017).

Vice is larger than *BuzzFeed*, and, according to several news reports, employs approximately 3000 people worldwide (e.g. Spangler 2017; Steel 2018). Similar to *BuzzFeed*, *Vice* has expanded into numerous new territories over the last few years, with one recent study estimating that the company now has a news presence in 25 different countries³⁷ (Painter et al. 2017). In the UK, *Vice* employs 208 staff ³⁸ (Vice 2016b). One report from the *Guardian* estimated that around 18 of these staff worked solely in news (Quinn and Jackson 2016), though this number is likely to have increased based on *Vice UK's* more recent investments in video production (Clarke 2018).

6.3.1. Staff Demographics

Compared to industry averages, there are several notable differences in the demographic composition of news workers at *BuzzFeed* and *Vice*. The first and most obvious is age. While in the US and the UK most journalists tend to be middle-aged (National Council for the Training of Journalists 2013; Willnat and Weaver 2014; Thurman et al. 2016), at *Vice*, the average age of a journalist is reported be around 25³⁹ (Adams 2013). Despite the lack of accurate information about the average age of the *BuzzFeed's* editorial department, anecdotal evidence points to an equally young workforce (Peretti 2018a). This was corroborated somewhat in my own research, with more than half of the people I interviewed from *BuzzFeed* being in their 20s.

³⁷ *Vice's* "about" page on its main website lists 25 physical addresses for different *Vice* offices, three of which are part of the same jurisdiction (United States). This would give *Vice* 22 foreign bureaus (Vice 2017a)

³⁸ Note: this data is based on financial statements filed to *Companies House* by *Vice UK* in 2016.

³⁹ Information based on an interview conducted between Shane Smith, CEO of *Vice*, and Tim Adams of the *Guardian*. In a 2016 interview, Shane Smith also repeated this fact, stating that "the average age at *Vice* is 25 across the board, globally" (Smith 2016a). However, I have not come across any statistics or publically-available data to support these statements.

At *BuzzFeed*, gender and ethnicity also appear to diverge from the standard. In data published in 2017, Jonah Peretti, the founder and CEO of *BuzzFeed*, reported that the company was made up of 55% women globally, and that 35% of its employees identified as non-white (Peretti 2017c). These percentages map very accurately to the findings of my content analysis⁴⁰, which indicated that 53% of *BuzzFeed's* news stories were authored by women, and 17.8% of stories were published by non-white authors. Moreover, of the 14 staff that were interviewed from *BuzzFeed*, 10 were women and three identified as non-white.

Generally speaking, there is a shortage of publically available information on the demographics of *Vice's* employees⁴¹, though the company has been criticised for its perceived lack of diversity in the past (Ip 2015a; Martinson 2015; Vice 2017b). This was also reflected in the content analysis, which showed that male journalists authored 60.8% of stories in total, and non-white journalists authored 6.2% of news stories. Of the eight *Vice* employees I interviewed, all but one was male.

In summary, news workers at *BuzzFeed* and *Vice* are generally younger in age compared to the rest of the journalistic field. *BuzzFeed* has a more female-dominated editorial team and employs significantly more people from non-white ethnic backgrounds. *Vice*, on the other hand, appears to be more male-dominated and less diverse. Collectively, these characteristics would seem to partially reflect the demographic composition of the audience, with both organisations reaching a core demographic of young consumers (Moses 2014), skewed toward women in the case of *BuzzFeed* (Morrison 2015; Hwong 2018), and men in the case of *Vice* (Vice 2016a). This also corresponds with Bourdieu's (1984) observation that the producers and audiences of cultural products tend to be homologous with one another and thus share a similar *habitus*⁴².

⁴⁰ See Chapter 5 (section 5.3) for a breakdown of news stories by author.

⁴¹ Vice has rarely been forthcoming about revealing the diversity of their own employees (e.g. James 2015), though the company has been criticised for its perceived lack of diversity in the past (Ip 2015a).

⁴² As a reminder, *habitus* refers to one's "predispositions, assumptions, judgements and behaviors" which are accumulated via an ongoing process of socialization; first

6.3.2. Hiring Practices

Statistically speaking, evidence would suggest that the demographic composition of staff at *BuzzFeed* and *Vice* differs from industry standards. These findings, however, do not explain *why* both organisations have chosen to populate their news departments with young journalists, nor do they provide a complete picture of specifically *who BuzzFeed* and *Vice* decided to hire in the early stages of their development. Based on data from interviews conducted with staff, this section looks in more detail at the motivations behind *BuzzFeed* and *Vice's* hiring practices, focusing on the development of their news operations in the UK and US. In line with Bourdieu, staff are imagined as possessing different forms of *capital*, co-opted by both organisations in the pursuit of different material and symbolic rewards. What emerges from this analysis is a hiring strategy marked strongly by *hybridity*, where the desire for distinction is strongly constrained by the need to be recognised as legitimate by fellow peers and the public at large.

6.3.3. Senior / Experienced Staff

"I think when you're recruiting young reporters the thing that people really want is an editor. That's a huge luxury. We have very strong editors and that's a big part of how we operate" – Ben Smith, editor-in-chief of *BuzzFeed US* (Smith 2014a).

Despite evidence indicating that *BuzzFeed* and *Vice* employ an overwhelmingly young staff, in the early stages of their development, they caught the attention of the wider field by hiring a number of high-profile and experienced journalists to help lead their news departments. As agents already in possession of high amounts of cultural and symbolic capital, these journalists seemed to be

in the family, and then through other social structures such as education and work (Benson and Neveu 2005: 3).

principally valued for their reputations and sense of professionalism that they brought to their organisations As one editor from *BuzzFeed* explained:

"If you look at the profile of our news editors it's extremely traditional... The Head of Breaking News spent twelve years at the New York Times, our investigations editor has two Pulitzer prizes...[these people] are obsessed with not getting things wrong and are also very good at verifying" (Respondent K, editor, *BuzzFeed UK*).

At *Vice*, similar connections were made between experience (in the field) and professionalism. For example, one senior editor saw it has his role to make the news department more of a "viable journalistic operation" by "getting staff up to speed on the fundamentals [of journalism]" (Respondent E, editor, *Vice US*). This included procedures associated with fact-checking, accuracy, and performing due diligence with sources. Several journalists I spoke with seemed to take a deep pride in their commitment to certain ethical principles. As one senior staff member from *Vice* argued: "News is a discipline. And [at *Vice*] there were a lot of conventions I was comfortable with throwing away, but not the ethics and the proper way to gather news" (Respondent J, reporter, *Vice US*).

The professional ideology of more experienced journalists seemed to have a trickle-down effect on younger journalists at *Vice*, who felt they benefited from the mentorship of certain senior editors in the organisation. As one reporter from the UK office explained:

"To be honest, I owe a lot to [name redacted], he's been incredible, he made Vice News what it is, [and] almost single-handedly, brought [the whole team] up to speed and really taught us all a lot. [Name redacted] is brilliant, I wouldn't know half of what I know now...He's very rational, very decisive; really what you need from a boss, really *experienced*, you know?" (Respondent A, reporter, *Vice UK*, my emphasis).

While some young staff seemed grateful for the opportunity to work with more experienced journalists, some senior editors from *Vice* highlighted the 90

frustrations that arose from trying to mentor a relatively youthful staff. As one editor said:

"[At Vice News] there was a lack of experience and a lot of naivety, and hubris, you know. You needed to mentor most of the people on staff because this was their first rodeo, you know. And a lot of them could grasp that, and their writing would improve, and their grasp of certain fundamentals would improve. But they'd still neglect to call this source, or you'd point out inaccuracies in their stories...So definitely that was a cultural difference and not just limited to one or two individuals" (Respondent E, editor, *Vice US*).

Beyond the sense of professionalism that senior journalists brought to news, younger journalists could also appreciate the social and symbolic value of having high-profile figures working for their organisations. For one reporter from *BuzzFeed UK*, this was a persuasive factor in them deciding to join the organisation as a reporter:

"I never thought about going for [the job at *BuzzFeed*] until I saw Heidi Blake had been appointed editor. And obviously she's got a brilliant reputation and is really well-known to the field. So, I thought wow, they must be serious about investigative journalism" (Respondent A, reporter, *BuzzFeed UK*).

These responses helpfully draw attention to the different kinds of *capital* that experienced journalists brought to *BuzzFeed* and *Vice*. These journalists were valued by staff and their organizations at large because of their *illusio*, or strong "feel for the game" (Bourdieu 1990: 66). Experienced journalists brought important cultural capital to *BuzzFeed* and *Vice* – knowledge, expertise, and skills, accumulated from time spent in the field – which were put to the service of advancing journalistic professionalism in their organizations⁴³. Among

⁴³ In a survey conducted with journalists from 243 newspapers in the US, Gade (2008: 384) made a similar observation, noting that editors played an important role as

younger interviewees, their own inexperience heightened the importance of this function, with senior editors acting as mentors to those less socialised to the field's dominant norms and practices. In this way, senior journalists acted as a strong force of conservation inside their organisations, ensuring a level of consistency in journalism's traditional norms and practices – even if, at times, this pedagogical role could lead to cultural clashes with younger members of staff.

There is a strong case to make that experienced journalists also had an important symbolic function, bringing a sense of credibility and prestige to their organisations by virtue of their general reputations in the field. As Bourdieu (1998: 70) notes, it is those who "internalize most completely the internal 'values' or principles of the field" that stand the best chance of being recognised as legitimate by peers. Here, experienced journalists played an important role in communicating internally and to the rest of the field that *BuzzFeed* and *Vice* were 'serious' about journalism. This was apparent in conversations with certain reporters and has also been noted in reactions from other members of the field (Tandoc and Jenkins 2017).

Finally, and though not directly acknowledged in interviews, it is likely that experienced journalists also brought a certain degree of social capital to *BuzzFeed* and *Vice*. As a reminder, social capital refers to one's membership in different formal and informal networks (Siapera and Spyridou 2012). Based on their experience in the field, many senior journalists develop networks of sources and other contacts which can be utilised during various stages of the news process; particularly at the initial stage of access and observation (see Domingo et al. 2008). This type of capital is arguably more valuable to new entrants in the field, which tend to be poorer in social capital compared to more experienced members of the field.

[&]quot;organizational bridges and buffers, attempting to advance journalism professionalism throughout [their organizations]".

6.3.4. Young Staff

"[A] 'healthy newsroom' [combines] "a mix of very experienced people who know what they're doing and extremely hungry people who didn't learn any rules and don't know they're breaking them" – Janine Gibson, editor-in-chief of *BuzzFeed UK* (Gibson 2016).

As mentioned towards the beginning of this chapter, *BuzzFeed* and *Vice* have entered the field of journalism with the dual ambition of growing their audience share and advertising revenue (economic capital) and adding social status and prestige to their brands (symbolic capital). While the previous section demonstrated the important role played by senior journalists in enhancing both organisations' *symbolic* capital, this section is more concerned with exploring the role played by young journalists in *BuzzFeed* and *Vice's* news departments. As newcomers to the field, younger journalists seemed to be valued for being more amenable to accepting new methods and styles of working, and also for their basic homology with the audience. These two qualities both seemed to support *BuzzFeed* and *Vice's* goal of increasing their economic capital via *public recognition*⁴⁴.

One area where this was particularly apparent was tastes and aesthetics. Many interviewees recognised the important role played by younger journalists in helping their organisations "adopt a distinctive editorial voice" (Benson 1999, 468) that appealed to their specific audience. As one editor from *Vice UK* put it: "You know, our style was different to everyone else, because...the hosts, the presenters, the people doing it [were] much younger, because our audience [was] much younger" (Respondent F, editor, *Vice UK*). One senior editor from *Vice US* was more explicit, suggesting that "by [hiring] kids who were bred and moulded in the network news system we were able to genuinely and authentically create our own voice" (Respondent J, reporter, *Vice US*). This "voice" seemed partly a reaction to the objective style of traditional journalism,

⁴⁴ As Bourdieu (1998: 70) states, public recognition, "is measured by numbers of readers, listeners, or viewers, and therefore, in the final analysis, by sales and profits".

which one editor criticised for "lecturing" the reader, and "present[ing] material in a kind of didactic way" (Respondent E, editor, *Vice US*).

Owing to the homology between producers and audiences, some journalists also reflected on how they found it "easy" to write for the audience in a way that appealed, because they shared similar tastes and interests (Respondent H, reporter, *BuzzFeed UK*). Another reporter commented on this, while drawing attention to the cultural differences between older and younger journalists:

"I think the internet can sniff out authenticity versus disingenuousness. You can feel when there's a sad journalist who's been told by their organization: 'You have to write like BuzzFeed! You have to write internet speak', and it's so uncomfortable. Whereas... [our staff] they are so natural" (Respondent G, reporter, *BuzzFeed UK*).

The relative ease at which young reporters could communicate using "internet speak" was beneficial because according to one editor, it allowed *BuzzFeed* to "speak to specific audiences with more *authentic* voices" (Respondent D, reporter, *BuzzFeed UK*, emphasis added). In this context, authenticity seemed to be closely related to new norms of online media discourse, that, as Chadwick notes (2013: 176), "privilege conversational styles of expression and a certain communicative egalitarianism". As one *BuzzFeed* reporter explained: "If you were telling someone about a story you had read, how would you actually say it to another human being? ... it's about sound[ing] more human and not talking from your lofty perch of being a 'serious' journalist" (Respondent H, reporter, *BuzzFeed*).

Aside from tastes and dispositions, younger journalists also seemed to be valued by their organisations for their knowledge of, and expertise with, different digital platforms and technologies. As one *BuzzFeed* reporter argued: "We've all had Tumblr accounts, Twitter, Instagram, etcetera. We know how these [social media platforms] work a lot better than someone who's maybe in their early 30s...I think that's something that's quite innate in younger journalists" (Respondent M, reporter, *BuzzFeed UK*). Mastery over such

platforms was often linked to evolving practices in news gathering, giving *BuzzFeed* and *Vice* reporters a competitive advantage over traditional media, which were critiqued for their supposed lack of proficiency with digital technologies. As one editor from *BuzzFeed* said: "[Our journalists] will spot the [story] before it trend[s] is on Facebook. By the time something is on Facebook it is done. And that's often when The Independent will write a story about it" (Respondent K, editor, *BuzzFeed US*). A reporter from *Vice* made a similar point, suggesting that "a lot of [*Vice's*] sourcing" for stories had come from young journalists in the organisation "being able to network really well and build trust with different people [online]" (Respondent C, editor, *Vice US*).

Based on these responses, it was apparent that younger journalists played an important role in supporting *BuzzFeed* and *Vice's* goal of introducing new forms of cultural capital to the field. This capital consisted of "technical expertise, general knowledge, verbal abilities, and artistic sensibilities" (Benson 2006: 189), some of which arrived in "embodied" form (Siapera and Spyridou 2012), and some which seemed strongly conditioned by broader endogenous and exogenous forces. In terms of embodied cultural capital, young journalists came equipped with specialist skills, knowledge, and know-how concerning digital technology, which seemed to complement the exigencies of news production at BuzzFeed and Vice. By virtue of their inexperience, young journalists also seemed to share less of an *Illusio* or investment in the "game" of journalism, which made them ideologically predisposed to accepting new methods and styles of working. This supported BuzzFeed and Vice's goal of adopting a distinctive editorial voice, which seemed strongly tied to emergent norms in online discourse; based around ideas of journalism being less detached, and more emotionally authentic, subjective, unvarnished, and personal (see Wahl-Jorgensen 2016)⁴⁵.

⁴⁵ Note: I discuss the issue of emotion in news in Chapter 8 (section 8.3) of this thesis.

6.4. Topics and Beats

"At BuzzFeed, we really think of identity as being very complex, and trying to understand and have empathy for the broadest range of experience we can and trying to make media that allows people to have their own identity validated but also to have appreciation and understanding of other people's identities. That, I think, is why we have done so much reporting on things like sexual assault on campuses or LGBT issues or why diversity in media is so important to us. Being able to be broadly empathetic and broadly sensitive to different people's experiences I think is hugely important. I think that will lead to less polarisation if we do that well" (Peretti 2017d).

In this final section, I wish to consider the role played by news coverage in BuzzFeed and Vice's development. As mentioned earlier, upon entering the journalistic field, both organisations invested resources into covering a mixture of traditional and non-traditional beats, focusing on subjects like politics, crime, and business, as well as alternative topics like civil rights, gender rights, and mental health. This was also supported in the content analysis, which demonstrated that both organisations divided resources between a mix of traditional and niche news topics⁴⁶. As with staff recruitment, this balance of coverage can helpfully be understood from the perspective of field theory, with different beats working to generate different forms of *capital* for their respective organisations. Mirroring the findings on hiring practices, the results of this research demonstrate strong hybridity at both organizations. While from the outset, there has been a clear interest in reporting on alternative subjects relevant to young people, the desire for legitimacy and credibility has been a strong countervailing force, pushing BuzzFeed and Vice towards covering traditional "beats"; albeit in a manner more tailored to the specific needs and wants of their target audience. Before beginning the analysis, it is worthwhile providing a rough description of how news and other content is divided at both organisations.

⁴⁶ See Chapter 5 (section 5.3) for a more detailed breakdown of news content by topic.

6.4.1. Vice

Beginning with Vice, the division of the company into different content "verticals" makes it more complex to isolate where precisely news begins and ends. For example, although Vice News constitutes a separate vertical, there is also *Motherboard*, which focuses on technology and science stories, *Broadly*, which focuses on gender and women's issues, and Tonic, which mainly explores wellness and health issues. Consequentially, many stories covered by Vice's other channels bleed into news and current affairs. At the time of writing, the US and UK version of the Vice.com is divided into eleven distinct categories: Politics, Entertainment, LGBTQ (Lesbian, Gay, Bisexual, Transgender, Queer), Drugs, Opinion, NSFW (Not safe for work), Photos, Shop, Magazine (Vice 2017c). If a US or UK user requests to only see content from the *Vice News* vertical⁴⁷, they are taken to a page which is divided into eight discrete categories: Vice News Tonight, Money, Donald Trump, Technology, Immigration, Drugs, Features, and Terrorism (Vice News 2017d). As this suggests, Vice organise news in an atypical way, approaching both traditional and alternative topics thematically rather than by employing a clear "beat" structure. This was also confirmed in various interviews with reporters and editors from the UK and US office of *Vice*. As one US reporter explained:

"Organisationally, I think what I found almost everywhere...was that most newsrooms have some sort of beat designation, where you're the Middle East guy or you're the homeless guy, or covering courts, police. Things definitely seem to be more general assignment type-stuff here. You do what you want to do, so there's less specialisation at Vice, definitely" (Respondent J, reporter, *Vice US*).

Similarly, one reporter from the UK office explained how the office was not organized into "specific beats" (Respondent A, reporter, *Vice*), rather, individual

⁴⁷ There is now no specific page for *Vice News* on the UK version of *Vice's* website. Users are re-directed to the US news homepage if they click the '*Vice News*' tab on *Vice's* UK website. This is a fairly recent change.

reporters tended to be generalists; flexible and adept at covering various issues deemed relevant to *Vice's* audience.

6.4.2. BuzzFeed

BuzzFeed is less complex than Vice in that the company is broadly divided into three divisions: BuzzFeed News, BuzzFeed Entertainment, and BuzzFeed International (which includes BuzzFeed UK). The US news operation is by far the most developed, and compared to Vice, appears to have a higher degree of beat specialisation, employing reporters with expertise in areas such as politics, technology, the media, business, and health. However, similar to Vice, BuzzFeed has also pursued more thematic coverage, focusing on alternative topics like lesbian, gay, bisexual and transgender (LGBT) issues, gender equality, and civil rights (Ip 2015b). The UK operation of BuzzFeed is significantly smaller, but still employs a dedicated team of political and investigative reporters, as well as more experienced editors with expertise in science, business, media, and European affairs. As of 2015, they have also had an LGBT editor, and employed several reporters with experience writing about issues affecting different minority groups. Outside of its UK and US offices, BuzzFeed have smaller teams of reporters based around the world, working for various international editions of the brand. They also have a dedicated world news team, which is partly composed of foreign correspondents covering the Middle East, East Africa, West Africa, Asia and Latin America. Again, these reporters appear divided between more traditional subject areas like politics and health, as well as alternative subjects such as gender and LGBT rights (Herman 2017).

Despite the higher degree of beat specialisation at *BuzzFeed*, the general impression given to me by interviewees was that, like *Vice*, *BuzzFeed* largely favoured generalist reporters with particular interests or passions, as opposed to traditional beat reporters who were viewed as less flexible. As one reporter from the UK office explained:

"We have a lot of general reporters grouped into junior reporters and senior reporters depending on your level of expertise. And we sort of cover everything. We're not beat reporters, we do have particular subject areas of interest and people will gravitate towards different things...In the same day I might write about the EU referendum, and then a local issue in London, a crime story, or something funny...So it's quite a range" (Respondent B, reporter, *BuzzFeed UK*).

This ability of *BuzzFeed* reporters to "code-switch", as one editor from the US put it to me, was highly valued in a context where reporters might be asked to cover a range of subjects, some more serious than others, on any given day (Respondent K, editor, *BuzzFeed US*). Lisa Tozzi, global news director for *BuzzFeed*, alluded to this in a recent interview, suggesting that "at BuzzFeed News specifically, and for journalists in general, it really helps to be flexible and to experiment with different types of reporting and writing styles" (Tozzi 2018). Similarly, *BuzzFeed's* investigations editor, Mark Schoofs, has previously spoken about the importance of placing "bets on people, not beats" by giving reporters the "latitude to tackle a wide variety of stories" (Schoofs 2016).

6.4.3. Traditional Beats

Since the launch of their news operations, *BuzzFeed* and *Vice* have both devoted a significant proportion of their editorial resources to covering traditional beats. For *BuzzFeed*, perhaps the two most significant areas of investment have been political and investigative reporting⁴⁸. According to Ben Smith, editor-in-chief of *BuzzFeed*, it was investigative reporting, along with reporting on politics, that offered *BuzzFeed* the greatest opportunity to "break through" and achieve wider recognition (Smith 2016b). In a similar sense, Shane Smith of *Vice* seemed to recognise that traditionally esteemed genres of reporting might be the best way for his organisation to make an impact. This was most apparent in *Vice's* early international current affairs coverage, which,

⁴⁸ This is true of news both in the UK and US.

as discussed, often focused on dangerous locations and hostile territories around the world to bring viewers so-called "news from the edge" (Reid 2014).

In interviews, journalists at BuzzFeed and Vice frequently discussed these areas of coverage using terms like "reputation" and "credibility", drawing attention to the symbolic benefits that these highly-regarded genres of reporting brought to their organisations. For example, considering the value of BuzzFeed's investigations unit in the UK, one reporter reasoned that "we want people to take BuzzFeed news seriously as a brand...the aim is to get BuzzFeed's name everywhere, to get awards, to make an impact" (Respondent A, reporter, *BuzzFeed UK*)⁴⁹. A *Vice* reporter spoke about foreign reporting in similar terms, arguing that despite it always being a "loss-leader" you did it to "earn a reputation", which also helped "sell advertising" (Respondent D, reporter, Vice US). Beyond the promise of symbolic rewards, then, others recognised the commercial potential of investing in these subject areas: "there is obviously brand reputation. [But] there is [also] a business case for it in that advertisers...[they] really want the full package of coverage. Before, it was harder to sell our product against just 'here's funny stuff we found on the Internet type content'" (Respondent L, editor, *BuzzFeed UK*).

Aside from foreign affairs and investigative reporting, journalists from *BuzzFeed*, and to a lesser extent, *Vice*, also spoke about the importance of politics coverage. Here though, journalists were keen to extoll other benefits, beyond just symbolic and economic rewards. As one editor from *BuzzFeed* explained, regular contact between reporters and political sources delivered more tangible benefits such as *access* (to elites) and exclusive information ("scoops" in journalistic parlance):

⁴⁹ In one interview, Heidi Blake, investigations editor for *BuzzFeed UK*, put forward a similar argument about the importance of investigative reporting. As she explained: "The project we are all engaged in is building BuzzFeed into a major serious news brand that people trust, and feel is a vehicle if they are whistle-blowers with stories to tell" (Blake 2016).

"You have to display a certain level of commitment [to political reporting] ... there is an extent to which it is a demand for people who control access to politicians... we want to have credibility with those people and politics coverage helps you with that, because credibility once again is how you get scoops... it's because your name is trusted by somebody [that] you get your stories" (Respondent L, editor, *BuzzFeed UK*).

This same editor explained that an organizational focus on scoops⁵⁰ was "partly a carry-over from America", where the editor-in-chief, Ben Smith, had continually emphasised "scoop-getting" as a way of breaking into the political establishment and "building [one's] reputation very quickly" (Respondent L, editor, *BuzzFeed UK*).

The reciprocity between credibility, scoops, and access is perhaps best illustrated in the number of high-profile and exclusive interviews *BuzzFeed* and *Vice* have managed to secure in recent years. High-profile political figures such as Barack Obama and Justin Trudeau have both granted exclusive interviews to *BuzzFeed* and *Vice*. Meanwhile, in the UK, both organisations have conducted interviews and ran several live events with figures from many of the major political parties. Although journalists I interviewed rarely made references to these specific events, one producer from *Vice UK* expressed a certain degree of scepticism about *Vice's* courting of high-profile figures, questioning whether their reputation as an "edgy" news brand might ultimately be harmed:

"Vice is seen a little outside the mainstream sphere so hasn't had to play the [political] game so much. It's only really the past few years that it's got involved in this kind of thing, so it's early days for that kind of reporting for it as a company...We obviously have spoken to politicians and there have been some trying interviews that the company has put out with senior politicians that people weren't necessarily happy with. Erm, both within and outside the company... Vice often looks at things

⁵⁰ Note: I discuss the relevance of scoops at *BuzzFeed* (and *Vice*) in more detail in Chapter 8 (section 8.1) of this thesis.

in terms of what a high-profile interview could do for the company, as in 'will this get us loads of press, great', is that the only thing we care about? Does [the company's] relationship with high-profile figures change because they actually want to speak to them and have a relationship with them? So, you know, that may change. Whether that's a good or a bad thing I guess we'll have to wait and see" (Respondent F, editor, *Vice UK*).

This seemed to be less of an issue for journalists at *BuzzFeed*, perhaps because the organisation has never branded itself as particularly "edgy" or outside the mainstream. Rather, breaking into the political field seems to have been an ambition for *BuzzFeed* since the launch of its news operation; Ben Smith, editor-in-chief of *BuzzFeed US*, being an important early proponent of this strategy (Smith 2014a).

In the same vein as their hiring of experienced journalists, BuzzFeed and Vice's investment in traditional areas of journalistic coverage seemed to be rooted in a desire for public and professional legitimacy. This helps explain both organisations' focus on culturally revered genres and forms – high in cultural and symbolic capital - and therefore capable of conferring journalistic excellence as well as social status and prestige (Tunstall 1971). As Lowrey (2011: 67) notes, "a legitimated journalism can sanctify and render credible...[an organisation's] messages", which, from the perspective of BuzzFeed and Vice, has been particularly important given their status as new entrants in the field. Yet based on the comments of some interviewees, it also seemed clear that these high-status forms of journalism were being mobilized by both organisations for commercial purposes, as a means of boosting the status of their brands to attract premium advertisers. Naturally, this should invite a degree of scepticism about the purity of *BuzzFeed* and *Vice's* motives, or, indeed, their long-term interest in funding costly beats such as foreign affairs and investigative journalism. As one former editor reflected: "Vice News was a promotional vehicle for Vice... Rather cynically now I can reflect that it wasn't like 'oh shit, we need to save journalism from itself'" (Respondent E, editor, Vice US).

Beyond foreign affairs coverage and investigative journalism, the *political* beat was also referenced by some journalists as a high-status journalistic genre, capable of bringing intangible, symbolic benefits like prestige and credibility to their organisations. Others, however, drew attention to the tangible benefits that came with producing regular politics coverage, including access to powerful actors and the ability to get scoops. In this way, the political beat appeared to fulfil mixed reporting goals. As Skovsgaard and van Dalen (2013: 375) note, the political beat is also attractive to organisations because it tends to be "less affected by commercial pressure than...other prestigious beats". Hence, it can function as a particularly cost-effective means for organisations to "get themselves known and recognized [and]...make a name for themselves" (Bourdieu 1993: 59). Collectively, these factors help explain why *BuzzFeed* and *Vice* have made the political beat a central feature of their news coverage.

6.4.4. Niche Topics

"Now we do a lot of research, globally, on our hundreds of millions monthly users. And do you want to know what their top passions points are? Music, number one; number two: the environment; number three: civil rights; number four: income and inequality; number five: social justice; and number six: LGBT issues. Now Gen Y knows what side of history they want to be on. But are they getting the kind of media that serves these passion points? Simply put, no" – Shane Smith, co-founder and former CEO of Vice (Smith 2016c).

"If you look at the way media is structured, around both ethnic minorities and LGBT people, you have traditional, kind of strong niche publications, and these are treated as second-tier stories at mainstream publications. And really, I had always covered marriage very aggressively, because I think for a lot of people of our generation, that isn't a niche story. And when I was at *Politico*, there was this guy, Chris Geidner, who was just beating me on stories quite regularly, so he was one of the first people I hired at BuzzFeed. And then, you know, 2013, marriage — it was probably the biggest story in the country" – Jonah Peretti, CEO and founder of BuzzFeed (Peretti 2015).

Alongside traditional areas of coverage, *BuzzFeed* and *Vice* have invested editorial resources into covering alternative news topics, often based around progressive political issues including civil rights, gender equality, LGBT issues, and mental health. For *Vice*, stories on drugs (particularly cannabis legislation) and about strikes and protests have also featured prominently. This was supported in the content analysis, which showed that stories about social rights, issues, and protests accounted for around a fifth of total news coverage⁵¹.

Unlike traditional areas of reporting, these areas of coverage appeared more explicitly tied to the affordances of digital technology, and the needs and wants of the audience. In interviews, journalists were cognisant of this too, recognising that covering topics like mental health, LGBT issues, and gender issues, was partly a "deliberate strategy" aimed at reaching younger audiences, who had different interests compared to "standard paper audiences" (Respondent H, reporter, *BuzzFeed UK*). As one freelance journalist for *Vice* surmised:

"Vice is slightly younger and are quite progressive and tuned into the kinds of subjects that young people are interested in. Things like – you mentioned mental health and LGBT issues – which are being talked about much more generally among young people" (Respondent G, freelancer, *Vice UK*).

One reporter from *BuzzFeed UK* offered a similar summary of the audience: "We obviously know our audience is millennials, and they're often left-leaning. There's a complete gender mix. But they're quite open, liberal, and young basically" (Respondent A, reporter, *BuzzFeed UK*). As a consequence, one reporter explained that *BuzzFeed* had taken quite a "dedicated position" to issues like LGBT rights and mental health, partly because of their "general importance", but also because they "always [did] well with [*BuzzFeed's*]

⁵¹ See Chapter 5 (section 5.3) for a detailed breakdown of news by topic.

readers" (Respondent B, reporter, *BuzzFeed UK*). That some journalist seemed to possess distinct knowledge about what subjects "did well" with readers seemed to point to the growing importance of data and metrics in news (Zamith 2018). One reporter from *BuzzFeed UK* for example, spoke about how, in a quantifiable sense, issues like "mental health, LGBT, housing" performed very well with audiences, which reinforced the need to "do more" of these stories (Respondent H, reporter, *BuzzFeed UK*). Another reporter from *BuzzFeed* who specialised in social media news, suggested that stories based around identity issues (race, gender, sexuality, etc.) could be particularly effective at driving online engagement, because of their ability to "provoke emotion...whether it's finding it funny, finding it sad, or making [the audience] feel as though we're making them angry and wanting to support a social justice cause" (Respondent M, reporter, *BuzzFeed UK*).

At Vice, journalists also recognised the mutually reinforcing relationship between certain subjects and audience engagement. One senior editor, for instance, spoke about the decision to hire journalists who specialised in covering "civil rights... gender bathroom rights...queer rights", who had "a sense of what was shareable [on social media]". When asked if journalists sometimes selected stories based on what data and analytics had shown to be successful in the past, one editor responded: "I mean, to a little extent. For example, our audience really like stories about weed, so like, fairly often, we'll say 'hey, what's up on the weed beat'" (Respondent C, editor, Vice US). There was also a certain commonality or "homology" (Benson 1999), between producers and audiences that gave reporters a perceived advantage when writing about these subjects. For this reason, interviews with reporters on issues like civil rights, LGBT rights, and mental health, frequently transitioned into more personal reflections on their general importance, sometimes because reporters themselves identified as a member of the particular community they were writing about, or, more generally, because of their emotional investment in a particular topic or cause. As one reporter from *BuzzFeed* explained:

"I like to think that there's stories that we cover that other people wouldn't. Like, we write about different communities, for instance, we 105 have [name redacted], who reports on the British Muslim community, and not as erm, like, a white person looking in. But writing from within a community. We write just not about things that matter to our audience, but they are written by people who also care about it from within those stories...What I want to write about as a topic, often [reflects] how other people feel also, especially about things like race, gender...[and] it's not a pandering thing. You're not going to write about something well unless you also care about it" (Respondent G, reporter, *BuzzFeed UK*).

Accordingly, journalists from *BuzzFeed* felt they were able to speak on certain subjects with more "*authentic* voices" (Respondent D, editor, *BuzzFeed*, emphasis added). At *Vice*, the reporters I interviewed seemed less willing to personally identify with particular topics, however, one editor from *Vice* made the observation that it was "very difficult to distinguish the interests of the staff with those of the audience" which helped *Vice* appear more authentic on certain subjects⁵² (Respondent E, editor, *Vice US*).

Beyond these technological and audience-related factors, some senior journalists discussed these subjects in more strategic terms, recognising their importance in the context of *BuzzFeed* and *Vice's* struggle to distinguish themselves from other members of the field. As one editor from the US office of *BuzzFeed* argued: "*BuzzFeed* has certain beats that we do better than anyone else, like LGBT, and we would often break news relating to what's going on with transgender rights in the US that would not be featured anywhere else" (Respondent K, editor, *BuzzFeed US*). Similarly, an editor from *Vice* made the case that the organisation had been at the "vanguard" of civil rights coverage in the US, reporting on various protests and the activist group *Black Lives Matter*, before other media picked up on the phenomenon (Respondent E, editor, *Vice US*).

⁵² These references to authenticity by *Vice* journalists connect with comments made by Kevin Sutcliffe, managing director of *Vice Studios UK*, who described authenticity, rather than trust, as the "new battleground" for online media (Sutcliffe 2015). I return to the idea of authenticity later in this section and also in the section on objectivity, which can be found in section 8.3 of this thesis.

Taking these various responses in to account, it is worth briefly retuning to Bourdieu (1984) who, writing in *Distinction*, argues:

"[E]very change in tastes resulting from a transformation of the conditions of existence and of the corresponding dispositions will tend to induce, directly or indirectly, a transformation of the field of production, by favoring the success, within the struggle constituting the field, of the producers best able to produce the needs corresponding to the new dispositions" (Bourdieu 1984: 231).

Unpacking this observation, Bourdieu (ibid.) makes the case that exogenous forces condition individual dispositions (judgements, tastes, habits, and behaviours), which agents in the field of cultural production can respond to by producing goods that correspond with these emerging needs. This provides a useful lens through which to understand BuzzFeed and Vice's investment in alternative news subjects, which can be interpreted as a response to broader technology-based changes, as well as corresponding transformations in the sociocultural conditions of news consumption. Specifically, the focus on broad issues pertaining to identity seem strongly informed by the new, "connective affordances" of digital media, which have given rise to what Papacharissi (2016: 310) describes as "networked" publics - online communities populated by "likeminded individuals" and "sustained by feelings of belonging and solidarity". It would seem no coincidence that these online spaces are heavily populated by young people – BuzzFeed and Vice's principle audience – who, according to research, increasingly turn to news for self-validation, fulfilment, and sense of belonging (Bennett 2008; Costera Meijer and Kormelink 2014; Drok et al. 2017).

As new entrants to the field, both organisations have been well placed to seize upon these transformations, "rethink[ing] beat structures from scratch" (Smith 2014a), and in the process, introducing *new* forms of cultural capital to the field. This capital has helped *BuzzFeed* and *Vice* differentiate themselves from competitors, many of which have remained wedded to news subjects 107

associated with the traditional 'beat' structure of news⁵³. A vital part of this equation has been younger journalists, whose homologous relationship with audiences and personal interest in such issues has enabled them to appear more *authentic* when discussing these subjects. That *personal* experience was sometimes valued over *professional* experience speaks to the complex ways in which personal and professional identities tend to converge in the digital environment⁵⁴ (Steensen 2016). This would also seem indicative of a broader trend towards more personal, emotional, and identity-driven forms of news gaining currency online (Ip 2015b; Beckett and Deuze 2016; Wahl-Jorgensen 2016).

Finally, it is important to not neglect the political-economic factors behind *BuzzFeed* and *Vice's* decision to focus on these specific news subjects. As several journalists alluded to in interviews, part of the appeal of issues like LGBT rights, gender equality, and civil rights was that they were particularly effective at driving audience engagement. As Carlson (2018: 411) notes, this interest in engagement "results in the quantified audience – the reduction of a news user to a set of monitored and recorded characteristics in order to extract economic value". This also supports that niche subjects at *BuzzFeed* and *Vice* played an important role in growing their public recognition, "measured by numbers of readers, listeners, or viewers, and therefore, in the final analysis, by sales and profits" (Bourdieu 1998: 70).

6.5. Conclusion

Fields are generally defined by two forms of power, which exist in opposition to one another: economic capital and cultural capital. As Benson (2006: 189-90) writes, "economic capital, on the whole, is more powerful but cultural capital is always needed to transform good fortune into 'legitimate' fortune". This chapter

⁵³ This situation, however, does gradually seem to be changing (e.g. lp 2015b).

⁵⁴ In one interview, Jonah Peretti, CEO of *BuzzFeed*, seemed to allude to this, suggesting that "whole areas of coverage" had been opened up because of *BuzzFeed's* emphasis on their employees "lived" experience, as well as their professional experience in the field (Peretti 2015).
has described *BuzzFeed* and *Vice's* attempts at managing this feat. As discussed, the structure of both organisations would seem to reflect a duality of concerns with achieving two forms of recognition: public recognition, "measured by numbers of readers, listeners, or viewers, and therefore, in the final analysis, by sales and profits" (Bourdieu 1998: 70), and peer recognition, afforded to those who "internalize most completely the internal 'values' or principles of the field" (Bourdieu 1998: 70). This fits with DiMaggio and Powell's (1983: 149) general observation that organisations "compete not just for resources and customers, but for political power and institutional legitimacy, for social as well as economic fitness" (DiMaggio and Powell 1983: 149). It has also had a demonstrable effect on the way *BuzzFeed* and *Vice* have structured news: their funding and partnerships, recruitment strategies, and the division of news coverage – which, in Bourdieu's terms, collectively generate and are constitutive of, different forms of *capital*.

Dealing first with strategies related to public recognition, it is worth recalling that in moments of upheaval, success tends to favour those producers "best able to produce the needs corresponding to the new dispositions" of the audience (Bourdieu 1984, 231). The disruption caused by digital technology has produced such a moment, which BuzzFeed and Vice have capitalised on by introducing new forms of capital to the field. The hiring of young staff and the pursuit of non-traditional subject areas has helped both organisations develop a distinct editorial voice, colonise new areas of expertise, and differentiate themselves from competitors in the field. Collectively, these strategies have helped BuzzFeed and Vice reach new audiences, whose particular dispositions seem to reflect some of the broader social and structural changes that are reshaping expectations about the style, subject, and form of news online (Steensen 2016).

While these more market-driven forms of legitimation have supported BuzzFeed and Vice's desire for public recognition, a simultaneous desire to be recognised as legitimate by peers has led to an emphasis on traditional journalistic norms and practices. This is evident in the hiring of experienced journalists, along with an investment in traditionally-esteemed areas of 109 reportage, which, in contrast to strategies aimed at public recognition, appear more strongly connected to the promise of intangible rewards such as status, credibility, and prestige. This reinforcement and avowal of journalism's traditional cultural capital ultimately has a conservative effect on the field, ensuring a degree of continuity in journalism's established "rules" or doxa. There is an important practical basis for this too. As so-called "agents of innovation" (Carlson and Usher 2016), digital natives often introduce new practices to journalism, which, by virtue of being *new*, lack the legitimacy afforded to more traditional methods of gathering and producing news. In this way, an emphasis on long held norms and practices helps mitigate against some of the risk posed by innovation.

While these findings suggest a strong degree of *hybridity* in the way *BuzzFeed* and Vice construct norms and practices, it is important to try and avoid viewing these changes in simple, binary terms; in terms of "new" versus "old" (Zelizer 2015). Rather, a more accurate term to describe these changes comes from the early part of this chapter, which discussed the emerging norm of *integration*, and the way established editorial values have become more closely imbricated with commercial interests (Gade and Raviola 2009; Cornia et al. 2018). This was initially observed in *BuzzFeed* and *Vice's* use of native advertising as a primary source of funding; a practice which has since found purchase among other members of the field. However, there is also a strong case to make that other features of news production at BuzzFeed and Vice – the rise in generalist rather than specialist reporters, the transition away from a traditional "beat" structure, and the proclivity for more personal and emotional news subjects and narrative styles⁵⁵ – similarly reflects a growing imbrication of editorial and business interests. While I am cautious about falling into the trap of dismissing more emotional and personal styles of discourse as somehow less valid, it is also the case that many of the aforementioned characteristics of news production point to a form of journalism more beholden to market principles, and based around less substantive forms of journalistic expertise. Ultimately,

⁵⁵ Note: I discuss this specific aspect of news coverage in more detail in Chapter 8 (section 8.3) of this thesis.

this has the effect of undermining journalistic autonomy, as journalists become more subjugated to market demands and the need to drive higher (quantifiable) forms of engagement in the news (expressed via common metrics like *page views, time spent, shares, likes,* and *referred traffic*).

7. Producing News: The Routines of News Production at BuzzFeed and Vice

Having spent the previous chapter exploring some of the structural and organisational factors that influence news content at *BuzzFeed* and *Vice*, this chapter moves one layer deeper to study news production at the level of routines – defined by (Shoemaker and Reese 2014: 165) the "patterned, repeated practices, forms, and rules that media workers use to do their jobs". Routines emerge from a practical need to manage media production, helping organisations ensure that stories are produced on time and in an efficient, predictable manner. However, as many substantive ethnographies of the 1970s and 80s demonstrated, routines also play a more unconscious role in structuring news content, helping to explain "the relatively standardised form of news produced across news outlets, and also the ideological nature of news and its orientation towards social and political elites" (Cottle 2007: 4; see, for example Tuchman 1972; Hall et al. 1978; Schlesinger 1978; Soloski 1989; Gans 2004).

In the ensuing years, and with a few notable exceptions (Boczkowski 2004; Born 2004; Ryfe 2012; Usher 2014), routines have not been studied with the same energy or enthusiasm, due in no small part to the well-documented problems of gaining access to sites of media production (Paterson and Zoellner 2010). Nonetheless, from what little information can be gleaned, it is clear that digital technology has had a profoundly disruptive effect on the established rhythms and routines of news production (e.g. Deuze 2004; O'Sullivan and Heinonen 2008; Phillips et al. 2009). This has been most pronounced at legacy news organisations, where, contrary to the expectations of some scholars, organisations have "clung to institutionalized routines that traditionally have served news professionals", or only engaged in change efforts that are "fleeting, skin deep, [and] merely ceremonial...leaving core practices, competencies, and products intact" (Lowrey 2011: 65-67). In short, the particular affordances and cultures of digital technologies have not necessarily incited major shifts in the field's cultural capital or doxa, as was once predicted (see Borger et al. 2013 for a discussion of some of the normative ideas underpinning these predictions).

However, as Bourdieu (1998: 41) observes, fields are competitive spaces where actors engage in a more or less *constant* struggle "for the transformation or preservation of the field". As the subfield of online journalism has matured, there have been signs that the cultural capital of journalism might be slowly shifting as new norms and practices begin to gain purchase (Karlsson 2011; Hellmueller et al. 2013; Zeller and Hermida 2015; Beckett and Deuze 2016). In this context, digital native organisations have emerged as major sites of speculation about changes to news production, both among industry commentators and scholars (e.g. Bakker 2012; Riordan 2014; Küng 2015; Carlson and Usher 2016; Nicholls et al. 2016; Usher 2017). Although empirical evidence remains in short supply, research has indicated that the professional routines of journalists working for these organisations might be different from those working in other mediums, or in workplaces with more traditional cultures. For example, digital journalists tend to approach stories more holistically compared to their print counterparts (Ferrucci and Vos 2017); they tend to value speed over rigor (Cassidy 2005); and also display greater loyalty to audiences than sources (Steensen 2009; Agarwal and Barthel 2015).

This chapter interrogates these claims by shedding light on the organisational dynamics of news work at *BuzzFeed* and *Vice*, focusing predominantly on the routine practices of news reporters and editors. Four research questions have been developed to help steer the discussion:

- How do the three major sources of routines⁵⁶ audiences, sources, and organisational factors, shape news work at *BuzzFeed* and *Vice*?
- 2) What role does technology play in shaping routines at BuzzFeed and Vice?
- 3) How do routines both constrain and enable certain forms of news work at BuzzFeed and Vice?

⁵⁶ As defined by Shoemaker and Reese (2014: 168).

4) In what way do routines at *BuzzFeed* and *Vice* challenge or reinforce the entrenched cultural capital of the journalistic field?

To address these questions, I rely on findings from qualitative interviews with reporters and editors from both organisations. The primary interest of this chapter is to explore how routines are discursively constructed and articulated by journalists and editors at *BuzzFeed* and *Vice*. If any changes are occurring to journalism's core professional practices and norms, then, as Vos and Singer (2016: 145) suggest, this should be evident in the "discourse of the journalistic field, which will cast some new practices and ideas as legitimate and some as illegitimate". This chapter consists of one overarching section (7.1) on routines. This section is then divided into several subsections which explore key aspects of daily news work: the structure of routines (7.1.1); multiskilling (7.1.2); reporters' autonomy from internal influences (7.1.3); reporters' autonomy from external influences (7.1.4); and the impact of audience analytics and data in the news process (7.1.5). Unless otherwise stated, the following account of routines applies to both the US and UK offices of *BuzzFeed* and *Vice*.

7.1. Routines

Perhaps the first and most obvious thing to note about *BuzzFeed* and *Vice* is that they mainly produce news content for the web, which has a significant impact on the spatial and temporal dimensions of daily news work. In contrast to the print or television news cycle, where news production tends to follow an "industrial" logic, the rhythm and organization of online news is more fluid, decentralised, and flexible, with publishing taking place continuously throughout the day (Gade 2011; Anderson et al. 2012). This was certainly the impression given to me by journalists from *Vice* and *BuzzFeed*, though the unpredictable and erratic quality of the news cycle arguably created a stronger organisational imperative to manage and routinise certain aspects of news work.

7.1.1. The Structure of Daily Work

At both organisations, the day typically began with the morning editorial meeting, which took place somewhere between 9am and 10am. In the case of BuzzFeed (UK), I had the opportunity to observe one of these meetings first hand when I spent a morning inside the company's London office. In this instance, Stuart Millar, the head of news at *BuzzFeed UK*, led the discussion, with Janine Gibson, editor-in-chief, overseeing. The meeting began with a discussion of the day's previous news. Stories that had performed particularly well across BuzzFeed's various social media channels were singled out for special praise. At the time, this included a major investigation into corruption in tennis (Blake and Templon 2016), which had been published that week by the investigative team and shared widely on social media. Various news staff then took it in turns to update the room on what stories they were working on, with editors occasionally chipping in to offer their own suggestions. Some staff also offered ideas for new stories, which they received feedback on, or flagged up for a more detailed conversation later in the day. Editors also used the morning meeting as an opportunity to guide other staff on the day's news, mentioning particular events to look out for that might be newsworthy. At this stage, there was less discussion of logistics. Instead, following a brief discussion of stories and potential events, editorial staff dispersed and returned to their own desks.

Based on interview responses from various editors and reporters, a similar routine appeared to be in place at *Vice*. Early on in the morning, editorial staff would gather and writers would pitch ideas for stories to editors. Editors would also assign stories to writers and set deadlines for publication. On occasion, discussions of longer term projects would also take place, with reporters offering rolling updates on the status of their stories. Perhaps one more unusual feature of the pitching process at *Vice*, was that writers were encouraged to suggest headlines for stories during the editorial meeting and *before* stories were formally commissioned. As one editor from *Vice* explained:

"One thing that we do which I think is a common new media thing, is we decide the headline before the article...So we'll decide the focus of the

article and how we're going to package it before it's written, which I think is a new thing, and definitely not what I was taught to do in journalism school!" (Respondent C, editor, *Vice US*).

This same editor was keen to delineate between what they saw as "clickbait"⁵⁷ and practices such as the one described above which they viewed as a legitimate strategy for maximising the impact of a news story. However, they also recognised that this practice was not what they had been "taught to do in journalism school", implying these new criteria for newsworthiness (based on the audience) might be interpreted negatively because they did not conform to normative ideals of journalistic autonomy.

Following on from the morning meeting, news stories would move into the development stage, where individual reporters would work on their particular assignments, occasionally drawing on the expertise of technologists (e.g. website developers, data scientists, videographers), presentation staff (e.g. designers, illustrators), and various editors (e.g. homepage editors, social media editors), to assist with the development of news stories. The presence of continuous deadlines meant that there was technically no "end-product", and thus stories developed "all the time and in the moment" as one reporter from *BuzzFeed* explained (Respondent C, reporter, *BuzzFeed UK*).

To help with the coordination of this work, staff from both *BuzzFeed* and *Vice* spoke in real-time, using the instant messaging software, *Slack*⁵⁸. This helps explain why during the day I spent at *BuzzFeed UK*, the office seemed remarkably quiet. Many reporters wore headphones plugged into computer monitors, and often, the only audible sound was typing on keyboards⁵⁹. In a

⁵⁷ "Clickbait" is a colloquial term in the news industry often used to describe content that is deliberately misleading or fails to deliver on what it promises the reader (e.g. Hamblin 2014)

⁵⁸ For a broad overview and description of *Slack*, refer to Betters (2016).

⁵⁹ My own observations have been further corroborated by one employee from the UK office, Andy Dangerfield, the social media editor for *BuzzFeed UK*. In an opinion piece written for *The Drum*, he described *BuzzFeed's* newsroom as "far less noisy than the BBC, as most communication – from the sharing of gifs to breaking news – takes place on messaging system Slack" (Dangerfield 2017).

conversation with one editor, he described *Slack*, as a "rolling news conference [and]...the beginning and end of news production" (Respondent L, editor, *BuzzFeed UK*). Another editor from *BuzzFeed* described it as "instant chat on steroids" and particularly essential for communication between offices based in different time zones (Respondent E, editor, *BuzzFeed UK*). This, for example, would include the London office reporting to the New York office on the day's news events, including any stories that might merit further development or need additional promotion. At *BuzzFeed, Slack* also functioned as a space for communication between specific groups of employees (e.g. the politics team), as well as a space for activities such as sharing important information on stories, allocating editorial resources, and discussing different performance-related aspects of news like impact.

Similarly, although *Vice* staff mentioned *Slack* less often to me in interviews, in a report published in May 2016, *Motherboard*, the technology-focused vertical of *Vice* explained how the organisation sent approximately "5000 messages a week in eight *Slack* channels" (Jeffries 2016a). *Vice* described *Slack* as a space for reporters and contributors to "pitch story ideas and ask for edits", as well as "share interesting links" and even "vote on headlines" for stories (Jeffries 2016a). Perhaps nothing demonstrates the centrality of *Slack* to the production process better than the fact one team at *Vice* experimented with abandoning *Slack* because it was deemed too much of a distraction to reporters (Jeffries 2016a; Jeffries 2016b).

While constant communication via these online channels certainly helped structure the daily routines of news workers, beyond fixed events like the morning editorial meeting or individuals making occasional appointments to discuss production or scheduling issues, the general impression given in interviews was of a news production process that seemed less tightly controlled and more fluid and decentralized⁶⁰.

⁶⁰ This has been acknowledged publically by various senior members of *BuzzFeed* and *Vice*. For example, describing the organization of news at *BuzzFeed*, Ben Smith, editor-in-chief, explained: "You know, I think we're very decentralized, and that's the reason we've been able to build very fast, with very high quality. For instance, I don't

At times, this was contrasted with "older" methods of organising news work, which were often criticised for being slow, inefficient, and overly-bureaucratic; "a tottering, petering pile of management on top of management", as one editor put it to me (Respondent N, editor, *BuzzFeed*). As a consequence, journalists frequently made out the news process at *BuzzFeed* and *Vice* to be more nimble, reactive, and efficient, partly because a "flat[ter] organisational structure" meant stories that did not have to move through "as many layers of hierarchy" (Respondent L, editor, *BuzzFeed*). At Vice, one US editor offered a slightly more haphazard description, suggesting that the news process was a "soupy mess of collaboration...of having various people doing various things and having it all just kind of come together; that's what Vice News was" (Respondent E, editor, *Vice US*).

7.1.2. Multiskilling

Working under fewer layers of hierarchy (e.g. mid-level management such as assignment, assistant, pictorial, and copy editors), individual reporters at *BuzzFeed* and *Vice* appeared to approach stories "more holistically" (Ferrucci and Vos 2017: 879), taking on more responsibility for their own work. As one reporter from *BuzzFeed* explained:

"When you work for print you just have to write, so you send off your stuff and then you've got the subs desk to re-read it and put the headline on it, you've got the picture desk and stuff. And now, obviously we have editors who check our own work but I have to correct my own typos and I'm the one creating the article in the CMS [Content Management System] so I have to put in my own pictures and headlines as well" (Respondent J, reporter, BuzzFeed).

necessarily approve what any team publishes. They have the power to do that. People know what we want, but also the decision-making is very distributed" (Smith 2015a). Similarly, Kevin Sutcliffe, former head of news programming for *Vice*, explained how *Vice* was "leaner and quicker... [and less] subject to bureaucratic pressures" compared to traditional media organizations, which he suggested made the news process more efficient (Sutcliffe 2015).

Similarly, another reporter from *BuzzFeed* suggested their routine consisted of "mostly independent work", apart from the occasions when they would seek out "help and advice" from superiors, or contact specialists to develop story assets like interactives, pictures, and illustrations (Respondent C, reporter, *BuzzFeed*). For one reporter, the nature of work at *BuzzFeed* was indicative of a broader trend in journalism, toward "multiskilling"⁶¹:

"Now, I think [journalists] are not just specialist trainees [in one medium], but have these skills to write stories for every platform...So it's changed in that you're expected to do more. You can't just write news stories [in one medium] ...You've got to be much more multi-skilled and better with technical equipment" (Respondent H, reporter, *BuzzFeed UK*).

An emphasis on multiskilling, was also evident at *Vice*, with individual reporters taking ownership over different aspects of the news process. Two news producers from the UK and US offices discussed with me at length their working process for creating video dispatches:

"In Vice, we are always operating in small teams, partly due to budget constraints, partly due to working in a hostile environment. As a producer at VICE you have all the responsibilities basically. It's not like we often got to work with APs: assistant producers, associate producers – because there weren't any, at least in the UK office. [So] if it's your story, you've got to own it...you've got to write the treatment, you've got to put together a budget, you've got to figure out fixers, drivers security, if necessary, you've got to make sure you've all got the right safety equipment...If you are doing dispatches from the field you're bringing back footage to the editor in the hotel and you're there with them, cutting

⁶¹ "Multiskilling" is a term that generally describes how, as a result of declining resources, journalists have had to become more adept at taking on new roles and managing new tasks in the news process (Bromley 1997; Deuze 2004; Klinenberg 2005).

it, putting it together, making sure it goes out" (Respondent F, reporter, *Vice UK*).

"[I would spend] my days reading, researching, phone calls, picking [stories]... a lot of meetings, a lot of trying to win people over to what I wanted to do...If I had video I shot, getting back, trying to edit it, working with an editor, writing scripts...it differed a lot. Some days or weeks were laid back...But I would go off somewhere and work very very hard, really strain myself and then take a little break and then edit" (Respondent D, reporter, *Vice US*).

Another news reporter from the UK office described a similarly onerous set of responsibilities during the news production process:

"I do everything from reporting on the ground to the research to the sorting things out. If I have a story I've got to research it all myself, get all the contacts myself, go out, do it; it's really a lot of work man" (Respondent A, reporter, *Vice UK*).

While journalists from *Vice* admitted this could be taxing work, they generally seemed to appreciate the being given additional responsibilities. According to one editor, journalists' early exposure to different "methods, workflows, and ways of doing things" was an inherent part of *Vice's* appeal. As he put it, "I think a lot of people were attracted to coming and writing for [*Vice*] because you could [wear] so many different hats...at other places it would be way more regimented. You'd be way more typecast as a video person, or whatever" (Respondent E, editor, *Vice US*).

7.1.3. Autonomy (Part I)

In the context of these working arrangements, reporters I interviewed from both organisations generally felt they had a significant degree of autonomy in dayto-day work. Often, this was discussed in the context of pitching stories, with journalists determining they generally had more opportunities to pick stories inline with their own preferences, rather than those of their superiors. As two journalists from *Vice UK* explained:

""[Vice] are always open to new ideas, so if you're keen [on a story], they'll let you give it a go [and] that was always a big appeal. Knowing friends at other places who didn't have that kind of freedom you always felt you were having a better time of it in that regard.... You had this freedom...being able to go to these places and do what I always wanted to do was such a massive appeal for me" (Respondent F, editor, *Vice UK*).

"The upward mobility here, if you put the work in, if you work hard [then] you can get places. Within a month [of joining Vice] I'd made a doc...I mean, we [still] have a lot of checks which is good...So if you're pitching to the UK office then the US office have to sign it off. But, compared to other places. You know, I've got friends who are like 'oh it took me a year to get this signed off' and I'm like 'it took me two weeks'...We don't know how lucky we are...It's really fluid [here]. You can literally get up, walk to your boss, and be like 'here you are, I've got this idea' and he's like 'yep, that sounds good, write me a one-sheet' and then the next day, budget, booked, gone...It can be that quick...I pretty much cover whatever I want to really" (Respondent A, reporter, *Vice UK*).

As above, these examples were sometimes contrasted with the fortunes of journalists at other media, who were perceived to enjoy less individual autonomy as a result of stricter hierarchies and additional layers of bureaucracy.

This being said, feelings of autonomy were not necessarily uniform across the whole organisation of *Vice*. On the contrary, several journalists I interviewed

from the US office implied stronger top-down control from news editors⁶². As one reporter commented:

"Sometimes I produced [stories] myself. I would pitch stories, I would be assigned stories, it was about fifty-fifty...I would [spend time] trying to win people over to what I wanted to do" (Respondent D, reporter, *Vice US*).

In fact, one editor cited the lack of time for reporters to work on their own stories as one reason why the US office had experienced a general dip in morale:

"[The] people who we had on staff, who we had in our newsroom, they were cracking it...and people got tired of that pace. We did what we could, morale-wise, to ensure that each member of our staff, would, apart from the daily stuff, get to work on original stuff that had conceived themselves, that they were personally invested in...I think that worked for a while, but ultimately, morale began to dip" (Respondent E, editor, *Vice US*).

At *BuzzFeed*, journalists also suggested that despite editors having the "last word on everything" (Respondent J, reporter, *BuzzFeed UK*), they were given an ample degree of latitude in news work. As one UK reporter said:

"If there's a big breaking news story or we see something that has been developing for a while, my editor might say 'maybe watch this and see if there's something we can do with it'. Less so in my role on story day-today and being like 'can you write this up' A lot of the time, I'm very lucky to have my editors believe in my judgement. If I see something that I

⁶² One possible explanation for this is that by virtue of being larger and better resourced, *Vice US* could afford to employ more staff (including editors), compared to *Vice UK*. Another reason for higher degrees of editorial control might relate to the fact that most senior members of *Vice's* news team were based in its US office, in Brooklyn, New York. On a couple of occasions, *US* editors spoke about this, suggesting that these managerial figures sometimes played a part in decisions about what news stories to commission.

think will work, or they come to me and say – 'do you think this will work on the website?' – they'll trust me. Mainly because being a 20-something female, I kind of represent a demographic of BuzzFeed readers" (Respondent M, reporter, *BuzzFeed UK*).

As per the comment above, *BuzzFeed* reporters were perhaps more inclined than *Vice* reporters to view their autonomy as a direct consequence of their homology with the audience. An editor I spoke with elaborated on this point during one interview:

"[At BuzzFeed, we think] it's okay to employ talented people in their early 20s and give them a go of it and give them good editing and guidance and they'll have ideas. And so, from those background assumptions that the traditional media maybe does not have... The traditional media likes people to serve their time a bit because all the senior people did, so bloody hell, everyone else is going to have to, and the traditional media likes to say...'what stories we'll cover what stories are important is'...tends to flow top-down. And...[while] we value experience and our editor-in-chief and our head of news do have final say about what stories we do, there is a slight inversion of that...Reporters should be able to pursue things they think will work. Hire people that are like your target audience, you know. They can...they will have good ideas and will do them well and trust them to do that" (Respondent L, editor, *BuzzFeed*).

In this light, autonomy was not only an outcome of the specific conditions of news production, but also a product of the similarity in dispositions between reporters and audiences. Reporters were trusted to self-select news because of their ability to integrate audience preferences into their own sense of news judgement. This would seem to stand in direct contrast to much of the earlier literature describing the relationship between reporters and editors, which has typically portrayed reporters as more source-oriented and editors as "more conscious of what will have audience appeal" (Reese 2009: 288; also see Gans 2004: 89-90).

7.1.4. Autonomy (Part II)

For the most part, then, autonomy was understood by journalists to mean a degree of independence from *internal* or organisational influences. Yet this leaves the influence of external actors largely unaccounted for. As the literature on news production routines has shown, both *sources* (e.g. Fishman 1980; Gans 2004; Strömbäck and Nord 2006), and, more recently, *audiences* (Steensen 2009; Agarwal and Barthel 2015; Ferrer-Conill and Tandoc 2018), have played an important role in shaping autonomy in news organisations. To address these issue, interviewees were asked questions about their daily routine, including how they generated news stories and what kind of influence the audience had on their daily work (see Appendix B).

Here, perhaps the first general observation to make is that the relative autonomy of journalists at *BuzzFeed* and *Vice* seemed partially dependent on 1) their specific role within the organisation, and, 2) the particular events of the day. For example, those reporters with a beat specialisation who experienced regular contact with sources, likely encountered more constrains in news work, both via informal means (e.g. off-the-record conversations, leaks) and formal means (e.g. interviews, press conferences, briefings). One reporter from *BuzzFeed* seemed to acknowledge this, suggesting that reporters working in beats like politics were "slightly [more] anchored" by "fixed events" compared to generalist reporters (Respondent H, reporter, *BuzzFeed UK*). Beyond beats, routines could also be constrained by random occurrences like breaking news events. For example, incidents like natural disasters and terrorist attacks typically demanded a rapid diversion of organisational resources to focus efforts on time-sensitive activities like researching and verifying information, and on-the-ground reporting.

Aside from these more obvious constraints, journalists were not particularly forthcoming in speaking about other external factors that encroached upon their autonomy and news judgement. That one of journalism's core normative commitments involves maintaining a degree of independence from political and economic pressures (Kovach and Rosentiel 2001), this made sense as a rhetorical strategy. At the same time, it was possible to glean from interviews that journalist's image of independence in news work might be more illusory than they wished to imply.

This was most apparent when journalists spoke about their relationship with the audience. Several reporters and editors I interviewed seemed to possess specific knowledge about the habitus of the audience – their particular wants, habits, interests etc. – that they habitually factored into their decisions about news selection. This newfound knowledge was often attributed to digital technologies, which provided new tools and techniques for quantifying and measuring audience behaviour (see Carlson 2018; Zamith 2018). Reporters from *BuzzFeed* were perhaps most explicit on this subject, explaining how:

"The huge advantage of writing digitally is that we know exactly what our readers are reacting to because you can see it for each individual story...We can see what people care about and what people want to read and cater to that and that's super important" (Respondent I, reporter, *BuzzFeed UK*).

That catering to audience's needs was seen as "super important" was partly attributed to a rise in audience autonomy⁶³. Reporters were aware that the "people formerly known as the audience" (Rosen 2006) had acquired more power in self-selecting the news, and consequentially, felt a stronger need to tailor content to their particular needs and preferences. As the same reporter above argued: "The big difference with the Internet...is that you can't tell people what they want any more...You have to work out what people want and give it to people in the form they think they want it" (Respondent I, reporter, *BuzzFeed UK*).

⁶³ The term audience autonomy, as Napoli (2012: 84) suggests "refers to how contemporary characteristics of the media environment—ranging from interactivity, to mobility, to on-demand functionality, to the increased capacity for user-generated content—all serve to enhance the extent to which audiences have control over their interactions with media".

In contrast to *BuzzFeed*, reporters at *Vice* were generally less revealing about how the audience factored into their sense of news judgement. However, there were definite signs that the audience were treated as an important constituent in the news process. One reporter, for instance, was critical of the apparent insularity of traditional news organisations, which he argued were preoccupied with "making journalism for other journalists", while at the same time, neglecting the more immediate needs of the audience (Respondent A, reporter, *Vice UK*). However, it was not completely clear from interviews precisely *how* the audience shaped *Vice* reporters' sense of news judgement. Individuals were certainly cognisant of the audience in terms of their tastes and interests, which appeared to affect certain decisions regarding the style and aesthetics of news. As one reporter commented:

"Vice had gathered this young audience who were mainly here for cultural stuff, art, music...and my feeling [with news] ...was that we need to tell [stories] in a way that they get it. Give them the information they need but also make sure they are invested in it emotionally...And so, you know, our style is different because we're making this stuff for sixteen to maybe thirty-year-olds...our style was very different to everyone else" (Respondent F, editor, *Vice UK*).

On the other hand, perhaps because of the basic homology between reporters and audiences, it was also assumed that "what interested [reporters] would interest the audience" (Gans 2004: 229). In this way, reporters from *Vice* appeared less audience-oriented than reporters at *BuzzFeed*⁶⁴.

However, while some *reporters* from *Vice* conveyed a degree of independence from external influences, senior figures offered a different assessment. One reason for this might relate to the basic division of labour at *Vice*, with editors

⁶⁴ One important caveat here is that several of the journalists I interviewed from *Vice* seemed to spend more time producing video content than textual content. It is possible that reporters producing text-based coverage would have been subject to different kinds of pressures compared to video reporters, owing to their substantially different production schedules.

appearing to take on more responsibility for making the news "acceptable to the audience" (Gans 2004: 89). This covered a range of activities, from deciding the appropriate framing of an article, to monitoring the performance of stories using *Vice's* internal analytics. Collectively, these activities were about ensuring stories "reach[ed] their full potential in terms of impact", according to one editor (Respondent C, editor, *Vice US*).

With editors from *Vice* appearing to shoulder most of the responsibility for making stories appealing to the audience, reporters were more insulated from integrating the audience into their own sense of news judgement. As the editor above noted:

"I think all editors pay attention [to analytics] but we actually try have our writers not to be too invested in how many hits their stories get because I don't think that's what they should be worrying about" (Respondent C, editor, *Vice US*).

"Worry" was a suitable word here, because by and large, most journalists I spoke with – and in particular, more senior figures from *Vice* – seemed troubled by the audiences' newly acquired power in the news production process. As one former editor from *Vice US* reflected:

"So, your point about the audience...and this is particularly true of digital upstarts, or upstarts in the news media world, that they're keenly aware of the audience and keenly aware of catering to the audience to some degree...And I don't necessarily mean in terms of puppy videos or listicles or things of that sort, but nevertheless, being like, 'look, we've got mouths to feed'. I'm not sure how much the New York Times' editors wake up and be like 'we've got mouths to feed'. They're more like, 'look, we've got stories to cover' and then [the production process] sorts itself out...Whereas at Vice, it was like, 'this is a business', and we have phantom subscribers...that subscribe to this brand of Vice and we need to make them feel at home" (Respondent E, editor, *Vice US*).

From the perspective of this editor, *Vice's* responsibility to the audience was partly a consequence of being a new entrant to the field, which produced a deficit in certain assets, that other, more established news brands could rely upon as capital⁶⁵. Here, it was telling that *Vice* was contrasted with the *New York Times,* which was cited favourably as an organisation where news judgement remained under the jurisdiction of journalists. By contrast, *Vice* was pejoratively referred to as a "business", implying that commercial pressures (or having "mouths to feed") might be encroaching upon editors' professional sense of news judgement. Similarly, another reporter from *Vice* expressed concerns that journalism had become more driven by a "looming idea of audience engagement", meaning organisations had to work harder to be "seen and heard more…through titillation, some sort of ideological affinity…or [by establishing] a voice or brand that appeals to a particular segment" (Respondent J, reporter, *Vice US*).

At *BuzzFeed*, interviewees similarly noted a link between their status as a new entrant and the need to cater to the interests of the audience. However, they generally appeared less concerned about this development compared to journalists from *Vice*. As two reporters explained:

"I think one thing that you're aware of [at *BuzzFeed*] is that there is no automatic audience. We don't get a tonne of front-page [traffic] and so you've got to actually think, 'is this important? ... Is this a story that people want to hear? Or should hear and might share?" (Respondent H, reporter, *BuzzFeed UK*).

"Yeah, so someone from [legacy publication] – I asked him recently what he liked better about writing at the [legacy publication] versus for us, and he said what he liked at [legacy publication] was that if his news story was put on the homepage, it was guaranteed 100k traffic or something,

⁶⁵ As Nicholls et al. (2016: 11-12) argue, "digital-born news media organisations…have to establish themselves without the assets that legacy media have, including brand reputation, loyal audiences, and revenues generated by print and broadcast activities".

and his editor saying to him 'well done'. And it was ultimately the editor's choice what went on the homepage. [Whereas] here, he actually has to think about what people actually want to read. *The world is your editor*" (Respondent G, reporter, *BuzzFeed UK*, emphasis added).

Given there was no "automatic audience" for news, reporters at *BuzzFeed* appeared relatively comfortable with the idea of factoring audience preferences into news judgement, connecting this positively with a greater sense of independence⁶⁶. Moreover, unlike many journalists from *Vice*, reporters from *BuzzFeed* appeared to find the idea of a journalism more tailored around the needs and wants of audiences *normatively desirable*. In this manner, audience choice functioned as the "seemingly more democratic…substitute for the internal standards" by which journalists judged news stories (Bourdieu 1998: 73). In a more recent article discussing the impact of audience analytics, Tandoc and Thomas (2015: 248) discussed this feature of contemporary news production, noting how the "broader trend in media toward customization and choice", was interfering with journalists' ability to independently select the news.

7.1.5. Audience Analytics and Metrics

Whether or not journalists were supportive of these changes, new considerations about the needs and wants of audiences at *BuzzFeed* and *Vice* brought new questions to the fore regarding how to judge the newsworthiness of particular issues or events. Often, such considerations were expressed in the language of audience metrics⁶⁷, with terms like *traffic*, *hits*, *clicks*, *views*, and *shares*, being regularly deployed in discourse. Yet in the same way opinions on audiences diverged between journalists at *BuzzFeed* and *Vice*, so did attitudes regarding audience data and metrics and their implementation in

⁶⁶ This comes across slightly paradoxical, but, as mentioned, reporters' ability to integrate audience preferences into news judgement seemed to provide them with a greater degree of freedom from superiors in deciding for themselves what stories to cover.

⁶⁷ A metric refers to a single, discrete type of data (e.g. page views; number of comments; number of shares), usually derived from a broader suite of analytics software (see Cherubini and Nielsen 2016 for a good overview).

the news process. At *Vice*, there was a definite awareness among editors and some reporters of audience metrics, the most prominent being web traffic, which subsequently had an influence on decisions about news selection. As one reporter from *Vice* explained:

"As far as traffic numbers, [there was] not day-to-day [pressure], but there was definite concern...it wasn't necessarily relayed on the news staff but on the editors or supervisors. There were weekly traffic meetings where they looked at the traffic numbers, analysed, figured out where they were deficient and so there was some pressure there, I think. I just don't know how it was projected on day-to-day work" (Respondent J, reporter, *Vice US*).

Editors from *Vice* were helpful in expounding further how traffic pressures manifested in daily work. One interviewee from the US office, worth quoting in full, described their routine as follows:

"We had to get cracking on the site immediately and so [stories] would be assigned, deadlines would be assigned. Much of the turnaround would be day of...We had to be highly selective. Whatever we thought was trending [on social media] the most in the morning that wasn't already spent [or] something that had hit the wires within a reasonably short amount of time; there was still room to carve some space out on that. That was the rationale for much of morning assignments...We [also] did a fair amount of enterprise reporting...but that wasn't going to generate 10 million monthly uniques...If you're going to get 10 million unique visitors a month, you're doing it by covering Obama day of, you're doing it by covering Trump day of. The Italian earthquake day of, like, within an hour or two. You're jumping on every mass shooting that's taking place in the United States, which we did, you know what I mean... So, we needed to be on top of breaking news. If something happened, the quicker we were on top of that with incremental coverage we would publish and push out on Twitter, the more we would get out of it, you know what I mean... That's how you get indexed on Google News. We 130

were trying to get our SEO [search engine optimisation] up. Otherwise, with your original enterprise reporting, the very Vice-y, sort of niche, you know, when you find the real under the radar story, unless it is very explicitly tied to something super relevant in people's minds, you've got to promote the hell out of it for people to know that it's a thing! It's going to be harder to make hay out of that stuff consistently whereas the stuff that's however many thousand tweets on Twitter, you know, if you can get something in there, that's in the flow, in that feed, in the continuum where millions of people are already looking for that sort of thing, then you're going to get a lot more aggregated traffic out of it. So, it's just logic" (Respondent E, editor, *Vice US*).

As this description suggests, the presence of data and metrics in the newsroom (and the concomitant pressures of meeting preordained traffic targets), encouraged an emphasis on editorial practices oriented around the coverage of breaking news events and trending topics, both of which promised to deliver high returns in traffic and attention. By contrast, "enterprise" or original reporting was deemed riskier, given its unpredictability in attracting a large audience. From this perspective, it is easier to appreciate why editors at *Vice* expressed scepticism about the involvement of the audience in the news process. As a "quantifiable, rationalizable, largely consumptive aggregate" (Anderson 2011b: 550), the audience were primarily viewed through the prism of data and metrics, using reductive terms like "traffic", which had material consequences on the daily work of journalists by stripping them of some of their autonomy in deciding what news to cover. As the editor above explained:

"There were always traffic targets, but then as far as the extent to which they would always be made plain to us, or there would always be a fire under our sheets, that wasn't really the case you know. Sometimes we would be just shy, sometimes we'd hit it better, and we understood that whenever a traffic target was made plain to us, it always seemed very aggressive, if you know what I mean...Ultimately, we were always trying to tell interesting stories and predicating our day to day on the idea that doing what we did well would result in the kind of hits that they were 131 looking for... And we knew what kind of material spiked and we would jump on those kinds of stories and we would want to tell them well" (Respondent E, editor, *Vice US*).

Given that editors seemed to possess most of the responsibility for fulfilling these targets, reporters rarely spoke about data and metrics in interviews; and when they did, it was chiefly to note that they paid little attention to them.

At *BuzzFeed*, journalists' attitudes towards data and metrics appeared to flow from their attitudes towards the audience, with reporters seemingly happy to integrate their knowledge of the audience into decisions about different aspects of news stories. Their relative contentment also seemed to feed a general curiosity in the uses of data and metrics, beyond broad references to metrics such as "traffic". This was most evident in discussions on the metric of "sharing", which, as Jonah Peretti (2014) has suggested, underpins much of *BuzzFeed's* thinking about content creation:

"[For *BuzzFeed*] sharing has always been the biggest metric because it shows that someone thinks a piece of content is worth passing on to a friend...Word of mouth is the distribution [of content] ...You are getting so much data back about what people like and what people share and that can immediately inform the media you create".

For journalists at *BuzzFeed*, sharing was held in similar esteem and recognised as distinct from other metrics, which were often reproached for being linked to cruder forms of audience measurement and more culturally debased practices like news aggregation (Anderson 2011c; Bakker 2012). As editor-in-chief of *BuzzFeed UK*, Janine Gibson, commented once in an interview:

"It's a fundamental mistake that people make about BuzzFeed that we are all about getting traffic. If an article is shared by relatively fewer people but really speaks to them, that's incredibly important to us...BuzzFeed journalists, editors, and video-makers study relentlessly what will cause their audience to share their content. We don't deal in big round numbers, we deal in metrics of sharing...The key to the success of news is working out how to marry those learning with more traditional journalistic skills of running down a story, finding out who's doing a bad thing and trying to cover it up" (Gibson 2016).

Reporters I spoke with generally echoed these sentiments, discussing how factors related to the metric of sharing fed into decisions about newsworthiness:

"Seventy-five percent of our traffic comes through social media referrals. Not homepage clicks which is extremely rare. Like, the BBC, Guardian; most of their traffic comes through homepage clicks. But BuzzFeed is not a destination homepage. You don't wake up in the morning and be like 'Oh I'm interested in the news, let's see what BuzzFeed is writing about'. You're going on Twitter, on Facebook, and then you go 'oh that story's interesting' and click through that way. So, when we're doing investigations or commissioning one, we're always thinking 'okay, is this something people are going to talk about? Is this something people are going to want to share?'" (Respondent A, reporter, *BuzzFeed UK*).

It would be fair to assert that working out what news people felt compelled to share was not an exact science, but rather, an evolving and ever-changing discipline that relied on user data being "captured, analysed, and manipulated in a perpetual loop of analysis, interpretation, experimentation, feedback, and refinement" (Küng 2015: 58). As a result, many journalists carried certain notions about what made stories shareable, ranging from the broad to the specific. One reporter, for example, emphasised that sharable news typically had to contain a "human relatable element", whether this concerned an issue relating to personal identity (e.g. race, gender, age), or a subject likely to provoke a strong emotional reaction in readers⁶⁸ (Respondent C, reporter, *BuzzFeed UK*). Beyond these broad observations, it was apparent that many reporters also carried a mental stock of more detailed knowledge about

⁶⁸ I discuss the use of emotion in news in more detail in the next chapter of this thesis.

particular tactics and strategies that heightened the possibility a news story being shared. Various reporters discussed the reinforcing relationship between sharing and, for example:

a) particular news subjects

"[It's partly] about working out what articles have done well in the past...So like, if it's something to do with mental health that tends to do quite well. If it's something to do with women then that tends to do quite well. Erm, periods do quite well! There's little things, like, little tick boxes in your heard, like 'yes, it's a woman, it's about mental health'; you create tick boxes as you go along" (Respondent C, reporter, *BuzzFeed UK*).

b) images

"Stories that are visual [tend to do well]. Obviously, every story that we publish we create an image thumbnail for it, and we try, with that thumbnail, to either put an image which will hook someone or explain the story quite quickly. A lot of the time, stories with a very strong image, like the image of a woman, or something that will make people emote...do very well" (Respondent M, reporter, *BuzzFeed UK*).

c) headline framings

"We're always thinking about headlines, different headline framings. We have these spreadsheets and documents that we're putting them in, down to like, what adjective seems to be doing really well...We're always learning, and then if we hit on something, we will keep using it. We know it's something that works, and then eventually people kind of get sick of it. But also everyone's on the lookout for the next formulation that works, or topic, erm, or anything like that" (Respondent G, reporter, *BuzzFeed UK*).

"BuzzFeed is known for having good internal analytics. It's about keeping a bit of a sense of it, you know: what's connected and what hasn't. You know, anything that fact-checks an anti-Corbyn story [and says it was] wrong is going to do brilliant because Corbyn is shared. Corbyn supporters share, SNP supporters share, Tories and mainstream Labour don't...Women share more than men, gay people share more than straight people...straight white men tend to be a less important demo for BuzzFeed than for others because they share less on social media...and since everywhere else is catering to them anyway, it doesn't matter if they're slightly less served here" (Respondent H, reporter, BuzzFeed UK).

Based on interviews alone, it was difficult to decipher how this knowledge was operationalized in every day routines. However, it seemed clear that data and metrics were a significant factor, if not a "primary ingredient" (Anderson 2011: 561) in news judgement at *BuzzFeed*, affecting the manner in which journalists thought about stories, ranging from what subjects to cover to what presentational devices to employ in order to maximize the reach and impact of stories. As one *BuzzFeed* reporter said: "We are build[ing] up this institutional knowledge of what works and what doesn't" (Respondent G, reporter, BuzzFeed UK).

7.2. Conclusion

This chapter has sought to provide an account of news production routines at BuzzFeed and Vice, paying close attention to the three major constituents of routines – audiences, sources, and organisational factors – and the role they play in enabling and constraining certain journalistic practices (Shoemaker and Reese 2009). This discussion has taken place against a backdrop of broader social, economic, cultural, and technological trends. The shift from modernity to postmodernity (Gade 2011); from industrialisation to "informationalization" (Hardt and Negri 2000); from solidity to liquidity (Deuze 2009b); all suitably describe a societal transition away from more centralized, hierarchical,

bureaucratic, and stable forms of industrial production, towards more fluid, decentralised, flexible, networked, and individualised modes of work (Deuze and Witschge 2018).

While these exogenous pressures have brought disruption to traditional routines and ideals (Robinson 2007) – that is, the established cultural capital of the field – digital natives, by virtue of being native to these conditions, have been able to rethink routines from scratch, developing new practices and workflows that, over time, "generate identities, behaviors, roles and values that are seen as appropriate" (Ryfe 2006: 140). This was certainly borne out in my observation of routines at BuzzFeed and Vice, which had less in common with the classic "assembly line" model of news (Gans 2004), but rather, seemed to bear many of the commonly cited traits of multimedia content production (Deuze 2009a). Accordingly, day-to-day news work at *BuzzFeed* and *Vice* was characterised by a high degree of fluidity, flexibility, integration, and collaboration, with individual journalists assuming more responsibility for their own work and operating under only occasional supervision from editors and other senior staff. This being said, the news process was still subject to constant evaluation, revision, and refinement, but most of this activity took place behind the domain of a computer screen, via virtual messaging applications like Slack.

Under this arrangement, and as observed in a multitude of other newsrooms, practices like speed, flexibility, and multiskilling were important facets of news work (Mitchelstein and Boczkowski 2009). While much of the literature so far has largely focused on the negative impact of these changes in traditional newsrooms, among the journalists I interviewed it was clear that these nascent trends were "less a source of disruption than...a potential new source of identity" (Ferrucci and Vos 2017: 3). Partly owing to their limited professional trajectories in the field, journalists from *BuzzFeed* and *Vice* were less cognitively and emotionally attached to "old" ways of making news, which benefited their organisations in the sense they were more ideologically predisposed to accepting new methods and styles of working. This was evident in the way that these practices were naturalised in discourse, with younger 136

journalists, in particular, recognising that the ability to working with new technologies, publishing across multiple platforms, managing competing workloads, and – accepting these were inevitable came with digital journalism - features of news work were important emerging forms of cultural capital in online journalism (Siapera and Spyridou 2012). Accordingly, certain news processes linked to the established cultural capital of the field were disparaged for being overly rigid, bureaucratic, and hierarchical.

A major part of the appeal of these new forms of news work was that they appeared to imbue individual reporters with a greater sense of ownership and control over news stories. Somewhat paradoxically, then, these new practices connected with a deeply traditional desire for autonomy (Maras 2013). At first glance, these findings seemed to contradict much of the literature on contemporary news work, which has shown how multiskilling, flexibility, and accelerating production cycles has led to more pressurised working arrangements, tighter deadlines, and less time to do the "core" work of journalism such as research and on-the-ground reporting (Ornebring 2009). Yet in making the distinction between internal and external autonomy, this chapter has shown how perceptions of increased autonomy were primarily judged against the threat of *internal* constraints, such as oversight from superiors or formalised publishing protocols. In the absence of these forms of top-down control, reporters often felt emboldened to independently pursue and pitch stories to editors that they found personally interesting, based on the knowledge that "what interested them would interest the audience" (Gans 2004: 229).

However, while these newer forms of cultural capital led to increased *perceptions* of autonomy, it is important to recall that values like speed, flexibility, and multiskilling in contemporary news work have not emerged in a vacuum, but rather, are closely linked to broader processes of commercialisation aimed at rationalising resources, reducing costs, maximising productivity, and being more responsive to the needs of the market (Cohen 2015). These processes have occurred in tandem with an "almost exclusive emphasis on individualised responsibilities" in the workplace; a trend often 137

cloaked in organisational and broader cultural discourses of 'enterprise', which stress empowerment, flexibility, entrepreneurialism, and autonomy (Deuze 2009a: 84). While much of the literature uncritically celebrate these new facets of work (Kreiss and Brennen 2016), for Deuze (2009: 85), these discourses amount to a "deliberate managerial attempt to regulate professional identity as a form of organizational control", leading workers to uncritically accept their own subjugation and normatively embrace more precarious, contingent, pressurised, and exploitative modes of labour. Among journalists at *BuzzFeed* and *Vice*, there was a definite sense of these values being internalised in discourse, so much so, that they seemed to form an integral part of their professional identities. Perhaps this also helps to explain why some senior journalists I interviewed seemed more perturbed by these changes, connecting them with a loss, rather than gain, in autonomy.

Aside from these internal constraints, external factors, filtered through organisational imperatives, played a major role in shaping journalistic routines and individual understandings of autonomy. A key constituent in this process was the audience, whose particular "informational wants and needs" (Beam 2001: 467) seemed to have an important role in shaping routines at *BuzzFeed* and Vice. This would appear indicative of audience-oriented publications in general, where, as Van Zoonen (1998: 136) notes, "it is it is not so much the relation with sources that circumscribes journalistic practice, but the requirements of the market or the community that is catered for". Nonetheless, what distinguishes the current era from earlier periods is the relative abundance and availability of digital technologies that "give journalism outlets unprecedented ability to track, measure, and quantify audience activity" (Cohen 2015: 108). Consequentially, older conceptions of the audience as the 'missing link' in news production (Schlesinger 1978) have been rendered less apposite, as journalists' intuitions about the audience have steadily been replaced by a rhetoric of the 'active' audience, "laying the groundwork for a vision of the professional reporter that is less autonomous in his or her news decisions and increasingly reliant on audience metrics as a supplement to news judgment" (Anderson 2011: 550).

At BuzzFeed and Vice, this rhetoric animated many conversations about the audience, albeit in ways that were specific to both organisations. While at Vice, journalists seemed generally more sceptical about metrics, connecting them with a loss of autonomy in work, BuzzFeed journalists seemed to embrace them, viewing metrics like "sharing" as effective substitutes for audience wants. These developments inevitably raise concern about the nature of editorial autonomy, suggesting that journalist's "gut feeling" about news is being encroached upon by the "quantified" audience (Schultz 2007; Anderson 2011; Carlson 2018). Although there are many legitimate reasons to criticise professional news judgement as a means for deciding the news (e.g. Hall 1978; Soloski 1989), this tacit knowledge has long been considered an "essential skill for journalists" and thus an important part of their cultural capital (Cohen 2015: 109). Moreover, there are legitimate questions to ask regarding the extent to which metrics can be interpreted as a genuine reflection of what the audience want, or indeed, need (Boczkowski and Peer 2011). Rather, metrics would seem to support a very narrow definition of consumer choice (Tandoc and Thomas 2015), ultimately driven by an institutional need to extract more (economic) value from the news process (Anderson 2011; Cohen 2015; Carlson 2018).

8. Making News: News Production Practices at *BuzzFeed* and *Vice*

Having outlined some of the key constituents shaping routines at *BuzzFeed* and *Vice*, this chapter goes deeper to investigate different facets of news content. Despite the very limited evidence, a general claim made about digital natives is that the content and character of their news is fundamentally different from other news providers (Riordan 2014; Küng 2015). Based on the findings presented so far, these claims would appear to have some purchase, though this must be balanced with the knowledge that certain elements of *BuzzFeed* and *Vice's* news operation remain quite traditional in structure and scope.

To move the discussion forward, this chapter addresses what many scholars consider to be *the* defining tension in the field at present, namely, that between the emergent participatory culture and journalism's professional logic of control over content (Lewis 2012). While journalism has long been understood as a "very weakly autonomous field" (Bourdieu 2005: 33), perhaps what sets apart the present era is "the *ease* with which individuals may participate in the creation and distribution of media" (Lewis 2012: 846, emphasis in original). This shift is seen to strike at the core of journalism's authority and self-definition (Netzer et al. 2014: 620), which, for the better part of a century, has relied on maintaining a strong degree of exclusivity and control over information to earn its autonomy and power.

To explore this broad issue, the following discussion examines some of the key sites where the participatory-professional boundary is being negotiated, paying close attention to the role of journalistic norms in mediating the acceptance, reconfiguration, and rejection of more audience-centric practices. To guide the discussion, three research questions have been developed:

 To what extent do journalists at *BuzzFeed* and *Vice* express a desire for gatekeeping control over news content? How is this manifest in news production?

- 2) Do BuzzFeed and Vice challenge or uphold traditional sourcing techniques?
- 3) What is *BuzzFeed* and *Vice* journalists' relationship with the objectivity norm? How is this manifest in news content?

In answering these questions, this chapter hopes to shed light on the various ways journalism's cultural capital might be responding to the participatory logic of digital media. The remainder of this chapter is divided into three main sections. The first examines gatekeeping and exclusivity (8.1), looking at the various processes involved in gathering, selecting, and filtering the news. The second section (8.2) examines sourcing practices at *BuzzFeed* and *Vice*. The third and final section examines journalists' relationship with objectivity (8.3); both at a rhetorical and practical level.

8.1. Exclusivity and Gatekeeping

Despite the relative abundance of information and increased means that ordinary people have acquired to participate in the news process, it would be fairly uncontroversial to suggest that news selection - or 'gatekeeping' -"remains one of the fundamental activities of any journalist or editor" (Bruns 2008: 175). Decisions about what news to cover; how to frame and collate information; where to invest resources; and what sources to use; all contribute to the process of gatekeeping, which, at a basic level, is concerned with what information is selected and rejected in the process of becoming news (Shoemaker and Vos 2009). While the contemporary media environment has multiplied and expanded the number of gates and gatekeepers through which information reaches the public (Singer 2008), this should not detract from the central role still played by the news media in producing most of the publics information about news and current affairs (Nielsen 2016). It is for this reason, however, that many scholars have expressed concern about the current state of information provision (Fenton 2011; McChensey and Pickard 2011; Phillips 2012; McChesney 2013). Contrary to early predictions that the internet would improve the quality and diversity of news content (see Curran et al. 2012), many empirical studies have pointed to a rise in what Bourdieu (1998:20) would call

the "uniformity and banality" of news. This has been attributed to a combination of greater pressures on speed and immediacy in reporting; fiercer competition between news providers; and a general lack of resources available to produce in-depth, well-researched, original content (e.g. Boczkowski and De Santos 2007; Paterson 2007; Davies 2008; Quandt 2008; Franklin and Carlson 2011; Phillips 2011).

These practical challenges aside, the shift to a "networked" system of communication has raised philosophical questions about journalism's role and purpose (Lewis et al. 2010; Peters and Broersma 2013; 2017). As Deuze (2005: 451) writes, "new media technologies challenge one of the most fundamental "truths" in journalism, namely: the professional journalist is the one who determines what publics see, hear and read about the world". For some, this loss of exclusivity and control – or "jurisdiction" over the news (Abbott 1988; Lewis 2012) - make it imperative that journalism begins to "redefine its relevance" (Broersma and Graham 2013: 461). These suggestions range from the modest to more radical, but tend to be predicated on the same assumption: that journalism needs to reimagine its relationship with the audience, and, to varying degrees, become more participatory, open, and inclusive (Kreiss and Brennen 2016). While many of these ideas predate digital media (Anderson 2011), they have often been borne from observations made at traditional news organisations. This has come at a cost of providing a more holistic picture of how gatekeeping might be changing in the digital era. Little is known, for instance, how digital natives have dealt with the "perceptual and practical threat" of participatory culture – that is, if participation is perceived to be a threat at all (Lewis 2012: 850).

8.1.1. Scoops and Exclusives

Dealing first with the more practical elements of gatekeeping, it was clear that journalists from *BuzzFeed* and *Vice*, at least rhetorically, placed some emphasis on exclusivity in news production, suggesting it remained an important doxic news value (Schultz 2007). At *Vice*, discussions about exclusivity frequently centred on long-form content such as features and

investigations, where significant resources were diverted to gathering and producing original news⁶⁹. It was thus harder to determine what kind of emphasis was put on original reporting in every day news production. When asked to describe their daily routine, one journalist from *Vice UK* mentioned that one of their main responsibilities was to "come up with exclusives" for either themselves or other reporters, though it was not clear if this was a daily activity or took place within another timeframe (Respondent B, reporter, *Vice UK*). When asked what distinguished news production at *Vice* from traditional media, another reporter from the UK was more revealing:

"I think in many ways it's the same, you know, because ultimately we want to be covering big news stories, so story selection isn't that different. If we see something like [the civil war in] Ukraine, for example, we're like 'okay, will this is definitely a big story. We should be doing something'. But then the question is what are we doing that will be different. So, for news, Vice always wanted to look at underreported, or complete unreported stories when it was making its films and doing its journalism. But obviously when you move into the news realm, you have to do that but you also have to cover what everyone else is doing, because those are the biggest stories of the day, or of the week, or whatever" (Respondent F, reporter, *Vice UK*).

This description quite artfully describes *Vice's* dual interest in originality and similarity in news. That this journalist saw it as "obvious" to cover what everyone else was doing adds further credence to claims that the "pack" mentality of journalism, first described by Crouse (1973), has maintained its importance as a routine in the digital news environment, despite a proliferation of new actors. This was corroborated by another journalist from the *US* office of *Vice*, who similarly alluded to the need to "balance the benefits derived from pack routine with the benefits of the 'exclusive'" (Shoemaker and Reese 2014: 184). To repeat an earlier quote:

⁶⁹ Please refer back to Chapter 5 (section 5.3) for a breakdown of news by format.

"We did a fair amount of enterprise reporting, [but] the thing is it took time to get off the ground, and then you push publish, and push it out on Facebook and Twitter or whatever, and sometimes the piece would hit maybe great, and other times less...We had to be highly selective. Whatever we thought was trending [on social media] the most in the morning that wasn't already spent [or] something that had hit the wires within a reasonably short amount of time; there was still room to carve some space out on that. That was the rationale for much of morning assignments...If you're going to get 10 million unique visitors a month, you're doing it by covering Obama day of, you're doing it by covering Trump day of. The Italian earthquake day of, like, within an hour or two. You're jumping on every mass shooting that's taking place in the United States, which we did, you know what I mean... So, we needed to be on top of breaking news. Otherwise, with your original enterprise reporting, the very Vice-y, sort of niche, you know, when you find the real under the radar story, unless it is very explicitly tied to something super relevant in people's minds, you've got to promote the hell out of it for people to know that it's a thing! It's going to be harder to make hay out of that stuff consistently whereas the stuff that's however many thousand tweets on Twitter, you know, if you can get something in there, that's in the flow, in that feed, in the continuum where millions of people are already looking for that sort of thing, then you're going to get a lot more aggregated traffic out of it. So, it's just logic" (Respondent E, editor, Vice US).

As this comment suggests, enterprise coverage had to be balanced with pack coverage because they essentially served different purposes⁷⁰. While original reporting was primarily valued for its cultural and symbolic function – as a means of enhancing the prestige and status of the *Vice* brand and

⁷⁰ The impression given to me in interviews was that this balance was far from even. As the editor quoted above acknowledged, the pressure of trying to "keep pace with the daily news juggernaut" often meant that reporters had very limited time to work on "original stuff they had conceived themselves" (Respondent E, editor, *Vice US*).
communicating journalistic excellence – the hit-or-miss nature of exclusives seemed to warrant a strong focus on news that could attract large audiences. In this way, pack coverage was primarily valued for its ability to generate economic capital, in the form of public recognition, "measured by numbers of readers, listeners, or viewers, and therefore, in the final analysis, by sales and profits" (Bourdieu 1998: 70).

At *BuzzFeed*, journalists were arguably more dismissive of pack coverage. As one reporter from *BuzzFeed UK* explained: "You can't just report on news that everyone else is reporting on because it doesn't guarantee people will come to your article when everyone has already written about it" (Respondent C, reporter, *BuzzFeed UK*). Instead, there appeared to be a stronger imperative to seek out exclusives – both in "enterprise" journalism, and in daily reporting via "scoops". Several journalists I spoke with traced this ethos back to Ben Smith, editor-in-chief of *BuzzFeed US*, who has spoken on several occasions about his affinity for scoops, maintaining that they remain a valuable part of the journalistic routine.

"I think I don't, really. I think reporters who really get this ecosystem get that a piece of churned, aggregated content, nobody's going to read it, nobody's going to share it, nobody's tweeting, "Wow, you guys did a great job writing this thing The New York Times broke. It's half as good, and it only took you three hours. Congrats...The stuff that breaks through is scoops and maybe that can be very, very fast and very short, but it's something that adds real value" (Smith 2014a).

"[What I'm looking for in journalists] is just basically raw aggression...that you're breaking stories, that you're breaking news, getting scoops, other people haven't gotten, that's the core of it right?" (Smith 2016d).

This ethos seemed to have a trickle-down effect among other *BuzzFeed* journalists, with various editors and reporters discussing the importance of

scoops in daily news coverage. As McKay Coppins, a former political reporter for *BuzzFeed US*, once explained:

"The thing is, Ben Smith came from a fairly traditional journalism background. I mean, he was a blogger right before he came to BuzzFeed but he had worked at the New York Observer, and had written for a lot of newspapers, The New York Daily News, and he wanted scoops. Like, his whole things was, like, 'just break news. It can be the tiniest scooplet but we just want new information...I don't care about aggregation, I don't care about writing the thing that everyone else is writing'...there's a certain genre of campaign journalism which is writing the 'day' story, which is like, 'this is what the candidate did today'...He was like 'I don't need any of that. I just want new information. So, like, short little scooplets and then we'll occasionally take big swings, like profiles and features, or whatever'" (Coppins 2016).

Similarly, journalists I interviewed from *BuzzFeed* suggested that the pursuit of new information was a "huge priority" in daily reporting, both in the US and UK (Respondent B, reporter, *BuzzFeed UK*). According to some reporters, the benefit of this focus was two-fold. One, *BuzzFeed* were able to counteract the "pack" mentality of traditional media, by avoiding the reproduction of stories being covered elsewhere:

"Papers still act as if they have to cover all the issues of the day, even though a lot of that is a 600-word commodity news story that PA [Press Association] has done, that every other newspaper has done, and they still have to put a reporter on it for three hours because they don't like having wire [copy] in the paper. But that fills so much time and so much production effort...If you have 15 versions of the same story then what's the point? [Our editors] are like 'yeah, this is quite important, but everyone's done it, so let's do our own thing'" (Respondent H, reporter, *BuzzFeed UK*).

"I think we're more focused [at BuzzFeed]...Thriving original reporting and original angles on stories, versus just repeating something that's already been covered elsewhere" (Respondent D, editor, *BuzzFeed UK*).

And two – in a similar manner to *Vice* journalists' discussions of "enterprise" reporting – *BuzzFeed* journalists' saw publishing small and large scoops as an effective means of quickly growing the reputation and credibility of the *BuzzFeed* brand:

"[The focus on scoops] is partly a carry-over from America I think. Obviously the Editor-in-Chief...Ben Smith, his background is as a scoopgetting...the term scooplet or micro-scoop gets banded around, that is his mode...[That was] the way they built up that credibility in America...You build up a huge amount of credibility when you're getting these scooplets, and that spreads out to a very big [political] industry over there. I think our [political system] is smaller, not just in terms of country but I think our industry is just smaller...[so] a lot of those microscoops are not in fact, quite as useful. [But] if you get yourself in a position where you are a trusted provider of those little nuggets of information that can build your reputation very quickly" (Respondent L, editor, *BuzzFeed UK*).

"But there's also people, like, our politics team, with the goal of trying to make sure people take us seriously within British political reporting, being on Sunday morning politics shows, erm, having scoops that other media organizations are forced to credit, even though they hate crediting us. Like, someone to be like 'BuzzFeed reported this', erm, they're building up credibility, erm, and so for them, being taken seriously in those circles is important" (Respondent G, reporter, *BuzzFeed UK*).

As these two quotes imply, the pursuit of scoops was deemed particularly important in the arena of politics; an area of reporting that journalists surmised

to offer the best opportunities for *BuzzFeed* to quickly grow their reputation and earn important symbolic capital. This aside, another potential reason why the political beat was deemed valuable is that it could efficiently provide a steady stream of information, including scoops, but at a fraction of the cost compared to other areas of reporting⁷¹ (see Skovsgaard and Dalen 2013).

In the interest of learning more about how exclusivity was routinised in daily newswork, a randomised sample of 40 news stories⁷² from the UK and US websites of *BuzzFeed* and *Vice* was compared with news stories produced by two legacy news organisations: the *BBC* and *The Guardian*. In the first instance, the primary source of each story was coded for exclusivity, with the aim of determining how often stories were based on material obtained directly or first-hand by journalists:

	BuzzFeed (n=40)	VICE (n=40)	Guardian (n=40)	BBC (n=40)
Direct / exclusive (%)	25	55	22.5	17.5
Indirect / second- hand (%)	70	45	70	70
Ambiguous / Can't tell (%)	5	0	7.5	10
Not applicable (%)	0	0	0	2.5
TOTAL	100	100	100	100

Table 7: Source exclusivity in news articles

⁷¹ For an in-depth discussion of the political beat, please refer back to Chapter 6 (section 6.4) of this thesis.

⁷² Note, the stories selected in this sample were daily formats, meaning features and investigations were excluded from the sample.

As the data shows, both *Vice* and *BuzzFeed* used more exclusive primary sources in daily news stories compared with the *Guardian* and *BBC*. This was most strongly pronounced at *Vice*, where over half of coverage (55%) was based on exclusive primary sources; more than double any other outlet. One explanation for why this percentage was so high relates to the coding procedure, which categorised the primary sources of comment and opinion pieces – mostly the organisations' own journalists – as exclusive. At *Vice*, these formats made up 12.5% of total coverage⁷³ – considerably higher than any other outlet – which affected the results of this particular sample. When the primary sources of comment and opinion pieces were re-coded as non-exclusive, a slightly different set of results emerged:

<u>Table 8:</u>	Source	exclusivity	in	news	articles	(opinion	and	comment
articles r	e-coded	as non-exclu	usiv	<u>/e)</u>				

	BuzzFeed	VICE	Guardian	BBC
	(n=40)	(n=40)	(n=40)	(n=40)
Direct / exclusive (%)	25	30	20	10
Indirect / second- hand (%)	70	70	72.5	77.5
Ambiguous / Can't tell (%)	5	0	7.5	10
Not applicable (%)	0	0	0	2.5
TOTAL	100	100	100	100

Although *Vice* still used the most exclusive primary sources in news stories, this percentage (30%) was much closer to *BuzzFeed* (25%) when opinion pieces were recoded as non-exclusive. However, given the emphasis that

⁷³ Please refer back to Chapter 5 (section 5.3) for a breakdown of news by format.

BuzzFeed journalists placed upon scoops, it was somewhat surprising to see the organization being outperformed by *Vice*, where journalists from both the UK and US had given the impression that chasing exclusives was less of a priority because of the pressure of trying to stay on top of pack coverage. Nonetheless, both digital natives outperformed the two traditional media organisations in the sample; by a small factor compared to *The Guardian* and a larger degree in the case of the *BBC*.

In a more general sense, these findings would appear to at least partially corroborate the opinion of journalists at *BuzzFeed* and *Vice* that beyond more time-consuming formats like features and investigations, there was some emphasis on generating scoops and exclusives in daily news work. This supports observations made in other (mainly traditional) organisational settings, which have shown journalists' strong normative desire for gatekeeping control leads to an emphasis on exclusivity in the news process (Schultz 2007; Lasorsa et al. 2011; Wiliams et al. 2011; Messner et al. 2012). It would also appear to contradict the rather trite claim that "online sites...simply [disseminate] what [is] being produced by traditional old media" (McChesney 2012: 685). In fact, there is a case to make that exclusives took on a heightened importance at *BuzzFeed* and *Vice*, owing to their status as new entrants to the field. This was evident in journalistic discourse, which suggested that exclusives performed a vital symbolic and cultural function, helping both organisations differentiate themselves from competitors, enhance their status and prestige, and increase their recognition among fellow peers.

Yet, despite there being at least some emphasis on gathering exclusives in news production at *BuzzFeed* and *Vice*, this should not conceal the fact that the basis for most stories was information obtained indirectly or second-hand from other sources. Across all four outlets, journalists habitually used second-hand information as the basis for news stories in daily reporting. Indirect or second-hand sources were used by all four outlets a minimum of 70% of the time, with *BBC* journalists relying on non-exclusive primary sources the most (77.5%) out of all four organisations. This is consistent with general

observations made about online news, which show journalists often rely on second-hand information as a means of coping with increased pressures in daily work (Davies 2008; Lewis et al. 2008; Phillips 2011). This situation has arguably worsened since the advent of social media, with the "ambient" social media-news nexus (Hermida 2010) putting greater pressure on individual journalists to "monitor and match" the activities of competitors (Chadha and Wells 2016: 1027). Ultimately, these trends have the effect of undermining the diversity of news and journalism's public service ideals, increasing "content homogenization" (Boczkowski and De Santos 2007: 168), and depriving people of new information and knowledge about the world⁷⁴.

8.1.2. Redefining Exclusivity

This, however, calls to attention a second and less-discussed aspect of exclusivity, which concerns the way news making now involves "managing multiple fast-moving flows of information already in circulation", in addition to "locating and sharing 'new' news" (Boyer 2013: 2). While this aspect of news work has often been subsumed by concerns about the overall decline in original reporting, others have taken more nuanced position, recognising the emergence of these practices as a logical response to the changing spatial and temporal dimensions of news. As Van Hout and van Leuven (2016: 118) observe:

"Technological innovations support the transition from traditional, onthe-spot reporting to a combination of filtration and curation of existing information, and, to a lesser extent, slow journalism or long form journalism...analysis and contextualization of events are increasingly viewed as the main task of journalists, while the direct reporting of facts is 'outsourced' to wire services and PR services"

⁷⁴ In his book, *On Television,* Bourdieu (1998) also remarks on this issue, noting how the news media often engage in the practice of monitoring each other's stories. For a journalist, he argues, "to know what to say, you have to know what everyone else has said" (Bourdieu 1998: 24).

Under these conditions, Anderson et al. (2012: 22) suggest that the journalist has been "moved higher up the editorial chain from the production of initial observations to a role that emphasises verification and interpretation, bringing sense to the streams of text, audio, photos and video" that circulate continuously online. This emphasis on "sense-making" – "helping people understand, interpret, and use information, rather than merely giving them access to it" (Singer 2008: 65) – is said to be transforming notions of exclusivity, and could augur a shift in journalism's gatekeeping role, as journalists become less focused on being "first", and more concerned with presenting existing information in novel or "fresh" ways (Pavlik 2001; Matheson 2004; Møller Hartley 2013; Usher 2014; Agarwal and Barthel 2015).

At *BuzzFeed* and *Vice*, it was apparent that these practices were integral to every day news work. Due to the speed of the news cycle and a finite amount of resources, journalists acknowledged it was often impossible to be first to break a news story, which seemed to lead them to redefine exclusivity as in terms of being able to offer a fresh take on information already in circulation, in ways that made sense to their particular audience. As one *BuzzFeed* reporter explained:

"Our priority isn't to quickly get something up first before the BBC and Guardian, because everyone's going to have it up within half-an-hour anyway. With some breaking news, there is an exception to that. Or if it's a subject we've covered quite heavily then we will have someone on top of that all the time, but it's generally, 'how can we move this forward?' Original reporting is a huge priority for us. If we do miss the breaking line, it doesn't mean we won't do the story, it will be a reaction with our own quotes and our own take on something" (Reporter B, reporter, *BuzzFeed UK*).

Thus, journalists seemed to mitigate against a loss of exclusivity by thinking of ways to repurpose existing news, sometimes by looking for "different angles" into ongoing stories (Respondent H, reporter, *BuzzFeed UK*), or by "getting original quotes" (Respondent C, reporter, *BuzzFeed* UK) to build on stories in 152

an iterative manner. A similar process seemed to be in place at *Vice*, where reporters were often required to: "[Find] the fresh angle into the ongoing story; show the particular view that [wasn't] shown, and make a story their own, aesthetically, stylistically, or whatever" (Respondent E, editor, *Vice US*). Or, as another reporter from the *UK* office of *Vice* explained:

"So, once you've identified the story [published elsewhere], you then have to, sort of, think what you're going to be doing that's different. And, you know, our style was always different to everyone else...because we're making this stuff for maybe sixteen to maybe thirty-year olds" (Respondent F, reporter, *Vice UK*).

As this quote implies, the audience were an important factor in journalists' decisions about how to repackage existing content. This was certainly true at *BuzzFeed* too, with two reporters describing the process of taking already published information, and repurposing it to make it more useful and relevant to their own readers:

"You might cover the same issues [as traditional media] but it's whether you can lead on it or not. So, on the doctor's strike: Firstly, junior doctors are in the BuzzFeed age group, they read us, they already read us, but our audience are interested too...So it was very easy for us to get traction with the doctors as well as the audience. And so, it's an issue all the papers are covering but we weren't just doing every slip and move...We could sort of decide which bits we were going to do, and what days we were really going to go in on an issue...And so, some things you decide to do but do them in a BuzzFeed way and others you just go 'yeah, this story is good, it's interesting, but it's not for our audience and everywhere else has done it quite well" (Respondent H, reporter, *BuzzFeed UK*).

"We like to do things that are 'debunky', things that...There was this awful phrase that The Spectator used...'news you can use', but if it's just sort, things that are explained or put into context...[Like] 'here's what the 153 scientists say and what that risk actually means for you'. For example, there were a few papers last year reporting on some woman who had stage one breast cancer and was trying to treat it with a vegan diet, so it's things like that...any [situation] where we feel like somebody is getting a lot of attention but the readers aren't actually being served useful information" (Interviewee N, editor, *BuzzFeed UK*).

BuzzFeed, then, could afford to be more selective in news that they covered. These decisions routinely centred on whether particular issues could be made fresh and relevant for the audience at hand. In interviews with online journalists from various US-based news outlets, Agarwal and Barthel (2015) made a similar observation, nothing that one way journalists dealt with "the difficulty of producing new information in the online environment" was by defining exclusivity as "*exclusive to* [their] *particular audience*" (Agarwal and Barthel 2015: 382, emphasis in original). This, they argue, "points to a normative environment in which speed or freshness has replaced exclusivity. The goal is not to be first, but to be the most useful to the audience" (Agarwal and Barthel ibid. 382).

This aside, other scholars have drawn attention to the practical advantages of approaching stories in this manner. For Møller Hartley (2013: 579), the process of journalists posting "additional facts and angles [to stories] in an accumulative manner" was a strategy that helped journalists manufacture a sense of urgency and relevance in news stories, even if these stories concerned subjects that did not correspond with classic definitions of breaking news. In Usher's (2014) case, the desire for "freshness" in news effectively operated as a synonym for immediacy, whereby "fresh" came to denote "a quality that Web producers and others charged with online journalism associated with their presumed sense of what the audience wanted: something new, something different" (Usher 2014: 109). According to Usher (ibid: 109), determining what was "fresh" was "one way to explain how journalists tried to make sense of the constant presence of a never-ending deadline in the digital age".

In terms of how these ideas manifested in news coverage⁷⁵, it was apparent that many stories published by *BuzzFeed* and *Vice* placed a strong emphasis on news analysis or interpretation (the *why* of an event), rather than plainly descriptive stories (the *what* of an event). To provide a few sample headlines from interpretive news stories:

Table 9: Example headlines in news articles published by BuzzFeed and <u>Vice</u>

Outlet	Headline	Date
Vice (UK)	Why the huge increase in the number of UK graduates is bad for everyone	11/10/16
Vice (UK)	Female immigrants explain what shocked them the most about Britain	08/03/17
Vice (US)	The second presidential debate, decoded	10/10/16
Vice (US)	Is Trump prepared for the next epidemic?	09/03/17
BuzzFeed (UK)	6 things you need to know about the budget	08/03.17
BuzzFeed (UK)	Here's why Australia just took a big step away from marriage equality	12/10/16
BuzzFeed (US)	9 reasons why Samsung is in big trouble	12/10/16
BuzzFeed (US)	This is how people around the world feel about travelling to the US right now	09/03/17

As these examples demonstrate, some stories produced by *BuzzFeed* and *Vice* seemed concerned with moving "beyond descriptive, fact-focused and sourcedriven journalism" to embrace forms characterized by "a prominent journalistic voice; and by journalistic explanations, evaluations, contextualizations, or speculations going beyond verifiable facts or statements by sources" (Salgado and Strömbäck 2012: 154). To add further empirical weight to these preliminary observations, stories from all four outlets in the sample were analysed for evidence of interpretive journalism. Following the approach of Salgado and Strömbäck (2012), stories were coded as "mainly descriptive" if they described events in a straightforward manner, while stories were coded as "mainly

⁷⁵ Note: I am referring here to *all* news coverage, rather than any specific formats.

interpretive" if the journalist "analyzed, evaluated or explained a situation while also describing it" (Salgado and Strömbäck 2012: 148).

	BuzzFeed (n=179)	VICE (n=174)	Guardian (n=200)	BBC (n=200)
Mainly descriptive (%)	59.8	25.9	73	70
Mainly interpretive (%)	37.4	64.4	21	20.5
Ambiguous / can't tell (%)	1.1	1.1	6	7
Not applicable (%)	1.7	8.6	0	2.5
Total	100	100	100	100

Table 10: Presence of interpretive journalism in news articles

As shown above, *BuzzFeed* and *Vice* routinely employed interpretive styles of storytelling in news. This was most pronounced at *Vice*, where nearly two-thirds of stories (64.37%) we coded as "mainly interpretive". One explanation for this relates to the high number of features, investigations, and opinion pieces that appeared in both *BuzzFeed* and *Vice's* sample⁷⁶, with these formats appearing to lend themselves to more interpretive styles of storytelling. At *The Guardian* and *BBC*, interpretive styles of storytelling appeared in around one fifth of news articles – a lower percentage than the two digital natives but still significant enough to suggest that both organisations saw value in this approach⁷⁷. In an opposite sense to *Vice*, these higher percentages might be explained by *The*

⁷⁶ See Chapter 5 for a more complete breakdown of news by format.

⁷⁷ This finding is consistent with other empirical research that has highlighted the growing importance of "interpretive" or "contextual" reporting in journalism over the last few decades (e.g. Patterson 1993; Barnhurst 2011; Fink and Schudson 2014).

Guardian and *BBC*'s routine use of the "report" format, which contributed to 83.5% and 85% of coverage, respectively⁷⁸.

To gain a better understanding of the *context* in which interpretive reporting appeared, stories were subsequently broken down by subject:

	BuzzFeed	VICE	Guardian	BBC
	(n=67)	(n=112)	(n=42)	(n=41)
Government /	32.0	23.2	52.4	65.8
politics				
Social issues /	27.6	25	17.9	9.7
rights / protests				
Crime /	6.0	10.7	10.7	4.9
terrorism				
Science /	14.9	1.80	2.3	2.4
technology				
Health	3.0	1.80	0	4.9
Business /	6.0	1.80	9.5	7.4
economy				
Entertainment /	6.0	17.0	0	4.9
arts				
Lifestyle	0	8.9	0	0
(trends, fashion,				
travel)				
Accidents/	1.5	0	0	0
disasters				

Table 11: Interpretive journalism versus topic

⁷⁸ See Chapter 5 (section 5.3) for a more complete breakdown of news by format.

Energy / nature	1.50	0	7.2	0
/ environment				
Other	1.50	9.8	0	0
TOTAL	100	100	100	100

For three out of the four outlets in the sample, the topic of government and politics was subject to the most interpretive reporting. This was particularly pronounced at the two traditional news organisations, where news about government and politics accounted for more than half of interpretive stories in total. At *BuzzFeed* and *Vice*, news about government and politics, as well as social issues, rights, and protests, made up nearly half of both organisations' interpretive coverage. This was in accord with findings presented earlier in this thesis, which signalled these were two important news topics for both organisations⁷⁹. Beyond these subjects, *BuzzFeed* also featured a high number of interpretive stories focused on science and technology. Meanwhile, *Vice* used interpretive storytelling fairly frequently in stories about entertainment and arts. Finally, in the interest of seeing how often interpretive reporting featured in the daily news cycle, articles by all four outlets that followed the standard "report" format were isolated and assessed for the presence of interpretive reporting.

⁷⁹ See Chapter 5 (section 5.3) and Chapter 6 (section 6.4).

	BuzzFeed (n=127)	Vice (n=60)	Guardian (n=167)	BBC (n=170)
Mainly descriptive (%)	81.1	65	80.2	78.8
Mainly interpretive (%)	18.1	28.3	12.6	10.6
Ambiguous / can't tell (%)	0.8	0	7.2	8.2
Not applicable (%)	0	6.7	0	2.4
TOTAL	100	100	100	100

Table 12: Interpretive journalism in news reports

As these results demonstrate, interpretive storytelling was far less prevalent in daily reporting across all four outlets. Naturally, this raises questions about the extent to which interpretive journalism was routinised in every day news work. For *Vice,* this might have partially been offset by the regular publication of comment and opinion pieces, all of which were coded as "mainly interpretive" and made up 12.5% of total coverage. This aside, the absence of interpretive journalism in daily reporting heightens the possibility that, at the routine level, all four organisations were regurgitating information published elsewhere, rather than adding anything "new" by including further interpretation, analysis, or explanation for their particular audience.

These findings add an additional layer of complexity to the broader discussion around gatekeeping. In accord with other research, journalists at *BuzzFeed* and *Vice* seemed interested in redefining exclusivity to mean offering something "fresh" or novel – a new take on events – fashioned explicitly around the particular tastes and interests of their audience. Although much of literature has treated this is a recent development (Møller Hartley 2013; Usher 2014; Agrwal and Barthel 2015), it is in fact a decades-old phenomenon. For example,

discussing the decline of event-centred reporting in three American newspapers, Barnhurst and Mutz (1997) write that:

Most predictions on the demise of newspapers depend entirely on the loss of novelty (Katz 1994). Upon losing their monopoly on timeliness, newspapers found a way to prop themselves up. They made their narratives richer by adding more news analyses (Donovan and Scherer 1992). If they could not scoop competitors with what editors call the 'first day story', which tells what happened, they could instead write a 'second-day story', which keeps the story alive and maintains its value in the marketplace, by telling the story better, with greater depth, explanation, and background" (Barnhurst and Mutz 1997: 46).

There are clear parallels between this description and the contemporary challenges to journalistic exclusivity; which has seen journalists double-down on the idea of "telling the story better, with greater depth, explanation, and background" (Barnhurst and Mutz ibid: 46). Yet the specific affordances of new technology have had consequences on how journalists articulate these ideas, with words like "freshness" being used to describe the process of making stories exclusive for one's own audience (Agarwal and Barthel 2015). More cynically, this could be seen as journalists' rationalising their own loss of autonomy, whereby market pressures are leaving less time and money for original reporting and forcing journalists to become more creative with the limited resources they still possess. As one *BuzzFeed* reporter put it to me:

"You have to be quite creative as well. So, when something happens on Twitter, for example, everyone can report on it, The Independent, The Daily Mail, even Twitter now has Twitter Moments, and people are talking about it so it's no longer news. So, you have to sort of find ways to make people want to come to your article instead" (Respondent C, reporter, *BuzzFeed UK*).

This, though, is premised on journalists being given time to *be* creative. As shown in this section, despite journalists' rhetorical emphasis on "freshness", it 160

seemed most daily reporting still involved a more straight-forward and descriptive account of information already in circulation. The motivation for covering such stories seemed strongly driven by audience and market considerations (economic capital), with breaking news and trending topics offering higher returns on traffic and attention compared to more under-the-radar stories.

While these observations would appear to reinforce notions that the field is becoming further subjugated by the "forces of commercial heteronomy" (Bourdieu 2005: 43), it is important to view the trend towards content homogenization and continuation of "pack" behaviours in the context of broader sociocultural and technological trends. For instance, in a news environment where consumption is increasingly atomized (Associated Press 2008), and young people, in particular, feel alienated from traditional forms of politics and civic participation (Banaji and Buckingham 2013), perhaps certain forms of "pack" journalism can usefully raise awareness among certain publics of issues that might otherwise have been ignored. As Phillips (2011: 50) argues "if routine reporting was abandoned and public relation professionals ignored, citizens would undoubtedly be deprived of a great deal of the information they need to stay informed about the operations of government and business". In respect to BuzzFeed and Vice, this argument would seem to carry more weight, given both organisations' interest in reporting on various social and political issues news seen to be in the "public interest" - as opposed to the usual gamut of news topics that one might associate with strongly market-oriented or readerdriven news (see Hanitzsch 2007).

8.2. Sourcing Practices

Moving away from the subject of exclusivity, a second, essential element of gatekeeping concerns the diversity of sources in news coverage. As Phillips (2010: 87) argues, "the question of who journalists speak to, how they obtain information, how they evaluate it and whose stories they choose to repeat is critical to any examination of the changing role of the news media". The arrival of new information and communication technologies is seen to have changed

relations between journalists and sources on several fronts. The internet, according to Heinrich (2012: 767), is a space where:

"...many information providers meet in a digitally connected global arena. A large array of potential new sources can now be reached via many connection points other than (traditional) official sources such as governmental institutions or press offices. Instead of a rather 'closed' system of newsgathering, production and distribution, in which only a limited number of partakers had the power to make and shape news, the network journalism sphere is an open space of information exchange".

Changes to the communications infrastructure have fuelled hopes of journalism becoming more inclusive, diverse, and participatory, thus tipping the balance of news coverage away from elites, toward previously disadvantaged and marginalised social actors (Downey and Fenton 2003; Dahlgren 2005). In particular, the rise of social media platforms has fuelled hopes of a more egalitarian and "networked public sphere" (Benkler 2006). While on the one hand, this is seen to "[threaten] journalism's claim to provide an authoritative and legitimate representation of the social word", on the other hand, it has the potential to "open up journalism to new voices, topics and publics" (Broersma and Graham 2013: 448).

Yet, despite the internet theoretically opening up opportunities for journalists to include a "different cast of voices" in news (Harcup 2003: 360), empirical research has shown that journalists largely remain intent on "guarding the gates" (Singer et al. 2011) and sticking to their traditional doxa. As a consequence, elite sources still tend to dominate news coverage, meanwhile, newer technologies like social media are "normalized" to fit with traditional routines and ideals (Singer 2005; Singer et al. 2011; Knight 2012; Lasorsa et al. 2012; Moon and Hadley 2014; Lecheler and Kruikemeier 2015). The continuation of traditional sourcing techniques has dampened hopes of the internet reinvigorating and extending the public sphere (Hardy 2014). Only under very specific circumstances does the traditional "hierarchy of credibility" (Becker 1967) in sourcing appear to become upended. Breaking news events,

for example, have been shown to facilitate the formation of new, "networked publics" that can promote non-elite voices and counter-hegemonic narratives that challenge and disrupt the dominant discourse (e.g. Lotan et al. 2011; Papacharissi and de Fatima Oliveira 2012; Hermida et al. 2014).

In the admittedly small pool of research on online sourcing practices, digital native media have hardly featured. Some research has drawn attention to potential differences between online and traditional journalists in how they might perceive and practice sourcing (Shin and Cameron 2003; Cassidy 2007; Agarwal and Barthel 2015). However, with a couple of minor exceptions (Painter et al. 2017; Tandoc 2017) these claims have rarely been investigated empirically.

8.2.2. Sourcing Rhetoric

On the subject of *BuzzFeed* and *Vice*; their staff, news coverage, and target audience⁸⁰ all suggest potential avenues for different approaches to sourcing. Rhetorically, both organisations have emphasised the importance of reaching audiences "ignored" by the mainstream media, particularly young people of which women and minority groups have sometimes received lip service. In a memo written to staff at BuzzFeed in August 2017, editor-in-chief Ben Smith reasoned that diversity was "obviously an asset in the core business of getting and telling killer stories from diverse sources to a diverse audience" (Smith 2017a). Similarly, in a blog post entitled "2017 update on diversity at BuzzFeed", CEO and founder of *BuzzFeed*, Jonah Peretti, explained that (staff) diversity was not "just a moral imperative", but a "competitive advantage" because it allowed *BuzzFeed* to build a "closer connection [with audiences]" and "communicate across diverse cultures... and understand the experience of [the audience] and how they use content in their actual lives" (Peretti 2017c). Shane Smith, CEO of Vice, has similarly promoted the organisation as the "voice" of 'Generation Y' - "a highly educated, ethnically diverse, global-

⁸⁰ Please refer to Chapter 6 for a detailed breakdown of each of these organisational features.

thinking, hard-to-reach generation" (Smith 2016c), which he has claimed to attract by focusing on underreported stories and non-traditional subject areas such as LGBT rights, civil rights, and gender issues⁸¹.

In interviews, general questions about newsgathering processes and routines often led to journalists touching upon different aspects of sourcing and the wider issue of diversity in news. For instance, speaking on the role of journalism, one reporter from *BuzzFeed UK* argued that:

"Journalism...can be a voice for many more minority groups than it was before [sic]. A lot of the reason I like doing my job at BuzzFeed...is that I can talk about slightly nicher stories, slightly weirder stories, that maybe don't typically make the headlines" (Interviewee M, reporter, *BuzzFeed UK*).

Another reporter from *BuzzFeed* made a similar case, though was more explicit in highlighting the link between *BuzzFeed's* staff, their audience, and organisational sourcing practices:

"Often, you'll hear a voice [at BuzzFeed] you don't hear elsewhere. Obviously young people is one, [but] we have a very diverse news team as well. So, we're very aware of not just speaking to middle-aged white men... So, whether we do a quick story or get the reaction to something...[the] voices we represent in our pieces is different and when we go out to cover stories we'll seek out different voices on the ground that reflects our audience" (Interviewee A, reporter, *BuzzFeed UK*).

For one editor, this approach meant *BuzzFeed* was able to act as a form of "alternative" journalism (Atton and Hamilton 2008), by providing a platform to individuals or groups marginalised by traditional media:

⁸¹ This claim was part corroborated in the content analysis of this study, which showed that stories about social rights, issues, and protests, accounted for around a fifth of total news coverage (see section 5.3).

"I would say one thing [BuzzFeed are successful at] is giving a voice to individuals or groups who do not traditionally get their voice heard, allowing them to speak in their own words, rather than mediated through...a traditional media lens" (Interviewee L, Editor, *BuzzFeed UK*).

At *Vice*, interviewees similarly recognised the homology between staff and audiences. One *Vice* editor hinted this affected how the organisation approached sourcing, but was not totally clear in their description:

"The hosts, presenters, the people doing it [at Vice] are much younger, because our audience is much younger. So, you know...our style is different because we're making this stuff for 16 to maybe 30-year-olds. We'd focus on young people's stories, young people's interests... [The audience] don't want to hear from me the whole time, you know...Ultimately it's about having other voices in news and making sure that [our] audience got their information from them" (Interviewee F, editor, *Vice UK*).

Aside from this remark, *Vice* journalists generally spoke about sourcing in vague terms, or in regard to other aspects of the news process such as verification and accuracy.

8.2.3. Sourcing Practices

To get a better understanding of how sourcing materialised in practice, an extensive examination of sources was conducted on news content produced by *BuzzFeed* and *Vice*. This was compared with sources that appeared in news content published by *The Guardian* and *BBC*. To help limit the number of articles being examined, a randomised sample of 160 news stories was studied (40 per organisation, 20 from their respective the UK and US websites)⁸². Given the interest in exploring how sourcing occurred at the *routine* level, long-form

⁸² Note: this is the same subset of data that was used in the previous section of this chapter to analyse exclusivity in sources.

content (features, investigations) and as well as aggregated content and live blogs were excluded from the sample. Any instance where significant differences were noted between the UK and US websites of each outlet are highlighted.

8.2.3.1. Basic information

To establish some basic statistics about sourcing at all four outlets, the mean and median number of sources per article was recorded, along with the mean and median number of sources from social media that appeared in each news article:

	BuzzFeed	Vice	Guardian	BBC
	(n=40)	(n=40)	(n=40)	(n=40)
Total number of	175	191	211	201
sources	175		211	201
Mean number	4.4	4.8	5.2	5.0
of sources			0.2	
Median number	3.0	3.0	5.0	5.0
of sources				
Mean number				
of sources from	0.8	1.0	0.5	0.5
social media				

Table 13: Summary of sources in news articles

As the data shows, both legacy media used a higher number of sources overall, which corresponded with a higher mean and median number of sources per article compared to *BuzzFeed* and *Vice*. This difference was most notable in the median, with *The Guardian* and *BBC* quoting an average of two more sources per article. This would suggest journalists from both legacy media generally approached daily reporting with more rigour, consulting a greater

number of sources in the newsgathering process⁸³. Practically, this might be explained by the amount of resources made available for news reporting, with journalists from *The Guardian* and *BBC* being given more time to work on news stories compared to their digital native counterparts. There also might have been a greater discrepancy at *BuzzFeed* and *Vice* between the resources made available for daily reporting, and those provided for in-depth formats like features and investigations. This aside, there is the issue of access. As two established media, it would be safe to presume that *The Guardian* and *BBC* possess higher amounts of social capital⁸⁴ compared to *BuzzFeed* and *Vice*, permitting them access to a greater range and depth of sources, particularly from the government and other major institutions.

Moving on to sources from social media, all organisations made some use of social media in daily sourcing. This accords with other research suggesting these platforms have become an "important part of the everyday toolkit of journalists" (Hermida 2013: 296). Both digital natives used a higher average number of sources from social media per article, although this number was not significantly higher than either legacy outlet. Given the centrality of social media to news production at BuzzFeed and Vice, this finding was somewhat surprising. However, it is also possible that that social media functioned more as an "awareness system" for journalists, alerting them "to trends or issues hovering under the news radar", rather than a reliable or consistent source of quotes (Hermida 2010: 302). Indeed, only 25% of stories from Vice, and 35% of stories from *BuzzFeed*, featured any quotes from social media. This was fairly similar to The Guardian and BBC, who used quotes from social media in 20% and 30% of stories, respectively. One reason for journalists' reluctance in using quotes from social media seemed to relate to the problem of verification - an aspect of digital reporting that other scholars have highlighted (e.g. Bruno 2011; Schifferes et al. 2014). As one BuzzFeed editor put it: "Social news

⁸³ To some extent this was supported in the content analysis, with *The Guardian* publishing stories with the highest word count among all four outlets. *The BBC*, however, was more comparable with *BuzzFeed* and *Vice* (see section 5.4).

⁸⁴ Social capital is defined by Bourdieu (1986: 248) as "the aggregate of the actual or potential resources which are linked to possession of a durable network of more or less institutionalized relationships of mutual acquaintance or recognition"

stories can be very, very difficult to verify because people lie on social media quite a lot...a lot of the time we try to stick to stuff that uses images or video⁸⁵ because it's a lot easier to verify" (Respondent K, editor, *BuzzFeed US*).

8.2.3.2. Source Characteristics

To examine sources in more detail, the individual characteristics of sources were coded. Sources were analysed according to *type, age, gender,* and *ethnicity.* To begin, I provide a brief and largely descriptive account of this data, followed by a more in-depth analysis and discussion of the key findings.

	BuzzFeed (n=40)	VICE (n=40)	Guardian (n=40)	BBC (n=40)
Government / politicians / law	38.9	34.6	44.1	45.3
Police / military	1.1	4.2	0.5	8.5
Business people / organisations	9.1	4.2	6.6	1.0
Civil society members / organisations	9.1	5.8	7.1	3.5
Own media	4.0	4.7	1.9	6.0
Traditional media	10.9	12.6	6.6	9.5
Other media	2.3	6.3	1.9	6.0
Celebrities / entertainers	0.6	1.6	2.8	3.5

Table 14: Type of sources in news articles

⁸⁵ On this point, another reason why a lower number of social sources might have been recorded is that only textual content from social media was counted as a source in stories, not video or images.

News agencies	1.1	0.5	3.3	0.5
Analysts / experts	3.4	4.7	13.3	6.5
Ordinary citizens	17.1	19.9	10.4	7.5
Other	2.3	1.0	1.4	2.5
TOTAL	100	100	100	100

As the results show, government officials, politicians, and legal professionals were quoted significantly more in news reports than any other source type. This was true of all four outlets, with only a small difference being observed between the two legacy outlets and digital natives. The relative high number of government sources used by *BuzzFeed* and *Vice* would seem to fit with other findings of this study, which have demonstrated the importance of the political beat to both organisations, in journalists' rhetoric and directly in news coverage⁸⁶. To a lesser extent, all organisations routinely drew from other elite groups and institutions such as the police, military, analysts and experts, and business leaders. Collectively, these results would appear to support recent research on sourcing practices that has highlighted the durability of established professional standards and routines in dictating sourcing practices (De Keyser and Raeymaeckers 2012; Moon and Hadley 2014; Van Leuven et al. 2014; Lecheler and Kruikemeier 2015).

This being said, all four outlets, but particularly *BuzzFeed* and *Vice*, showed an interest in quoting from non-elite sources. This was most apparent in the number of ordinary citizens quoted in news coverage, with *BuzzFeed* and *Vice* featuring ordinary citizens nearly twice as often as *The Guardian*, and more than twice as often as the *BBC*. These sources were best represented in the news about social issues, rights, and protests. Sources from civil society such as charity workers and NGOs were featured the most by *BuzzFeed* (9.1%), followed by *The Guardian* (7.1%), *Vice* (5.8%), and the *BBC* (3.5%). Finally, all

⁸⁶ See Chapter 5 (section 5.3) and Chapter 6 (section 6.4).

organisations routinely relied upon quotes produced by other media in news reporting, particularly traditional media sources.

Secondly, sources were coded by gender:

	BuzzFeed (n=40)	VICE (n=40)	Guardian (n=40)	BBC (n=40)
Male (%)	42.3	46.6	59.7	59.2
Female (%)	28.0	28.3	20.4	19.9
Can't tell (%)	12.6	9.4	3.3	4.5
Not applicable (%)	17.1	15.7	16.6	16.4
TOTAL	100	100	100	100

Table 15: Gender of sources in news articles

As the data indicates, there was a fairly high degree of homogeneity in news coverage in terms of gender representation. Among all four outlets, male sources were quoted the most, though they appeared more frequently in stories produced by *The Guardian* and *BBC* (59.7% and 59.2% of the time, respectively). *BuzzFeed* and *Vice* quoted marginally more female sources than both legacy media (about eight percent). Owing to this gender imbalance, male sources generally dominated coverage across specific news topics. In stories about government and politics, for instance, females made up under one-fifth of sources in coverage by *The Guardian* and *BBC*, and around a quarter of sources in coverage by *BuzzFeed* and *Vice*. One exception to this pattern was stories on the subject of social rights, issues, and protests. In this category, *Vice* featured almost as many female sources than male sources. This was not true of *The Guardian* and *BBC*, which quoted at least twice as many male sources as female sources in stories on this subject.

Regardless, none of the four organisations produced coverage resembling the gender of the UK or US population as a whole, which, according to census data, is split fairly evenly between males and females (Office for National Statistics 2011; United States Census Bureau 2017)⁸⁷. These findings tend to corroborate the literature on gender in journalism, which has shown news is consistently biased in favour of male sources (e.g. Liebler and Smith 1997; Zoch and Turk 1998; Armstrong 2004). In the case of *BuzzFeed*, these results are arguably more surprising, given the relatively high number of female journalists employed by the organisation⁸⁸. While there is mixed evidence for the link between gender and sourcing (Steiner 2009; Hanitzsch 2012), this would suggest that the female journalistic habitus was not necessarily an accurate predictor of source selection, with other organisational and institutional factors playing a more determinative role.

Thirdly, sources were also coded by age, which produced the following results:

	BuzzFeed	VICE	Guardian	BBC
	(n=40)	(n=40)	(n=40)	(n=40)
0-10	0	0	0.5	0
11-20	0	0.5	0	0
21-30	2.9	2.1	3.8	3.0
31-40	3.4	4.2	12.3	2.0

Table 16: Age of sources in news articles

⁸⁷ There is a question, of course, of the extent to which *BuzzFeed* and *Vice* might be expected to produce news that is representative of the population as a whole when their main target audience is young people. Data suggests, however, that gender parity exists *across* age categories in the UK and US, meaning the demographic composition of young people is also roughly even in terms of gender (Office for National Statistics 2011; United States Census Bureau 2017).

⁸⁸ See Chapter 5 (section 5.3).

41-50	5.7	10.5	12.3	11.9
51-60	9.7	4.7	9.0	11.4
61-70	15.4	13.6	13.7	24.9
71+	0.6	6.8	8.1	3.5
Can't tell	45.1	41.9	30.3	26.9
Not applicable	17.1	15.7	16.6	16.4
TOTAL	100	100	100	100

Sources aged 61-70 were quoted the most in news reports by all four outlets, in particular, by the BBC (24.9% of total sources). The next best represented age category was 41-50 for *Vice, The Guardian*, and *BBC*, and 51-60 for *BuzzFeed*. Overall, then, middle-aged to older sources were quoted the most in news coverage. This was somewhat surprising in regard to *BuzzFeed* and *Vice*, which have both branded themselves publically as a "voice" of youth culture. As above, this was also supported in my interviews with journalists, who spoke about the importance of providing a platform to those traditionally marginalised by the media, including young people. One important caveat, however, is that it was often difficult to tell the age of sources, particularly individuals categorised as "ordinary citizens", who were frequently quoted by *BuzzFeed* and *Vice* in news coverage. In fact, only 6.67% and 15.8% of ordinary citizens were assigned age categories, with the remainder being coded as "can't tell".

In instances where it was possible to code younger sources, one potentially revealing point of difference was the *context* in which these sources appeared. In coverage by *the Guardian* and *BBC*, sources aged 40 and under were quoted

the most in news about crime and terrorism⁸⁹; at least three times as often as any other news topic. By contrast, in *BuzzFeed* and *Vice's* coverage, more than half of quotes by sources aged 40 and under appeared in news about social rights, issues, and protests.

Finally, sources were analysed according to ethnicity. In some instances, this was difficult to determine, hence the slightly higher number of sources coded as "can't tell":

	BuzzFeed (n=40)	VICE (n=40)	Guardian (n=40)	BBC (n=40)
White (%)	51.4	53.9	59.7	62.2
Black (%)	0.6	4.2	2.8	5.5
Asian and Indian (%)	4.0	2.6	8.1	7.0
Hispanic (%)	0	0.5	1.4	0
Can't tell (%)	26.9	23	11.4	8.0
Not applicable (%)	17.1	15.7	16.6	17.4
TOTAL	100	100	100	100

Table 17: Ethnicity of sources in news articles

White voices dominated news coverage across all four outlets. Again, this fits with other research which has found evidence of a bias towards whites in news coverage (e.g. Campbell 1995; Kurpius 2002). The *BBC* (62.2%) followed by the *Guardian* (53.9%), featured the highest number of white sources, while

⁸⁹ These findings connect with other research that has shown that young people tend to be overrepresented by traditional media in stories about crime (Griffin 2004; Devlin 2006; Wayne et al. 2008).

BuzzFeed (51.4%) and Vice (53.9%) featured between six and eleven percent less. This, however, did not correspond with an increase in non-white voices. In fact, the *Guardian* (12.3%) and *BBC* (12.5%) used more non-white sources⁹⁰ in news reports compared to BuzzFeed (4.6%) and Vice (7.3%). This can be explained partly by The Guardian and BBC's more extensive coverage of international politics, which quoted a relatively high portion of Asian actors⁹¹. This aside, the *BBC* guoted a relatively high number of black sources in stories about government and politics. However, two-thirds of these sources came from a single story examining statements made by the American politician, Ben Carson. Given the low number of non-white sources used by *BuzzFeed* and Vice in news coverage, it was more difficult to identify any discernible pattern to their coverage. News by Vice on the subject of social rights, issues, and protests, featured half of all non-white sources, closely followed by news about government and politics (46.2%). In *BuzzFeed's* coverage, non-white sources were fairly evenly split between news on government and politics, and social rights, issues, and protests.

8.2.4. Analysis and Discussion

Examined collectively, the empirical data on sourcing practices at *BuzzFeed* and *Vice* demonstrates a significant chasm between rhetoric and practice. Despite claiming to be a "voice" for young people, these findings demonstrate a clear "pattern of access, credibility, and news practices" (Kurpius 2002: 853) across coverage, resulting in the dominance of a fairly "narrow white male perspective" in news (Costera Meijer 2007: 178). That white, male, middle-aged officials appear to dominate news coverage is consistent with the historic literature on sourcing practices, which has shown that the media routinely provide a platform for cultural and political elites, privileging those in power over other, "non-elite" actors such as activists and ordinary people (e.g. Sigal 1973; Hall et al. 1978; Tuchman 1978; Soloski 1989; Gans 2004). Added to the fact that a relatively low number of sources from social media were used in news

⁹⁰ Note this is a cumulative percentage taken from combining the three non-white source categories together.

⁹¹ These stories were focused in particular on North Korea.

coverage, it would seem that, even at *BuzzFeed* and *Vice*, there was not a fundamental shift occurring in journalism's gatekeeping role or professional sourcing standards and routines.

Perhaps the only exception to this was *BuzzFeed* and *Vice's* use of ordinary citizens in news coverage, who made up nearly a fifth of total quoted sources. The high prevalence of these sources in coverage about social rights, issues, and protests suggests that, in some circumstances, *BuzzFeed* and *Vice* might have played a role in extending the public sphere by stimulating a form of bottom-up, "sub-politics" (Beck 1997), whereby ordinary citizens and activists were given voice in coverage based around particular social causes and issues (see Dahlgren 2005). This, however, must be parsed with the knowledge that *BuzzFeed* and *Vice* featured very few sources from civil society, and a scarce number of young and/or non-white sources. As a result, it would be difficult to herald *BuzzFeed* and *Vice's* reporting as a truly "radical popular" style of news – the kind commonly associated with alternative forms of journalism (see Harcup 2003; Atton and Wickenden 2005; Atton and Hamilton 2008).

These findings generally problematize claims made by some scholars that the internet will necessary "expand the public sphere by introducing more balanced news access for a wider range of sources" (Van Leuven et al. 2014: 851; see Dahlgren 2005; Castells 2008; Gans 2011). They also feel especially discordant in relation to *BuzzFeed* and *Vice*, given they are "native" to the digital environment and carry reputations for being ideologically progressive, editorially diverse⁹², and interested in a broad spectrum of social rights issues that concern ethnic minorities and other marginalized groups. On this note, I wish to end this section by considering some of the major reasons why these factors do not seem to have significantly impacted sourcing practices.

The first and perhaps the most obvious reason relates to efficiency. As media sociology has explicated, news is less an individual activity than an "organisational accomplishment", designed to guarantee news stories are

⁹² Here, I am specifically referring to BuzzFeed.

produced "on time and to a predetermined form" (Cottle 2007: 3). Institutional sources such as government spokespersons and politicians assist with this process by providing a reliable, steady stream of easily verifiable and credible information to journalists, which they proffer in exchange for publicity⁹³ (e.g. Sigal 1973; Hall et al. 1978; Gans 1979; Gandy 1982). Related to journalists' reliance on official sources, another possible explanation for an imbalance in representation relates to what Armstrong (2004) describes as a "mirror" effect in news coverage. Sources from key institutions such as the government, police and courts tend to be disproportionately white, middle-aged and male – with this imbalance being reproduced by journalists from *BuzzFeed* and *Vice* would consult a broader range of sources in non-routine, or "enterprise" reporting, such as features and investigations, which were not analysed in this particular study (see Hansen 1991).

Outside the immediate locale of the newsroom, another potential influence on source selection relates to the broader journalistic field. News routines, as established several decades ago, are as much the product of a shared "professional ideology", as they are a consequence of individual and organizational factors (Hall et al 1978: 249). As alluded to above, ethics and standards are one important facet of journalistic professionalism, but an arguably more significant norm is objectivity, which, as Soloski (1989) contends, functions like a "control mechanism", guiding journalists on what counts as newsworthy and who counts as a valid news source. The literature in this area suggests that journalists regularly seek out quotes from authoritative sources, because the act of outsourcing opinion and facts to outsiders, helps insinuate journalists from criticism or accusations of bias in reporting (e.g. Cook 1998). This "strategic ritual" (Tuchman 1972) also seemed to be in place at *BuzzFeed* and *Vice*, with journalists possibly using institutional sources as a source of "facts" to avoid accusations of inaccuracy or bias. In Vice's case, their reputation for producing highly subjective news seemed

⁹³ Somewhere around here, I just want to add a Tuchman reference, when she talks about the 'news net'.

incongruous with this finding. At the same time, the mixing of "facts" and "opinion" might not necessarily have precluded the ability of journalists to still present news in a more subjective and opinionated style.

One final and important facet of the relationship between journalists and elite sources rests in the symbolic power of news language. As both Bourdieu (1991) and Carlson (2017) have observed, one reason why journalists are recognised as culturally authoritative is because of their adherence to certain narrative conventions - "[objective writing] ... the inverted-pyramid style-account with few adjectives and frequent attribution to sources" (Carlson 2017: 59) - which journalists deploy to "promote themselves as authoritative and credible spokespersons of 'real life' events" (Zelizer 1992: vii). In a Bourdieusian sense, these now familiar narrative conventions have become an integral part of the field's cultural capital and doxa. One consequence of this, as Vos (2016: 388) writes, is that journalists are inclined to accept that "authoritative news comes from authoritative sources", leading to a reproduction of elite opinion in news texts. There is a case to make that these practices are even more relevant to BuzzFeed and Vice, which, lacking in the symbolic capital of traditional organisations, follow the established journalistic doxa to heighten their chances of being recognised as legitimate – both by fellow peers and the public at large.

8.3. Objectivity

Having spent some time discussing the norm of gatekeeping and the various processes of gathering, selecting, and filtering the news, the remainder of this chapter looks more closely at the *form* of news; that is, the various design elements, participatory affordances, and narrative conventions that undergird news stories at *BuzzFeed* and *Vice*. While the routines associated with these different elements are worth studying in their own right, understanding the materiality of news is doubly important in that it helps reveal how journalists structure knowledge and convey their authority. As Bourdieu (1991) recognised, news discourse is equipped with an "almost magical" symbolic power: the "power of constituting the given through utterances, of making

people see and believe, of confirming or transforming the vision of the world and, thereby, action on the world and thus the world itself" (Bourdieu 1991: 170).

In the digital era, numerous critics, commentators, and scholars have expressed concern that this symbolic power might be waning. As discussed earlier in this chapter, this partly relates to journalism's apparently loss of gatekeeping control; or diminished capacity to determine what the public "see, hear, and read" about the world (Deuze 2005: 451). Another aspect concerns the occupational norm of objectivity, which, by many estimations, has been *the* defining norm of Anglo-American journalism over the last century (Schudson 1978; 2001; Lichtenberg 1996; Mindich 1998). In recent years, a growing mass of critics, both from inside and outside the industry, have begun to question the viability of objectivity as an authoritative means of truth-telling (e.g. Singer 2008; Rosen 2010; Peters 2011; Hellmueller et al. 2013; Wahl-Jorgensen 2013). At its core, this critique is concerned with problematising the traditional epistemology of journalism, or journalism's established "ways of knowing" (Wahl-Jorgensen 2016: 129), which is seen to clash with the open, participatory logic of digital media and culture (Lewis 2012).

While these boundary disputes have sparked concern about journalism's ongoing role and relevance (Broersma and Graham 2013), more recent evidence would suggest that the cultural capital of journalism might be shifting in response to the exogenous "shock" (Nielsen 2012) of digital technology. Scholars highlight new and emerging forms of journalism, premised on the affordances of digital technology, that emphasise values such as emotion, authenticity, transparency, and subjectivity (Karlsson 2011; Zeller and Hermida 2015; Beckett and Deuze 2016; Wahl-Jorgensen 2016). From an epistemological standpoint, these forms diverge from objectivity in that they all seek to establish a different "authority relation" with the audience (Carlson 2017). While objectivity has been viewed as a relatively "closed process" that accentuates distance, detachment, and didacticism (Hellmueller et al. 2013), the values of transparency, participation, and subjectivity, imply openness and accountability, congregating around the idea of the journalist speaking "from a

position of parity with an informed audience rather than as a member of a detached elite" (Chadwick 2013: 176). In this section, I consider these values in turn, exploring how journalists from *BuzzFeed* and *Vice* discursively constructed their relationship with the objectivity norm, both in rhetoric and in practice.

8.3.1. Transparency

While it was evident that journalists were broadly supportive of objectivity as a "strategic ritual" (Tuchman 1972)⁹⁴, they appeared to harbour stronger doubts about objectivity as an epistemic basis for news. This seemed partly borne from the recognition that the internet had provided ordinary people with more means to criticise, question, and interrogate journalistic methods and styles of constructing news stories. As one editor from *BuzzFeed* put it:

"What [the internet] has done is made much more visible how the sausage gets made, and that makes a lot of people very uncomfortable...I don't think there has been a universal collapse of the wall between journalists and the audience. [But] in general, more journalists are exposed to audience feedback in more ways than they were before" (Respondent K, editor, *BuzzFeed US*).

Similarly, a reporter from *BuzzFeed UK* felt that journalists had to be "much more accountable" online, not just because it was "good practice and ethics" but because people had the ability to "Google every bit of [a story] and tell you when it's not [correct]" (Respondent I, reporter, *BuzzFeed UK*). Yet while *BuzzFeed* journalists could identify some of the potential roots of discontent between journalists and audiences, they were less forthcoming in explaining *how BuzzFeed* practically aimed to address these issues. Ben Smith, editor-inchief of *BuzzFeed US*, has perhaps been most explicit, arguing for the need to embrace *transparency* in news reporting, "sharing…knowledge and sources",

⁹⁴ Please refer to the previous section (8.2) on sourcing practices.

while rejecting "traditional reporting procedures of ostentatious, and sometimes false, balance and voice-of-God authority" (Smith 2017b).

At Vice, journalists similarly recognised the need for transparency, but spoke about this more in regard to the form of news. One editor, for example, contrasted Vice's approach with the objective style and presentation of traditional nightly news broadcasts:

"What I don't like about nightly news that I think turned a lot people off was this artifice, and I believe it is an artifice, of a reporter at the top of the segment saying here's what happened today. Their hair looks great, they're probably wearing a flack jacket if it's a warzone, they're probably standing away from the action...what you're losing is a lot of the interesting stuff, what it [feels] like to be there, the process of actually gathering news...I think that's [partly] why people...gravitated to Vice. I think people want to be informed, and I think they want to feel like they're not being patronised...you can't legislate authenticity but I do think Vice felt authentic and people liked that" (Respondent J, editor, *Vice US*).

Providing readers or viewers of *Vice News* with "a sense of how the story came to be and why it was presented as it was" (Kovach and Rosentiel 2001: 83), was considered an important aspect of reporting by many *Vice* journalists. One reporter felt that news at *Vice* portrayed "reality as it is [sic]", even if, at times, this was "unpleasant for the audience" (Respondent H, reporter, *Vice EU*). In fact, one editor went as far as to label this the "big innovation" that *Vice* introduced to the wider field of journalism. As they put it:

"I think the big innovation was being transparent and showing a reporter gathering the news or trying to make sense of what was going on in the world and bringing viewers along for the ride...sort of seeing the process...that's what I think *Vice* did really well (Respondent J, editor, *Vice US*).
While most discussions about transparency seemed to take place in the context of *Vice's* video content, another editor spoke about this value in regard to *Vice's* textual output:

"I think the posture of [*Vice*] and the presentation of a lot of the material, was on a level, if you will, with the audience. More so than your typical newswire copy is going to seem at first blush, or whatever" (Respondent E, editor, *Vice US*).

In this sense, transparency also emerged from a desire to establish a different kind of "authority relation" with the audience (Carlson 2017). As one reporter explained, being transparent was about bringing a more "raw" aesthetic to reporting; one which aspired to present information "truthfully" by getting closer to the subject at hand, rather than adopting a more distanced, "objective" posture. This style, he argued, particularly resonated with younger audiences, who were apparently bored of conventional news formats that "overlooked them", and were seeking a more "in-depth [and] personal guide" to current affairs (Respondent A, reporter, *Vice UK*).

8.3.1.1. Disclosure Transparency

While from a rhetorical standpoint, transparency seemed to be valued by journalists at *BuzzFeed* and *Vice*, this does little to reveal how it was routinised in practice. Given the lack of an adequate framework for studying transparency, the content analysis portion of this research settled on comparing several manifest features of news stories from *BuzzFeed* and *Vice*, with content produced by *The Guardian* and *BBC*. The purpose here was to ascertain a) how transparency was routinized in every day practice, and b) how, if at all, transparency features varied between *BuzzFeed* and *Vice* and traditional media organisations.

To begin, two very basic features of news stories were analysed: *authorial information* and *timestamps*. The objective here was to see how much

information each organisation routinely provided about the origin and province of their published news articles.

	BuzzFeed (n=179)	Vice (n=174)	Guardian (n=200)	BBC (n=200)
Author's name only (%)	3.4	37.4	56	12.6
Author's name with contact details (%)	96.0	51.7	33.5	2.0
No authorial information (%)	0.6	10.9	10.5	85.4
TOTAL	100	100	100	100

Table 18: Authorial information in news articles

Table 19: Time stamps in news articles

	BuzzFeed (n=179)	Vice (n=174)	Guardian (n=200)	BBC (n=200)
Date of publication only (%)	0	38.5	0	10.6
Time of publication only (%)	0	0	0	88.9
Date and time of publication (%)	100	60.3	100	0.5
No time stamp (%)	0	1.2	0	0

TOTAL	100	100	100	100
-------	-----	-----	-----	-----

These findings provide some evidence that the transparency norm had a material impact on the routines of news production at each organisation. This was most obvious in the use of timestamps in news articles, with all four outlets making some use of the affordances of digital technology to display the time (and date) of publication of a story. *BuzzFeed* and *The Guardian* were the most transparent in this respect, always displaying the time and date of publication of their news articles. *Vice* were generally more inconsistent, sometimes listing a date of publication and other times listing a date and time of publication for their stories. There was no discernible pattern in regard to why some stories were more transparent in their use of timestamps compared to others. At the *BBC*, partly because of the frequency at which news stories were published, most articles indicated just their time of publication. This, however, might also have been a strategic attempt by the *BBC* to emphasise the immediacy of news, giving the impression to readers that they were a source of the most up-to-date information on breaking news and recent events.

Moving on to the use of authorial information in news stories, both digital natives generally provided readers more opportunities to contact a news story's author(s) directly, though *BuzzFeed* did this far more consistently than *Vice*. This would seem to fit with previous assertions made by journalists from both organisations, particularly those at *BuzzFeed*, who felt that the online environment had given people more opportunities to "monitor, check, criticize and even intervene in the journalistic process" (Deuze 2005: 455). This aside, there is a technology-based argument to make why *BuzzFeed* and *Vice* might have been more forthcoming about the authorship of their news articles. As Steensen (2016) has observed, social media platforms are contributing to an individualization, intimization, and personalization of journalists above the "institutional brands" of organisations. This might also explain why *BuzzFeed*,

more so than *Vice*, was transparent about the authorship of its articles, given its seemingly heavier dependence on social media as source of traffic.

In contrast to the two digital natives, both The Guardian and BBC appeared more wedded to maintaining a degree of distance between reporters and the audience, downplaying a "subjectivization of facts" (Schudson 1978: 144). In the BBC's case, an organisational emphasis on impartiality in reporting might have been one contributing factor, with the organisation looking to minimise perceptions of authorial jurisdiction over individual news stories (see BBC 2018a). In addition, the BBC has a separate area on their news website for readers to leave feedback, report errors, or make complaints about news stories, which might have mitigated against the need to delegate some of these responsibilities to individual journalists (BBC 2018b). Likewise, at The Guardian, interactions seemed more formalised, with a "small pool of dedicated moderators" mediating communications between journalists and readers, primarily through the activity of commenting on articles, rather than contacting journalists directly (Guardian 2009). This might explain why most stories published by The Guardian only listed the name of an author and omitted their contact details. This also draws to attention the importance of resources in circumscribing transparency practices. In the absence of support from more formal feedback mechanisms, it is possible that journalists at *BuzzFeed*, and to a lesser extent, Vice, were required to take more responsibility for their own stories, becoming more accountable by default, rather than because of some overarching ideological commitment.

Moving beyond these more basic features of news articles, transparency was also operationalized by examining the use of hyperlinks in news stories. Hyperlinks are commonly regarded as one of the pre-eminent features of digital communications and have been frequently lauded for their potential to improve the quality and credibility of journalism online. As Steensen (2011: 313) observes: "The general assumption of researchers interested in hypertextual online journalism is that if hypertext is used innovatively it would provide a range of advantages over print journalism: no limitations of space, the possibility to offer a variety of perspectives, no finite deadline, direct access to sources, personalized paths of news perception and reading, contextualization of breaking news, and simultaneous targeting of different groups of readers".

Several of the above features of hypertext have been connected with ideas of journalistic transparency. Coddington (2014: 141) argues that "links can help reinforce a report's facticity by connecting readers directly with sources and showing readers how journalists know what they know". In this way, hyperlinks are seen to bridge the epistemological divide between journalistic expertise and the lay audience, letting ordinary people peek "behind the curtain" (Lowrey and Anderson 2005) to observe and assess news gathering practices. Research generally suggests that hyperlinking has become an institutionalised practice in online journalism, although differences have been observed between the way different actors use links in news stories (e.g. Quandt 2008; Tsui 2008; Napoli 2010; Coddington 2012; 2014).

For this study, two very basic factors relating to hyperlinks were examined: the number of links used per news article and the source of these links (internal or external to the news site).

	BuzzFeed (n=179)	Vice (n=174)	Guardian (n=200)	BBC (n=200)
Total no. of hyperlinks	1236	994	1167	816
Mean no. of hyperlinks per article (%)	7.0	5.8	6.2	4.1

Table 20: Hyperlinks in news stories

Beginning with the total number of hyperlinks used in news stories, all four organisations routinely used links in news stories; though some more than others. *BuzzFeed* had an average of 7.0 links per news story – the highest in the sample – although this result was slightly skewed by the appearance of two stories that contained 143 links, and 112 links, respectively. This was also an issue in the sample of stories by *The Guardian* (US) and *Vice* (US), which both contained one story with an unusually high number of links. With these outliers discounted, a slightly different set of results emerged:

	BuzzFeed (n=177)	Vice (n=173)	Guardian (n=199)	BBC (n=200
Total no. of hyperlinks	981	911	1065	816
Mean no. of hyperlinks per article (%)	5.6	5.3	5.7	4.1

Table 21: Hyperlinks in news stories (adjusted)

Based on these results, *The Guardian*, not *BuzzFeed*, published stories containing the highest average number of links per article (5.7). This was very closely followed by *BuzzFeed* and *Vice*, which both still featured over five links per article on average. Only the *BBC* contained less than five links per article (4.1), but nevertheless made routine use of links in coverage. In addition, this number would have likely been higher if the coding procedure had included links that appeared *outside* the body text of news articles. At the *BBC*, it was common for links to appear in widgets next to articles, rather than as part of the main text of a story. One final observation relates to differences in the total number of hyperlinks between the UK and US websites of all four news organisations in the sample. Once again, discounting the four stories with the unusually high number of hyperlinks, the following results were obtained:

	BuzzFeed UK	Vice UK	Guardian UK	BBC UK
	(n=96)	(n=82)	(n=100)	(n=100)
Total no. of hyperlinks	4.10	362	455	352
Mean no. of hyperlinks per article (%)	4.3	4.4	5.0	3.6
	BuzzFeed US (n=81)	Vice US (n=91)	Guardian US (n=99)	BBC US (n=100)
	(11-01)	(11-31)	((
Total no. of hyperlinks	571	549	610	464

Table 22: Hyperlinks in news stories (UK and US split)

As the results show, the US websites of all four organisations used a higher mean number of links per article compared to their UK counterparts. This difference was most pronounced at *BuzzFeed*, where stories published by *BuzzFeed US* used on average nearly three more links per article compared to *BuzzFeed UK*. One simple explanation for this difference was that journalists from the US had a much larger pool of external sources to draw upon, given the size and diversity of the US news ecosystem compared to the UK news ecosystem. In respect to *The Guardian* and *BBC*, another reason for this discrepancy might have related to more resources (compared to their much larger UK operations), with both organisations having to rely on more external sources of information to make up for a shortage in editorial resources. To add more nuance to these findings, links from each story were examined according to whether they pointed to an external or internal source:

	BuzzFeed (n=179)	VICE (n=174)	Guardian (n=200)	BBC (n=200)
Total no. of internal hyperlinks	243	275	766	577
Mean no. of internal hyperlinks per article (%)	1.4	1.6	4.1	2.9
Total no. of external hyperlinks	993	717	392	232
Mean no. of external hyperlinks per article (%)	5.7	4.2	2.1	1.2

Table 23: Internal and external hyperlinks in news stories

As the data shows, there were clear differences between the two digital natives and two traditional news organisations in the use of internal and external links in news stories. Beginning with *The Guardian* and *BBC*, internal links were used on average twice as often in news stories compared to external links. This chimes with findings from other research, which has shown that traditional news organisations generally prefer to link to internal sources of information, rather than direct users to external online sources (e.g. Barnhurst 2002; 2010; Dimitrova et al. 2003; Termayne 2005). This is often attributed to professional journalism's normative desire for control over news content, which has proven an awkward fit with the participatory logic of digital media and networked aspects of communication (Singer 2008; Lewis 2012).

Another reason, however, might relate to the fact that *The Guardian* and *BBC* imposed stricter internal constraints on linking practices, encouraging journalists to link to "supposedly dispassionate, factual information provided by established, authoritative, and largely professional media sources" in an attempt to minimise the conflict arising from including more "opinion voices and

partisan sources...that might be perceived as a threat to the norm of journalistic objectivity" (Coddington 2012: 149). This was certainly apparent in the *BBC's* case, who state explicitly in their editorial guidelines that external links must be "editorially justifiable" by normally being "factually accurate and of high quality", for example, pointing to newspaper sites and "reputable" blogs (BBC 2018a). The *BBC's* commitment to impartiality in reporting also means that any source that has a clear "editorial standpoint" should be balanced with a "reasonable range of [other] editorial perspectives" so as to avoid bias (BBC ibid.).

At *BuzzFeed* and *Vice*, the majority of links in news stories pointed to external sources, rather than internal content, which was true for both the UK and US websites of each organisation. This difference was sharpest at BuzzFeed, where external hyperlinks appeared roughly four times as often as internal hyperlinks in news articles. One explanation for this is that compared to The Guardian and BBC, both BuzzFeed and Vice had a far less extensive archive of internal sources to draw upon, leading to a higher dependence on external sources to add information and context to stories. Unlike both traditional news organisations, there also appeared to be less formalised procedures for linking to sources at *BuzzFeed* and *Vice*, which seemed to correspond with a higher average number of external links appearing in each news article. The practice of linking out also helped validate the view from journalists at BuzzFeed and Vice that openness and transparency were normatively desirable values in online journalism. Rather than linking inwards and "erecting walls" around journalistic products and processes (Singer 2008: 75), journalists at BuzzFeed and Vice seemed comfortable letting audiences "peek behind the curtain" to observe news gathering processes (Lowrey and Anderson 2005). This approach can be traced back to the antecedent practices of bloggers, whose "culturally embedded obligation to link to the [external] sources of their information" has been an integral part of their doxa and cultural capital since the earliest days of web-based writing (Carlson 2015: 8). Ben Smith, editor-inchief of *BuzzFeed US* has acknowledged this previously, discussing in one post entitled "My Life in the Blogosphere" the influence of blogging on the news culture at *BuzzFeed*:

"Indeed, the strongest new news outlets...have also co-opted and professionalized the tools and ethos of bloggers — fast, direct publishing; an informal voice; a commitment to transparency. We've pulled in some of the adaptable stars of that era. And we believe those people, tools, and values can serve our unchanging commitments to immediate, well-told, fearless, compelling, and independent journalism" (Smith 2015b).

While *Vice* have followed a somewhat different trajectory into journalism (see Chapter 5), the principles of blogging that Smith (ibid.) refers to above – "fast, direct publishing; an informal voice; a commitment to transparency" – would similarly seem to bear an imprint on the style and methods that *Vice* use to construct news stories, including their linking practices.

Based on these results, it was clear that certain aspects of disclosure transparency were routinised in daily news work at *BuzzFeed* and *Vice*. These processes were normatively valued by journalists because they supported a collective desire to appear more transparent by "being open [with audiences] about how news [was] being produced" (Karlsson 2010: 537). Based on this evidence, there were signs that objectivity, at least from an epistemological standpoint, was ceding some ground to transparency.

8.3.1.2. Participatory Transparency

Nonetheless, it is important to recall that transparency is not one "thing", but rather, a set of practices and values that exist on a spectrum (Karlsson 2010). If disclosure transparency is about revealing more to the audience about how news is gathered and assembled, then participatory transparency is more radical in that it "aims at getting the audience *involved* in the news production process in various ways" (Karlsson 2010: 578, emphasis added). Ideas of journalism becoming more participatory connect with broader normative arguments advanced by various scholars and public intellectuals in the literature on digital journalism (e.g. Gillmor 2004; Jenkins 2004; Jarvis 2006; Rosen 2006; Shirky 2008). This work – which has its basis in theories of

participatory and deliberative democracy; Habermasian and Deweyan ideals of public communication; as well as the antecedent practices of public journalism (Anderson 2011) – generally argues that greater participation of audiences in the news "harbors the potential to democratize both journalism and society at large" (Borger et al. 2013: 125).

Despite these bold claims, research generally indicates that journalists have been unwilling to surrender this level of gatekeeping control over the news, viewing the encroachment of audiences in the news process as a threat to their professional status and autonomy (e.g. Domingo et al. 2008; Hermida and Thurman 2008; Wardle and Williams 2010; Hermida et al. 2011; Jönsson and Örnebring 2011). As a result, organisational efforts to resolve the professional-participatory tension have mainly focused on offering users' greater means to participate in the news *after* its publication, through activities like commenting, voting, and sharing. Whether such forms of participation promote "greater inclusivity or "thicker" forms of citizenship" is highly debatable (Peters and Witschge 2015: 20). On top of this, there is the more difficult but oft-neglected question of whether handing control to "the people formerly known as the audience" will necessarily improve the quality of journalism or diversify to the public sphere at large (Kreiss and Brennen 2016).

To explore these issues, news articles were assessed for different forms of participation. While not an exhaustive list of features⁹⁵, these categories provide an insight into the ways in which *BuzzFeed* and *Vice* have attempted to accommodate audiences in the news process – that is, beyond being more "open about how news is being produced" (Karlsson 2010: 537).

⁹⁵ Please refer to Appendix D for an explanation of each of these categories.

	BuzzFeed (n=179)	VICE (n=174)	Guardian (n=200)	BBC (n=200)
	((11 200)	(11 200)
Share (%)	100	100	100	100
Comment (%)	100	43.2	37.1	12.9
Poll (%)	1.9	0	0	0
Quiz (%)	0	0	0.6	0
Submit content (%)	1.3	0	0.6	2.3
Reader contributions published (%)	0.6	0	0.5	0
Reader news (%)	0	0	0	0

Table 24: Participatory features in news articles

As the above table illustrates, audience interaction with news stories was largely restricted to activities like sharing and commenting across all four organisations. Readers were very rarely invited to submit content and their contributions⁹⁶ were almost never featured in news stories. *BuzzFeed* offered the option to comment and share on every one of its news stories while for *Vice* and *The Guardian*, the ability to comment was offered in less than half of news articles. At the *BBC*, users were given the option to comment on only 12.9% of

⁹⁶ By audience contributions, I am referring to instances where readers have been previously asked to send in their own content, and these contributions now explicitly form part of the news story. This is different to journalists simply quoting ordinary citizens as sources in news stories – an aspect of reporting covered more extensively in the previous section (8.2) on sourcing practices.

news stories⁹⁷. No news stories were presented as wholly written by readers, with journalists generally retaining control over the writing and editing phase of news production.

Thus, in line with findings from other research (Domingo et al. 2008; Hermida et al. 2011), it appeared that *BuzzFeed* and *Vice* were predominantly interested in facilitating participation at the *distribution* (sharing news on social media) and *interpretive* (commenting on articles) phases of news production. This was also reflected in interviews, where journalists generally spoke about their interactions with the audience in terms of receiving feedback or comments on articles. It was not always clear whether journalists valued these forms of feedback. As two *BuzzFeed* reporters commented:

"You can't hide from people online! [laughs]. They find you. I get emails, sometimes. It's easier for them to interact with you as well. And as a woman, I would say, you get much more criticism. You get used to it [laughs]... That said,, it is nice when people acknowledge the work you're doing...I do get some positive feedback" (Respondent C, reporter, *BuzzFeed UK*).

"I mean, a lot of the time when people share stuff we get mentions and we see it and its 90% positive... It's people going like "look at this story...you'll love this". I try not to read the comments on a lot of the articles. If someone directly messages me on twitter or directly emails me I'll look at it. But I feel like, I've learnt from writing online, people who actually comment on the articles a lot of the time are just looking for fights and will say something deliberately inflammatory or go deliberately

⁹⁷ This relatively low percentage would seem to accord with Williams and Wardle's (2008) research into the integration of user-generated content (UGC) in news production at the *BBC*, which found that journalists were generally sceptical of the value of audience comments in news. It should also be noted that the *BBC* have a separate, dedicated website called 'Have Your Say', which actively invites readers to submit content (BBC 2018b). This also might help explain why users were not generally invited to comment on daily news stories.

against the idea of your story...A lot of the time, I try stay away from that" (Respondent M, reporter, *BuFeed UK*).

Two reporters from *Vice* appeared to have similarly mixed feelings about feedback from the audience:

I do [interact with the audience] now and then. I get a lot of people tweeting me saying "I really like that" and I'll go "oh thanks, what do you like about it? Because it's nice to know. But you always get, like, a lot of morons as well you know. Like I actually enjoy reading YouTube comments sometimes because it's really funny. You're just like "what kind of maniac is writing this stuff. But every now and again you get a real insight" (Respondent A, reporter, *Vice UK*).

"Thanks to social media, we get feedback often... There was a lot of kind of mad stuff... [But] I think it makes you better at your job too, and understand the responsibility you have because it's obviously having an effect...Sure, you'll get some abuse, but if you're used to covering a warzone a bit of internet abuse isn't a problem. So yeah, always a lot of feedback that you take on board, whether it's positive or negative" (Respondent F, editor, *Vice UK*).

In some cases, then, journalists were sceptical, even hostile towards certain forms of feedback – particularly direct comments on articles. Other times, they seemed to value working in an "ambient" (Hermida 2010) media environment, with various social media heightening incidental exposure to feedback from the audience. Nonetheless, there was a distinct lack of interest, both rhetorically and structurally, in facilitating more maximal forms of participation. As a result, transparency only seemed to lead to subtle shifts in journalists' cultural capital – not the "full-blown epistemological shift" (Vos and Craft 2016: 2) imagined by some advocates of participatory journalism.

8.3.3. Emotion

Aside from providing new opportunities for journalists to be more open and accountable, the rise of digital media also appears to have engendered a cultural shift towards forms of storytelling that privilege more personal, emotionally involved, and subjective modes of discourse (Zelizer 2009; Peters 2011; Wahl-Jorgensen 2016). In the wake of the so-called "affective" turn in social sciences (Clough and Halley 2007), there has been a growing interest in these forms of expression, extending to ideas about "affective" news, "constructed out of subjective experience, opinion, and emotion, all sustained by and sustaining ambient news environments" (Papacharissi 2015: 34). For some, these changes could portend an epistemological shift in journalism's "ways of knowing", as more 'objective' claims to knowledge and truth-telling give way to more "personalized, subjective, and emotional forms of narrative" in news (Wahl-Jorgensen 2016: 132; also Pantti 2010; Peters 2011; Beckett 2015; Beckett and Deuze 2016).

At *BuzzFeed* and *Vice*, journalists expressed clear rhetorical support for this shift in journalism's epistemology. Here, older journalistic forms were frequently pitted against newer styles of storytelling, as a way of explaining the superiority of more emotionally involved, personal, and subjective forms of news. As one reporter from *BuzzFeed UK* explained:

"I think we try to write in quite an engaging style. We don't want to be like a *Times* article; we try not to write in a dry style. If you compare *BBC* online copy, for example, which is known to be very straight and 'just the facts' compared to our style [sic]. We almost try to write with a bit more vigour and colour...to try and make it a bit more engaging" (Respondent A, reporter, *BuzzFeed UK*).

In a similar manner, one editor from *Vice* used the example of traditional news to explain how *Vice* looked to establish more of a "peer relationship" with readers, in an effort to draw them closer into stories:

"The narrative [of our news] would sort of compel the viewer along...it wasn't necessarily one that 'lectured', you know, or presented material in a kind of didactic way. It was a lot more casual, it was more of a peer relationship in a way [while] at the same time still serving the purposes of being informative" (Respondent E, editor, *Vice US*).

On the subject of the journalist-audience relationship, another editor from *Vice* explained how in news coverage, reporters were encouraged to act as "representatives, or avatars, for the viewers", which helped make them appear more "authentic and not omniscient" (Respondent J, editor, *Vice US*). At *BuzzFeed*, reporters similarly strived to establish a closer connection with readers, but spoke about this more in the context of the *language* of news stories, suggesting that more informal, conversational styles of reporting could help craft a greater "experience of involvement" in the news (Peters 2011). As two reporters explained:

"If you were telling someone about a story you had read, how would you actually say it to another human being?... It's not like it's dumbed down, it's just why not say it like someone would say it...makes [the story] sound more human and not talking from your lofty perch of being a 'serious' journalist" (Respondent H, reporter, *BuzzFeed UK*)

"People still want expertise, but it's about presenting it to the way people are thinking. And that's the same with news. People don't want to feel like they've been dumbed down to" (Respondent B, reporter, *BuzzFeed UK*).

Here, "objective" styles of news were criticised for their seemingly top-down, elitist posture, while styles of discourse that aspired for greater parity with the reader were favoured. Considering the appeal of these forms, some journalists moved beyond the realms of their own habitus to consider the relationship between these kinds of news and the broader communications environment. Some, for example, drew attention to the link between more personalised and emotional forms of news and the growth of social media. To repeat an earlier quote from Chapter 6:

"Seventy-five percent of our traffic comes through social media referrals, not homepage clicks...so when we're doing investigations we're always thinking 'okay, is this something people are going to talk about, is this something people are going to want to share? And generally, you just want something really shocking or really emotive...People share something if they feel something" (Respondent A, reporter, *BuzzFeed UK*).

An emphasis on emotional news was partly, then, a result of *BuzzFeed's* own publishing model, which was heavily dependent on social media for traffic and in particular, the metric of *sharing*⁹⁸. In this way, emotionality in news was also shaped by organisational imperatives, which seemed to push reporters towards integrating emotion into their own sense of news judgement. Comparing their experiences with working for a legacy news organisation, one reporter from *BuzzFeed* explained:

"At [legacy news organisation] I wasn't really thinking about readers having an emotional reaction to [stories]...Where [sic] with *BuzzFeed*, I'm thinking of an emotion that I want to get out of the person, like, of this is really bad, this is really funny, that sort of thing...You want to get a reaction out of people because you want them to share" (Respondent C, reporter, *BuzzFeed UK*).

At *Vice*, journalists also made the link between the status of emotion in reporting and the broader goals of their own organisation. One such goal was reaching younger audiences, who, according to one *Vice* editor, were "overwhelmingly male, with a sort of appetite for adventure, for inanity and absurd things, for really shocking scenes...[and] vivid coverage" (Respondent J, editor, *Vice US*). An awareness of these tropes was linked with a need to

⁹⁸ For a more detailed discussion of sharing, please refer back to section 7.1.5.

create entertaining and viscerally engaging content. On the subject of *Vice's* war coverage, for example, one reporter said:

"People might [look at our coverage] and say, 'I think there shouldn't be so many gunshots in there', and I get it...But it's the Internet age man, the Playstation generation. We're making stuff that's valid for them and we're still making it accurate and right" (Respondent A, reporter, *Vice UK*).

Another, more experienced reporter, who had spent time working for a range of American news publications, compared the practices at *Vice* to those of tabloid outlets, who would "manipulate things and do whatever they could do to get punch, to be sensational, to get as much attention as possible" (Respondent D, reporter, *Vice US*). In this manner, conversations about emotion at *Vice* were sometimes inflected with a small, but creeping sense of cynicism about journalism becoming overly reader-oriented. One editor, for example, seemed to bemoan the ways that journalism had become more driven by a "looming idea of audience engagement", which had put pressures on news organisations to capture audiences' attention, whether through "titillation, some sort of ideological affinity...or a voice or brand that's appealing to [a particular] segment" (Respondent J, editor, *Vice US*). Other *Vice* interviewees, however, seemed to take a longer view, with one editor arguing that journalism had always "sought to sensationalise, to inform, to sway, to persuade, and to entertain" (Respondent E, editor, *Vice US*).

Taking these contributions into account, the content analysis portion of this study sought to explore how emotion was operationalised in practice at *BuzzFeed* and *Vice*. As discussed in the methods chapter of this thesis, there is no clear procedure for studying emotion in journalism⁹⁹. Perhaps the most

⁹⁹ While this is an issue that extends beyond journalism (e.g. Edwards 1999), it would be reasonable to suggest that the lack of an established procedure for studying emotion in journalism has been strongly hindered by scholars' historic aversion to non-objective news forms (see Peters 2011: 299-303 for a good overview).

developed methodology for studying emotion in journalism has come from Wahl-Jorgensen (2013), whose study of emotionality in Pulitzer prize-winning articles has provided the basis for some the analytical constructs used in this section.

To begin with, the language of headlines used in news articles was examined. In traditional journalism, "objective" news headlines tend to focus on 'just the facts' (Mindich 1998), offering a straightforward, impersonal, and mainly descriptive account of news events. By contrast, more emotive headlines employ a number of different discursive features to create a heightened "experience of involvement" in the news (Peters 2011). This includes techniques like "forward-referencing" (Blom and Hansen 2015), where a headline alludes to information contained within a story, typically using words like "this", "how", and "why" to arouse curiosity in the reader. For example:

Outlet	Headline	Date
Vice (UK)	What does Nigel Farage get out of supporting Donald Trump?	11/10/16
Vice (US)	What Wells Fargo knew	10/10/16
BuzzFeed (UK)	Meet the people who say their firms were destroyed by RBS	12/10/16
BuzzFeed (US)	How anti-science forces thrive on Facebook	07/03/17

Table 25: Examples of forward referencing in headlines

Aside from this, headlines were categorised as "emotive" if they used *personalisation*, by addressing the reader in the second-person (e.g. '6 things *you* need to know about the budget' – *BuzzFeed UK*), or, if they used special punctuation such as questions, or exclamation points to get readers' attention (e.g. Surprise! Republicans' Obama Care replacement would hurt the poor – *Vice US*).

	BuzzFeed (n=179)	Vice (n=174)	Guardian (n=200)	BBC (n=200)
Straight (%)	54.8	31.0	90	82
Emotive (%)	41.9	62.1	8.0	17
Ambiguous / can't tell (%)	3.3	6.9	2.0	1.0
TOTAL	100	100	100	100

Table 26: Headline language in news articles

As the data shows, both digital natives used emotive headlines far more frequently in news stories compared to *The Guardian* and *BBC*. This was most pronounced at *Vice*, where more than half (62.1%) of news articles used emotive headlines. In contrast, less than a fifth of news stories from the *BBC* used emotive headlines (17%), and, at the *Guardian*, emotive headlines were used only 8% of the time¹⁰⁰. These findings would appear to reaffirm comments made by journalists from *BuzzFeed* and *Vice*, who spoke about embracing more conversational and personal forms of language in news, in lieu of more objective, impersonal discourses. To get a better understanding of how emotion was *routinised* in news, the use of different headlines was cross-referenced with data on news formats. The aim here was to see to what extent emotionality extended beyond "narrative" forms of journalism like features and investigations, and bled into daily news formats such as reports and opinion pieces.

¹⁰⁰ On this note, there was also a small, but perceptive difference between the UK and US editions of *The Guardian* and *BBC*. While the UK editions of *The Guardian* and *BBC* employed emotive headlines only 5% and 13% of the time, respectively, these percentages rose to 11% and 21% for their US editions.

Table 27: Emotive headline versus news format

	BuzzFeed	Vice	Guardian	BBC
	(n=179)	(n=174)	(n=200)	(n=200)
Report (%)	25.2	50	4.8	8.2
Feature / investigation (%)	86.8	75	28.6	100
Interview (%)	50	77.8	0	100
Comment / opinion (%)	0	81.8	100	100
Review (%)	0	50	0	0
Listicle (%)	100	0	0	100
Quiz (%)	0	0	0	0
Game (%)	0	0	0	0
Aggregated content (%)	100	10	0	50
Live blog (%)	0	0	0	0
Other / can't tell (%)	0	55.6	0	66.7

As the data shows, emotive headlines were more commonly employed in 'formats like features and investigations, which was true for all outlets. In terms of daily reporting there was a significant difference in the use of emotive headlines between the two traditional news organisations and the two digital natives. While emotive headlines featured in just 4.7% and 8.2% of news reports by *The Guardian* and *BBC*, respectively, this number was higher at *BuzzFeed* (25.2%), and significantly higher at *Vice* (50%). Still, nearly three-quarters of news reports used a straight headline at *BuzzFeed*, which would seem to suggest that emotionality, at least at the level of daily reporting, was the exception for journalists, rather than the rule. At *Vice*, on the other hand,

emotionality appeared more strongly embedded in routines, with journalists employing emotive headlines in half of all published news reports.

In the interest of providing a more comprehensive analysis of emotionality, two further analytical constructs were developed. Building on the work of Wahl-Jorgensen (2013) and others (e.g. Thomson et al. 2008; Fink and Schudson 2014; Moon and Hadley 2014) news stories were analysed for whether they employed an 'emotive' lead paragraph or used personalised storytelling, in an effort to add emotional resonance to a particular subject and heighten readers' sense of involvement in the news. An 'emotive' lead paragraph was defined as a paragraph that jettisoned the "inverted pyramid" style of news presentation¹⁰¹, and instead opened with an anecdote or story to engage the reader. News stories that featured "personalized storytelling" focused on the emotions or experiences of individuals to add context and emotional weight to a particular subject or issue.

	BuzzFeed (n=179)	Vice (n=174)	Guardian (n=200)	BBC (n=200)
Inverted Pyramid (%)	58.7	18.4	74	85
Emotive (%)	30.7	62.1	15	8.0
Ambiguous / Can't tell (%)	9.5	19.0	6.0	6.0
Not applicable (%)	1.1	0.5	5.0	1.0
TOTAL	100	100	100	100

Table 28: Lead paragraph in news articles

¹⁰¹ The "inverted pyramid" is widely regarded as a common discursive feature of journalistic objectivity (Mindich 1998; Pöttker 2003). According to this style, the most important information, or the 'who-what-why-when' of a story comes first, with other information then appearing in descending order of importance.

	BuzzFeed	Vice	Guardian	BBC
	(n=179)	(n=174)	(n=200)	(n=200)
Yes (%)	14.5	24.1	4.5	4.0
No (%)	83.3	74.7	90.5	95
Ambiguous / Can't tell (%)	1.1	0.6	0	0
Not applicable (%)	1.1	0.6	5.0	1.0
TOTAL	100	100	100	100

Table 29: Personalised storytelling in news articles

As these two tables show, *BuzzFeed* and *Vice* used emotive narrative features in news stories more frequently compared to *The Guardian* and *BBC*. This difference was most pronounced at *Vice*, where nearly two-thirds of stories used an emotive lead paragraph (62.1%), and nearly a quarter of stories made use of personalised storytelling (24.1%). At *BuzzFeed*, these percentages were demonstrably lower, but news stories nevertheless made frequent use of emotive lead paragraphs (30.7%), and some use of personalised storytelling (14.5%). By comparison, *The Guardian* and *BBC* featured non-conventional lead paragraphs in 15% and 8% of stories, respectively, and generally made minimal use of personalised storytelling in news.

To add further nuance to these findings, data on lead paragraphs and personalised storytelling was cross-referenced with data on news formats. Here, the purpose was to observe the level to which emotive narrative techniques were routinised in daily reporting:

Table 30: Emotive lead paragraph versus news format

	BuzzFeed	Vice	Guardian	BBC
	(n=179)	(n=174)	(n=200)	(n=200)
Report (%)	11.8	41.7	10.2	3.5
Feature /				
Investigation	81.6	84.6	71.4	40
(%)				
Interview (%)	100	66.7	100	0
Comment /	100	86.4	25	71.4
Opinion (%)				
Review (%)	0	50	0	0
Listicle (%)	57.1	0	0	0
Quiz (%)	0	0	0	0
Game (%)	0	0	0	0
Aggregated	100	0	0	25.0
content (%)		-		
Live Blog (%)	0	0	0	0
Other / can't tell	0	77.8	50	33.3
(%)				

Table 31: Personalised storytelling versus news format

	BuzzFeed	VICE	Guardian	BBC
	(n=179)	(n=174)	(n=200)	(n=200)
Report (%)	0	8.3	0.6	2.9
Feature /				
Investigation	65.8	50	35.7	40
(%)				
Interview (%)	50	11.1	100	0
Comment /	0	22.7	25	14.3
opinion (%)	°			1 110
Review (%)	0	0	0	0
Listicle (%)	0	0	0	0
Quiz (%)	0	0	0	0
Game (%)	0	0	0	0
Aggregated	0	10	0	0
content (%)				
Live Blog (%)	0	0	0	0
Other / can't tell	0	33.3	50	0
(%)				

Much like the cross-referenced data on emotive headlines, findings here demonstrate that emotive techniques were reserved in principle for news formats that lent themselves to more "narrative" styles of storytelling, such as features and investigations¹⁰² (Bird and Dardenne 2009). That being said, *Vice*

¹⁰² This finding also might help account for the large difference in use of emotive techniques between *BuzzFeed* and *Vice*, and the two traditional news organisations in the sample. At the two digital natives, formats like features and investigations made up a *higher portion of total coverage*, compared to *The Guardian* and *BBC*, and it was

still displayed a proclivity for routinising emotion in daily reporting, with emotive leads appearing in close to half (41.7%) of all news reports. By comparison, *BuzzFeed, The Guardian*, and *BBC*, used emotive leads and personalised storytelling fairly infrequently in daily reporting, across both their UK and US websites. This suggests that, by and large, objectivity remained an important epistemic basis for news writing, at least at the day-to-day level of news production. One explanation for this might relate to efficiency, with the core discursive elements of objectivity providing a reliable and predictable toolkit for journalists to use to craft stories in a timely and efficient manner.

One final observation worth making relates to the use of emotive techniques across different news *subjects*.

	BuzzFeed (n=179)	VICE (n=174)	Guardian (n=200)	BBC (n=200)
Government / politics (%)	28.3	62.2	23.0	14.8
Social issues, rights, and protests (%)	42.4	62.2	31.6	16.7
Crime and terrorism (%)	25.0	58.3	13.8	2.5
Science, education, and technology (%)	18.8	33.3	12.5	14.3
Health (%)	50	100	0	40
Business and economy (%)	13.3	80	7.4	4.5

Table 32: Emotive story lead versus news format

these types of formats that most commonly employed emotive features (see Chapter 5).

Entertainment and arts (%)	31.8	75	5.9	5.6
Lifestyle (trends, fashion, travel) (%)	0	91.7	0	0
Accidents and disasters (%)	20	0	0	0
Energy and environment (%)	66.7	0	0	0
Other / can't tell (%)	57.1	15.3	0	0

Table 33: Personalised storytelling versus news subject

	BuzzFeed (n=179)	VICE (n=174)	Guardian (n=200)	BBC (n=200)
Government / politics (%)	28.3	62.2	23.0	14.8
Social issues, rights, and protests (%)	42.4	62.2	31.6	16.7
Crime and terrorism (%)	25.0	58.3	13.8	2.5
Science, education, and technology (%)	18.8	33.3	12.5	14.3
Health (%)	50	100	0	40
Business and economy (%)	13.3	80	7.4	4.5
Entertainment and arts (%)	31.8	75	5.9	5.6
Lifestyle (trends, fashion, travel) (%)	0	91.7	0	0

Accidents and disasters (%)	20	0	0	0
Energy and environment (%)	66.7	0	0	0
Other / can't tell (%)	57.1	15.3	0	0

Perhaps the first observation to make is that emotive strategies were not restricted to "softer" news subjects (for want of a better term¹⁰³), with emotive leads and personalised storytelling featuring in news on government and politics, as well as topics like international affairs, science, and technology. That being said, only *Vice* frequently used non-conventional leads for stories relating to business affairs and economics, with the other three outlets mostly covering stories of this nature using the conventional inverted pyramid structure. Somewhat ironically, it was "softer" stories relating to entertainment and the arts that were mostly written using conventional lead paragraphs and without personalised storytelling techniques. The exception to this, again, was *Vice*, which used emotive leads in 75% of its coverage on entertainment.

These findings would seem to complicate notions of a clear line between "hard" and "soft" news; or information and entertainment. Rather, they add weight to the growing consensus that in the 21st century, the "traditional dichotomies associated with journalism…are becoming progressively blurred" (Peters 2011: 298). In interviews, a few *BuzzFeed* journalists I spoke with alluded to this. One editor, for example, suggested that they found the supposed division between entertainment and news "overwhelmingly arbitrary and unhelpful" (Interviewee K, editor, *BuzzFeed US*). Similarly, a reporter from the UK argued that one of *BuzzFeed's* specialities was writing "high-brow stuff in a low-brow way, and low-brow stuff in a high-brow way", further complicating any sense of a line between serious and non-serious forms of news (Interviewee G, reporter, *BuzzFeed UK*).

¹⁰³ See Reinemann et al. (2011) for a detailed overview of the theoretical debate over "hard" and "soft" forms of news.

Finally, among all four news outlets, stories covering social rights issues, strikes, and protests, most frequently employed emotional leads and personalised storytelling. This makes sense in light of the fact that many of these stories were centred around the experiences of individuals; often members of minority groups whose own opinions were used to dramatize broader issues relating to race, gender, mental health, and sexuality. The mutually beneficial association between these subjects and emotionality offers another reason as to why *BuzzFeed* and *Vice* might have chosen to build a large portion of their coverage around these issues.

8.4. Conclusion

The goal of this final empirical chapter has been to explain how *BuzzFeed* and *Vice* approach content selection and presentation; paying close attention to the connection between rhetoric and practice. As stated in the introduction, there is little doubt that digital media "bring new potentialities to how journalists connect to news audiences as well as create new challenges in managing the flood of available content" (Carlson 2017: 74). Yet precisely how these changes affect the shape of news is highly contingent on various endogenous and exogenous factors, including organisational structures, working practices, understandings of the audience, and broader "social, especially political and economic, conditions in which [journalistic production] is organized" (Champagne 2005: 50).

Taking into account these factors, this chapter has made an important contribution by demonstrating, once again, the important role of traditional norms in structuring practices at *BuzzFeed* and *Vice*. Thus, while efforts were made by both organisations to introduce *new* forms of capital to the field, these attempts were strongly circumscribed by the "latent ideals" of journalistic professionalism (Anderson 2011b: 552), or those shared presuppositions regarding institutional roles, epistemologies, and ethical ideologies that constitute the cultural capital of the field (Vos et al. 2012). This was perhaps most evident in journalists' rhetorical desire for gatekeeping power over

content, based on the assumption that the "process of making decisions about what is news and how that news should be reported", should remain under their jurisdiction and control¹⁰⁴ (Hermida et al. 2011: 16). This was bolstered by a news process that favoured elite sourcing routines and only provided minimalist means for users to participate in the news. Thus, much like their traditional counterparts, *BuzzFeed* and *Vice* seemed committed to upholding a gatekeeping role, despite the internet theoretically opening up new opportunities for audiences to participate in the news.

While media sociology would attest to the importance of the 'routine' in determining this approach (Cottle 2007), a major contention of this chapter has been that in respect to *BuzzFeed* and *Vice*, the reproduction of these traditional discourses and practices of professional control are deeply connected to the organisational pursuit of prestige and status, or symbolic capital; the kind earned by, for example, breaking exclusive stories or being seen in close proximity with elite sources. As Williams et al. (2011: 86) note, the routines associated with traditional newswork "are durable partly because they are so important to the continuing authority of journalism: they not only construct external events as news, but they are self-legitimating practices". This would also seem to be the case for *BuzzFeed* and *Vice*, where professionalism (and its associated routines) "provides an extra-organizational set of guidelines" for achieving journalistic legitimacy, or *peer recognition*¹⁰⁵, to use Bourdieu's (1998) terms.

Yet it is the very presence of these professionalised norms and practices that expose *BuzzFeed* and *Vice* to the same sorts of criticisms levelled at traditional media: that they are elitist; overly reliant on official voices; fail to promote diversity and foster civic participation; and generally reinforce, rather than challenge, the dominant status quo (Cottle 2007). These criticisms are hard to

¹⁰⁴ Although as shown in the previous chapter, editorial judgement is being encroached upon by the 'quantified' audience (Anderson 2011), expressed via data and metrics.

¹⁰⁵ As a reminder, peer recognition is generally afforded to those who "internalize most completely the internal 'values' or principles of the field" (Bourdieu 1998: 70).

parse with the knowledge that *BuzzFeed* and *Vice* target a predominantly young audience, employ a diverse editorial staff¹⁰⁶, and have shown an interest in covering progressive political issues tied to the concerns of specific minority groups. These factors alone might be seen as logical precursors for an approach to news less focused on the institutional side of society, or one that might promote more diverse viewpoints from non-elite actors. Yet this would only seem to underline the importance of the professional paradigm in structuring digital news – *particularly* for new entrants, who seek to "[mirror] dominant practices and forms" to bolster their public and professional legitimacy (Lowrey 2011: 66). As Benson (1999: 468) notes, "entry into the journalistic field requires acceptance of the basic rules of the game, which themselves are a powerful force of inertia".

This being said, new entrants also seek to differentiate themselves by introducing "dispositions and position-takings which clash with the prevailing norms of production and the expectations of the field" (Bourdieu 1993: 57). This was evident in *BuzzFeed* and *Vice's* orientation toward the audience, which despite being rhetorically and structurally opposed to more direct forms of audience participation, seemed concerned with fostering participation in epistemological terms, signalling the desire to create a different "authority relation" with the audience (Carlson 2017). This was manifest in journalists' interest in transparency and emotionality, both of which were viewed as superior "ways of knowing" compared to the dominant epistemology of objectivity. In practice, however, efforts by journalists to "craft an experience of involvement" in the news were uneven (Peters 2011: 299). Perhaps the clearest endorsement of a shift in journalism's epistemology was in the use of certain transparency techniques to construct news stories. These practices were more pronounced at *BuzzFeed* and *Vice*, where the relatively "closed process" of objectivity seemed to give way to greater accountability and openness (Hartley 2013). For want of a better phrase, journalists seemed concerned with demonstrating a kind of epistemic humility, recognising their own limits, and

¹⁰⁶ As shown in Chapter 5 (section 5.3), *BuzzFeed* was significantly more diverse than *Vice*.

indeed, the limits of news texts themselves to provide a definitive account of what was happening in the world.

This was also evident in journalists' rhetorical embrace of emotionality in news discourse, which borrowed from certain ideas of participatory journalism to imagine the journalist-audience relationship as less didactic or lecturing, and more of a dialogical conversation between two equal parties (e.g. Benkler 2006; Jarvis 2006; Bruns 2008). Yet in practice, more emotional and personalized discourses were highly situational, and varied significantly according to news subject and format. Accordingly, objectivity still retained its relevance as a "strategic ritual" (Tuchman 1973), particularly in daily reporting. This would suggest that claims of a full-blown epistemological shift in journalism have been overstated. As Kreiss and Brennen (2016: 305) note, "there is, a simple question of the resources available for the routine and reliable provision of public information that haunts much of the literature around digital journalism". Hence, even if digital media might emphasise the need for more personal and emotional discourses in journalism, the problem of routinising these practices creates a need for the more stable and efficient routines of objectivity in newswork.

9. Conclusion: BuzzFeed and Vice in the Field of Journalism

"Men make their own history, but they do not make it as they please; they do not make it under self-selected circumstances, but under circumstances existing already, given and transmitted from the past" – Karl Marx, The Eighteenth Brumaire of Louis Napoleon, (1852: 115).

If a week is a long time in politics then what constitutes a long time in digital journalism? Nearly four years have passed since the commencement of this research project, and in that time the industry has continued to experience profound and unsettling change. *BuzzFeed* and *Vice* are still making news, but many of their contemporaries have disappeared or remain in decline. Technology platforms have continued to grow more powerful at the expense of publishers. There has been a 'pivot to video', quick followed by a 'pivot to reality' (Moses 2017). And, perhaps more profoundly, there has been a growing cultural backlash against elites and experts, including journalists, driven by a populist political surge in many Western democracies.

Taking stock of these rapid transformations, one would be forgiven for thinking the job of studying change in journalism is an exercise in futility. However, if my experience in the field has taught me anything, it is that "nothing has changed and yet everything is different", to quote the famous line of Jean-Paul Sartre. Indeed, even a cursory glance at previous technological revolutions – be it television, the telegraph, or radio – suggests that change is far more gradual than conventional discourse often implies (Zelizer 2015). Moreover, the tendency to frame the subject in before/after terms, often neglects what new institutionalists call "path dependency", which describes the way decisions about the future are historically contingent, shaped strongly by longstanding norms, which are themselves the product of previous struggles. This is as true for new entrants as it is for established members of the field. As Bourdieu (1996: 206) observes:

"The stakes of the struggle between dominants and pretenders [within any given field of cultural production, including journalism], the issues they dispute . . . depend on the state of the legitimate problematic, that is, the space of the possibilities bequeathed by previous struggles, a space which tends to give direction to the search for solutions and, consequently, influences the present and future of production".

This is an insight I have found myself returning to, time and again, during the course of this research project. One challenging aspect of researching digital natives is their imbrication with normative discourse on "innovation" in digital journalism – what Kreiss and Brennen (2016: 306) describe as "a 'catch-all' term that spans the development of novel business models, collaborations, technologies, practices, and content". Consequentially, digital natives have frequently been celebrated for "disrupting" the field; introducing new ideas and practices that are viewed as normatively better than older ways of doing journalism, simply by virtue of being "new" (Kreiss and Brennen 2016). This discourse has come at the cost of providing a more balanced assessment of digital natives, which are undoubtedly concerned with changing the field, but do so within historically bound parameters and continued recourse to the field's established cultural capital and doxa (Møller Hartley 2013).

Based on this knowledge, this thesis has sought to provide an in-depth case study of news production at two of the largest and most renowned digital native news organisations: *BuzzFeed* and *Vice*. The research has been based on interviews conducted with 24 journalists and a comparative quantitative analysis of news content published by *BuzzFeed*, *Vice*, and two legacy news providers: *The Guardian* and *BBC*. In very simple terms, this study has been interested in exploring what, if anything, makes news production at *BuzzFeed* and *Vice different* from traditional journalism. Three primary research questions have guided this enquiry:

 What are the major characteristics of news production at *BuzzFeed* and Vice? And how, if at all, do they differ from traditional journalism?

- 2) What do these findings tell us about the cultural, economic, social and symbolic capital of the journalistic field? And how, if at all, it might be shifting?
- 3) How do various endogenous and exogenous forces shape BuzzFeed and Vice's approach to journalism?

In answering these questions, this project has looked to provide a fresh perspective on how journalism, and the various activities involved in its production, might or might not be changing in the digital environment. The remainder of this chapter is divided into two sections. The first section provides an overview of my key research findings, while the second adopts a more macro-level perspective, considering the subject of journalistic autonomy and the role of normative theory in structuring how we think about the field and its core values.

9.1. Overview of research findings

In the ongoing struggle to define the "legitimate principles" of the field (Bourdieu 1985: 734), new agents can take up positions "aimed either at conserving or transforming the structure of relations of forces that is constitutive of the field" (Bourdieu, 2005, p. 30). A major finding of this thesis is that in regard to digital natives, this process is less dichotomous than one might first assume. While frequently lauded by scholars and industry commentators for bringing new forms of cultural capital to the field, this study has shown that digital natives emphasise historically contingent norms to: 1) mitigate against some of the risks posed by innovation 2) and, enhance their legitimacy, credibility, and status among other field members and the public at large. BuzzFeed and Vice bring a hybrid approach to news (Chadwick 2013) in that they are concerned with both conserving and transforming the established cultural capital of the field. This overarching conclusion is based on a detailed analysis of the structure and organisation of news production at *BuzzFeed* and *Vice*. By identifying the various forms of resources or "capital" intrinsic to their news operations, this research has been able to demonstrate precisely where both organisations might be challenging the traditional boundaries of journalism, and indeed, where the traditional cultural capital and doxa of the field is being preserved.

This dynamic was initially studied in the context of *BuzzFeed* and *Vice's* entry into the journalistic field. Notwithstanding vague rhetorical gestures toward making a "difference" or an "impact", it appeared that both organisations were primarily drawn to journalism because of its high symbolic status and potential to add "reputation and credibility to their brands" (Painter et al. 2017: 18). At the same time, there was a clear economic basis for entering the field, with young people – *BuzzFeed* and *Vice's* primary audience – representing a potentially lucrative and neglected demographic of consumers to target with news. As observed in Chapter 6, this dual pursuit of economic and cultural/symbolic capital seemed to correspond with an approach to journalism that strived for two different forms of legitimacy; one intellectual and the other economic (Champagne 2005). Dealing first with practices oriented around the intellectual pole, this was evident in BuzzFeed and Vice's focus on culturally sanctified practices like in-depth reporting and investigative journalism, as well as the hiring of several high-profile journalists, who brought important embodied cultural capital to their respective organisations. In principle, these strategies seemed oriented around the pursuit of symbolic capital, bolstering *BuzzFeed* and Vice's prospects of being recognised as legitimate by their fellow peers.

Yet as Carlson and Usher (2016: 7) note, "digital news startups must balance institutional mimicry with being explicitly innovative and forward-looking. It is not enough that they exist as entities within a competitive news market; they need to differentiate themselves from existing news". In this regard, *BuzzFeed* and *Vice* also seemed eminently concerned with introducing *new* forms of capital to the field, based more explicitly around public recognition, and/or the pursuit of economic goals. Evidence of this was found in the shift from traditional "beat" subjects to issues based more explicitly around reader-identified interests. This was supported by the hiring of a young, diverse, and technically-proficient staff, who seemed to occupy a homologous position to their own readers, which gave *BuzzFeed* and *Vice* am advantage in appearing "authentic" on certain issues of pertinence to their own readers. By virtue of
being "new" to the field, young journalists also seemed more willing to break the "rules" or doxa, embracing, for instance, more conversational and emotional forms of journalistic storytelling while rejecting more conventional, "objective" styles of news.

However, it was also young journalists' lack of socialization that led them to uncritically accept different methods and routines of newswork. This aspect was discussed at length in Chapter 7, which explored the effects of postindustrialism on the organisation of news work at BuzzFeed and Vice. In contrast to the "assembly line" model of work most commonly associated with mass media production (Gans 2004), news production at BuzzFeed and Vice was more decentralised, defined by values like flexibility, adaptability, fluidity, and collaboration. Under these conditions, journalists were expected to take more responsibility for their own work, "multiskilling" to engage in a variety of news processes and produce a range of news content across multiple platforms throughout the day. While these changes have created major cultural clashes inside traditional news organisations (Singer 2004; Deuze 2008; Paterson and Domingo 2008), for journalists at *BuzzFeed* and Vice, these new modes of work seemed "less a source of disruption than...a potential new source of identity" (Ferrucci and Vos 2017: 870). Indeed, there is a strong case to make that values like flexibility, adaptability, fluidity, and collaboration now constitute a legitimate part of the field's cultural capital (Siapera and Spyridou 2012), as expressed by numerous journalists from BuzzFeed and Vice who seemed to embrace these new modes of work, connecting them positively with a greater sense of individual freedom and autonomy. A second, related finding of this chapter was that reporters' autonomy at BuzzFeed and Vice seemed to also stem from their homologous relationship with audiences. Accordingly, news judgement was less of an individualised activity and instead, something that journalists felt increasingly comfortable sharing with the "quantified" audience (Anderson 2011b; Carlson 2018). This seemed more pronounced at BuzzFeed, whose interest in the metric of *sharing*, in particular, seemed to constitute an important and ever-evolving factor in news selection. At Vice, it was apparent that editors, rather than reporters, took more responsibility for managing audience metrics and they generally seemed more sceptical about their impact.

While these findings suggested that journalists were willing to cede some control to the audience, Chapter 8 reverted back to emphasising the importance of traditional journalistic norms in structuring the selection and presentation of news at *BuzzFeed* and *Vice*. From an ethical standpoint, objectivity, with its ethical connotations of accuracy, fairness, and independence, seemed to retain its status as a "sacred belief" among journalists at both organisations (Norderstreng, 1995: 115). Similarly, journalists expressed a strong desire to maintain gatekeeping control over content, which was supported, structurally, by a production process that only provided very limited means for users to participate in the news. These factors were also shown to contribute to the preservation of traditional sourcing techniques, with newer technologies being 'normalized' by journalists at *BuzzFeed* and *Vice* to fit with established routines (Singer 2005). Perhaps the strongest evidence of a shift in journalism's cultural capital was observed in journalists' discourse on transparency and emotionality as alternative epistemologies to objectivity. Here, journalists expressed a certain dissatisfaction with the relatively "closed process" of objectivity (Hellmueller et al. 2013), as well as criticising it for its tendency to create unnecessary distance between reporters and readers. Consequentially, journalists sought to develop a new authority relation with readers, premised on greater openness and accountability, and by presenting facts and information using more emotional, subjective, and personal narrative styles of storytelling. In practice, however, engagement with these new forms of discourse were uneven, with emotionality, in particular, generally being reserved for certain formats and news subjects, rather than routinized in every day news work. These findings suggest that claims about a shift in the epistemology of journalism are perhaps overstated, and only changing incrementally in response to digital media and its attendant culture.

9.2. Journalistic Autonomy or "What is Happening to the Field?"

To understand what is happening in the field, writes Bourdieu (2005: 43) "one has to understand the degree of autonomy of the field and, within the field, the degree of autonomy of the publication that a journalist writes for". Based on this

assertion, and in an effort to tie the various strands of this thesis together, I wish to reflect on what is happening in the field by considering the broad normative issue of journalistic autonomy. As noted in Chapter 2, "[a]utonomy is a core ideal of the standard model of professional journalism" (Waisbord 2013: 43). It is also *the* defining component of field theory, with Bourdieu valuing "most highly those forms of intellectual and artistic production sheltered from external economic or political pressures" (Benson 2008: 3). However, owing to journalism's status as a "very weakly autonomous field" (Bourdieu 2005: 33), it is important to recall that the field has always been "strongly dictated by the social, especially political and economic, conditions which it is organized" (Champagne 2005: 50). For this reason, journalism has always had to construct its own sense of autonomy against the external forces of heteronomy, with the respective power and influence of neighbouring fields rising and falling at different points in history (Schudson 2005; Vos 2016).

Certainly, since the passing of the "high-modernism" period of journalism in the 1960s (Hallin 1992), the field has become more homologous (or overlapping) with the economic field (Benson 1999). As a result, journalists' exclusivity or "jurisdiction" (Abbott 1988) over the news has diminished, contributing to an overall decline in the cultural capital of the field (Bourdieu 2005). Following the preceding period of "post-professional" journalism (Nadler 2016), there is a strong case to make that we stand amidst entering a "third wave" of techno-commercial development, "characterized by the development and rapid proliferation of low-cost, automated systems that can capture, link, and organize large amounts digital trace data that reflect non-purposive feedback from all consumers of digital media products" (Zamith 2018: 421).

These developments, combined with the lack of an adequate business model for digital news (Nielsen 2016), seem to be paving the way for a new set of "hybrid" news practices –premised on the new norm of *integration* – that "[combine] established editorial values with values such as collaboration, adaptation and business thinking". The growth and entrenchment of native advertising is perhaps the most visible example of this trend (Carlson 2015). This aside, general observations made about news production at *BuzzFeed* 219 and *Vice* would seem indicative of a shift toward the closer imbrication of editorial and business interests; or the cultural and economic capital of the field. This was evident in journalists' routines; in practices such as multiskilling and flexibility, and in their willingness to abdicate some of their own editorial autonomy to the "quantified" audience, which, at times, became the "seemingly more democratic...substitute for the internal standards" by which [journalists] selected the news (Bourdieu 1998: 73).

Taken together, there should be little doubt that these various facets of news work threaten journalistic autonomy and expertise. Perhaps the worst-case outcome of this situation, as Tandoc and Thomas (2015: 251) note, is a complete "drift away from journalism ethics, toward an audience-centered free-for-all governed by market logic". At the risk of sounding reductionist, this could leave journalism as "nothing but" the emanations of underlying economic forces (Schudson 2005: 214), contributing to a loss of jurisdiction that would risk causing irrevocable damage to the autonomy of the field and its wider institutional authority.

Leaving aside these more apocalyptic predictions, it is important to also consider that the cultural capital of the field might be *shifting* in response to these aforementioned pressures, rather than declining altogether (Anderson 2008). As Vos and Craft (2016) recognise, this can be challenging to parse with field theory, which understands journalism's cultural capital, autonomy, and legitimacy to all be mutually constitutive of one another. Paradoxically, new practices like emotionality and transparency represent "a form of cultural capital whereby the field willingly foregoes a measure of autonomy to gain legitimacy" (Vos and Craft 2016: 13). This helpfully forces us to consider the ways in which a less autonomous journalism (i.e. one more subservient to its readers), might in fact be a prelude to news playing a more relevant and useful role in people's lives.

To illustrate this point, it is helpful to think about *BuzzFeed* and *Vice's* coverage, which despite being strongly influenced by the logic of the market, did not focus on the usual gamut of subjects associated with heavily commercialised 220

media¹⁰⁷ (e.g. sex, scandal, celebrity, sports). Rather, both organisations chose to make news about progressive political issues – civil rights, gender equality, lesbian, gay, bisexual, and transgander rights (LGBT) – central to their news coverage. Naturally, this complicates arguments that market-driven news necessarily leads to a "race-to-the-bottom" in news coverage, or that marketoriented editorial strategies are inherently incompatible with a commitment to public service journalism (McManus 1994). Instead, various trends such as the ongoing fragmentation of audiences (Napoli 2012); the growing mobility and personalization of the media (Beckett and Deuze 2016); the rise in personal and emotional forms of discourse (Wahl-Jorgensen 2016); and the proliferation of hybrid spaces that blur "public and private, civic and consumption based, collective and personal narratives" (Papacharissi 2014: 25), have made it possible – and more crucially, profitable – to target audiences with content that might stimulate alternative forms of "sub-politics" (Beck 1997; Downey and Fenton 2003; Dahlgren 2005). In this regard, it is important to not foreclose the possibility of more heteronomous forms of journalism making a positive contribution to citizenship, despite them being ultimately beholden to the logic of the market. As one editor from *BuzzFeed* put it to me:

"[In terms of the subtler] goods that [journalism] can do...to the extent that a cohesive society relies on there being a degree of empathy with people who are not you and not like you, people in different situations to yourselves. To the extent that making an informed decision in electoral terms would ideally, involve considering more than just what is specifically better for you...to the extent that all of these things are the case, there is a low-level but very important good that journalism, along with other things like popular drama, can achieve in terms of...telling other people's stories, of letting people understand that lives are lived differently to theirs but with as much hope and fear and joy and validity as their own. Just gently nudging up the degree of empathy and awareness of people in society" (Interviewee L, editor, *BuzzFeed UK*).

¹⁰⁷ See Franklin (1997); Sparks (2000).

These specific contradictions underline the need to study the field *in context*, whether this be particular situations or specific social actors. Simply put, journalism is too heterogeneous, complex, and inconsistent to be discussed unitarily any longer, or using simple binary terms: hard versus soft, popular versus serious, emotion versus reason, professional or amateur, fact versus opinion, commercially or publically oriented, information versus entertainment. Rather, like Witschge et al. (2018: 6), I am advocating for an understanding of journalism as made up of both continuities and inconsistencies – the kind best appreciated by reflexive and empirically grounded research.

Appendix A: Sample email requesting newsroom access

Dear [insert name],

My name is Paul Stringer and I am a PhD student from the School of Media and Communication at the University of Leeds. I was getting in touch because I think you might be a good person to speak to regarding my research project.

My PhD is about how news journalism is changing because of technology, and I am interested in examining how news and current affairs stories are produced at [Vice/BuzzFeed]. The ultimate aim of my project is to contribute to clearer understandings of how leading net-native news organisations are challenging traditional understandings of journalism.

The research would ideally involve me spending some time inside the key news production locations of [Vice/BuzzFeed], speaking to journalists and other staff, and learning more about their day-to-day work. In the first instance, it would be great to see if you think this might be possible. I would welcome the opportunity to meet to discuss this research in more detail, and explain what I am hoping to accomplish.

Thank you for your time, and I'll look forward to hearing from you at your convenience.

All the best,

Paul Stringer

Appendix B: Interview guide

For the purposes of clarity, the following interviewee is understood to be an employee of BuzzFeed:

- Opening questions (to establish basic information about interviewee)
- 1. Let's start with your job...Could you tell me what your job title is, and describe to me what you do for BuzzFeed?
- 2. How long have you been working for the company?
- 3. And more generally, how long have you been working as an ____ (insert job title here)
- 4. Have you had any sort of formal training for your role? (university, diplomas etc.)
- Company questions (to establish some basic information about BuzzFeed)
- 5. Based on your experience, could you tell me about what it is you think that BuzzFeed do?
- 6. And if you can, could you describe BuzzFeed's approach to news? Are there any specific features of a BuzzFeed news story that distinguishes it from content produced by other news organizations?
- 7. How are decisions made regarding what news stories BuzzFeed cover? Are there certain topics / issues that tend to be deemed more important than others?
- Questions about routines and practices
- 8. What is your daily working routine? Can you describe it to me? What does a typical working day look like for you?
- Social Media
- 9. Could you provide a breakdown to me of how you use different social media platforms in your day-to day work? What purpose does each platform serve?

- 10. Is social media is an important tool for generating news stories? If yes, then could you describe to me how social media helps you decide what news stories to write? Is it a matter of looking at what's trending, where the prominent conversations are, etc.?
- 11. In what other ways do BuzzFeed generate news stories? Do you use wires / press releases / scheduled media events? Are any reporters assigned to particular beats?
- 12. Do you think it's necessary for BuzzFeed journalists to have a significant social media presence? If so, why?
- Questions about the audience
- 13. How would you describe an average reader of BuzzFeed news?
- 14. What topics or issues do you think your audience is most interested in reading about?
- 15. And what about journalists at BuzzFeed, what kinds of news stories do you think matter most to them?
- 16. Do you interact with your audience regularly in your work? If so, how?
- 17. Does audience feedback have any impact on the stories you write? Can you give me an example?
- Data / Analytics
- 19. Do you use data or analytics in your work? If so, how?
- 20. What kind of data matters most to you when trying to evaluate the success of a news story? Is it shares, number of minutes spent on article, or another kind of metric?
- Questions about journalism / news in general
- 21. What purpose do you think journalism serves in society?
- 22. Do you think the news BuzzFeed produces matches up to that ideal?
- 23. Do you think the role of journalists has changed over the last decade, or has it stayed the same?
- 24. Are you optimistic about the future of journalism?
- 25. Finally, is there anything else you'd like to add or share with me?

Appendix C: Sample email requesting interview

Dear [insert name],

I hope you're well. My name is Paul Stringer and I am a PhD student from the School of Media and Communication at the University of Leeds. I was getting in touch because I think you might be a good person to speak to about my research project.

My PhD is broadly about how journalism is changing because of technology, and the main focus of the project is on how net-native news organisations like BuzzFeed are challenging traditional understandings of journalism.

As part of the research, I am looking to interview several members of the [Vice/BuzzFeed], team. If it can be arranged, I would love the opportunity to be able to meet and speak with you about your role as a senior reporter for BuzzFeed. I wouldn't need much of your time and your contribution would be enormously valuable to my research.

I'm based in London at the moment, and can be very flexible in terms of organising times / dates to meet. Alternatively, if it's easier to talk via email, or over the phone, then that's no problem.

Thank you for your time and all the best,

Paul Stringer

Appendix D: Participant Information Sheet and Consent Form

Research Project: Making News at Digital-Natives [insert date]

Project Information for Participants

- You are being invited to participate in a research project about how news production and journalism is changing because of technology.
- Please read the information in this document carefully to understand why the research is being done and what being a participant on this research project might involve for you.
- If you have any questions about your own participation, or this project in general, then please get in touch with me using the contact details provided in this document.

What is this project about?

In journalism studies, a great deal has been written about the impact of the Internet on news production. However, most academic research has focused on how news production has changed at traditional news organizations like *The New York Times* and *The Guardian*.

In contrast, relatively little remains known about how news and current affairs are produced at digital native organizations like [*BuzzFeed/Vice*]. Understanding how news is made at such organizations is critical if we want to gain a better understanding of how online journalism is evolving and how the day-to-day work of journalists is changing.

This research will involve me interviewing you about your work, asking you about different aspects of journalism and the news production process at [*BuzzFeed*/*Vice*].

Why have I been chosen?

You have been chosen to take part in this project because you are an employee of [*BuzzFeed*/*Vice*] and contribute to the news production process.

What do I have to do?

If you are interviewed for this project, then I will require between 30 minutes and an hour of your time. During the interviews I will be asking questions about your day-to-day work, and also more general questions about your role as a journalist (if you are one) and what you think that means. You are free to refuse to answer any interview questions that you do not feel comfortable answering.

What are the possible disadvantages and risks of taking part?

I will require a small amount of your time if you are being interviewed for this research project. In terms of fixing a date and a time for this to take place, I will always try and be as accommodating as possible to limit any disruption this might cause.

What are the possible benefits of taking part?

Whilst there are no immediate benefits for those participating in this research project, by taking part you will be helping to advance academic and public knowledge about journalism and the changing nature of news production.

Do I have to take part?

No. It is completely up to you to decide whether or not you wish to take part in this research project. You do not have to give a reason to withdraw from this research project.

Will my taking part in this project be kept confidential?

Personal information (your full name, email address, and job title) that is gathered as part of this project will be stored securely and only be accessible to the project research team.

In line with previous research, I will not identify by name any people unless I have their specific permission to do so. For participants who are anonymised in my research, one risk worth acknowledging is that you might still be identified based on your job title, particularly if you are a more senior employee.

What will happen to the results of the research project?

The results of this research will be published in academic journals, which will be freely available on-line. You can request a copy of the published results from the researcher (Paul Stringer). The data collected during the course of the project might be used for additional or subsequent research; however, you will not be identified by name in any report or publication unless you have given your specific permission for this. This research data will be kept for up to three years after the original findings of the research are published.

Who is organising / funding this research?

The University of Leeds is organising this research. The Economic and Social Research Council (ESRC) of England and The University of Leeds fund this research.

Contact for further information

Thank you for taking the time to read through this information. If you have any questions or would like any further information you can contact me using either the email or telephone number you have been provided.

Thank you for taking time to read this sheet and for considering taking part in the study

Consent Form for Research Project: "Making News at Digital Natives"	Add your initials next to the statement if you agree
I confirm that I have read and understand the information sheet dated 28/04/16 explaining the above research project and I have had the opportunity to ask questions about the project.	
I understand that my participation is voluntary and that I am free to withdraw at any time during the data gathering stage of the research, without providing any reason and without there being any negative consequences. In addition, should I not wish to answer any particular question or questions, I am free to decline. If I do wish to withdraw my data I will contact Paul Stringer either by email or telephone.	
I give permission for the researcher (Paul Stringer) and his project supervisors to have access to my responses. I understand that my name will not be linked with the research data unless I have specified that this be allowed. I accept the risk that I am potentially identifiable in the report or reports that result from the research, based on my name and/or job title. I understand that my responses will be kept strictly confidential.	
I agree for the data collected from me to be stored and used in relevant future research in an anonymised form, up to three years after the original findings of the study are published.	
I understand that other genuine researchers will have access to this data only if they agree to preserve the confidentiality of the information as requested in this form.	
I understand that other genuine researchers may use my words in publications, reports, web pages, and other research outputs, only if they agree to preserve the confidentiality of the information as requested in this form.	
I agree to take part in the above research project and will inform the lead researcher should my contact details change.	

Name of participant	
Participant's signature	
Date	
Name of researcher	

Signature	
Date	

Appendix E: Interviewee List

Respondent	Title	Length	Organisation
А	Reporter	1:00: 10	BuzzFeed
В	Reporter	54:53	BuzzFeed
С	Reporter	1:04:32	BuzzFeed
D	Editor	18:09	BuzzFeed
E	Editor	31:11	BuzzFeed
F	Editor	34:02	BuzzFeed
G	Reporter	1:05:15	BuzzFeed
Н	Reporter	46:28	BuzzFeed
1	Reporter	1:28:32	BuzzFeed
J	Reporter	36:59	BuzzFeed
К	Editor	57:53	BuzzFeed
L	Editor	1:38:32	BuzzFeed
М	Reporter	44:45	BuzzFeed
N	Reporter	38.53	BuzzFeed
0	Reporter	1: 13: 24	BuzzFeed / Freelance
А	Reporter	47:05	VICE
В	Reporter	1:04:22	VICE
С	Reporter	38:39	VICE
D	Reporter	41:53	VICE
E	Editor	1:41:39	VICE
F	Editor	1:22:24	VICE
G	Reporter	47:53	VICE / Freelance
Н	Reporter	Typed	VICE EU
J	Editor	70:51	VICE

Appendix F: Coding Scheme

Variable	Sub-category	Description
Basic Article		
information		
News outlet	1- BuzzFeed UK	Indicate which website the article was
	2- BuzzFeed US	published on.
	3- Vice UK	
	4- Vice US	
	5- Guardian UK	
	6- Guardian US	
	7- BBC UK	
	8- BBC US	
Headline		Please write the headline of the article.
Date		Please write the date of publication of the article.
Author gender	1- Male	Where possible, indicate the author's
	2- Female	gender. If this is not possible, use the code labelled 'can't tell".
	3- Can't tell	
Author ethnicity	1- White	Where possible, indicate the author's
	2- Asian or Indian	ethnicity. If this is not possible, use the code labelled 'can't tell'.
	3- Black	
	4- Hispanic	
	5- Can't tell	
Word Count		Record the word count of the article
		(excluding any text that appears
		outside the <i>main body text</i> of the article)
URL		Record the website URL of the article
Advanced article		
information		
Торіс	1- Government and politics	Code the main topic of the article from
	2- Social issues, rights,	the adjacent list. If you are unsure, please use the code 'other/can't tell'.
	and protests	
	3- Crime and terrorism	
	4- Science, education, and	
	technology	
	E 11. 10	
	5- Health	

	7- Entertainment (including	
	celebrity news and gossip)	
	8- Lifestyle (trends,	
	fashion and travel)	
	9- Accidents and disasters	
	10- Energy and	
	environment	
	11- Other / can't tell	
Format	1- Report	Code the article according to its
	2- Feature or investigation	format. If you are unsure, please use the code 'other/can't tell'.
	3- Interview	
	4- Comment or opinion	
	5- Review	
	6- Listicle	
	7- Quiz	
	8- Game	
	9- Aggregated / curated	
	content	
	10- Live blog	
	11- Other / can't tell	
Storytelling style	1- Mainly descriptive	Code the article either as mainly
	2- Mainly interpretive	<i>descriptive</i> or <i>mainly interpretive</i> . If you encounter stories where neither
	3- Ambiguous /can't tell	variable seems to fit, please code
Main source name		these as 'ambiguous/can't tell'. Indicate the name (if a human source)
		of the main source of the article. If it is
		a non-human source like a leaked document, then write down N/A.
Main source		If the source is affiliated with a
organisation		particular institution or organisation, please record its name. Otherwise, put
	4 0 40	N/A.
Main source	1-0-10	Indicate the age of the source (note: you may have to manually search for
age	2- 11-20	their age on the internet if this is not
	3-21-30	made obvious in the story). If this is not possible, use the code labelled 'can't
	4-31-40	tell'. If you cannot identify a main
	5-41-50	source for the news story, then use the label 'not applicable'.
	6-51-60	
	7- 61-70 8- 70+	
	9- Can't tell	
	9- Can t tell	
	10 Not applicable	
Main Source gender	10-Not applicable	Indicate the gender of the source

	2- Female	search for their age on the internet if
	3- Can't tell	this is not made obvious in the story). If
		this is not possible, use the code
	4- Not applicable	labelled 'can't tell'. If you cannot identify a main source for the news
		story, then use the label 'not
Main Source	1- White	applicable'. Indicate the ethnicity of the source
ethnicity	2- Asian or Indian	(note: you may have to manually
etimoty		search for their age on the internet if
	3- Black	this is not made obvious in the story). If this is not possible, use the code
	4- Hispanic	labelled 'can't tell'. If you cannot
	5- Can't tell	identify a main source for the news story, then use the label 'not
	6- Not applicable	applicable'.
Main source	1- Government officials,	Indicate the type of source. If this is
type	politicians, or legal	difficult to determine, please use the label 'other / can't tell'. If you cannot
	professionals	identify a main source for the news
	2- Police, law	story, then use the label 'not applicable'.
	enforcement, or military	
	3- Business people and	
	organisations	
	4- Civil society members	
	and organisations (NGOs,	
	trade unions, charities,	
	etc.)	
	5- Own media (where the	
	news organization	
	references their own	
	content)	
	6- Traditional media	
	7- Other media	
	8- Celebrities and other	
	famous people (artists,	
	musicians, etc.)	
	9- News agencies/wires	
	10- Analysts or experts	
	(including academics)	
	11- Ordinary citizens	
	12- Other/can't tell	
	13- Not applicable	
Main Source	1- Exclusive	Indicate if the main source is being
exclusivity	2- Indirect / second-hand	quoted exclusively by the journalist in question, or have his/her/its quotes
	3- Ambiguous/can't tell	have been obtained indirectly, or
		second-hand from another source. You

	4- Not applicable	may have to use the internet and Google the text from individual quotes to accurately determine whether they are exclusive or indirect/second-hand. If this is too difficult to determine, please use the label 'ambiguous / can't tell'. If you cannot identify a main source for the news story, then use the label 'not applicable'.
Main Source	1- Yes	Indicate if the main source being
social media	2- No	quoted from social media (e.g. <i>Twitter, Instagram, YouTube, Facebook,</i>
	3- Can't tell	Linkedin). Use the code 'can't tell'
	4- Not applicable	when this is not clear. If you cannot identify a main source for the news story, then use the label 'not applicable'.
Source 1 name		Indicate the name (if a human source) of the first quoted source of the article. If it is a non-human source like a leaked document, then write down N/A.
Source 1 organisation		If the source is affiliated with a particular institution or organisation, please record its name. Otherwise, put N/A.
Source 1	1- 0-10	Indicate the age of the source. If the
age	2- 11-20	source is not a person (e.g. a document), then please use the code:
	3- 21-30	'not applicable'. If it is not possible to
	4- 31-40	identify the age of the source, then use the code: 'can't tell'.
	5- 41-50	
	6- 51-60	
	7- 61-70	
	8- 70+	
	9- Can't tell	
	10- Not applicable	
Source 1 gender	1- Male	Indicate the gender of the source. If
	2- Female	the source is not a person (e.g. a document), then please use the code:
	3- Can't tell	'not applicable'. If it is not possible to identify the gender of the source, then use the code: 'can't tell'.
Source 1 ethnicity	1- White	Indicate the ethnicity of the source. If
	2- Asian or Indian	the source is not a person (e.g. a document), then please use the code:
	3- Black	'not applicable'. If it is not possible to
	4- Hispanic	identify the ethnicity of the source, then use the code: 'can't tell'.
	5- Can't tell	
Source 1 type	1- Government officials,	Indicate the type of source. If this is
	politicians, or legal	difficult to determine, please use the label 'other / can't tell'.
	professionals	

	2- Police or military	
	3- Business people and	
	organisations	
	4- Civil society members	
	and organisations	
	(Campaigners, NGOs,	
	Charities, etc.)	
	5- Own media (where the	
	news organization	
	references their own	
	content)	
	6- Traditional media	
	7- Other media	
	8- Celebrities / famous	
	people	
	9- News agencies/wires	
	10- Analysts or experts	
	(including academics)	
	11- Ordinary citizens	
	12- Other /can't tell	
Source 1 exclusivity	1- Exclusive	Indicate if the source is being quoted
	2- Indirect / second-hand	exclusively by the journalist in
	3- Ambiguous / can't tell	question, or have his/her/its quotes have been obtained indirectly, or
	Ŭ	second-hand from another source. You
		may have to use the internet and Google the text from individual quotes
		to accurately determine whether they
		are exclusive or indirect/second-hand. If this is too difficult to determine,
		please use the label 'ambiguous / can't
Source 1	1- Yes	tell'. Indicate if the source being quoted
social media	2- No	from social media (e.g. Twitter,
	3- Can't tell	Instagram, YouTube, Facebook, Linkedin). Use the code 'can't tell'
		when this is not clear.
Source 2, 3, 4, 5, 6,		NOTE: Please repeat the above
7, 8, 9, 10, etc.		coding procedure for <i>every</i> quoted source that appears in the news
		article.
Total number of		Write down the total number of quoted sources that were used in the news
sources		article.
Total number of		Write down the total number of sources from social media that were
social media		used in the news article.
sources		

Objectivity and		
Emotionality		
Headline Language	 Straight / factual Emotional Ambiguous / can't tell 	Indicate if the article uses a straight/factual headline or an emotional headline. If you cannot determine the headline, please use the eade 'ambiguous(can't tall'
Storytelling Lead	 Inverted pyramid Narrative / anecdotal lead Other / can't tell Yes 	code 'ambiguous/can't tell'. Indicate if the lead paragraph of the story uses an inverted pyramid structure or a narrative / anecdotal lead. If you cannot determine the lead paragraph, please use the code 'other/can't tell'. Indicate if the news article uses
Personalized storytelling	2- No 3- Ambiguous / can't tell	personalized storytelling. If you are unsure, use the code 'ambiguous/can't tell'.
Disclosure		
Transparency		
Time stamps	 Date of publication only Time of publication only Date and time of publication No timestamp 	Indicate the level of information provided about the date and/or time of publication of the news article.
Authorial information	 Author's name Author's name and contact details (email and / or social media links) No author information 	Please indicate how much information is provided about the author of the news article.
Number of hyperlinks		Indicate the total number of hyperlinks used in the news article.
Number of internal hyperlinks		Indicate the total number of internal hyperlinks used in the news article.
Number of external hyperlinks		Indicate the total number of external hyperlinks used in the news article.
Participatory		
Transparency		
Participation index (stories can contain zero, or 1 or more of the following values)	 1- Sharing 2- Commenting 3- Poll 4- Quiz 5- Submit your own content 	 Indicate how many of the following features are contained within the news article: 1) Sharing: readers are encouraged to share the news item (via social media or email) 2) Commenting: readers are invited to comment on the news item 3) Poll: readers are invited to submit

 6- Reader contribution published in text 7- Reader news (news item presented as wholly being written by a reader 	 an answer to a poll 4) Quiz: readers are invited to answer questions in a quiz 5) Submit your own content: readers are encouraged to send in their own content, including reports, images, and videos, to contribute to a news story 6) Reader contribute to a news story 6) Reader contribution is published: where readers have been previously asked to send in their own content, these contributions now explicitly form part of the news story. NOTE: this is different to journalists simply using ordinary people as sources, such as tweets compiled from Twitter, or comments taken from Facebook. 7) Reader news: news item is presented as being wholly written by a reader

Appendix G: Codebook

Codebook: Making News at Digital Natives

Analysing News Content Produced by BuzzFeed, Vice, The Guardian and BBC

1. Overview

1.1 Introduction

This quantitative content analysis is part a larger project examining news production at digital-native news organizations.

The codebook provides instructions for how to code online news articles published by *BuzzFeed, Vice, The Guardian* and *BBC*. Each article will be coded according to a set of predefined variables, which are described in more detail below. An accompanying coding sheet is also provided for coders to input the results of the analysis. If the coder has any questions or encounters any issues during the coding process, they should consult the lead researcher of the project (Paul Stringer) who will aim to answer any queries or questions.

1.2. Sample

The sample comes from news articles collected from the UK and US websites of the four aforementioned news organizations:

http://www.bbc.co.uk/news (UK site) http://www.bbc.com/news (US site) http://www.buzzfeed.com/news (UK site) https://www.buzzfeed.com/news/?country=en-us (US site) https://www.buzzfeed.com/news/?country=en-us (US site) https://www.buzzfeed.com/news/?country=en-us (US site) https://www.buzzfeed.com/news/?country=en-us (US site) https://www.theguardian.com/uk (UK site) https://www.theguardian.com/us?INTCMP=CE_US (US site) https://www.vice.com/en_uk/news (UK site)

To help narrow down the sample, the top 10 most prominent articles that featured on the pages of each of these websites were collected at the same time each day, over two different week-long periods. The number of total articles gathered was 800. To help preserve the news content in the form it was captured, each article was downloaded using *NCapture* (part of the *NVivo* software package) and also using a third-party web browser application called *'Full Page Screenshot Capture'*, which saved articles in the PDF file format.

1.3. Unit of Analysis

The unit of analysis for this research project was defined as a single news article.

2. Coding Procedure and Variables

What follows is a detailed guide for coding online news articles from *BuzzFeed*, *Vice*, *The Guardian* and *BBC*. While some variables should be relatively self-explanatory, other variables are supplanted with a description to help provide more clarity. If anything remains unclear please speak to Paul Stringer, the project lead.

2.1. Basic Article Information

To begin the procedure, you will code basic information about the news article using the following variables.

Variable 1: News Outlet -On what website was the article published?

1- BuzzFeed (UK)	5- Guardian (UK)
2- BuzzFeed (US)	6- Guardian (US)
3- VICE (UK)	7- BBC (UK)
4- VICE (US)	8- BBC (US)

Variable 2: Headline

-What is the headline of the news article? Please write this down in full.

Variable 3: Date

-What date was the news article published? Please write this down in a consistent format.

Variable 4: Author Gender

-What is the gender of the author(s) who has/have published the news article? If there is more than one author, please indicate their gender(s) too. In any instance where you cannot tell the gender of the author, please code this as "can't tell".

1- Male 2- Female 3- Can't tell

Variable 5: Author Ethnicity

-What is the ethnicity of the author(s) who has/have published the news article? If there is more than one author, please indicate their ethnicity too. In any instance where you cannot tell the ethnicity of the author, please code this as "can't tell".

1- White 2- Asian or Indian 3- Black 4- Hispanic 5- Can't Tell

Variable 6: Word Count

-What is the total word count of the article? Please count the body text only; not headlines, sub-headlines, captions, or embedded content from social media (e.g. Tweets and Facebook posts).

Variable 7: URL

-Please note the URL address of the article.

2.2. Advanced Article Information

Having coded some basic information, you will move on to coding some more specific information about the news article using the following variables.

Variable 8: Topic of News story

-Please record the main topic of the news story. If you are unsure of the main topic, or you feel that the article gives equal weight to multiple topics, then use the code "other / can't tell".

- 1- Government and Politics
- 2- Social Issues, rights, and protests
- 3- Crime and terrorism
- 4- Science, education, and technology
- 5- Health
- 6- Business and economy
- 7- Entertainment (including celebrity news and gossip)
- 8- Lifestyle (trends, fashion, and travel)
- 9- Accidents and disasters
- 10- Energy, nature, and environment (including weather)
- 11- Other / Can't Tell

Variable 9: Format of News Story

-Please record the format of the news story. If you are unsure of the format, please use the category "other / can't tell".

1- Report	2- Feature / investigation	3- Intervi	ew 4-
Comment/Opi	nion		
5- Review	6- Listicle	7-Quiz	8- Game
9- Aggregated	I / curated content	10- Live blog	11- Other /
can't tell			

Note: sometimes news formats such as reports or features might *contain* quizzes or games, but these should still be coded as reports or features.

Variable 10: Storytelling Style

-This variable is interested in whether the main body text of the news story is primarily descriptive or interpretive. Stories that are *mainly descriptive* are primarily concerned with describing *what* has happened in a rather straightforward and uncomplicated style. On the other hand, stories that are *mainly interpretive* will typically involve the writer trying to analyse, evaluate or explain a situation while also describing it. On some occasions, it might be difficult to determine whether a story is mainly descriptive or mainly interpretive. When this occurs, the value "other / can't tell" can be used.

The coding procedure for sources is more complex. For this reason, an explanation of what constitutes a news 'source' is provided, before the coding procedure for sources is described in more detail.

2.3.1. Introduction

In the broadest sense, a source is an individual, organization, or document, that provides a journalist with information for a news story. Traditionally, journalists have gravitated towards 'elite' sources like lawmakers, politicians, and business leaders, when constructing news stories. Sources are often represented in quoted speech in news stories, for example:

R.C. Hammond, a senior State Department official, said such a dialogue would be out of the question as long as North Korea continues its pattern of "test launches, over flights," and "threats of detonations over the Pacific."

However, on other occasions, a source's presence in a news story may be less obvious. For example, in the passage below, two sources (Tim Selatey and 'Business Filings') are mentioned but neither are attributed speech in quotations:

A donate button was only added recently to the Citizens for Trump website to help offset outof-pocket expenses, Selaty said. The button links to a PayPal page for Patriotic Strategies, which, according to business filings in Texas, incorporated as an LLC last March, listing Earl and James Lee Brown as officers. Selaty said he would be surprised if the group has raised

In rare cases, quoted speech might also be used in a news story that is unattributable to a particular source:

The constitutional monarch had been treated by doctors for "<u>water on</u> the brain" and a chest infection in August 2015. On Sunday, the palace

For the purposes of clarity, this project defines a source as a provider of *attributed* textual information (e.g. the president *said*, it was *reported*, the officer *believed*), that appears in *quotation* marks in a news story (any attributed text that is *not* quoted should *not* be recorded). The only exception to this is textual content from social media, for example, Facebook posts and tweets, which can often appear embedded in news stories. For example:

Papadopoulos, who was earlier this month sentenced to 14 days in prison for lying to the FBI, has turned his attention to publicly bashing Downer. He has posted nine tweets about Downer in 12 days, accusing the former high commissioner of spying and secretly recording their conversation.

George Papadopoulos @GeorgePapa19	y
The notion that Downer randomly reached out to me just to h a gin and tonic is laughable. Some organization or entity ser him to meet me. For the sake of our republic and the integrit this investigation, I think it's time Downer is as exposed as Christoper Steele.	nt
2:56 AM - Sep 11, 2018	
\bigcirc 14.3K \bigcirc 8,701 people are talking about this	0

Papadopoulos said in another tweet: "I think it's time Downer is as exposed as Christopher Steele," referring to the former British intelligence officer behind the Trump dossier. He has also tweeted at the heads of the US Senate Intelligence Committee that he's willing to testify about Downer.

Providing the embedded social media content contains a textual element, it can be counted as a source. The source should be noted as the social media account from which the content was posted.

2.3.2. Main Source Variables

Who or what is the *main source* of the news story. This could be an individual, an organization, or a document such as a press release, a report, or a statement. While often the case, the main source of the story is not necessarily the source that is quoted *the most* in a news article. In very rare instances, you might even encounter an article that includes no quotes from a main source. In this regard, it is helpful to read the article headline and news story in its entirety, to understand the full context of the story before making a judgement about the main source.

Variable 11: Main source name (if a person)

-What is the name of the source? Note: if the main source is a document or organization then write N/A.

Variable 12: Main source organization

-If the source is an individual working for a particular organization, then please write down the name of the organization. Sometimes stories will also be based on an organizational document such as a press release or statement. If this is the case please also record the name of the organization that has published the document.

Variable 13: Main source age (if a person)

-Please record the age of the source. Often, this will not be obvious and so you might have to use the internet (e.g. Googling a person's name) to help determine the age of a source. If the main source is not a person (e.g. a document), then please use the code: "not applicable". If it is not possible to identify the age of the main source, then use the code: "can't tell".

1- 0-10 (years-old)	4- 31-40	7- 61-70
2- 11-20	5- 41-50	8-71+
3- 21-30	6- 51-60	9- Can't tell
10-Not applicable		

Variable 14: Main source gender (if a person)

-Please record the gender of the source. If the main source is not a person (e.g. a document), then please use the code: "not applicable". If it is not possible to identify the gender of the main source, then use the code: "can't tell".

1- Male	3- Can't tell
2- Female	4- Not applicable

Variable 15: Main source ethnicity (if a person)

-Please record the ethnicity of the source. Often, this will not be obvious and so you might have to use the internet (e.g. Googling a person's name) to help determine the ethnicity of a source. If the main source is not a person (e.g. a document), then please use the code: "not applicable". If it is not possible to identify the ethnicity of the main source, then use the code: "can't tell".

1- White	3- Black	5- Can't tell
2- Asian or Indian	4- Hispanic	6- Not applicable

Variable 16: Main source type

-Please record the *type* of main source. If this is difficult to determine, please use the label "other / can't tell". If you cannot identify a main source for the news story, then use the label "not applicable".

1-Government officials, politicians, or legal professionals

2-Police, law enforcement, or military

3-Business people and organisations (note: this includes spokespeople who act in their capacity as representatives of businesses. This does not include ordinary people working for businesses unless they are claiming to speak on behalf of their organisation)

4-Civil society members and organisations (these are individuals or organisations that are distinct from government and business, for example: charities, trade unions, co-operatives, and foundations)

5-Own media (if the main source is from the organisation publishing the story) 6-Traditional media (traditional media refers to established media organisations, often which started out making news in other mediums such as newspapers, magazines, radio, and television)

7-Other media (other media refers to newer media organisations that produce news and other content for a predominantly online audience, for example: *BuzzFeed, Slate, Huffington Post, BuzzFeed, Vox, The Intercept*) 8-Celebrities and other famous people (artists, musicians, etc.)

9-News agencies and wires (e.g. *Reuters, Associated Press, Agence France-Presse, Bloomberg*)

10-Analysts or experts (including academics)

11-Ordinary citizens ("ordinary" people refers to individuals not affiliated to any particular organisation or institution. This can include eyewitnesses, protestors and activists, individual workers, and volunteers)

12-Other / can't tell

13-Not applicable

Variable 17: Main source exclusivity

-Please code whether the main source is exclusive or has been obtained indirect / second-hand by the journalist. This can sometimes be difficult to determine and might require you to Google quotes from a source to see where they appear and how they are attributed in other contexts. Sometimes, news organizations will make it clear when they have obtained exclusive information. Either because the story will be labelled as an 'exclusive', or quotes will use words like "Speaking to the *Guardian*, Dr Rogers said...", or "Miss Smith told *BuzzFeed News* that she..." Most often, however, you are likely to find that information from a source has been received indirectly or second-hand by the journalist – from scheduled events like press briefings or via reports from other media, to give two examples. If, after conducting your own research, you are still unsure whether a source is exclusive or indirect / second-hand, please use the code "ambiguous / can't tell". If you cannot identify a main source for the news story, then use the label "not applicable".

1-Exclusive 2-Indirect / Second-hand 3-Ambigious / can't tell 4-Not applicable

Variable 18: Main source social media

-Is the main source from social media? Sometimes this can take the form of embedded social media content, other times, a journalist may reference a quote came directly from social media (e.g. the President *tweeted*). If you are unsure, please use the label "can't tell". If you cannot identify a main source for the news story, then use the label "not applicable".

1-Yes 2-No 3-Can't tell 4-Not applicable

2.3.3. Story Source(s) Variables

You should record all sources used by the journalist in the news story, that is, any individual, organization or document that is quoted by the journalist using speech marks, or embedded text from social media. This procedure is largely the same as the process for coding the main source of the news article.

Variable 19: Source name (if a person)

-What is the name of the source? Note: if the source is a document or organization then write N/A.

Variable 20: Source organization

-If the source is an individual working for a particular organization, then please write down the name of the organization. Sometimes stories will also be based on an organizational document such as a press release or statement. If this is the case please also record the name of the organization that has published the document.

Variable 21: Source age (if a person)

-Please record the age of the source. Often, this will not be obvious and so you might have to use the internet (e.g. Googling a person's name) to help determine the age of a source. If the source is not a person (e.g. a document), then please use the code: "not applicable". If it is not possible to identify the age of the source, then use the code: "can't tell".

1- 0-10 (years-old)	4- 31-40	7- 61-70
2- 11-20	5- 41-50	8-71+
3- 21-30	6- 51-60	9- Can't tell
10- Not applicable		

Variable 22: Source gender (if a person)

-Please record the gender of the source. If the source is not a person (e.g. a document), then please use the code: "not applicable". If it is not possible to identify the gender of the source, then use the code: "can't tell".

1- Male	3- Can't tell
2- Female	4- Not applicable

Variable 23: Source ethnicity (if a person)

-Please record the ethnicity of the source. Often, this will not be obvious and so you might have to use the internet (e.g. Googling a person's name) to help determine the ethnicity of a source. If the source is not a person (e.g. a document), then please use the code: "not applicable". If it is not possible to identify the ethnicity of the source, then use the code: "can't tell".

1- White	3- Black	5- Can't tell
2- Asian or Indian	4- Hispanic	6- Not applicable

Variable 24: Source type

-Please record the *type* of source. If this is difficult to determine, please use the label "other / can't tell".

1-Government officials, politicians, or legal professionals

2-Police, law enforcement, or military

3-Business people and organisations (note: this includes spokespeople who act in their capacity as representatives of businesses. This does not include ordinary people working for businesses unless they are claiming to speak on behalf of their organisation)

4-Civil society members and organisations (these are individuals or organisations that are distinct from government and business, for example: charities, trade unions, co-operatives, and foundations)

5-Own media (if the source is from the organisation publishing the story)

6-Traditional media (traditional media refers to established media organisations, often which started out making news in other mediums such as newspapers, magazines, radio, and television)

7-Other media (other media refers to newer media organisations that produce news and other content for a predominantly online audience, for example: *BuzzFeed, Slate, Huffington Post, BuzzFeed, Vox, The Intercept*)

8-Celebrities and other famous people (artists, musicians, etc.)

9-News agencies and wires (e.g. *Reuters, Associated Press, Agence France-Presse, Bloomberg*)

10-Analysts or experts (including academics)

11-Ordinary citizens ("ordinary" people refers to individuals not affiliated to any particular organisation or institution. This can include eyewitnesses, protestors and activists, individual workers, and volunteers)

12-Other / can't tell

Variable 25: Source exclusivity

-Please code whether the source is exclusive or has been obtained indirect / second-hand by the journalist. This can sometimes be difficult to determine and might require you to Google quotes from a source to see where they appear and how they are attributed in other contexts. Sometimes, news organizations will make it clear when they have obtained exclusive information. Either because the story will be labelled as an 'exclusive', or quotes will use words like "Speaking to the *Guardian*, Dr Rogers said...", or "Miss Smith told *BuzzFeed News* that she..." Most often, however, you are likely to find that information from a source has been received indirectly or second-hand by the journalist – from scheduled events like press briefings or via reports from other media, to give two examples. If, after conducting your own research, you are still unsure whether a source is exclusive or indirect / second-hand, please use the code "ambiguous / can't tell".

1-Exclusive 2-Indirect / Second-hand 3-Ambigious / can't tell

Variable 26: Source social media

-Is the source from social media? Sometimes this can take the form of embedded social media content, other times, a journalist may reference a quote that came directly from social media (e.g. the President *tweeted*). If you are unsure, please use the label "can't tell".

1- Yes 2- No 3-Can't tell

Please repeat this coding procedure for every quoted source or embedded piece of text-based social media content that appears within the body text of an article.

Variable 27: Number of sources

Please record the total number of quoted sources that are used in the news article.

Variable 28: Number of social sources

Please record the total number of sources from social media that are used in the news article.

2.4. Objectivity and emotionality

-Three variables have been developed to analyse whether news articles are more 'objective' or 'emotional' in character.

Variable 29: Headline Language

-This variable is designed to assess whether the news article uses emotional or straight language in its headline. Emotional language is defined as language that includes one or more of the following features:

Forward referencing: headlines that use forward referencing use words that imply the article must be read in full in order to gain comprehension of a particular subject. They often build in a *promise* designed to create curiosity in the reader, and typically contain words like 'how', 'why', and 'this' that suggest the article will explain to the reader a particular topic or issue. Examples include: "*This* is how much plastic humans have made" (*BuzzFeed*); "People on Twitter are freaking out about *this* life-changing corn hack" (*BuzzFeed*); "Here's how legal experts say the Grenfell inquiry can avoid becoming a whitewash" (*BuzzFeed*); "*Why* the mentally ill keep getting shot by the cops" (*Vice*); "*What* Wells Fargo knew" (*Vice*); "*These* women feel excluded from the strike" (*Vice*); "The second presidential debate, decoded" (*Vice*)

Personalization: headlines that address the reader directly, typically using the word 'you', or variations of this. For example: "Jeff Sessions just made it easier for the cops to take *your* stuff" (*VICE*); "Flights are going to cost *you* more because of climate change" (*VICE*); "Google wants *you* to spend a lot more time in its app" (*BuzzFeed*)

Special punctuation and expressives: headlines that contain special punctuation like question marks and exclamation marks. For example: "Is microdosing the future of marijuana? (*Vice*). Or headlines that employ word games, alliteration, idioms, proverbs, slang / swear words and popular expressions. For example: "Money talks" (*BuzzFeed*), "Science of the Lambs" (*BuzzFeed*); "The latest 'Keeping up with the Kardashian's' Trailer is intense as fuck" (*BuzzFeed*); "So, you've decided to give a shit about climate change" (*VICE*).

Conversely, headlines that use straight language typically use language that reflects the content of the news article without trying to induce curiosity in the reader. They are often characterised by the use of straight, matter-of-fact language. For example: "UK and EU at odds on Brexit bill" (*BBC*); "Grandparents should leave homes and savings to millennials, says MP" (*VICE*); "Thailand's King dies at age 88" (*BuzzFeed*); "Baby killed and child injured in Colchester dog attack" (*BuzzFeed*).

Following this guidance, headlines should be categorized as either 'straight' or 'emotional'. In instances where it is difficult to tell, please use the code 'can't tell'.

1- Straight 2- Emotional 3-Can't tell

Variable 30: Storytelling Lead

-The lead paragraph of the news article is defined as the first paragraph of text that forms the main body of the article (not the headline or sub-headline). This variable aims to determine whether the lead paragraph uses the traditional 'inverted pyramid' style of news reporting, or the story begins with a 'narrative' or 'anecdotal' lead. The inverted pyramid style aims to convey the most important information in the first paragraph (the who, what, why, and when), and is typically written in a straight, matter-of-fact style. The narrative or anecdotal lead is more concerned with telling an emotionally engaging story, and often begins with an anecdote or story to draw the reader in.

Please code articles according to whether they adopt an 'inverted pyramid' lead or an 'anecdotal/narrative' lead. In instances where it is difficult to determine the style of the lead paragraph, please use the code 'can't tell'.

1- Inverted pyramid 2- Anecdotal/narrative 3- Ambiguous/other/can't tell

Variable 31: Personalized Storytelling

-This variable is interested in whether news articles use personalized storytelling to engage the reader in the story. Personalized storytelling is "a narrative form which draws on the personal experience of a particular individual caught up in a story to dramatize a broader social issue" (Wahl-Jorgensen 2013: 135). Sometimes, journalists will explain an event or issue from the perspective of an individual to add greater emotional depth to the story. This is a particularly popular trope in stories about war, accidents, and disasters, and in long-form articles like features and investigations. Please indicate if the article in question uses personalized storytelling with 'yes' or 'no'. If you are unsure if an article is using personalized storytelling, please use the code 'ambiguous/can't tell'.

1- Yes 2- No 3- Ambiguous/can't tell

2.5. Disclosure transparency

- This set of variables have been developed to see how 'open' the journalist is being with readers about the origin and province of their news story.

Variable 32: Time Stamps

-Please record how much information is present about the time and/or date an article was published.

1-Date of publication only	3-Date and time of publication
2- Time of publication only	4-No time stamp

Variable 33: Authorial Information

-Please record how much information an article contains about its the author(s)?

1- Author's name only2- Author's name and contact information (including links to the author's social media accounts)3-No authorial information

For articles that contain multiple authors, *all* authors must have their names and contact information displayed for you to use code two: 'author's name and contact information'. Otherwise, please select from the other two codes, as appropriate.

2.6. Hyperlinks

The coding procedure for hyperlinks is slightly more complex. For this reason, an explanation of what constitutes a hyperlink is provided, before the coding procedure for hyperlinks is described in more detail.

2.6.1. Introduction

A hyperlink is a word, image, or phrase that a user can click on to go to travel to a new document or web page. Hyperlinks are now recognised as an "established feature of online news" (Dimitrova and Neznanski 2006: 256) and are typically used to provide the reader with more information or context about a particular event. Often, hyperlinks will appear embedded in text, for example:

Starbucks baristas on Sunday became eligible for six weeks of 100% paid maternity leave, but that's still far less than the 18 weeks corporate employees get, prompting some shareholders to ask if the policy could be discriminatory.

The hyperlink above leads to an article on the website of Starbucks. Alternatively, hyperlinks are sometimes separated from the main body text of a news story:

It comes after Mr Johnson set out his Brexit "red lines" at the weekend.

- Conference live: Rolling text and video updates
- Kuenssberg: Fears of a downward spiral
- 'Uncomfortable truths' in PM's race audit
- Live: MEPs debate state of Brexit talks

Mrs May has been shown the foreign secretary's speech and is understood to be happy with it, the BBC's assistant political editor Norman Smith.

These hyperlinks appear in-line with the body text of the news story and link to other news stories published on the BBC's main website. Finally, hyperlinks can appear *outside* the main body text of a news story, in, for example, sidewidgets, below the main article, or elsewhere on the website:



Theresa May has brushed off questions about Boris Johnson's recent Brexit comments - saying she does not want to be surrounded by "yes men".

Features

Vincent

In the example above, links to other news stories can be found in the 'top stories' widget to the side of the article, and also above the article, in a series of links that take the user to other areas of the BBC news website (e.g. 'World', 'Business', 'Science', 'Health').

"There is no simple idea that is going to change all of these. It will take action in a number of areas."
₽ View comments
Related Topics
Boris Johnson Theresa May

A strange encounter with St



Is Mumbai India's most unliveable city?

Most Read



Links also commonly feature below the final paragraph of the news article, pointing the user to 'related topics' or encouraging them to 'share'.

In the interest of clarity, this study defines hyperlinks as links that appear either *embedded* in the main body text of a news article, or *in-line* with the body text of a news article (not, for example, in separate widgets or sidebars). Hyperlinks that appear either *before* or *after* the first and final paragraph of a news story should also not be recorded.

2.6.2. Hyperlinks Variables

Variable 34: Number of Hyperlinks -Please record how many hyperlinks appear in the news article.

Variable 35: Number of internal hyperlinks

-Please record how many internal hyperlinks are used in each article. An internal link is a link that points the reader *inwards* to an internal source of information on the same website (e.g. A hyperlink in a *BuzzFeed* article that takes the reader to another *BuzzFeed* news article).

Variable 36: Number of external hyperlinks

-Please record how many external hyperlinks are used in each article. An external link is a link that points the reader *outwards* to an external source of information, on another website.

2.7. Participatory transparency

- One variable has been developed to explore the extent to which readers are invited to participate in news articles.

Variable 37: Participation Features

-Please indicate how many of the following features are contained within the news article. Note: you may select more than one feature.

1-Sharing: tools are provided for readers to share the news item (for example, via social media or email) 2-Commenting: readers are invited to comment on the news item 3-Poll: readers are invited to submit an answer to a poll in the news item 4-Quiz: readers are invited to answer questions in a quiz in the news item 5-Submit your own content: readers are encouraged to send in their own content, including reports, images, and videos, to contribute directly to a news story

6-*Reader contribution published*: where readers have been previously asked to send in their own content, these contributions now explicitly form part of the news story. Note: this is different to journalists simply using ordinary people as sources, such as tweets compiled from Twitter, or comments taken from Facebook.

7-*Reader news*: news item is presented as being wholly written by a reader, rather than a journalist.
2.8. Conclusion

Thank you for your assistance. If you require any further clarification on any of the variables detailed above, then please contact the lead researcher (Paul Stringer) of the project.

9. Bibliography

Abbott, A. D. 1988. *The System of Professions: An Essay on the Division of Expert Labor*. Chicago: University of Chicago Press.

Adams, T. 2013. Shane Smith: "I want to build the next CNN with Vice – its within my grasp". *The Guardian*. [Online]. [Accessed 7th February 2016]. Available from: <u>https://www.theguardian.com/media/2013/mar/23/shane-smith-vice-interview</u>

Agarwal, S.D. and Barthel, M.L. 2015. The friendly barbarians: Professional norms and work routines of online journalists in the United States. *Journalism*. **16**(3), pp. 376-391.

Amazeen, A. M. and A.R. Muddiman. 2017. Saving media or trading on trust? The effects of native advertising on audience perceptions of legacy and online news publishers. *Digital Journalism*, 6(2), pp. 176-195.

Amazeen, A.M. and Wojdynski, B.W. 2018. [Post-print]. The effects of disclosure format on native advertising recognition and audience perceptions of legacy and online news publishers. *Journalism*. [Online]. [Accessed 8th May 2018]. Available from: <u>http://journals.sagepub.com/doi/abs/10.1177/1464884918754829</u>

Anderson, C. 2008. Journalism: Expertise, authority, and power in democratic life. In: Hesmondhalgh, D. and Toynbee, J. Eds. *The Media and Social Theory*. New York: Routledge, pp. 248-264.

Anderson, C.W. 2011a. Blowing up the newsroom. Ethnography in an age of distributed journalism. In: Domingo, D. and Paterson, C. eds. *Making Online News. Newsroom Ethnographies in the Second Decade of Internet Journalism*. New York: Peter Lang Publishing, pp. 151–160.

Anderson, C.W. 2011b. Between creative and quantified audiences: Web metrics and changing patterns of newswork in local US newsrooms. *Journalism*. **12**(5), pp. 550-566.

Anderson, C.W. 2011c. Deliberative, agonistic, and algorithmic audiences: Journalism's vision of its public in an age of audience transparency. *International Journal of Communications*. **5**, pp. 529-547.

Anderson, C. W., Bell, E. and Shirky, C. 2012. *Post-industrial journalism: adapting to the present—A report*. New York: Tow Centre for Digital Journalism. [Online]. [Accessed 22nd March 2017]. Available from: <u>http://towcenter.org/re search/post-industrial-journalism-adapting-to-the-present-2/</u>

Armstrong, C. 2004. The influence of reporter gender on source selection in newspaper stories. *Journalism & Mass Communication Quarterly*. **81**(1), pp. 139–154.

Associated Press. 2008. A new model for news: Studying the deep structure of young-adult news consumption. Associated Press. [Online]. [Accessed 11th November 2017]. Available from: <u>http://apo.org.au/node/15035</u>

Atton, C. and Wickenden, E. 2005. Sourcing routines and representation in alternative journalism: A case study approach. *Journalism Studies.* **6**(3), pp. 347–359.

Atton C. and Hamilton, J. 2008. *Alternative journalism*. London: Sage.

Bakker, P. 2012. Aggregation, content farms and Huffinzation. *Journalism Practice*. **6**(5-6), pp. 627-638.

Banaji, S. and Buckingham, D. 2013. *The civic web: Young people, the internet, and civic participation*. Cambridge, MA: MIT Press.

Banning, S. A. 1999 The professionalization of journalism: A nineteenth-century beginning. *Journalism History*. **24**(4), pp. 157–160.

Barnhurst, K.G. 2002. News geography and monopoly: The form of reports on U.S. newspaper internet sites'. *Journalism Studies* **3**(4), pp. 477-489.

Barnhurst, K.G. 2010. The form of reports on U.S. newspaper internet sites: An update. *Journalism Studies*. **11**(4), pp. 555?66

Barnhurst K.G. 2011. The problem of modern time in American journalism. *KronoScope*. **11**(1–2), pp. 98–123.

Barnhurst, K G., and Mutz,D.C. 1997. American journalism and the decline in eventcentered reporting. *Journal of Communication*, **47**(4) (Autumn), pp. 27–53.

Barnhurst, K. and Nerone, J. 2009. Journalism History. In: Wahl-jorgensen, K. and Hanitzsch, T. Eds. *The Handbook of Journalism Studies*. London: Routledge, pp. 17-28.

Bazaarvoice. 2012. Talking to strangers: millennials trust people over brands. Bazaarvoice. [Online]. [Accessed 27th May 2017]. Available from: <u>http://media2.bazaarvoice.com/documents/Bazaarvoice_WP_Talking-to-</u> <u>Strangers.pdf</u>

BBC. 2018a. Editorial guidelines. *BBC*. [Online]. [Accessed 14th January 2018]. Available from: <u>https://www.bbc.co.uk/editorialguidelines/</u>

BBC. 2018b. Have your say. *BBC*. [Online]. [Accessed 11th March 2018]. Available from: <u>https://www.bbc.co.uk/news/have_your_say</u>

Beam R.A. 2001. Does it pay to be a market-oriented daily newspaper? *Journalism &Mass Communication Quarterly* **78**(3), pp. 466–483.

Becker, H. 1967 Whose side are we on? Social Problems, 14, pp. 239–47.

Beckett, C. 2015. How journalism is turning emotional and what that might mean for news. *Polis Blog.* [Online]. [Accessed 28th May 2018]. Available from: <u>http://eprints.lse.ac.uk/63822/</u>

Beckett, C. and Deuze, M. 2016. On the role of emotion in the future of journalism. *Social Media* + *Society*, **2**(3), pp. 1–6.

Bell, E., Owen, T., Brown, P., Hauka, C. and Rashidian, N. 2017. *The platform press: How Silicon Valley reengineered journalism*. New York: Tow Center for Digital Journalism, Columbia Journalism School.

Benady, D. 2014. Native advertising faces Gen Y's bullshit detector. *Campaign*. [Online]. [Accessed 13th June 2017]. Available from: <u>https://www.campaignlive.co.uk/article/native-advertising-faces-gen-ys-bullshit-detector/1281378</u>

Benkler, Y. 2006. *The Wealth of Networks: How Social Production Transforms Markets and Freedom*. New Haven: Yale University Press.

Bennett, W.L. 2008. Changing citizenship in the digital age. In: Bennett, W.L. ed. *Civic Life Online: Learning How Digital Media Can Engage Youth*. Cambridge, MA: The MIT Press, pp. 1–24.

Benson, R. 1999. Field theory in comparative context: a new paradigm for media studies. *Theory and Society*, **28**(3), pp. 463-98.

Benson, R. 2006. News media as a "Journalistic Field": What Bourdieu adds to new institutionalism, and vice versa. *Political Communication*. **23**(2), pp. 187-202.

Benson, R. and Neveu, E. 2005. *Bourdieu and the journalistic field*. eds. Malden, MA: Polity Press.

Benson, R. and Neveu, E. 2005. Introduction: Field theory as a work in progress. In: Benson, R. and Neveu, E. Eds. *Bourdieu and the Journalistic Field*. Malden, MA: Polity Press, pp. 1–28.

Berelson, B. 1952. Content analysis in communication research. New York: Hafner.

Berger, A. 1998. Depth interviews. In: Berger, A. Ed. *Media research techniques* 2^{*nd*} *Ed.* London: Sage, pp.55-62.

Betters, E. 2016. What is Slack and how does it work? *Pocket Lint.* [Online]. [Accessed 27th March 2018]. Available from: <u>https://www.pocket-lint.com/apps/news/136472-what-is-slack-and-how-does-it-work</u>

Bingham, W. V. and Moore, B. V. 1931. How to interview. Oxford, England: Harpers.

Bird, E.S. 2009. Tabloidization: What is it, and does it really matter? In: Zelizer, B. Ed. *The changing faces of journalism.* London: Routledge, pp. 40-49.

Bird, S. E. and Dardenne, R. 2009. 'Rethinking news and myth as story-telling. In: Wahl-Jorgensen, K. and Hanitzsch, T. Eds. *The Handbook of Journalism Studies*. London and New York: Routledge, pp. 205-217.

Blake, H. 2016. BuzzFeed to Fleet Street – "We're figuring out the future of journalism". Interview with Ian Burrell from *The Drum*. [Online]. [Accessed 14th June 2017]. Available from: <u>http://www.thedrum.com/opinion/2016/06/09/buzzfeed-fleet-street-were-figuring-out-future-journalism</u>

Blake, H. and Templon, J. 2016. The tennis racket. *BuzzFeed*. [Online]. [Accessed 14th March 2017]. Available from: <u>https://www.buzzfeed.com/heidiblake/the-tennis-racket?utm_term=.rykNLzpjR#.uyq8zV6A1</u>

Blom, J.N. and Hansen, K.R. 2015. Click bait: Forward-reference as lure in online news headlines. *Journal of Pragmatics*. **76**, pp. 87-100.

Boczkowski, P. 2004. *Digitizing the news: Innovation in online newspapers*. Cambridge, MA: MIT Press.

Boczkowski, P. J. and J. A. Ferris 2005. Multiple media, convergent processes, and divergent products: Organizational innovation in digital media production at a European firm, *Annals of the American Academy of Political and Social Science*. **597**(1), pp. 32–47.

Boczkowski, P. and De Santos, M. 2007. When more media equals less news: Patterns of content homogenization in Argentina's leading print and online newspapers. *Political Communication*, **24**(2), pp. 167–180.

Boczkowski, P.J. and Peer, L. 2011. The choice gap: the divergent online news preferences of journalists and consumers. *Journal of Communication*. **61**(5), pp. 857–876.

Borger, M., van Hoof, A., Costera Meijer, I. and Sanders, J. 2013. Constructing participatory journalism as a scholarly object. *Digital Journalism*. **1**(1), pp. 117-134.

Born, G. 2004. *Uncertain vision: Birt, Dyke and the reinvention of the BBC*. London: Secker & Warburg.

Bourdieu, P. 1984. *Distinction: A social critique of the judgement of taste*. Cambridge, MA: Harvard University Press.

Bourdieu, P. 1985. The social space and the genesis of groups. *Theory and society*. **14**(6), pp. 723-744.

Bourdieu, P. 1986. The forms of capital. In: Richardson, J. ed. *Handbook of Theory and Research for the Sociology of Education*. New York, NY: Greenwood, pp. 241 – 258.

Bourdieu, P. 1990. The logic of practice. Stanford, CA: Stanford University Press.

Bourdieu, P. 1991. Language and symbolic power. Cambridge: Polity.

Bourdieu, P. 1993. *The field of cultural production. Essays on art and literature*. Cambridge: Polity Press.

Bourdieu, P. 1996. The rules of art. Cambridge, England: Polity.

Bourdieu, P. 1998. On Television. New York: New Press.

Bourdieu, P. 2000. Pascalian Meditations. Cambridge: Polity Press.

Bourdieu, P. 2005. The political field, the social science field and the journalistic field. In: Benson, R. and Neveu, E. eds. *Bourdieu and the Journalistic Field*. Cambridge: Polity Press, pp. 29–47.

Boyer, D. 2013. *The life informatic: Newsmaking in the digital era*. Ithaca, NY: Cornell University Press.

Brenner, M. E. 2006. Interviewing in educational research. In: Green, J.L., Camilli, G. and Elmore, P.B. Eds. *Handbook of complementary methods in education research 3rd ed*. Washington, DC: American Educational Research Association, pp. 357–370.

Brewer, J.D. 2000. Ethnography. Buckingham: Open University Press.

Broersma, M. 2010. The unbearable limitations of journalism: On press critique and journalism's claim to truth. *International Communication Gazette*. **72**(1), pp. 21–33.

Broersma, M. and Graham, T. 2013. Twitter as a news source: How Dutch and British newspapers used tweets in their news coverage, 2007-2011. *Journalism Practice*. 7(4), pp. 446-464.

Brooks, G. 2016. The subfield of online journalism: a study of the legitimizing practices of online news organizations. Ph.D. thesis, University of Cambridge.

Bromley, M. 1997. The end of journalism? Changes in workplace practices in the press and broadcasting in the 1990s. In: Bromley, M. O'Malley, T. Eds. *A Journalism Reader,* London: Routledge, pp. pp. 330–350.

Bruno, N. 2011 *Tweet first, verify later? How real-time information is changing the coverage of worldwide crisis events*. Oxford: Reuters Institute for the Study of Journalism.

Bruns, A. 2008. The active audience: Transforming journalism from gatekeeping to gatewatching. In: Paterson, C. and Domingo, D. Eds. *Making online news: The ethnography of new media production*. New York, NY: Peter Lang, pp. 171-184.

Buckingham, D. 2000. *The Making of Citizens: Young People, News and Politics*. London: Routledge.

Burrell, I. 2014. BuzzFeed to Fleet Street – "We're figuring out the future of journalism". *The Drum.* [Online]. [Accessed 13th January 2018]. Available from: <u>http://www.thedrum.com/opinion/2016/06/09/buzzfeed-fleet-street-were-figuring-out-future-journalism</u>

BuzzFeed. 2016. BuzzFeed UK limited financial statements 31 December 2016. *Companies House*. [Online]. [Accessed 22nd May 2018]. Available from: <u>https://beta.companieshouse.gov.uk/company/08318051/filing-history</u>

BuzzFeed. 2018a. Advertise. *BuzzFeed* [Online]. [Accessed 10th September 2017]. Available from: <u>https://www.buzzfeed.com/advertise</u>

BuzzFeed. 2018b. About. *BuzzFeed*. [Online]. [Accessed 10th March 2018]. Available from: <u>https://www.buzzfeed.com/about</u>

Campbell, C.P. Race, myth and the news. Thousand Oaks, CA: Sage.

Carlson, M. 2007. Blogs and journalistic authority: The role of blogs in US election Day 2004 Coverage. *Journalism Studies*. **8**(2), pp. 264-279.

Carlson, M. 2015. When news sites go native: redefining the advertising–editorial divide in response to native advertising. *Journalism*, **16**(7), pp. 849-865.

Carlson, M. 2017. *Journalistic authority: Legitimating news in the digital era*. New York: Columbia University Press.

Carlson, M. 2018. Confronting measurable journalism. *Digital Journalism*. **6**(4), pp. 4016-417.

Carlson, M. and Usher, N. 2016. News startups as agents of innovation: for-profit digital news startup manifestos as metajournalistic discourse. *Digital Journalism*, **4**(5), pp. 563-581.

Carlsson, B. and Nilsson, E. 2015. Technologies of participation: Community news and social media in northern Sweden. *Journalism.* **17**(8), pp. 1113-1128.

Cassidy, W. 2005. Variations on a theme: the professional role conceptions of print and online newspaper journalists. *Journalism & Mass Media Quarterly*. **82**(2), pp. 264–280.

Cassidy W.P. 2007. Online news credibility: an examination of the perceptions of newspaper journalists. *Journalism & Mass Media Quarterly*. **82**(2), pp. 264–280.

Castells, M. 2009. *Communication Power*. New York: Oxford University Press Costeltoe.

Chadha, K. and Wells, R. 2016. Journalistic responses to technological innovation in newsrooms. *Digital Journalism*. **4**(8), pp. 1020-1035.

Chadwick, A. 2013. *The hybrid media system: Politics and power.* London: Oxford University Press.

Chalaby, J. K. 1996. Journalism as an Anglo-American Invention: A comparison of the development of French and Anglo-American journalism, 1830s–1920s. *European Journal of Communication*. **11**(3), pp. 303–326.

Champagne, P., and Marchetti, D. 2005. The contaminated blood scandal: Reframing medical news. In: Benson, R. and Neveu, E. Eds. *Bourdieu and the Journalistic Field*. Cambridge: Polity Press, pp. 113-134.

Charmaz, K. 2006. Constructing grounded theory: A practical guide through qualitative analysis. Thousand Oaks, CA: Sage.

Cherubini, F. and Nielsen, R.K. 2016. *Editorial analytics: How news media are developing and using audience data and metrics.* Oxford: Reuters Institute for the Study of Journalism.

Clarke, S. 2018. Vice shakes up UK-based operation, announces new hires and layoffs. *Variety*. [Online]. [Accessed 24th March 2018]. Available from: <u>http://variety.com/2018/tv/news/vice-restructures-uk-and-emea-1202691927/</u>

Clough, P. T. and Halley, J. Eds. 2007. *The Affective Turn: theorizing the social*. Durham, NC: Duke University Press.

Coddington, M. 2012. Building frames link by link: The linking practices of blogs and news sites. *International Journal of Communication*. **6**, pp. 2007–2026.

Coddington, M. 2014. Normalizing the hyperlink. *Digital Journalism*. **2**(2), pp. 140–155.

Cohen, N.S. From pink slips to pink slime: Transforming media labor in a digital age. *The Communication Review*. **18**(2), pp. 98-122.

Colson, V. and Heinderyckx F. 2008. Do online journalists belong in the newsroom? A Belgian case of convergence'. In: Domingo, D. and Paterson, C. Eds. *Making Online News: The Ethnography of New Media Production*. New York: Peter Lang, pp. 143-156.

comScore. 2012. Next-generation strategies for advertising to millennials. [Online]. [Accessed 18th January 2018]. Available from: <u>https://www.comscore.com/Request/Presentations/2012/Millenials-Report-Download-January-2012?c=1</u>

comScore. 2017. comScore Ranks the Top 50 U.S. digital media properties for June 2017. [Online]. [Accessed 22nd September 2018]. Available from: <u>https://www.prnewswire.com/news-releases/comscore-ranks-the-top-50-us-digital-media-properties-for-june-2017-300491305.html</u>

Cook, T. E. 1998. Governing with the news. Chicago: University of Chicago Press.

Coppins, M. 2016. Interview with Evan Ratliff for *Longform Podcast.* [Online]. [Accessed 3rd January 2018]. Available from: <u>https://longform.org/posts/longform-podcast-207-mckay-coppins</u>

Corcoran, L. 2017. Interview with Vice's social media team. *Newswhip*. [Online]. [Accessed 8th September 2017]. Available from: <u>https://www.newswhip.com/2017/04/interview-with-vices-social-media-team/</u>

<u>Cornia, A., Sehl, A. and Nielsen, R.K. 2018. [Post-print].</u> "We no longer live in a time of separation": A comparative analysis of how editorial and commercial integration became a norm. *Journalism*.

Costera Meijer, I. 2007. The paradox of popularity: how young people experience the news. *Journalism Studies;* **8**(1), pp. 1–22.

Costera Meijer, I. 2013. Valuable journalism: The search for quality from the vantage point of the user. *Journalism* **14**(6), pp. 754–770.

Costera Meijer, I. and Kormelink, G.T. 2014. Checking, sharing, clicking and linking: Changing patterns of news use between 2004 and 2014. *Digital Journalism*. **3**(5), pp. 664–679.

Cottle, S. 2000. Rethinking news Access. Journalism Studies. 1(3), pp. 427 – 448.

Cottle, S. 2007. Ethnography and news production: New(s) developments in the field. *Sociology Compass*, **1**(1), pp. 1–16.

Crouse, T. 1973. The boys on the bus. New York, NY: Random House.

Curran, J. Fenton, N. and Freedman, D. 2012. *Misunderstanding the internet.* London: Routledge.

Craft, S., Barnhurst, K.G., Brennen, B., Glasser, T.L., Hanitzsch, T. and Singer, J.B. 2014. Trajectories. *Journalism Studies*. **15**(6), pp. 689-710.

Creswell, J. W. 2007. *Qualitative inquiry & research design: choosing among five approaches*: Thousand Oaks, CA: Sage.

Creswell, J. W. and Plano Clark, V. L. 2011. *Designing and conducting mixed methods research*. Thousand Oaks: CA: Sage.

Dahlgren, P. 1996. Media logic in cyberspace: Repositioning journalism and its publics. *Javnost/The Public*. **3**(3), pp. 59–72.

Dahlgren, P. 2005. The internet, public spheres, and political communication: Dispersion and deliberation. *Political Communication*. **22**(2), pp. 147-162.

Dangerfield, A. 2017. How BuzzFeed News compares to the BBC: Inside two very different newsrooms. [Online]. [Accessed 25th September 2018]. Available from: <u>https://www.thedrum.com/opinion/2017/01/17/how-buzzfeed-news-compares-the-bbc-inside-two-very-different-newsrooms</u>

Davies, N. 2008. Flat Earth News. London: Vintage.

Deacon, D., Pickering, M., Golding, P. and Murdock, G. 2007. *Researching communications: A practical guide to methods in media and cultural analysis 2nd ed.* London: Hachette UK.

De Keyser, J. and Raeymaeckers, K. 2012. The printed rise of the common man. How Web 2.0 has changed the representation of ordinary people in newspapers. *Journalism Studies*. **13**(5–6), pp. 825–835.

Denzin, Norman K. 1978. *The research act: A theoretical introduction to sociological methods*. New York: McGraw-Hill.

Deuze, M. 2003. The web and its journalisms: Considering the consequences of different types of news online. *New Media and Society*. **5**(1), pp. 203–26.

Deuze, M. 2004. What is multimedia journalism? *Journalism Studies*. **5**(2), pp. 139–52.

Deuze, M. 2005. What is journalism? Professional identity and ideology of journalists reconsidered. *Journalism*. **6**(4), pp. 442–64.

Deuze, M. 2009a. Technology and the individual journalist: agency beyond imitation and change. In: Zelizer, B. Ed. *The Changing Faces of Journalism: Tabloidization, Technology and Truthiness.* Abingdon: Routledge, pp. 82-97.

Deuze, M. 2009b. Journalism, citizenship, and digital culture. In: Papacharissi, Z. Ed. *Journalism and Citizenship: New Agendas in Communication*. New York, NY: Routledge, pp. 15 – 28.

Deuze, M. and Paulussen, S. 2002. Online journalism in the low countries: Basic occupational and professional characteristics of online journalists in Flanders and the Netherlands. *European Journal of* Communication. **17**(2), pp. 237-245.

Deuze, M. and Witschge, T. 2018. Beyond journalism: Theorizing the transformation of journalism. *Journalism*. **19**(2), pp. 165-181.

Dimitrova, D., Connolly-Ahern, C., Williams, A.P., Kaid, L.L. and Reid, A. 2003. 'Hyperlinking as gatekeeping: online newspaper coverage of the execution of an American terrorist'. *Journalism Studies*. **4**(3), pp. 401-414.

Dimock, M. 2018. Defining generations: Where Millennials end and post-Millenials begin. *Pew Research Center*. [Online]. [Accessed 21st September 2018]. Available from: <u>http://www.pewresearch.org/fact-tank/2018/03/01/defining-generations-where-millennials-end-and-post-millennials-begin/</u>

Dooley, P. 1997. *Taking their political place: Journalists and the making of an occupation.* Westport, CT: Greenwood Publishing Group.

Domingo, D. 2006. Inventing online journalism. Development of the internet as a news medium in four Catalan online newsrooms. PhD thesis, Universitat Autonoma de Barcelona.

Domingo, D., Quandt, T., Heinonen, A., Paulussen, S., Singer, J.B. and Vujnovic, M. 2008. Participatory journalism practices in the media and beyond: An international comparative study of initiatives in online newspapers. *Journalism Practice*. **2**(3), pp. 326-342.

Domingo, D., Masip, P. and Costera Miejer, I. 2015. Tracing digital news networks. Towards an integrated framework of the dynamics of news production, circulation and use. *Digital Journalism*. **3**(1), pp. 53-67.

Donsbach, W. 2008. Journalists' role perceptions. *The International Encyclopaedia of Communication*. [Online]. [Accessed 12th October 2017]. Available from: <u>http://0-</u>www.communicationencyclopedia.com.wam.leeds.ac.uk/subscriber/uid=385/tocnode ?id=g9781405131995 chunk g978140513199515 ss10-1&authstatuscode=202

Donsbach, W., and Patterson, T. E. 2004. Political news journalists: Partisanship, professionalism, and political roles in five countries. In: Esser, F. and Pfetsch, B. Eds. *Comparing Political Communication: Theories, Cases, and Challenges*, New York: Cambridge University Press, pp. 251–270.

Dooley, P.1997. *Taking their political place: Journalists and the making of an occupation.* Westport, CT: Greenwood Publishing Group.

Downey, J. and Fenton, N. 2003. New media, counter publicity and the public sphere. *New Media & Society* **5**(2), pp. 185–202.

Drok, N. Hermans, L. Kats, K. 2018. Decoding youth DNA: The relationship between social engagement and news interest, news media use and news preferences of Dutch millennials. *Journalism*, **19**(5), pp. 699-717.

Edwards, D. 1999. Emotion discourse. Culture & Psychology. 5(3), pp. 271–291.

Eisenhardt, K. M. 1989. Building theories from case study research. *The Academy of Management Review.* **14**(4), pp. 532-550.

Epstein, E.J. 1973. *News from nowhere: television and the news*. New York: Random House.

Fenton, N. Ed. 2010. *New media, old news: Journalism and democracy in the digital age*. London: Sage.

Fenton, N. 2011. Deregulation or democracy? New media, news, neoliberalism and the public interest. *Continuum: Journal of Media & Cultural Studies*, **25**(1), pp. 63-72.

Ferrer-Conill, R. and Tandoc, E.C. 2018. The audience-oriented editor. *Digital Journalism.* **6**(4), pp. 436-453.

Ferrucci, P. and Vos, T. 2017. Who's in, who's out? Constructing the identity of digital journalists. *Digital Journalism.* **5**(7), pp. 868-883.

Fink, K. and Schudson, M. 2014. The rise of contextual journalism, 1950s–2000s. *Journalism*. **15**(1), pp. 3-20.

Fishman, M. 1980. Manufacturing the News. Austin, TX: University of Texas Press.

Flyvbjerg, B. 2006. Five misunderstandings about case-study research. *Qualitative Inquiry*. **12**(2), pp. 219-245.

Franklin, B. and Carlson, M. 2011. *Journalism, sources and credibility*. New York: Routledge.

Franks, S. 2013. Women and journalism. London: I.B. Tauris.

Freedman, D. 2012. Web 2.0 and the death of the blockbuster economy. In: Curran J. Fenton, N. and Freedman, D. Eds. *Misunderstanding the internet*. New York: Routledge, pp. 69–94.

Fuchs, C. 2011. *Foundations of critical media and information studies*. New York, NY: Routledge.

Fullerton, L. 2017. Content advertising slated to grow into a multi-billion dollar industry by 2020. *The Drum*. [Online]. [Accessed 13th April 2017]. Available from: <u>http://www.thedrum.com/news/2017/02/14/content-advertising-slated-grow-multi-billion-dollar-industry-2020</u>

Gade, P. 2008. Journalism guardians in a time of great change: newspaper editors' perceived influence in integrated news organizations. *Journalism & Mass Communication Quarterly*, **85**(2), pp. 371-392.

Gade, P J. 2011. Postmodernism, Uncertainty and Journalism. In: Lowrey, W. and Gade, P.J. eds. *Changing the News: The Forces Shaping Journalism in Uncertain* Times. New York: Routledge, pp. 63-82.

Gade. P. J. and Raviola, E. 2009. Integration of news and news of integration: a structural perspective on news media changes. *Journal of Media Business Studies*, **6**(1), pp. 87-111.

Gandy, O. H. 1982. *Beyond agenda setting: Information subsidies and public policy*. Norwood, NJ: Ablex.

Gans, H. J. 2004. *Deciding what's news: A study of CBS Evening News, NBC Nightly News, Newsweek and Time*. New York: Pantheon.

Gans, H. 2011. Multiperspectival news revisited: Journalism and representative democracy. *Journalism*. **12**(1), pp. 3-13.

Garrahan, M. 2017. Vice Media gains \$5.7bn valuation with TPG investment. *Financial Times*. [Online]. [Accessed 14th September 2018]. Available from: https://www.ft.com/content/519391b0-5504-11e7-9fed-c19e2700005f

Gibson, J. 2016. BuzzFeed to Fleet Street – "We're figuring out the future of journalism". Interview with Ian Burrell from *The Drum*. [Online]. [Accessed 14th June 2017]. Available from: <u>http://www.thedrum.com/opinion/2016/06/09/buzzfeed-fleet-street-were-figuring-out-future-journalism</u>

Gieryn, T. F. 1983. Boundary-work and the demarcation of science from nonscience: Strains and interests in professional ideologies of scientists. *American Sociological Review*. **48**(6), pp. 781–795.

Gillmor, D. 2004. We the media. Grassroots journalism by the people, for the people. Sebastopol, CA: O'Reilly.

Glaser, B. G., and Strauss, A. 1967. *The discovery of grounded theory: Strategies for qualitative research*. London: Weidenfeld and Nicolson.

Gomm, R., Hammersley, M. and Foster, P. 2000. Case study and generalization. In: Gomm, R. Hammersley, M and Foster, P. Eds. *Case study method: Key issues, key texts*. London: Sage, pp. 98–115.

Griffith, E. 2014. Why YouTube is spending mega ad dollars to promote Vice News. *Fortune*. [Online]. [Accessed 20th September 2017]. Available from: <u>http://fortune.com/2014/06/19/youtube-vice-news/</u>

Guardian. 2009. Frequently asked questions about community on the Guardian website. *The Guardian*. [Online]. [Accessed 27th June 2017]. Available from: <u>https://www.theguardian.com/community-faqs</u>

Hackett, A. R. 1984. Decline of a paradigm? Bias and objectivity in news media studies. *Critical Studies in Mass Communication*. **1**(3), pp. 229 – 259.

Hall, S., Critcher, C., Jefferson, C., Clarke, N.J. and Roberts, B. 1978. *Policing the crisis: Mugging, the state, and law and order*. London: Macmillan.

Hallin, D. C. 1992. The passing of the "high modernism" of American journalism. *Journal of Communication*. **42**(3), pp. 14–25.

Hallin, D. and Mancini, P. 2004. *Comparing media systems: Three models of media and politics*. Cambridge: Cambridge University Press.

Hamblin, J. 2014. What is clickbait? Everything. *The Atlantic*. [Online]. [Accessed 14th May 2017]. Available from: https://www.theatlantic.com/entertainment/archive/2014/11/clickbait-what-is/382545/

Hampton, M. 2001. Understanding media: Theories of the press in Britain, 1850-1914. *Media, Culture & Society*. **23**(2), pp. 213-231

Hanitzsch, T. 2007 Deconstructing journalism culture: Toward a universal theory.

Communication Theory. **17**(4), pp. 367–385.

Hanitzsch, T. 2012. Does gender determine journalists' professional views? A reassessment based on cross-national evidence. *European Journal of Communication*. **27**(3), pp. 257-277.

Hansen, C. 1991. Source diversity and newspaper enterprise journalism. *Journalism Quarterly*. **68**, pp. 474-482.

Hansen, A., Cottle, S., Negrine, R. and Newbold, C. 1998. *Mass communication research methods*. London: Macmillan.

Harcup, T. .2003. The unspoken/ said: The journalism of alternative media. *Journalism: theory, practice and criticism.* **4**(3), pp. 356-376.

Hardt, M. and A. Negri 2000. Empire. Cambridge, MA: Harvard University Press.

Hardy, J. Critical political economic of the media. London: Routledge.

Hartley, J. 2004. Case study research. In: Cassell, C. and Symon, G. Eds. *Essential guide to qualitative methods in organizational research*. London: Sage, pp.323-333.

Hedman, U. 2016. When journalists tweet: Disclosure, participatory, and personal transparency. *Social Media* + *Society*. **2**(1), pp. 1-13.

Heinrich, A. 2012. Foreign reporting in the sphere of network journalism. *Journalism Practice*. **6**(5–6), pp. 766–775.

Hellmueller, L. Vos, P.T. and Poepsel, A. M. 2013. Shifting journalistic capital? Transparency and objectivity in the twenty-first century. *Journalism Studies*, **14**(3), pp. 287-304.

Herman, M. (2017) News orgs succeed with fresh approach to international coverage. *Columbia Journalism Review*. [Online]. [Accessed 11th September 2017]. Available from:

https://www.cjr.org/business_of_news/international_coverage_the_guardian_buzzfee d_topic_region.php

Hermida, A. 2010. Twittering the news. *Journalism Practice*. **4**(3), pp. 297–308.

Hermida, A. 2011. Tweet the news: Social media streams and the practice of journalism. In: Allan, S. Ed. *The Routledge Companion to News and Journalism*, Routledge, New York, pp. 671–682.

Hermida, A. 2013. '#Journalism: Reconfiguring journalism research about Twitter one tweet at a time. *Digital Journalism*. **1**(3), pp. 295–313.

Hermida, A. and Thurman, N. 2008. A clash of cultures: the integration of usergenerated content within professional journalistic frameworks at British newspaper websites. *Journalism Practice* **2**(3), pp. 343-356.

Hermida, A., Domingo, D. Heinonen, A. Paulussen, S. Quandt, T. Reich, Z. Singer, J.B. and Vujnovic, M. 2011. The Active recipient: participatory journalism through the lens of the Dewey-Lippmann debate. #ISOJ – The Official Journal of the International Symposium on Online Journalism **1**(2), pp. 129–152.

Hermida, A., Lewis, S. C. and Zamith, R. 2014, Sourcing the Arab Spring: A case study of Andy Carvin's sources on Twitter during the Tunisian and Egyptian revolutions. *Journal of Computer-Mediated Communication*, vol. **19(**3), pp. 479-499.

Herring, S. C. 2010. Web content analysis: Expanding the paradigm. In: Hunsinger, J., Klastrup, L. and Allen, M. *International Handbook of Internet Research*. London: Springer Science+Business Media B.V, pp. 233–249.

Hesmondhalgh, D. 2006. Bourdieu, the media and cultural production. *Media, Culture & Society*. **28**(2), pp. 211-231.

Hwong, C. 2018. Tracking the news: Reader demographics. *Verto Analytics.* [Online]. [Accessed 22nd March 2018]. Available from: <u>https://www.vertoanalytics.com/chart-week-tracking-news-reader-demographics/</u>

Ip, C. 2015a. The cult of Vice. *Columbia Journalism Review*. [Online]. [Accessed 20th May 2016]. Available from: <u>http://www.cjr.org/analysis/the_cult_of_vice.php</u>

Ip, C. 2015b. Why digital media and identity issues are a match. *Columbia Journalism Review*. [Online]. [Accessed 20th March 2017]. https://www.cjr.org/analysis/why digital media and identity issues are a match.p hp

James, B. 2016. Marketing to a millennial. *Huffington Post.* [Online]. [Accessed 19th April 2017]. Available from: <u>https://www.huffingtonpost.co.uk/bev-james/marketing-to-a-millennial_b_8781346.html</u>

Janowitz, M. 1975. Professional models in journalism: The gatekeeper and the advocate. *Journalism Quarterly*. **52**, pp. 618–626.

Jarvis, J. 2006. Networked journalism. *Buzzmachine.* [Online]. [Accessed 13th January 2018]. Available from: <u>https://buzzmachine.com/2006/07/05/networked-journalism/</u>

Jeffries, A. 2016a. We're taking a break from Slack. Here's Why. *Vice*. [Online]. [Accessed 22nd November 2017]. Available from: <u>https://motherboard.vice.com/en_us/article/aekk85/were-taking-a-break-from-slack-heres-why</u>

Jeffries, A. 2016b. An oral history of our week without Slack. *Vice*. [Online]. [Accessed 15th May 2017]. Available from: <u>https://motherboard.vice.com/en_us/article/gv5jpy/an-oral-history-of-our-week-</u> without-slack

Jenkins, H. 2004. The cultural logic of media convergence. *International Journal of Cultural* Studies. **7**(33), pp. 33–43.

Johnson, R.B., Onwuegbuzie, A.J. and Turner, L.A. Toward a definition of mixed method research. *Journal of Mixed Methods Research*. **1**(2), pp. 112-133.

Johnstone, J. W. C., Slawski, E. J. and Bowman, W. W. 1976. *The newspeople: A portrait of American journalists and their work*. Urbana: University of Illinois Press.

Jönnson, A.M. and Örnebring, H. 2011. User-generated content and the news: Empowerment of citizens or an interactive illusion? *Journalism Practice*, **5**(2), pp. 127–144.

Jurkowitz, M. 2014. The growth in digital reporting: what it means for journalism and news consumers. *Pew Research Centre*. [Online]. [Accessed 11th November 2017]. Available from: http://www.journalism.org/2014/03/26/the-growth-in-digital-reporting/

Kafka, P. 2016. NBC Universal is doubling its bet on BuzzFeed by investing another \$200 Million. *Recode*. [Online]. [Accessed 7th May 2017]. Available from: <u>https://www.recode.net/2016/10/20/13352900/nbcuniversal-buzzfeed-investment</u>

Kanter, J. 2017. BuzzFeed is about to make a third of its staff redundant. *Business Insider UK*. [Online]. [Accessed 27th December 2017]. Available from: <u>http://uk.businessinsider.com/buzzfeed-uk-to-make-45-staff-redundant-2017-12</u>

Kaplan, D. 2002. *Politics and the American Press: The rise of objectivity*. Cambridge: Cambridge University Press.

Karlsson, M. 2010. Rituals of transparency. *Journalism Studies*. **11**(4) pp. 535–545.

Karlsson, M. 2011. The immediacy of online news, the visibility of journalistic processes and a restructuring of journalistic authority. *Journalism*. **12**(3), pp. 279–295.

Karlsson, M. 2012. Charting the liquidity of online news: Moving towards a method for content analysis of online news. *International Communication Gazette*. **74**(4), pp. 385–402.

Karlsson, M. and Strömback, J. 2010. Freezing the flow of online news. *Journalism Studies* **11**(1), pp. 2–19.

Karlsson, M. and Sjøvaag, H. 2016. Content analysis and online news. *Digital Journalism.* **4**(1), pp. 177-192.

Karpf, D. 2016. Social science research methods in internet time. *Information, Communication & Society.* **15**(5), pp. 639-661.

Klinenberg, E. 2005. Convergence: News production in a digital age. *The Annals of the American Academy of Political and Social Science*. **597**(1), pp. 48-64.

King, N. Using interviews in qualitative research. In: Cassell, C. and Symon, G. Eds. *Essential guide to qualitative methods in organizational research.* London, New Delhi, Thousand Oaks, CA: Sage, pp. 11-23.

Klinenberg, E. 2005. Convergence: New production in a digital age. *Annals of the American Political Science Association.* **59**(7), pp. 48–68.

Knight, M. 2012. Journalism as usual: The use of social media as a newsgathering tool in the coverage of the Iranian elections in 2009. *Journal of Media Practice*. **13**(1), pp. 61–74.

Kormelink, T. and Meijer, C. I. 2014. Tailor-made news. *Journalism Studies*. **15**(5), pp. 632-641.

Kovach, B. and Rosentiel, T. 2001. *The elements of journalism: What newspeople should know and the public should expect.* New York: Crown Publishers.

Kreiss, D. and Brennen, S.J. 2016. Normative models of digital journalism. In: Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. eds. *The SAGE Handbook of Digital Journalism*. London, California, New Delhi, Singapore: SAGE Publications, pp. 299-315.

Küng, L. 2015. Innovators in digital news. London: I.B. Tauris.

Kurpius, D.D. Sourcing and civic journalism: Changing patterns of reporting? *Journalism & Mass Communication Quarterly*, **79**(4), pp. 853-866.

Lafrance, A. 2012. BuzzFeed with a press pass: What happens when the GIF king tries to take Washington? *Nieman Lab*. [Online]. [Accessed 22nd February 2016]. Available from: <u>http://www.niemanlab.org/2012/08/buzzfeed-with-a-press-pass-what-happens-when-the-gif-kings-try-to-take-washington/</u>

Lasorsa D., Lewis S.C. and Holton, A. 2011. Normalizing Twitter. *Journalism Studies*, **13**(1), pp. 19–36.

Lecheler, S. and Kruikmeier, S. 2015. Re-evaluating journalistic routines in a digital age: A review of research on the use of online sources. *New Media and Society*. **18**(1), pp. 156-171.

Lewis, S. C. (2012) The tension between professional control and open participation. *Information, Communication & Society.* **15**(6), pp. 836–866.

Lewis, J., Williams, A., Franklin, B., Thomas, J. and Mosdell, N. 2008. *The quality and independence of British journalism: tracking the changes over 20 years*. Mediawise: Cardiff University. [Online]. [Accessed 4th May 2017]. Available from: https://orca.cf.ac.uk/18439/1/Quality%20%26%20Independence%20of%20British%2 0Journalism.pdf

Lewis, S. C., Kaufhold, K. and Lasorsa, D. L. 2010. Thinking about citizen journalism: The philosophical and practical challenges of user-generated content for community newspapers. *Journalism Practice*, **4**(2), pp. 163-179.

Lewis, S.C., Zamith, R. and Hermida, A. 2013. Content analysis in an era of big data: A hybrid approach to computational and manual methods. *Journal of Broadcasting and Electronic Media*. **57**(1), pp. 34-52.

Lichtenberg, J. 1996. In Defence of Objectivity Revisited. In: Curran, J. and Gurevitch, M. Eds. *Mass Media and Society*. London, New York, Sydney, Auckland: Arnold, pp. 225–42.

Lichterman, J. 2015. Here's how BuzzFeed is thinking about its international growth. *Nieman Lab.* [Online]. [Accessed 12th June 2018]. Available from: <u>http://www.niemanlab.org/2015/04/heres-how-buzzfeed-is-thinking-about-its-international-growth/</u>

Liebler, C.M. and Smith S.J. 1997. Tracking gender differences: A comparative analysis of network correspondents and their sources. *Journal of Broadcasting & Electronic Media*. **41**(1), pp. 58–68.

Lindlof, T. 1995. *Qualitative Communication Research Methods*. Thousand Oaks, CA: Sage.

Lindlof, T.R. and Taylor, B.C. 2010. *Qualitative Communication Research Methods*. Thousand Oaks, CA: Sage.

Lotan, G., Graeff, E., Ananny, M., Gaffney, D., Pearce, I., and boyd, D. 2011. The revolutions were tweeted: Information flows during the 2011 Tunisian and Egyptian revolutions. *International Journal of Communication*, **5**, pp. 1375–1405.

Lowrey, W. 2006. Mapping the journalism-blogging relationship. *Journalism*. **7**(4), pp. 477–500.

Lowrey, W. 2011. Institutionalism, news organisations and innovation. *Journalism Studies*. **12**(1), pp. 64-79.

Lowrey, W. and Anderson, W. (2005) The journalist behind the curtain: Participatory functions on the internet and their impact on perceptions of the work of journalism. *Journal of Computer-Mediated Communication*. **10**(3). [Online]. [Accessed 1st November 2016]. Available from: <u>http://onlinelibrary.wiley.com/doi/10.1111/j.1083-6101.2005.tb00261.x/full</u>

Machill, M. and Beiler, M. 2009. The importance of the internet for journalistic research: A multimethod study of the research performed by journalists working for daily newspapers, radio, television and online. *Journalism Studies*. **10**(2), pp. 178–203.

Maras, S. 2013. Objectivity in journalism. Oxford: Polity Press

Marliére, P. 1998. The rules of the journalistic field: Pierre Bourdieu's contribution to the sociology of the media. *European Journal of Communication*, **13**(2), pp. 219–234.

Marshall, J. and Alpert, L.I. 2016. Publishers take on ad-agency roles with branded content. *The Wall Street Journal*. [Online]. [Accessed 3rd January 2017]. Available from: <u>https://www.wsj.com/articles/publishers-take-on-ad-agency-roles-with-branded-content-1481457605</u>

Martinson, J. 2015. The virtues of Vice: how punk magazine was transformed into media giant. *The Guardian*. [Online]. [Accessed 1st June 2017]. Available from: <u>https://www.theguardian.com/media/2015/jan/01/virtues-of-vice-magazine-transformed-into-global-giant</u>

Marx, K. 1852. *The Eighteenth Brumaire of Louis Bonaparte*. New York: International Pub. Co.

Matheson, D. 2004. Weblogs and the epistemology of online news: Some trends in online journalism. *New Media & Society*. **6**(4), pp. 443–468.

Mathison, S. 1988. Why triangulate? *Educational Researcher.* **17**(2), pp. 13-17.

McAteer, O. 2018. Behind BuzzFeed's bold move to launch 'AM to DM' show on Twitter. *Campaign*. [Online]. [Accessed 27th February 2018]. Available from: <u>https://www.campaignlive.co.uk/article/behind-buzzfeeds-bold-move-launch-am-dm-show-twitter/1457977</u> McManus, J. H. 1994. *Market-driven journalism: Let the citizen beware?* Thousand Oaks, CA: Sage.

McMillan, S. J. 2000. The microscope and the moving target: The challenge of applying content analysis to the world wide web. *Journalism & Mass Communication Quarterly*. **77**(1), pp. 80–98.

McChesney, R.W. 2012. Farewell to journalism? Time for a rethinking. *Journalism Studies*. **13**(5-6), pp. 682-694.

McChesney R.W. 2013. Digital Disconnect. New York: The New Press.

McChesney, R.W. and Pickard, V. eds. 2011. *Will the last reporter please turn out the lights: the collapse of journalism and what can be done to fix it.* New York, NY: The New Press.

Mellado, C. and Van Dalen, A. 2013. Between rhetoric and practice: Explaining the gap between role conception and performance in journalism. *Journalism Studies*. **15**(6), pp. 859-878.

Mendelson, H. 2000. Organizational architecture and success in the information technology industry. *Management Science*, **46**(4), pp. 513-529.

Merton, R. 1949. Social theory and social structure. Glencoe, NY: Free Press.

Messner, M., Linke, M. and Eford, A. 2012. Shoveling Tweets: An analysis of the microblogging engagement of traditional news organizations. *ISOJ Journal.* **2**(1), pp. 76–90.

Meyer, R. (2016) How many stories do newspapers publish per day? *The Atlantic.* [Online]. [Accessed 15th September 2017]. Available from: <u>https://www.theatlantic.com/technology/archive/2016/05/how-many-stories-do-</u> newspapers-publish-per-day/483845/

Mitchelstein, E. and Boczkowski, P. J. 2009. Between tradition and change: A review of recent research on online news production. Journalism. **10**(5), pp. 562–586.

Mindich, D. T. Z. 1998. Just the facts: How "objectivity" came to define American journalism. New York: NYU Press.

Mindich, DT.Z. 2005. *Tuned out: Why Americans under 40 don't follow the news*. Oxford: Oxford University Press.

Møller Hartley, J. 2013. The online journalist between ideals and audiences: Toward a (more) audience-driven, source-detached journalism? *Journalism Practice*, **7**(5), pp. 572-587.

Moon S.J. and Hadley, P. 2014. Routinizing a new technology in the newsroom: Twitter as a news source in mainstream media. *Journal of Broadcasting and Electronic Media*, **58**(2), pp. 289–305.

Morrison, K. 2015. Study: Female Facebook users prefer Yahoo, BuzzFeed, Men like The Guardian and Wired. *Adweek*. [Online]. [Accessed 19th April 2016]. Available from: <u>https://www.adweek.com/digital/study-female-facebook-users-prefer-yahoobuzzfeed-men-like-the-guardian-wired/</u> Morrissey, B. 2016. BuzzFeed's Dao Nguyen: "We don't think of ourselves as an island". *Digiday* [Online]. [Accessed 11th September 2017]. Available from: https://digiday.com/podcast/buzzfeed-dao-nguyen-digiday-podcast/

Morse, J. M. 1991. Approaches to qualitative-quantitative methodological triangulation. *Nursing Research*, **40**(2), pp. 120-123.

Moses, L. 2014. Which millennial news sites are really attracting millennials? *Digiday*. [Online]. [Accessed 18th September 2018]. Available from: <u>https://digiday.com/media/millennial-news-sites-really-attracting-millennials/</u>

Nadler, A. 2016. *Making the news popular: Mobilizing U.S. news audiences*. Chicago: University of Illinois Press.

Napoli P.M. 2010. Audience evolution: New technologies and the transformation of media audiences. New York: Columbia University Press.

Napoli, P.M. 2012. Audience evolution and the future of audience research. *The International Journal of Media management.* **14**(2), pp. 79-97.

National Council for the Training of Journalists. 2013. *Journalists at work: their views on training, recruitment, and conditions*. National Council for the Training of Journalists. [Online]. [Accessed 2nd June 2017]. Available from: http://www.nctj.com/downloadlibrary/jaw_final_higher_2.pdf

Netzer, Y., Teneboim-Weinblatt, k. and Shifman, L. 2014. The construction of participation in news websites: A five-dimensional model. *Journalism Studies*. **15**(5), pp. 619-631.

Nerone, J. 2009. The death and rebirth of working-class journalism. *Journalism*. **10**(3), pp. 350-355.

Neuendorf K.A. 2002. The content analysis guidebook. Thousand Oaks, CA: Sage.

Newman, N. and Austin, S. 2015. *Attitudes to sponsored and branded content (native advertising)*. Reuters Institute for the Study of Journalism: Oxford. [Online]. [Accessed 5th September 2017]. Available from: <u>http://www.digitalnewsreport.org/essays/2015/attitudes-to-advertising/</u>

Newman, N., Fletcher, R., Levy, D.A.L. and Nielsen, R.K. 2016. *Reuters Institute Digital News Report 2016*. Oxford: Reuters Institute for the Study of Journalism.

Newman, N., Fletcher, R., Kalogeropoulos, A., Levy, D.A.L. and Nielsen, R.K. 2017. *Reuters Institute Digital News Report 2017*. Oxford: Reuters Institute for the Study of Journalism.

Newman, N., Fletcher, R., Kalogeropoulos, A., Levy, D.A.L. and Nielsen, R.K. 2018. *Reuters Institute Digital News Report 2018*. Oxford: Reuters Institute for the Study of Journalism.

Nguyen, D. 2016. How BuzzFeed thinks about data, and some charts too. *BuzzFeed*. [Online]. [Accessed 11th November 2017]. Avilable from: <u>https://www.buzzfeed.com/daozers/how-buzzfeed-thinks-about-data-and-some-charts-too</u> Nicholls, T., Shabbir, N. and Nielsen, R.K. 2016. *Digital-born news media in Europe*. Oxford: Reuters Institute for the Study of Journalism.

Nicholls, T., Shabbir, N. and Nielsen, R.K. 2017. *The global expansion of digital-born news media*. Oxford: Reuters Institute for the Study of Journalism.

Nielsen, R.K. 2016. The business of news. In: Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. eds. *The SAGE Handbook of Digital Journalism*. London, California, New Delhi, Singapore: SAGE Publications, pp. 51-67.

Nielsen, R.K., Cornia, A. and Kalogeropoulos, A. 2016. *Challenges and opportunities for news media and journalism in an increasingly digital, mobile, and social media environment.* Oxford: Reuters Institute for the Study of Journalism.

Nielsen, R.K. 2016. The business of news. In: Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. eds. *The SAGE Handbook of Digital Journalism*. London, California, New Delhi, Singapore: SAGE Publications, pp. 51-68.

Oakes, O. 2015. BuzzFeed boss Peretti warns of using online ads that "don't care about readers". *Campaign.* [Online]. [Accessed 17th March 2018]. Available from: <u>https://www.campaignlive.co.uk/article/buzzfeed-boss-peretti-warns-using-online-ads-dont-care-readers/1372734</u>

Office for National Statistics. 2011. 2011 Census. ONS. [Online]. [Accessed 13th May 2018]. Available from: <u>https://www.ons.gov.uk/census/2011census</u>

Ornebring, H. 2009. *The two professionalisms of journalism: Journalism and the changing context of work*. Reuters Institute for the Study of Journalism: Oxford.

O'Sullivan, J. and Heinonen, A. 2008. Old Values, New Media. *Journalism Practice*. **2**(3), pp. 357-371.

Painter, J. 2008. *Counter-Hegemonic news: A case study of AI-Jazeera English and Telesur*. Reuters Institute for the Study of Journalism: Oxford.

Painter, J., Erviti, C.M., Fletcher, R., Howarth, C., Kristiansen, S., Bienvenido, L., Ouakrat, A., Russell, A. and Schäfer, S.M. 2017. *Something old, something new: digital media and the coverage of climate change*. Oxford: Reuters Institute for the Study of Journalism.

Palau-Sampio, D. 2016. Reference press metamorphosis in the digital context: clickbait and tabloid strategies in Elpais.com. *Communication & Society*. **29**(2), pp. 63-79.

Pantti, M. 2010. The value of emotion: An examination of television journalists' notions on emotionality. *European Journal of Communication*, **25**(2), pp. 168–181.

Papacharissi, Z. 2015. Towards new journalism(s). Affective news, hybridity, and liminal spaces. *Journalism Studies*. **16**(1), pp. 27-40.

Papacharissi, Z. 2016. Affective publics and structures of storytelling: sentiment, events and mediality. *Information, Communication & Society.* **19**(3), pp. 307-324.

Papacharissi, Z., and de Fatima Oliveira, M. 2012. Affective news and networked publics: The rhythms of news storytelling on #Egypt. *Journal of Communication*. **62**(2), pp. 266–282.

Paterson, C. 2007. International news on the internet: Why more is less. *Ethical Space: The international Journal of Communication Ethics.* **4**(1), pp. 57-66.

Paterson, C. 2008. Introduction: Why ethnography? In: Paterson, C. and Domingo, D. Eds. *Making Online News: The Ethnography of New Media Production*. New York: Peter Lang, pp. 1–11.

Paterson, C. and Zoellner, A. 2010. The efficacy of professional experience in the ethnographic investigation of production. *Journal of Media Practice*. **11**(2), pp. 37-41.

Paterson, C., Lee, D., Saha, A. and Zoellner, A. 2016. *Advancing media production research: Shifting sites, methods, and politics.* Basingstoke: Palgrave Macmillan.

Patterson T.E. 1993. Out of order. New York: Knopf.

Patterson, T. E. 1995. The American democracy 7th ed. New York: McGraw-Hill.

Pavlik, J. 2001. Journalism and new media. New York: Columbia University Press.

Peretti, J. 2013. Memo to the BuzzFeed team. *Linkedin*. [Online]. [Accessed 21st July 2017]. Available from: <u>https://www.linkedin.com/pulse/20130904212907-1799428-memo-to-the-buzzfeed-team/</u>

Peretti, J. 2014. a16z podcast: For Buzzfeed sharing is the metric that matters. Interview with Chris Dixon for *a16z* podcast. [Online]. [Accessed 11th June 2017]. Available from: <u>https://a16z.com/2014/09/18/a16z-podcast-for-buzzfeed-sharing-is-the-metric-that-matters/</u>

Peretti, J. 2015. BuzzFeed's Jonah Peretti and Ben Smith explain how they turned a "great cat site" into a powerful publisher. Interview with Peter Kafka from *Recode* magazine. [Online]. [Accessed 1st May 2018]. Available from: <u>https://www.recode.net/2015/6/12/11563480/buzzfeeds-jonah-peretti-and-ben-smith-on-the-dress-disappearing-posts</u>

Peretti, J. 2017a. 9 Boxes: Building out our multi-revenue model. *BuzzFeed*. [Online]. [Accessed 22nd August 2018]. Available from: <u>https://www.buzzfeed.com/jonah/9-boxes</u>

Peretti, J. 2017b. How BuzzFeed CEO Jonah Peretti took an instant messaging bot and turned it into a \$1.5 billion media empire. Interview with Alyson Shontell for *Business Insider UK*. [Online]. [Accessed 22nd February 2018]. Available from: <u>http://uk.businessinsider.com/buzzfeed-jonah-peretti-startup-success-how-i-did-it-interview-podcast-2017-5?r=US&IR=T</u>

Peretti, J. 2017c. 2017 Update on diversity. *BuzzFeed*. [Online]. [Accessed 19th January 2018]. Available from: <u>https://www.buzzfeed.com/jonah/2017-update-on-diversity-at-buzzfeed</u>

Peretti, J. 2017d. The interview: Jonah Peretti on taking BuzzFeed to the world, AI in media, and the rise of populist politics. Interview with Paul Colgan from *Business Insider Australia*. [Online]. [Accessed 21st May 2018]. Available from:

https://www.businessinsider.com/the-interview-jonah-peretti-on-taking-buzzfeed-tothe-world-ai-in-media-and-the-rise-of-populist-politics-2017-6?IR=T

Peretti, J. 2018a. 2018 update on diversity at BuzzFeed. [Online]. [Accessed 26th September 2018]. Available from: https://www.buzzfeed.com/jonah/2018-update-on-diversity-at-buzzfeed

Peretti, J. 2018b. Can BuzzFeed afford to keep funding BuzzFeed News? Interview with Matthew Ingram for *Columbia Journalism Review*. [Online]. [Accessed 12th February 2018]. Available from: <u>https://www.cjr.org/business_of_news/buzzfeed-business-model.php</u>

Peters, C. 2011. Emotion aside or emotional side? Crafting an "experience of involvement" in the news'. *Journalism*. **12**(3), pp. 297-316.

Peters, C. and Broersma, M. 2013. *Rethinking journalism: Trust and participation in a transformed news landscape*. London: Routledge

Peters, C. and Broersma, M. 2017. *Rethinking journalism again: Societal role and public relevance in a digital age*. London: Routledge.

Peters, C. and Witschge, T. 2015. From grand narratives of democracy to small expectations of participation. *Journalism Practice*. **9**(1), pp. 19-34.

Phillips, A. 2010. Old sources: new bottles. In: Fenton, N. Ed. *New Media, Old News, Journalism and Democracy in the Digital Age*. London: Sage, pp. 87–101

Phillips, A. 2011. Faster and shallower. Homogenisation, cannibalisation and the death of reporting. In: Lee Wright, P., Phillips, A. and Witschge, T. *Changing Journalism.* London: Routledge, pp. 81–98.

Phillips, A. 2012. Sociability, speed and quality in the changing news environment. *Journalism Practice*. **6**(5–6), pp. 669–679.

Phillips, A., Singer, J.B., Vlad, T. and Becker, L.B. 2009. Implications of technological change for journalists' task and skills. *Journal of Media Business*. **6**(1), pp. 61-85.

Picard, R. G. 2014. Twilight or new dawn of journalism? *Journalism Studies*. **15(**5), pp. 500–510.

Pöttker, H. 2003. News and its communicative quality: The inverted pyramid: When and why did it appear? *Journalism Studies*. **4**(4), pp. 501–511.

Puijk, R. 2008. Ethnographic media production research in a digital environment. In: Paterson, C. and Domingo, D. Eds. *Making Online News: The Ethnography of New Media Production*. New York: Peter Lang, pp. 29–41.

Quandt, T. 2008. (No) news on the world wide web? A comparative content analysis of online news in Europe and the United States. *Journalism Studies* **9**(5), pp. 717–738.

Quandt, T. and Singer, J. 2009. Convergence and cross-platform content production. In: Wahl-jorgensen, K. and Hanitzsch, T. Eds. *The Handbook of Journalism Studies*. London: Routledge, pp. 130-144. Quinn, B. and Jackson, J. 2016. Vice Media lays off 20 staff in restructuring plans. *The Guardian*. [Online]. [Accessed 25th February 2017]. Available from: <u>https://www.theguardian.com/media/2016/may/24/vice-media-lays-off-20-staff-in-restructuring-plans</u>

Reinemann, C., Stanyer, J., Scherr, S. and Legnante, G. 2011. Hard and soft news: A review of concepts, operationalizations and key findings. *Journalism*. **13**(2), pp. 1-19.

Reese S.D. 2009. Managing the symbolic arena: The media sociology of Herbert Gans. In: Holtz-Bacha C., Reus G., Becker L.B. Eds. *Wissenschaft mit Wirkung*. VS Verlag für Sozialwissenschaften, pp. 279-293.

Reich, Z. 2012. Journalism as bipolar interactional expertise. *Communication Theory*. **22**(4), pp. 339-358.

Reid, A. 2014. Vice launches news channel for "young, global audience". *Journalism.co.uk*. [Online]. [Accessed 25th March 2017]. Available from: <u>https://www.journalism.co.uk/news/vice-launches-news-channel-for-young-global-audience/s2/a556032/</u>

Reese, S. D. 1990. The news paradigm and the ideology of objectivity: A socialist at the Wall Street Journal. *Critical Studies in Mass Communication*. **7**(1), pp. 390–409.

Ridley, L. 2016. BBC-BuzzFeed tennis investigation shows how curious bedfellows became a perfect media match. *Huffington Post*. [Online]. [Accessed 12th June 2017]. Available from: <u>https://www.huffingtonpost.co.uk/2016/01/18/bbc-buzzfeed-tennis-match-fixing-investigation n 9008412.html</u>

Riffe, D., Lacy, S., and Fico, F. 2014. *Analyzing media messages: Using quantitative content analysis in research 3rd Ed.* New York, NY: Routledge.

Riordan, K. 2014. *Accuracy, Independence, and Impartiality: How legacy media and digital natives approach standards in the digital age.* University of Oxford: Reuters Institute for the Study of Journalism.

Robinson, S. 2007. Someone's gotta be in control here. *Journalism Practice*. **1**(3) pp. 305–21.

Rodriguez, A. 2016. How a single deal with a decidedly unhip tech company built the Vice media behemoth. *Quartz*. [Online]. [Accessed 12th February 2017]. Available from: <u>https://qz.com/776628/shane-smith-how-a-single-native-advertising-deal-with-intel-intc-built-the-vice-media-behemoth/</u>

Rosen, J. 2006. The people formerly known as the audience. *PressThink.* [Online]. [Accessed 27th February 2017]. Available from: http://archive.pressthink.org/2006/06/27/ppl_frmr.html

Rosen, J. 2010. The view from nowhere: questions and answers. *PressThink*. [Online]. [Accessed 13th March 2017]. Available from: <u>http://pressthink.org/2010/11/the-view-from-nowhere-questions-and-answers/</u>

Ryfe, D. M. 2006. Guest editors introduction: New institutionalism and the news. *Political Communication*. **23**(2), pp. 135–144.

Ryfe, D. M. 2012. *Can journalism survive? An inside look at American newsrooms*. Cambridge: Polity.

Ryfe, D.M. 2016. The importance of time in media production research. In: Paterson, C., Lee, D., Saha, A. and Zoellner, A. *Advancing media production research*. Basingstoke: Palgrave Macmillan, pp. 38-53.

Salgado, S. and Strömbäck, J. 2012. Interpretive journalism: A review of concepts, operationalizations, and key findings. *Journalism*. **13**(2), pp.144-161.

Sarfatti-Larson, M. 1977. *The rise of professionalism*. Berkeley: University of California Press.

Schauster, E.E., Ferrucci, P. and Neill, M.S. 2016. Native advertising is the new journalism: how deception affects social responsibility. *American Behavioral Scientist*, **60**(12), pp. 1408-1424.

Schifferes S., Newman, N. and Thurman, N. 2014. Identifying and verifying news through social media: developing a user-centred tool for professional journalists. *Digital Journalism*. **2**(3), pp. 406–418.

Schlesinger, P. 1978. Putting 'Reality' Together: BBC News. London: Constable.

Schlesinger, P. 1980. Between sociology and journalism. In: Christian, H. Ed. *Sociology of the Press and Journalism*. Keele: University of Keele, pp. 341–69.

Schoofs, M. 2016. Digital digging: how BuzzFeed built an investigative team inside a viral hit factory. Interview with Benjamin Mullin from *Poynter*. [Online]. [Accessed 24th March 2018]. Available from: <u>https://www.poynter.org/news/digital-digging-how-buzzfeed-built-investigative-team-inside-viral-hit-factory</u>

Schudson, M. 1978. *Discovering the news: A social history of American newspapers.* New York: Basic Book Publishing.

Schudson, M. 2001. The objectivity norm in American journalism, *Journalism: Theory, Practice & Criticism.* **2**(2), pp. 149–170.

Schudson, M. (2011) *The sociology of news*. 2nd Ed. New York & London: Norton.

Schudson, M. and Anderson, C. 2008. Objectivity, professionalism, and truth seeking in journalism. In: Wahl-jorgensen, K. and Hanitzsch, T. Eds. *The Handbook of Journalism Studies*. London: Routledge, pp. 88–101.

Schultz, I. 2007. The journalistic gut feeling. *Journalism Practice*. **1**(2), pp. 190-207.

Sigelman, L. 1973. Reporting the News: An Organizational Analysis. American Journal of Sociology, **79**(1), pp. 132–51.

Shin, J. and Cameron, G. 2003. The interplay of professional and cultural factors in the online source-reporter relationship. *Journalism Studies*. **4**(2), pp. 253–272.

Shirky, C. 2008. *Here comes everybody. The power of organizing without organizations*. New York: The Penguin Press

Shoemaker, P. and Vos, T.P. 2009. *Gatekeeping theory*. New York: Routledge.

Shoemaker, P. and Reese, S. 2014. *Mediating the message in the 21st century. A media sociology perspective*. 3rd ed. New York: Routledge.

Shontell, A. 2017. How BuzzFeed CEO Jonah Peretti took an instant messaging bot and turned it into a \$1.5 billion media empire. *Business Insider UK*. [Online]. [Accessed 27th June 2017]. Available from: <u>http://uk.businessinsider.com/buzzfeed-jonah-peretti-startup-success-how-i-did-it-interview-podcast-2017-5</u>

Siapera, E. and Spyridou, L.P. 2012. The field of online journalism: a Bourdieusian analysis. In: Siapera, E. and Veglis, A. eds. *The Handbook of Global Online Journalism*. Oxford: Wiley- Blackwell, pp. 77–97.

Singer, J. 2003. Who are these guys? The online challenge to the notion of journalistic professionalism. *Journalism*, **4**(2), pp. 139–163.

Singer, J. 2005. The political J-blogger: "Normalizing" a new media form to fit old norms and practices. *Journalism*. **6**(2), pp. 173–98.

Singer, J. 2008. The journalist in the network. A shifting rationale for the gatekeeping role and objectivity norm. *Trípodos*, **23**, pp. 61-76.

Singer, J. 2013. User-generated visibility: Secondary gatekeeping in a shared media space. *New Media & Society*. **16**(1), pp. 55–73.

Singer, J. B., Domingo, D., Heinonen, A., Hermida, A., Paulussen, S., Quandt, T. and Vujnovic, M. 2011. *Participatory Journalism: Guarding Open Gates at Online Newspapers*. Wiley-Blackwell, Malden, MA.

Skovsgaard, M. and van Dalen, A. 2013. The fading public voice: The polarizing effect of commercialization on political and other beats and its democratic consequences. *Journalism Studies*. **14**(3), pp. 371-386.

Smith, S. 2013. Vice CEO on old media: "They can go to hell quite frankly". Interview with Miguel Helft for *Fortune* magazine. [Online]. [Accessed 11th June 2015]. Available from: <u>http://fortune.com/2013/10/14/vice-ceo-on-old-media-they-can-go-to-hell-quite-frankly/</u>

Smith, B. 2014a. Ben Smith: The Complete Transcript. *Nieman Reports.* [Online]. [Accessed 27th November 2017]. Available from: http://niemanreports.org/articles/ben-smith-the-complete-transcript/

Smith, B. 2015a. "Here is the rabbit hole" Meduza talks to BuzzFeed chief editor Ben Smith about the future of new media. Interview with Ivan Kolpakov and Ilya Krasilshchik for *Meduza*. [Online]. [Accessed 22nd April 2017]. Available from: https://meduza.io/en/feature/2015/07/06/here-is-the-rabbit-hole

Smith, B. 2015b. My life in the blogosphere. *BuzzFeed*. [Online]. [Accessed 23rd May 2017]. Available from: <u>https://www.buzzfeednews.com/article/bensmith/this-is-my-blog</u>

Smith, S. 2016a. Vice's Shane Smith wants his company to be the voice for millennials. Interview with John Horn for *The Frame*. [Online]. [Accessed 22nd September 2017]. Available from: <u>https://www.scpr.org/programs/the-frame/2016/12/07/53546/vice-s-shane-smith-wants-his-company-to-be-the-voi/</u>

Smith, B. 2016b. Q&A with BuzzFeed News Editor-in-Chief Ben Smith. Interview with *Hearst Ventures*. [Online]. [Accessed 22nd March 2017]. Available from: http://www.hearst.com/newsroom/q-a-with-buzzfeed-news-editor-in-chief-ben-smith

Smith, S. 2016c. McTaggart Lecture. *Edinburgh International Television Festival*. [Online]. [Accessed 24th August 2017]. Available from: https://www.youtube.com/watch?v=ZSdkxXGQAPA

Smith, B. 2016d. Interview with Max Linsky for *Longform Podcast.* [Online]. [Accessed 14th April 2016]. Available from: <u>https://longform.org/posts/longform-podcast-185-ben-smith</u>

Smith, B. 2017a. BuzzFeed is slowly getting white and less male. *Splinter*. [Online]. [Accessed September 28th 2018]. Available from: <u>https://splinternews.com/buzzfeed-news-is-slowly-getting-less-white-and-less-mal-1797683969</u>

Smith, B. 2017b. Why BuzzFeed news published the dossier. *The New York Times*. [Online]. [Accessed 15th January 2018]. Available from: <u>https://www.nytimes.com/2017/01/23/opinion/why-buzzfeed-news-published-the-dossier.html</u>

Soloski, J. 1989. News reporting and professionalism: Some constraints on the reporting of news. *Media, Culture and Society*, **11**(2), pp. 207–28.

Spangler, T. 2017. Vice Media Laying Off 2% of Staff Amid International, Video Expansion (Exclusive). *The Atlantic* [online]. [Accessed 10th September 2017]. Available from: <u>http://variety.com/2017/digital/news/vice-media-layoffs-global-video-1202502249/</u>

Stake, R. 2003. Case studies. In: Denzin, N.K. and Lincoln, Y.S. Eds. *Strategies of qualitative inquiry 2nd Ed*. Thousand Oaks, CA: Sage, pp. 134 - 164.

Steel, E. 2018. Former Employee Sues Vice, Claiming It Pays Women Less. *The New York Times*. [Online]. [Accessed 22nd April]. Available from: https://www.nytimes.com/2018/02/13/business/media/vice-media-lawsuit.html

Steensen, S. 2009. The shaping of an online feature journalist. *Journalism*. **10**(5), pp. 702–718

Steensen, S. 2016. The intimization of journalism. In: Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. eds. *The SAGE Handbook of Digital Journalism*. London, California, New Delhi, Singapore: SAGE Publications, pp. 113-128.

Steensen, S. and Ahva, L. 2015. Theories of journalism in a digital age: An exploration and introduction. *Journalism Practice*, **9**(1), pp.1-18.

Steiner L .2009. Gender in the newsroom. In: Wahl-Jorgensen, K. and Hanitzsch, T. Eds. *Handbook of Journalism Studies*. London: Routledge.

Sterne, P. 2014. Former Vice Media editor says company killed stories over 'brand partner' concerns. *Politico*. [Online]. [Accessed 22nd February 2017]. Available from: <u>https://www.politico.com/media/story/2014/10/former-vice-media-editor-says-company-killed-stories-over-brand-partner-concerns-002932</u>

Strauss, A. L. and Corbin, J. M. 1990. *Basics of qualitative research: Grounded theory procedures and techniques*. Newbury Park, CA: Sage.

Strömbäck, J. and Nord, L.W. 2006. Do politicians lead the tango? A study of the relationship between Swedish journalists and their political sources in the context of election campaigns. *European Journal of Communication*. **21**(2), pp. 147-164.

Strömbäck, J. and Aalberg, T. 2008. Election news coverage in democratic corporatist countries: A comparative study of Sweden and Norway. *Scandinavian Political Studies*. **31**(1), pp. 91-106.

Strömbäck, J. and Shehata, A. 2008. Structural biases in British and Swedish election news coverage. *Journalism Studies*. **8**(5), pp. 798-812.

Summers, M. W. 1994. *The press gang: Newspapers and politics (1865–1878)*. Chapel Hill: University of North Carolina Press.

Sutcliffe, K. 2015. The virtues of Vice: how punk magazine was transformed into media giant. Interview with Jane Martinson from *The Guardian*. [Online]. [Accessed 1st June 2017]. Available from:

https://www.theguardian.com/media/2015/jan/01/virtues-of-vice-magazinetransformed-into-global-giant

Sweney, M. 2016a. Guardian announces partnership with Vice. *The Guardian*. [Online]. [Accessed 26th May 2017]. Available from: <u>https://www.theguardian.com/media/2016/dec/08/guardian-announces-partnership-with-vice</u>

Sweney, M. 2016b. BuzzFeed breaks UK ad rules over misleading advertorial. *The Guardian*. [Online]. [Accessed 21st March 2017]. Available from: <u>https://www.theguardian.com/media/2016/jan/13/buzzfeed-breaks-uk-ad-rules-over-misleading-advertorial</u>

Tandoc, E.C. 2013. Web analytics, social media, and the journalistic doxa: The impact of audience feedback on the evolving gatekeeping process. Ph.D. thesis, University of Missouri-Columbia.

Tandoc, E. 2014. Journalism is twerking? How web analytics is changing the process of gatekeeping. *New Media and Society.* **16**(4), pp. 559-575.

Tandoc, E.C. 2017. Five ways BuzzFeed is preserving (or transforming) the journalistic field. *Journalism*. **19**(2), pp.200-216.

Tandoc, E.C. and Thomas, R.J. 2015. The ethics of web analytics. *Digital Journalism*. **3**(2), pp. 243-258.

Tandoc, E.C. and Jenkins, J. 2017. The Buzzfeedication of journalism? How traditional news organizations are talking about a new entrant to the journalistic field will surprise you! *Journalism*, **18**(4), pp. 482-500.

Thomson E.A., White P.R.R. and Kitley, P. 2008. "Objectivity" and "hard news" reporting across cultures: Comparing the news report in English, French, Japanese and Indonesian journalism. *Journalism Studies*, **9**(2), pp. 212–228.

Thorson, E. 2008. Changing patterns of news consumption and participation. *Information, Communication and Society.* **11**(4), pp. 473-489.

Thurman, N. 2008. Forums for citizen journalists? Adoption of user-generated content initiatives by online news media. *New Media and Society*. **10**(1), pp. 139–57.

Thurman, N., Cornia, A. and Kunert, J. 2016. *Journalists in the UK*. Oxford: Reuters Institute for The Study of Journalism. [Online]. [Accessed 2nd June 2017]. Available from:

https://reutersinstitute.politics.ox.ac.uk/sites/default/files/Journalists%20in%20the%2 0UK.pdf

Tozzi, L. 2018. Q&A with Lisa Tozzi of BuzzFeed News. Interview with Andy Bechtel from *The Editor's Desk*. [Online]. [Accessed 7th May 2018]. Available from: <u>https://editdesk.wordpress.com/2018/03/01/qa-with-lisa-tozzi-of-buzzfeed-news/</u>

Tracy, S. 2013. *Qualitative research methods: Collecting evidence, crafting analysis, communicating impact.* Malden, MA: Wiley-Blackwell.

Tremayne, M. 2005. News websites as gated cybercommunities. *Convergence*. **11**(3), pp. 28-39.

Trotter, J.K. 2015. BuzzFeed deleted posts under pressure from its own business department. *Gawker*. [Online]. [Accessed 12th September 2017]. Available from: <u>http://tktk.gawker.com/buzzfeed-deleted-posts-under-pressure-from-its-own-busi-1697762873</u>

Tsui, L. 2008. The hyperlink in newspapers and blogs. In: Turow, J. and Lokman, T. Eds. *The hyperlinked society: questioning connections in the digital age*. Ann Arbor: The University of Michigan Press, pp. 70-84.

Tuchman, G. 1972. Objectivity as strategic ritual: An examination of newsmen's notions of objectivity. *The American Journal of Sociology*, **77**(4), pp. 660-679.

Tuchman, G. 1978. *Making News: A study in the social construction of reality*. New York: Free Press.

Tuchman, G. 2002. The production of news. In: Jensen, K.B. Ed. A Handbook of *Media and Communication Research: Qualitative and Quantitative Methodologies*. London, New York: Routledge, pp. 78-90.

Tumber, H. and Prentoulis, M. 2005. Journalism and the making of a profession. In: De Burgh, H. Ed. *Making Journalists*. New York: Routledge, pp. 58-74.

Tunstall, J. (1971) Journalists at work. London: Constable.

United States Census Bureau. 2017. Census Data. *USCB*. [Online]. [Accessed 11th November 2017]. Available from: <u>https://www.census.gov/data.html</u>

Usher, N. (2010) Why SEO and audience tracking won't kill journalism as we know it. *NiemanLab.* [Online]. [Accessed 1st May 2015]. Available from: <u>http://www.niemanlab.org/2010/09/why-seo-and-audience-tracking-won%E2%80%99t-kill-journalism-as-we-know-it-2/</u> Usher, N. 2014. *Making news at The New York Times*. Ann Arbor, MI: University of Michigan Press.

Usher, N. 2017. The appropriation / amplification model of citizen journalism. *Journalism Practice*. **11**(2-3), pp. 1-19.

Usher, N. 2017. Venture-backed news startups and the field of journalism: Challenges, changes, and consistencies. *Digital Journalism*. **5**(9), pp. 1116-1133.

Van Dijck, J. and Nieborg, D. 2009. Wikinomics and its discontents: A critical analysis of Web 2.0 business manifestos. *New Media & Society*. **11**(5), pp. 855-874.

van Dijck, J., and Poell, T. 2013. Understanding social media logic. *Media and Communication*, **1**(1), pp. 2-14.

Van Hout, T. and Van Leuven, S. 2016. Investigating "churnalism" in real time news. In: Franklin, B. and Eldridge II, S. *The Routledge Companion to Digital Journalism Studies*. London, UK: Routledge, pp. 117–125.

Van Leuven S., Deprez, A. and Raeymaeckers, K. 2014. Towards more balanced news access? A study on the impact of cost-cutting and Web 2.0 on the mediated public sphere. *Journalism.* **15**(7), pp. 850–867.

Van Zoonen, L. 1998. A professional, unreliable, heroic marionette (M/F): structure, agency and subjectivity in contemporary journalisms. *European Journal of Cultural Studies*. **1**(1), pp. 123-143.

Vice. 2015a. Vice and HBO enter major deal to expand news programming. *Vice.* [Online]. [Accessed 30th August 2017]. Available from: <u>https://news.vice.com/article/vice-and-hbo-enter-major-deal-to-expand-news-programming</u>

Vice. 2015b. VICE to Gawker: fuck you and fuck your garbage click-bait 'journalism'. *Vice*. [Online]. [Accessed 27th August 2017]. Available from: <u>https://www.vice.com/en_us/page/response</u>

Vice. 2016a. Vice Media Kit. Vice. [Online]. [Accessed 1st September 2017]. Available from: <u>https://upload-</u> assets.vice.com/files/2016/01/15/1452894236compressed.pdf

Vice. 2016b. Annual reports and financial statements for the year ended 31 December 2016. *Companies House*. [Online]. [Accessed 23rd November 2017]. Available from: <u>https://beta.companieshouse.gov.uk/company/04531415/filing-history</u>

Vice. 2017a. About. Vice. [Online]. [Accessed 13th January 2018]. Available from: https://www.vice.com/en_id/page/about-58477f133bbbf901f85613df

Vice. 2017b. Vice UK collective workers statement. *Medium*. [Online]. [Accessed 19th February 2018]. Available from: https://medium.com/@v1ceworkersuk/vice-uk-collective-workers-statement-7cd5f9538f24

Vice. 2017c. Vice homepage. *Vice*. [Online]. [Accessed 8th September 2017]. Available from: <u>https://www.vice.com/</u>

Vice. 2017d. Vice News homepage. *Vice*. [Online]. [Accessed 22nd September 2017]. Available from: <u>https://news.vice.com/en_us</u>

Vice. 2018. Vice Channel. YouTube. [Online]. [Accessed 24th May 2018]. Available from: <u>https://www.youtube.com/user/vice</u>

Vos, T.P. 2016. Journalistic Fields. In: Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. eds. *The SAGE Handbook of Digital Journalism*. London, California, New Delhi, Singapore: SAGE Publications, pp. 383-397.

Vos. T. and Craft, S. 2016. The discursive construction of journalistic transparency. *Journalism Studies*. **18**(12), pp.1505-1522.

Vos, T.P. and Singer, J.B. 2016. Media discourse about entrepreneurial journalism. *Journalism Practice*, **10**(2), pp. 143-159.

Wahl-Jorgensen, K. 2009. News production, ethnography, and power: On the challenges of newsroom-centricity. In: Bird, E. ed. *The Anthropology of News and Journalism: Global Perspectives*. Bloomington, IN: Indiana University Press, pp. 21–35.

Wahl-Jorgensen, K. 2013. Subjectivity and story-telling in journalism. *Journalism Studies*. **14**(3), pp. 305-320.

Wahl-Jorgensen, K. 2016. Emotion and Journalism. In: Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. eds. *The SAGE Handbook of Digital Journalism*. London, California, New Delhi, Singapore: SAGE Publications, pp. 128-145.

Waisbord, S. (2013) *Reinventing Professionalism: Journalism and News in Global Perspective.* Cambridge, U.K.: Polity Press.

Wardle, C. and Williams, A. 2010. Beyond user-generated content: a production study examining the ways in which UGC is used at the BBC. *Media, Culture & Society*. **32**(5), pp. 781–799.

Weare, C. and Lin, W.Y. 2000. Content analysis of the world wide web: Opportunities and challenges. *Social Science Computer Review*. **18**(3), pp. 272–292.

Weaver, D. H. Ed. 1998. *The global journalist: News people around the world*. New Jersey: Hampton Press.

Weaver, D.H. 2005. Who are the journalists? In: de Burgh, H. ed. *Making Journalists.* London and New York: Routledge, pp. 44-57.

Weaver, D. Beam, H. Randal, Brownlee, A. Bonnie J., Voakes, S. P. and Wilhoit, G. 2007. *The American journalist in the 21st Century: U.S. news people at the dawn of a new millennium*. New York: Routledge.

Welbers, K., van Atteveldt, W., Kleinnijenhuis, J., Ruigrok, N. and Schaper, J. 2016. News selection criteria in the digital age: Professional norms versus online audience metrics. *Journalism*. **17**(8), pp. 1037-1053.

White, M, D. 1950. The gatekeeper. *Journalism Quarterly*. **27**, pp. 383-390.

Widdicombe, L. 2013. The bad-boy brand. *The New Yorker*. [Online]. [Accessed 29th June 2016]. Available from: <u>https://www.newyorker.com/magazine/2013/04/08/the-bad-boy-brand</u>

Widholm, A. 2016. Tracing online news in motion. *Digital Journalism*. **4**(1), pp. 24-40.

Williams, A., Wardle, C. and Wahl-Jorgensen, K. 2011. Have they got news for us? Audience revolution or business as usual at the BBC? *Journalism Practice*, **5**(1), pp. 85–99.

Willig, I. 2012. Newsroom ethnography in a field perspective. *Journalism.* **14**(3), pp. 372-387.

Willig, I., Waltorp, K. and Hartley, M.J. 2015. Field theory approaches to new media practices: An introduction and some theoretical considerations', *Mediekultur*, **58**, pp. 1-12.

Willnat, L. and Weaver, D. H. 2014. *The American Journalist in The Digital Age: key findings*. School of Journalism, Indiana University. [Online]. [Accessed 5th June 2017]. Available from: <u>http://archive.news.indiana.edu/releases/iu/2014/05/2013-american-journalist-key-findings.pdf</u>

Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. 2016. Introduction. In: Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. *The SAGE Handbook of Digital Journalism*. London, California, New Delhi, Singapore: SAGE Publications, pp. 1-4.

Witschge, T., Anderson, C.W., Domingo, D. and Hermida, A. 2018. [Post-print]. Dealing with the mess (we made): Unraveling hybridity, normativity, and complexity in journalism studies. *Journalism*.

Yin, R.K. 1994. Case study research: Design and methods, 2nd ed. London: Sage.

Yin, R.K. 2011. *Qualitative research: From start to finish.* New York, London: The Guildford Press.

Zamith, R. 2018. Quantified audiences in news production. *Digital Journalism*. **6**(4), pp. 418-435.

Zeller, F. and Hermida, A. 2015. When tradition meets immediacy and interaction. *Sur le journalisme, About journalism, Sobre jornalismo*. **4**(1), pp. 106-119.

Zelizer, B. 1992. Covering the body: The Kennedy assassination, the media, and the shaping of collective memory. Chicago, IL: University of Chicago Press.

Zelizer, B. 2009. Journalism and the academy. In: Wahl-Jorgensen, K. and Hanitzsch, T. eds. *The Handbook of Journalism Studies*. New York and London: Routledge, pp. 29-42.

Zelizer, B. 2015. Terms of choice: Uncertainty, journalism, and crisis. *Journal of Communication*. **65**(5), pp.888-908.

Zoch L.M. and Turk J.V. 1998. Women making news: Gender as a variable in source selection and use. *Journalism & Mass Communication Quarterly*, **75**(4) pp. 762–775.

Zou, S. 2018. Emotional news, emotional counterpublic. 2018 [Post-print]. *Digital Journalism.*